National Criminal Justice Reference Service

ncjrs

This microfiche was produced from documents received for inclusion in the NCJRS data base. Since NCJRS cannot exercise control over the physical condition of the documents submitted, the individual frame quality will vary. The resolution chart on this frame may be used to evaluate the document quality.



MICROCOPY ESOLUTION TEST CHART NATIONAL BUREAU OF STANDARDS-1963-A

Microfilming procedures used to create this fiche comply with the standards set forth in 41CFR 101-11.504.

Points of view or opinions stated in this document are those of the author(s) and do not represent the official position or policies of the U. S. Department of Justice.

National Institute of Justice United States Department of Justice Washington, D.C. 20531 DATE FILMED 5/15/81



T

T

m

DRAFT SYSTEM IMPLEMENTATION PLANS FOR PARTICIPATING COMMUNITIES

PROJECT 16B

of the

CIATED PUBLIC-SAFETY COMMUNICATIONS OFFICERS, INC.

Prepared by

BOOZ•ALLEN & HAMILTON, INC.

LAW ENFORCEMENT ASSISTANCE ADMINISTRATION

Grant No.: 79-SS-AX-0013

November 15, 1980

DRAFT SYSTEM IMPLEMENTATION PLANS FOR PARTICIPATING COMMUNITIES

PROJECT 16B

of the

1,

ASSOCIATED PUBLIC-SAFETY COMMUNICATIONS OFFICERS, INC.

PREPARED BY

BOOZ ALLEN & HAMILTON Inc. NCJRS JAN 15 1981

ACQUISITIONS

776 SHREWSBURY AVENUE TINTON FALLS, NEW JERSEY 07724 747-9303 AREA CODE 201

This volume of draft System Implementation Plans for Participating communities was prepared by Associated Public-Safety Communications Officers (APCO) in response the LEAA Grant No. 79 SS AX 0013. APCO's Board of Officers, constituted as a project Task Group I, provided overall guidance for the project. Task Group II, composed of selected APCO members and representatives of the five local communities participating in the project, selected for their particular professional expertise, provided the technical and professional guidance necessary to assure that the document would realistically reflect the requirements of their communities. Booz, Allen & Hamilton provided consulting contractor services to APCO. The professional competence and dedication to the public's service shown by all of the participants in this project reflect great credit upon the telecommunications profession.

APCO wishes to express its appreciation to the following individuals for their contributions to the field of public safety:

William H. Bailey

TASK GROUP I

11

114

Sanford H. Smith

President-Elect of APCO Russell V. Robinson Detroit Police Department Detroit, Michigan

Joseph W. Gallelli

ACKNOWLEDGEMENT

LEAA Program Monitor

President of APCO 1979/1980 City of Greensboro, North Carolina

President of APCO 1978/1979 Henry L. Crutcher California Department of Parks and Recreation Sacramento, California

> First Vice President of APCO New York State Police Albany, New York

				8 . T N	
	Craig M. Jorgensen	Second Vice President of APCO Utah Department of Transportation			
		and Public Safety State of Utah Salt Lake City, Utah			I. INTRODUC Schedu Organi
TASK	GROUP II				II. LEXINGTO
	Philip Y. Byrd	State of Florida Communications Tallahassee, Florida			System
	Jerry Campbell	Telecommunications Consultants, Inc. Calimesa, California			System
	Martin G. Ficke	Bucks County Pennsylvania			IV. PHOENIX, System
		Communications Doylestown, Pennsylvania			V. SALT LAK System
	Sgt. William W. C. Lisanby	Lexington Fayette Urban County Police Department Lexington, Kentucky			APPENDIX A -]
	Charles L. McGill	Bucks County Pennsylvania Communications Doylestown, Pennsylvania			APPENDIX B - 1
	William L. Miller	Chicago Police Department Chicago, Illinois			
	Kent Rasmussen	Oklahoma Department of Public Safety Oklahoma City, Oklahoma		I	
	John E. Simmons	Phoenix Fire Department Phoenix, Arizona			
	Lt. Rex L. Vance	Salt Lake County Sheriff's Department Salt Lake City, Utab			
APCO	NATIONAL OFFICE				
	Ernest J. Landreville	Executive Director		(TT)	
	Bruce P. Fisher	Project Director			
CONSU	JLTING CONTRACTOR	Booz, Allen & Hamilton Inc.			
	Thomas B. Wiggins	Vice President			
	Gerald W. Bernas	Project Manager	е 44		

TABLE OF CONTENTS

CTION - VOLUME II ule of Activities ization of This Report

ON, KENTUCKY m Implementation Plan - Draft

OKLAHOMA Implementation Plan - Draft

ARIZONA Implementation Plan - Draft

KE COUNTY, UTAH n Implementation Plan - Draft

EXAMPLE PROBLEM ANALYSIS REPORT

BUCKS COUNTY, PA - FINAL LETTER REPORT

..... dia I. INTRODUCTION - VOLUME II Schedule of Activities Organization of This Report the project. SCHEDULE OF ACTIVITIES 1. Project 16B group.

I. INTRODUCTION

A major objective of APCO Project 16B is to assist the participating communities in the development of their individual plans for the implementation of a digitally addressed trunked communication system (DATCS). It is the responsibility of the Project Director in each community to generate these plans and to go forward with their implementation. The System Implementation Planning Guidelines, developed under APCO Project 16B, were used as a guide. The responsibility of APCO and its contractor, Booz, Allen and Hamilton Inc., is to assist the individual Project Directors during the grant period to the limit of available project funds.

This report contains the draft (opies of the System Implementation Plan (SIP) for Lexington, Kentucky; the State of Oklahoma, Phoenix, Arizona, and Salt Lake County, Utah. An SIP for Bucks County, Pennsylvania was not developed under this program. Time prevented these SIP documents from receiving final review and approval by the respective governmental bodies although the major supporting parts of each SIP was reviewed in depth by each community in the course of

The project was initiated in each community by developing a formal Support Agreement and Work Plan. This agreement was characterized by a Master Schedule of Activities for each community, a sample of this schedule typical of the schedules developed in each of the five participating communities, is shown as EXHIBIT I-1. This was the first step in the planned ten step program.

Data collection is implemented in two parts; the review of existing documentation, and personal interviews with potential users of DATCs. To provide maximum efficiency in collecting and recording only the data needed, interview guides were developed for each community. These quides also serve as permanent data records which can be varified in response to time dependent system changes and/or user attitudes. These data were collated and reduced to a written report reflecting the communication system and its inventory as it existed in each community. This report, in its entirety, appears as an Appendix to each SIP.

As part of the data collection activity, selected performance data for the existing communication system was acquired. These data served two purposes; they served as a quide for specifying the DATCs system design, and they provide benchmarks for later evaluating performance improvements achieved by the DATCS. These data elements were collected mainly by the potential users and provided to the

The documented data base highlighted existing and potential communications system problems. It also provided a list of requirements the potential user had for any communication system and for DATCS in particular. Each problem and each requirement has been addressed in each community and, where possible, a series of alter-

I-1



[----

I-2

native solutions was documented. A draft problem analysis report was developed for each community. This report was submitted for review and each community selected the preferred solution to each problem. A draft problem analysis report, typical of the similar report in each community, appears as Appendix A to this report. The selected preferred solutions to the problems in each community appears as an Appendix to each of the community SIPs.

The DATCS system concept for each community has been developed from the individual data bases and from the selection of preferred solutions to potential problems and system requirements. This system concept is unique to each community but has a high degree of similarity among the four conceptual DATCS systems. A draft report of the system concept was reviewed by each community and the system concept which resulted from this review is presented in its entirety as Part II of each SIP in this report.

A draft Statement of Work (SOW) for each community has been developed from the instructions set forth in the System Implementation Planning Guidelines. These draft SOWs are shown as Appendices to individual SIPS in this report.

The early review of the radio communication system problems and requirements for Bucks County, Pennsylvania led to their decision to pursue a conventional upgrade of their radio system and defer DATCS planning at this time. A final project report for Bucks County appears as Appendix B to this report.

2. ORGANIZATION OF THIS REPORT

Each draft SIP contained in this report retains the integrity of the draft System Implementation Plan (SIP) submitted to each participating community. Section II contains the Lexington, Kentucky draft SIP, Section III contains the State of Oklahoma, draft SIP, Section IV contains the Phoenix, Arizona draft SIP, and Section V contains the Salt Lake County, Utah draft SIP. Appendix A presents an example Problem Analysis Report of the draft reports presented to each community. Appendix B contains the final project report to Bucks County, Pennsylvania.

(T

II. LEXINGTON, KENTUCKY

System Implementation Plan - Draft



Sgt. William C. W. Lisanby Division of Police Lexington-Fayette Urban County Police 1409 Forbes Road Lexington, Kentucky 40505

....

- 6

10

DIGITALLY ADDRESSED TRUNKED COMMUNICATION SYSTEM

SYSTEM IMPLEMENTATION PLAN

LEXINGTON-FAYETTE URBAN COUNTY GOVERNMENT

November 17, 1980

BOOZ · ALLEN & HAMILTON Inc.

776 SHREWSBURY AVENUE TINTON FALLS, NEW JERSEY 07724 747-9303 AREA CODE 201

TABLE OF CONTENTS

T

.

 \square

 \mathcal{G} .

Γ.

II.

III.

APPENDICES:

	Page
INTRODUCTION	II-1
<pre>1.1 Project Background 1.2 Project Director and Participat Agencies</pre>	II-1 ing II-2
1.3 Summary of Current Status of Implementation	II-2
SYSTEM CONCEPT .	II-5
2.1 System Operation	II-5 TT-7
2.3 Facilities and Personnel 2.4 Cost Estimate	TI-16 TI-22
REQUIRED IMPLEMENTATION ACTIVITIES	II-35
3.1 Implementation Project Managemen	nt II-35
3.2 Agency Agreements	II-37
3.3 Funding	II-38
3.4 Operating Management, Personnel (Ind Training	II-38
3.5 Public Safety Communications Center (PSCC)	II-39
3.6 FCC Licensing	II-40
3.7 Procurement and Post Implementat Support	tion II-40

А	-	CURRENT SYSTEM	INVENTOR	RY	II-45
В		LEXINGTON DATCS	SYSTEM	REQUIREMENTS	II-49
C		FOREWORD	a the		II-59

ii

-----1.1 PROJECT BACKGROUND Ш part of a five phase program: Test and evaluation Î systems currently existing.

I. INTRODUCTION

For the past three years, the Association of Public-safety Officers (APCO) has been pursuing a program funded by the Law Enforcement Assistance Association (LEAA) to evaluate and implement a digitally addressed, trunked communications systems (DATCS)*. This program is the direct result of both the Federal Communications Commission (FCC) action to relieve the problem of frequency congestion in the publicsafety community and the LEAA's efforts to improve law enforcement techniques. The frequency congestion problem along with the FCC proceedings on this issue is addressed in FCC docket No. 1862. This document opened radio channels for "trunked" radio systems in the 800 MHz band to public-safety users. The LEAA recognized that the FCC action provided needed channels, and that trunked technology has the potential to provide enhanced radio service to public safety users. As a result the LEAA has funded Project 16 to further investigate trunked systems. This "Project 16B" effort has been undertaken as

Feasibility determination (Project 16A - completed)

Functional requirements development (Project 16A completed)

Implementation program (Project 16B - ongoing)

Technology transfer.

When APCO Project 16 was initiated the Lexington Police Communications Division was seeking ways to upgrade or replace the current mobile radio system. Members of the communications division monitored the progress of Project 16 closely to evaluate the possibility that DATCS could meet the city's police communications needs. As Project 16 evolved it became apparent that a DATCS system was a possible way to meet not only the needs of the police department, but also the needs of other city agencies that use mobile radio.

* The DATCS concept involves the sharing of radio channels by all mobile radio users in the City of Lexington. The number of user groups is larger than the number of channels in the system, and any of the available channels can be assigned to any user group as their need for communications arises, but no group will be able to listen to the conversations of other user groups. The sharing of the channel "pool" in DATCS is similar to the sharing of a number of long distance circuits in a telephone system. In a DATCS system, a large number of radio channels will be available to all users in Lexington instead of the several small isolated

The Mayor, recognizing the potential of DATCS, asked APCO to consider Lexington as a model city for developing a system implementation plan (SIP) under APCO Project 16B. To assist Lexington in developing this SIP, APCO provided the city with a consultant (Booz, Allen & Hamilton, Inc.) experienced in the design and implementation of technically advanced communications systems. The Lexington project director, Sergeant William Lisanby of the Police Communications Technical Unit with the assistance of Booz, Allen identified Lexington's mobile radio requirements. These requirements were analyzed along with the current system's ability to meet them. This analysis showed that DATCS is not only technically feasible but also the best solution to Lexington's mobile radio problems. A DATCS system concept for Lexington has been defined, reviewed, and favorably received by the City's users of mobile radio.

H

This system implementation plan describes the steps Lexington must take to procure, implement and evaluate the operation of a DATCS system. This document is a guide for the management team to use in implementing an advanced land-mobile communications system. It is being delivered to the city in draft form because it is a working document therefore, updated information will cause additions and changes to this document. Where options for implementation strategy are available, this document reflects strategies based on the best information available at this time. As a result of available information, the level of detail contained here varies.

1.2 PROJECT DIRECTOR AND PARTICIPATING AGENCIES

The Lexington project is under the direction of: . SGT. William C. W. Lisanby, Division of Police . Charles L. McGill, Division of Police

Participating agencies include:

Division of Police Division of Fire Fayette County Sheriff Fayette County Jailor Fayette School Security Public Works Park and Recreation Division of Housing Services Building Inspection.

1.3 SUMMARY OF CURRENT STATUS OF IMPLEMENTATION

The purpose of this section is to report progress to date in implementing Lexington's DATCS system concept.

To date, a DATCS system concept for Lexington has been defined, reviewed, and agreed to in principle by the users of mobile radio.

However, operational boundary issues will need to be resolved to obtain universal agreement on the PSCC system concept. To continue the project the City of Lexington should first agree in principle that the DATCS system concept is the way to resolve the city's mobile radio communications deficiencies. This agreement does not require that Lexington fund hardware at this time.

The next step will be to appoint an implementation advisory committee consisting of city commissioners. The commissioners would represent those city agencies that will be users of the system.

At this point, it will be necessary to form an implementation project management team. This team would include Lexington personnel skilled in specific areas of: procurement, finance, personnel, operations planning, communications system design and facilities design. This team should be formed as soon as possible so that project momentum from the planning phase is not lost.

On November 15, 1980, APCO's financial support for Project 16 along with the consulting support attached to it will end. At this time Lexington will need to find funds to continue the project. Immediate needs for funding are:

> Management team Continuing technical support

Continued funding for these areas will ensure Project continuity and allow for further progress in DATCS system implementation.

	This system c (Appendix B) will each agency. Cost included.
	2.1 SYSTEM OPERAT
	This section future mobile radi
	Current except a major de highly d
	Lexingto
	. Certain mutual i
	These elements of detail in the foll
	2.1.1 PHILOSOPHY
	The proposed ceived to closely improvements are a there exist certai system and that DA through technologi proposed operation
	overcome and that
\diamond	Lexington's m number of radio ch tions. Operations that fewer number
	that communication significantly incr
	<pre>Fayette Siban count</pre>
	in a conventional m "channels" exceeds in a conventional
	Preceding page hlank

II. SYSTEM CONCEPT

concept describes how each performance requirement be satisfied and the level of participation by c estimates based on the system concept are also

TION

presents the proposed system operation of Lexington's to system. It is proposed that:

operations be largely unaltered in the new system, as necessary to accommodate improvements to resolve eficiencies or new technology where found to be desirable

on develop a commonly owned and jointly managed safety communications center (PSCC) to serve on police, fire, and EMS communications

principles of operation be adopted to preclude Interference and maintain radio discipline.

the system operation are described in greater lowing paragraphs.

OF OPERATION

mobile radio system for Lexington has been conreplicate current operations, except where basic appropriate. The previous report* noted that in major deficiencies in the present mobile radio ATCS offers some highly desirable new capabilities ical advancement. It is the philosophy of the i, therefore, to suggest changes in present operation that the previously noted deficiencies will be new technology will be adopted, where applicable.

major mobile radio deficiencies are an inadequate nannels and insufficient interagency communicas will be changed therefore, only to the extent of users will be assigned to each "channel"** and ns among agencies and within departments will be ceased by the development of DATCS.

Mobile Radio System Deficiencies for Lexingtony Government, May 29, 1980."

ense used in DATCS is not equivalent to a channel mobile radio system. In a DATCS, the number of the number of frequency (pairs), which is impossible system.

Because DATCS also offers a few "highly desirable" features and options such as dynamic flexibility (reassignment of channels from a central location) and commonality of equipment, communications management will be able to respond to large-scale events of mutual concern in ways heretofore impossible. For example, in the event of a downed aircraft, selected public-safety units from Lexington Police, Fire, EMS, and Blue Grass Airport could be assigned to a common channel(s) for the duration of the event. Afterwards, each unit would return to its previously assigned channel. As a result, routine operations would be unaffected by the incident.

Similar flexibility is applicable to public works, for example, where snow plow activity would ordinarily disrupt day-to-day operations, but through channel segmentation made possible by DATCS, snow cleanup and removal could be coordinated separately. Also, because of equipment commonality made necessary by DATCS, mobile radio units among all city agencies would be unit-for-unit compatible and therefore interchangeable.

2.1.2 DEVELOPMENT OF A COMMONLY OWNED AND JOINTLY MANAGED PUBLIC SAFETY COMMUNICATION CENTER (PSCC)

At this time it is still unresolved whether Lexington will develop a commonly owned and jointly managed public safety communications center or will continue to operate complaint reception and dispatch individually and uncoordinated among Lexington Police, Fire, CD and EMS. For the purpose of the proposed system concept of operation, we have assumed that Lexington will elect to develop a communication center for public safety. Development of such a center is the only significant physical deviation from present operations recommended for the DATCS system.

This recommendation is based on the desirable objective to develop the universal telephone number 911 for all public safety. Also made possible by the development of central communications center is the development of computer-aided dispatch (CAD) in the future to assist in complaint reception, ranking of priority calls, management of field resources, and status monitoring of all the units. Development of 911 for Lexington is currently being pursued at this time, but not directly as part of this effort.

Development of a commonly owned and jointly managed public safety communication center will require the establishment of a separate management structure to oversee and operate communications for Lexington Police, Fire, CD and EMS. Planning should begin soon to develop a management structure acceptable to all users. In addition to operating the communication center, management could also have responsibility for radio maintenance for all agencies in Lexington. Establishment of a centralized radio maintenance capability would represent a large step towards eliminating the present fragmented arrangement for radio maintenance.

II-6

2.1.3 PRINCIPLES OF OPERATION

L

T

I

T

M

The principles described in this section are based on the assumption that complaint reception and dispatch for Lexington Police, Fire, CD and EMS and radio network control for all users will be conducted from a central location. The facilities required to support the operations associated with central communications are discussed in Section IV.

The principles of operation are expressed in terms of:

Group addresses or channels

Dynamic assignment of primary group address

location.

Each agency will be provided one or more unique channels or group addresses according to their specific requirements. The initial allocation of group addresses by agency is articulated in Section III - System Description. Mobile radio units will operate in a manner similar to today's operation, but there will be fewer mobile radio units per assigned channel. Only mobile radio units with a common purpose will share a particular channel. Mobile radios will possess generally one, but as many as five channels, with at least one channel capable of being dynamically reprogrammed from central network control. In general, supervisory persons will have more than one channel whereas line personnel will have access to only one channel, with the capability to change that channel on request.

Each agency with a dispatch location or access point will be equipped with all channels associated with its mobiles. In this way, each agency will be able to communicate with all of its mobiles regardless of which particular channel they maybe assigned. Each fixed access point will be also given access to the system operator (central network control) to make a request for change of channel.

The system operator will be responsible for maintaining discipline, managing the mobile unit changes of group addresses to preclude interference, and following agreed-to policies regarding the procedures to effect changes in the group address. In this way, the changes to an agency's radio configuration will not occur without the expressed approval of the particular agency. The dispatchers will not have the capability themselves to change group addresses. This will insure that dispatchers perform a dispatching function and not a computer operator function, which will be associated with the system operator.

2.2 SYSTEM DESCRIPTION

This section describes the future DATCS mobile radio system based upon Lexington's requirements. Specifically, addressed are

Accessibility to all appropriate channels by each dispatch

the performance requirements, external interfaces, control functions, radio spectrum and FCC licensing implications, and ancillary items associated with the central communication center. These are discussed individually in the following paragraphs.

៍ឲ

2.2.1 PERFORMANCE REQUIREMENTS

This section specifies the performance requirements that must be satisfied to implement a DATCS system in Lexington. Specific performance requirements that are discussed include:

DATCS capacity

Intra- and inter-agency communications

Mobile control head characteristics

Two-part digital address

Dynamically assignable group address

Talk-around capability

Interface with conventional portable radios

Radio coverage

Managment information

System protection measures.

2.2.1.1 DATCS Capacity

The DATCS system design must be able to support eventually 2,000 individual mobile and/or portable units. The specific number of frequency pairs that will be initially installed will be based upon the 70-percent-loading-within-five-years-of-licensing criteria established by the FCC. For example, if it is projected that there will be 365 mobile radio units on the air within five years, then seven radio frequency pairs will be developed, based on 75 mobile units per frequency pair corresponding to full loading.

The system must also be capable of supporting 150 unique group addresses with as few as one to as many as 100 individual units assigned to a specific group address. It is not clear at this time to what extent the interfacing of conventional portables can be considered legitimate loading for the purposes of meeting the FCC criteria.

2.2.1.2 Intra- and Inter-agency Communications

Intra- and inter-agency communications will be accomplished, depending upon specific circumstance, by one of three methods:

II-8

Limited Selection of Group Addresses - Supervisory level mobile radio users will be equipped with the group address of other users, in instances where access to other group addresses would be routinely required. This method provides selected users the option of operating on other group addresses without dispatcher initiative. It is expected that user-initiated access to other group addresses will be limited to supervisors and administrative personnel.

<u>Cross Patching</u> - Will be provided to most dispatchers. The dispatcher will be able to crosspatch two or more group addresses on an as-needed basis. In this way, the dispatcher can make or break a cross-patch arrangement, thereby maintaining channel discipline. Patching will be conducted generally among only those channels available to a particular agency; cross-patching between different public-safety agencies is not proposed.

Reassignment of Group Addresses - This method of interagency communications will be initiated only by network control personnel (system operator) when users need to temporarily become part of another group. It is anticipated that reassignment of group addresses will be used to respond to large-scale events of mutual responsibility and therefore, will probably occur infrequently. The reassignment of group addresses will be accomplished by one of two

Change in group address from a regularly assigned to an already existing group address of another user group

Creation of a "phantom" group whereby individual units would be assigned to an entirely new group address created to meet a temporary, one time need.

2.2.1.3 Mobile Control Head Characteristics

ways:

The mobile control head will be provided a variety of operational features, as determined by the specific needs of each agency. There will be two classes of mobile control heads:

> <u>Supervisory Control Head</u> - will provide a mobile radio user with up to five unique group addresses for interagency communications. Access to these group addresses will be user selectable. The primary group address, at the very least, will be dynamically reassignable, whereas the four other group addresses will be essentially fixed (but reprogrammable by maintenance technicians, if necessary).

Non-Supervisor Control Head - will provide a mobile radio user a single group address, dynamically assigned by the user's agency. If a user requires access to another group address, the user's dispatcher will request a change-ingroup-address status via the control operator or will cross-patch the user to the appropriate group address.

To improve dispatcher effectiveness, automatic reporting of user status* will be provided for Lexington police and optional for all other participants in DATCS. The status reporting capability will be incorporated external to the control head in a module specifically designed to convey status information; in this way, status reporting can be added or removed without major control head modification. A minimum of five status messages will be provided.

Occasionally the user may not be able to immediately access the desired channel because all channels will be temporarily allocated ir which case, a "channel busy" indication will be provided. The user will also automatically enter a queue set up by the controller to seize the next available channel, in the order of requests.

2.2.1.4 Two-part Digital Address

As a minimum, each unit will both be uniquely identified (unalterable) and identified as a member of a group address (alterable). The unit identification will be an integral part of the mobile unit, not readily changeable. On the other hand, the affiliated group address will be dynamically programmable to be able to satisfy changing needs.

The Lexington DATCS must have a capacity eventually for 2,000 individual mobiles and portables and 150 unique group addresses. Although current equipment offerings typically provide three levels of address (i.e., unit, subfleet, and fleet), no need exists to segment Lexington operations beyond the unit and subfleet level.

2.1.1.5 Dynamically Assignable Group Address

The capability will be provided to dynamically assign and change at least one group address of the mobile units from a central location; each agency will individually decide which group address (among field of five) will be dynamically assignable. However, this capability will only reside at a single location (network control) in the communications center. In this way, effective network control can be exercised.

The system operator, who performs the function of network control, will have available two options to dynamically assign a group address:

* Status reporting includes all typical responses frequently made by a unit to the dispatcher: in service, out of service, in car, out of car, on scene, acknowledge, etc.

II-10

Modify a particular unit's regularly assigned group address to an already existing group address (i.e., Streets and Roads to Traffic Engineering, etc.)

Create a "phantom" group, not regularly assigned, which will be dissolved when the need for its continued existance disappears; units will then return to the regularly assigned group address.

Changes in group addresses will follow firmly established operational procedures to maintain strict control; these procedures are best established jointly by the administrator of the PSCC and management board (composed of representatives of the user agencies (i.e., Police, Fire, Public Works, etc.)

2.2.1.6 Talk-around Capability

8

1 M B 1 A 1

Mobile units require communications independent of the base station facilities when beyond radio range or when the dispatcher is not involved in a message between close mobiles. This talk-around capability should be provided through a channel specifically set aside for the purpose of direct simplex communications (talk-around).

Talk-around will be mobile-user selectable as currently exists with the present radio systems. However, unlike the present systems, all units desiring to use talk-around must agree in advance to switch to talk-around; if only one unit switches communications will not be possible. While DATCS could provide automatic switching to talk-around under specific conditions (i.e., sense of "out-of-range" condition), manual switching is preferred to maintain predictable communications.

Although industry estimates vary for the introduction of trunked portable radio (3-6 years), there are no assurances that trunked portable radios will ever be available at an affordable cost or compatible with DATCS. Also, there is concensus that any type of trunked portables will not be available sooner than three years. For these reasons, a method is therefore required which will provide Lexington continued access to portable radio operations.

To provide for portable operations in the interim, the current existing conventional portable radio systems will be crossbanded with appropriate group addresses in the trunked systems; crossbanding will be under the control (enabling/disabling) of the appropriate dispatcher responsible for the particular group address. Conventional channels anticipated to be interfaced with DATCS are:

> Lexington Police (operations) Fire (operations)

2.2.1.7 Interface With Conventional Portable Radios

Public Works

Sheriff

Bluegrass Airport (public safety).

Because the possibility exists that all DATCS channels will be busy ("all channels busy") when a portable radio attempts to communicate over the DATCS system, the need exists to give priority access (first available channel) to the group address associated with the conventional portable. In any event, the portable communications will be received by the appropriate dispatcher regardless of the availability of the appropriate DATCS channel. This anomaly of an interface between conventional and trunked radio systems is a characteristic that can not be easily removed.

2.2.1.8 Mobile Radio Coverage

Reliable mobile radio coverage will be provided throughout Fayette County, at least to the same or greater reliability than currently realized by the existing Lexington Police mobile radio system. Quantitatively, mobile radio coverage reliability is proposed to exceed 95-97 percent, which is considered very good by industry standards.

It is not clear at this time whether more than one quality (high-elevation) site will be required to provided the required coverage. If more than one site will be necessary, then the Lexington's selected system engineer will be responsible for:

Indicating the need for the remote site(s)

Specifying which site and its necessary characteristics

Assisting Lexington procure the remote site (i.e., buy, lease, etc.), if necessary.

It is appropriate, however, to describe three alternative approaches to county-wide coverage, assuming that more than one quality site is necessary:

II-12

Simulcasting - mobiles receive identical radio communications from one or more sites simultaneously. This technique has the advantage that a mobile is assured of receiving a communication regardless of its location in the jurisdictional area. Conversely, there are disadvantages of high cost, technical complexity, and conspicuous use of frequency spectrum.

Separate Trunked Systems - mobiles would communicate with the trunked system associated with the mobile's specific location. This approach is simpler and less costly than simulcasting, but special measures are required to communicate over the entire jurisdiction simultaneously and to properly use the control channels.

Hybrid Trunked/Conventional System - same as the "Separate Trunked Systems" except a conventional radio system(s) would be used to cover remote, infrequently traveled areas of the county. However, FCC licensing eligibility is uncertain because of an absence of well-defined regulations.

2.2.1.9 Management Information

100

the second second second

Management information in the form of system performance data will be the primary tool to determine, on an ongoing basis, how well the DATCS system is meeting Lexington's mobile radio needs. System performance data will be provided to management of the DATCS in the following forms:

2.2.1.10 System Protection

One major intent of developing a DATCS for Lexington is to offer a mobile communication system which exhibits a very high degree of availability*, significantly improved over the present collection of individual systems. For example, loss of a single DATCS base station will only slightly reduce a mobile's chance for obtaining a channel, on request. Whereas today, loss of a single base station (or land lines, tower, antenna, etc.) completely inhibits further communications until repair.

In the event of the loss of a major component (i.e., controller or equivalent component), the DATCS system will revert to a conventional mode in which mobiles would be directed to pre-assigned channels. To accomplish this, the SOW will impose a number of system protection measures that must be satisfied to adequately insure Lexington against the consequences of catastrophic system failure.

Regarding alarm features, network control will be automatically alerted of subsystem and component failure. In this way, immediate action can be taken by the system operator, probably even before the field units notice any discernable change in performance.

Time-out timers will be provided on the mobiles to preclude inadvertent capture of the system for extended periods as well as on the base stations since unnecessarily long-duration conversations tend to deny channel access to other DATCS users. Long-duration conversations could potentially be a problem since the mobile radio user could falsely perceive that DATCS is a completely private, always clear-channel radio system which is unlikely to deny channel access to other users because of one's own usage.

2.2.2 EXTERNAL INTERFACES

This section describes the interfaces between DATCS and the systems external to DATCS:

1 Same

* Percentage of time the system is functional at full capability.

Telephone

Conventional portable radio

Computer-aided dispatch (CAD).

An interface between DATCS and the switched public telephone network will be established. As a minimum: administrative lines will be provided for the PSCC mangement and staff; 911* will be developed, if technically and cost feasible; and existing emergency telephone lines will be retained and re-terminated into the PSCC, at least until the public is familiar and confident in 911. A patch between DATCS and the telephone network is proposed at the system operator console.

Conventional portable radios are to be interfaced with the proposed DATCS system to enable portable radio users to communicate with DATCS mobiles, as necessary. A crossband connect between the DATCS and conventional repeaters, under the control of the appropriate dispatching agency, is proposed to accomplish this function. This solution is the least disruptive approach to dealing with a lack of trunked portable radios, for the interim. Each agency will decide individually to what extent their existing portable radio system will interface with DATCS on a routine basis; decisions may be different. For example, Lexington Police will probably elect to routinely cross connect whereas Bluegrass Airport may elect to cross connect only on an exception basis, because of differences in operations.

An interface with CAD will be established, provided CAD is developed for Police, Fire, CD and EMS in the future. If CAD is developed, then DATCS will interface in two ways:

- Use of automatic status reporting in the mobiles will apprise the CAD system of an individual's degree of availability to handle new complaints
- At the DATCS radio console, the CAD will provide the dispatcher the most current information on:
 - Complaints and their priority and status
 - Available resources to respond to new complaints.

Additional interfaces could be provided for the Sheriff, Jailer, School Security, etc. if the need arises during the course of further discussion and study.

* An independently conducted effort is expected to be initiated soon to solicit a proposal from GTE of Kentucky.

II-14

2.2.3 NETWORK CONTROL FUNCTIONS

-10

1

Network control represents the only other significant change in operations, besides development of the Lexington PSCC. However, unlike the development of the PSCC which is elective, establishment of a network control function is mandatory for DATCS. This is because monitoring system performance and managing a commonly owned, citywide mobile communication system must be centrally performed and cannot be distributed among several departments.

The responsibilities of a network control function will be divided into two categories:

Policy - responsible for: establishing guidelines for the use of DATCS, accommodating demands from various independent agencies impartially, and ensuring that day-to-day operations are appropriately performed

System Operations - responsible for the day-to-day operations of the network. The new or existing (system operations) agency would respond to the policies and guidelines established by the parent group. For example, system operations would handle requests for group address changes, but would not dictate under what circumstances a request for change in group address may be honored, since that is a matter of policy.

For further study, it is preliminarily proposed that policy be established by an all-new Management Board created to serve this specific function, comprised of representatives of the user agencies. System Operations, on the other hand, could be vested in an existing agency or developed independently. These choices warrant the closest attention.

2.2.4 DISPATCHING

Each agency that presently performs a dispatching function for itself or other agencies will be provided a dispatch capability under DATCS. As a minimum, individual agencies will be provided a desk-top type console containing all group addresses of responsibility and, in some cases, group addresses of mutual interest with other agencies/departments. These consoles will be capable of: radio patching among the channels of access; selecting the desired area coverage (if made necessary by multiple sites); communications with network control to request changes in group address; and receiving all messages from mobiles of responsibility. In addition, communications will be possible with at least two group addresses simultaneously; some agencies may have the capability to communicate with three or more group addresses simultaneously.

2.2.5 RADIO SPECTRUM AND FCC LICENSING

Presently, DATCS is only feasible at 800 MHz, which previously was a television-only band, but recently reallocated for mobile radio. For the foreseeable future, at least, sufficient 800 MHz spectrum will be available in the Lexington area to support DATCS.

Another favorable aspect is the current FCC demeanor which regards these kind of mobile radio systems with favor and something to be promoted. Therefore, if any negotiations are required with the FCC in the future to resolve regulatory issues (i.e., multiple base station sites, etc.) favorable rulemaking can be anticipated.

It is advisable, however, to begin establishing rapport with appropriate FCC staff to help "smooth the path" to successful licensing.

2.2.6 ANCILLARY ITEMS

Ancillary items for the purpose of the Lexington System Concept of Operation include the following:

Emergency Power

Maintenance.

Emergency power will be necessary for: the primary, downtown site; the remote site(s) in the periphery of the County (if necessary); and the CCC. Depending on the inherent emergency power capabilities of the downtown site, emergency power may not be required. However, emergency power will be necessary for both the remote site(s) and the CCC. Hydrocarbon-type fuel generators are proposed because of the substantial power requirements of each base station (1000 watts continuous each).

The maintenance strategy for the DATCS system in Lexington is two fold:

Initial Contract Maintenance - Initially maintain the DATCS system under a maintenance contract with a commercial shop, at least until a self-maintenance capability is established and sufficient confidence in the DATCS system has been achieved. This period will provide the primary basis for Lexington maintenance staff to acquire the appropriate new skills required.

2.3 FACILITIES AND PERSONNEL

This section describes the required facilities and personnel associated with the development of DATCS together with an all-new PSCC in Lexington.

2.3.1 FACILITIES

Lexington's physical facility requirements consist of two parts.

2.3.1.1 Public Safety Communications Center (PSCC)

This section describes the characteristics of the PSCC as well as provides estimated building interior square footage that will be required. The PSCC will possess sufficient space to accommodate the transfer of Lexington's current communications services as well as the new space requirements of DATCS. Lexington's PSCC will have the capability to service emergency calls via the emergency number "911", and to accommodate computer-aided dispatch (CAD) in the future*. Lexington's major public safety agencies of Police, Fire, Civil Defense (CD) and Emergency Medical Services (EMS) will be the primary agencies served by the center. The cost estimates, based on the square footage estimates included in this section, are provided in Section 2.4 and are intended for budgeting. A decision by Lexington as to whether to proceed in forming a PSCC should be based in part on these cost estimates. Many of the inputs used to estimate square footage for a PSCC have been provided by the Lexington Community Project Director (CPD).

Formation of a PSCC will enable Lexington to improve the delivery of emergency services to the public, minimize the cost of accommodating growth in services, and possibly lower ongoing costs. Lexington's public-safety agencies currently provide complaint reception and dispatch on an individual, agency-by-agency basis. By combining certain functions, such as complaint reception, system maintenance, and system management, and collocating dispatch, communications operations will be streamlined*.

The service PSCC are:

basis.

Public Safety Complaint Reception and Dispatch - will be 4.2 provided for all Lexington public safety agencies. The facility will be designed to eventually accommodate 911 as well as CAD. Introduction of 911 by Lexington for combined complaint reception will assure that citizens' request for assistance will be received by an operator who can offer assistance from one or all of Lexington's public-safety agencies. Moreover, overall response time for such centers has been shown to be reduced since confusion has been eliminated as to which agency to call and which responding agency is most appropriate. For calls involving multiple-agencies, savings in time will be dramatic. Future implementation of CAD will automatically route complaint information between the complaint operators and the appropriate Police, Fire, CD or EMS dispatch personnel, thus eliminating the manual complaint card system. CAD will also allow Lexington's public-safety agencies to adopt multiple zone dispatch with greater ease. CAD will route the call to the appropriate dispatcher or, if multiple agency services are needed, to the correct dispatchers. Development of CAD in Lexington will rqquire a separate planning effort. ** It is important to clarify that complaint reception is proposed to serve all public-safety agencies, whereas dispatch will be performed by personnel dedicated to a particular agency (e.g., fire). Other configurations are also conduct the initial dispatch, but these should be examined on a cases-by-case

II-16

The services that are proposed to be provided by the Lexington

Radio Maintenance, Citywide - will be provided for all agencies participating in DATCS. The inherent characteristics of DATCS, such as sharing of common equipment and compatible mobile radios make radio maintenance on a citywide basis highly desirable. System-level coordination of maintenance will be required regardless of whether Lexington decided to develop a PSCC.

Communications System Mangement and Administration - DATCS will integrate Lexington's uncoordinated, independent mobile radio systems into a coordinated, single mobile radio system. Presently each conventional system is managed on an individual system basis. However, the nature of DATCS a PSCC demands that management and administration of communications be centrally conducted. Therefore, for convenience, these functions should be located in the same physical facility. If a PSCC is developed the role of communications system management and administration would be commensurately increased. Since the proposed PSCC and DATCS concepts of operation are strongly linked, management should be located in the same facility.

The following estimates of the required square footage of building space have been prepared for budgeting and are, therefore based on a preliminary analysis of PSCC size, review of existing operations, and experience with similar systems. When a decision to proceed with development of a PSCC is made a more detailed analysis will be required. The analysis to date indicates that Lexington's PSCC must meet the following minimum functional requirements:

- Sufficient floor space for city-wide complaint reception, estimated at 560 square feet for 8 positions.
- Floor space for combined dispatch operations estimated at 1120 square feet for 8 positions.
- Floor space for radio maintenance estimated at 432 square feet for 3 work bench positions.
- Floor space for storage areas (spare radios, parts inventory, special equipment, etc.) estimated at 200 square feet.
- Floor space for installations and readio changes estimated at 1215 square feet for 3 vehicle bays.
- Administrative and overhead* office space to accommodate an estimated 6 staff members and support staff, requiring 1058 square feet.

II-18

* Closets, restroom, hallways, etc.

.

.

.

Sufficien sabotage.

•

.

man

outages.

These minimum requirements represent the concept to date. Clearly, additional analysis and planning will be required to move from the present concept of a PSCC to a smoothly operating PSCC.

Implementation of DATCS and a Public Safety Communications Center simultaneously is a worthy objective for Lexington. It is advantageous to combine the planning of DATCS and integrated dispatch and complaint reception at this time since many of the issues of DATCS are relevant to the PSCC.

2.3.1.2 Radio Sites

The unique propagation characteristics of radio signals at 800 MHz make selection of an appropriate site critical. A suitable site should be identified, prior to bid release, that will enable Lexington-Fayette County to receive extensive land area coverage from a single site. The key radio site facility to be selected must provide adequate floor space, limit unauthorized access, and be available for long-term lease or purchase.

Based on these site selection constraints the following specific requirements must be considered during the site selection process:

The site shall be located downtown, preferably on one of Lexingtons tallest buildings.

There must be sufficient floor space to accommodate at least 20 base station repeaters, control logic unit, spare units, and maintenance work area; 400 square feet of floor space is estimated.

The antenna elevation should be at least 400 feet above average terrain (AAT).

For security reasons the site should have limited access by the public.

A long-term lease to the site should be available and obtained soon.

An additional site(s) in the periphery of the county may be necessary to assure reliable county-wide coverage or, in the future, to provide reliable coverage for trunked portable radios, anticipated to be available in 3-6 years. However, whether additional sites will be required cannot be predicted at this time since the specific coverage realized by Lexington will depend on the specific vendor offering at time of bid. Since the acquisition of these remote sites can be appropriately made the responsibility of the vendor, a decision therefore on the necessity of a peripheral site(s) is not necessary at this time.

Sufficiently secure from unauthorized public access and

Backup emergency power to protect from commercial power

2.3.2 PERSONNEL

Personnel staffing will be critical to successful PSCC operation. Certain decisions regarding personnel are, however, not appropriate at this time. These are: 2 {

G-

10

- . Specific qualifications
- . Salary
- . Organizational structure
- . Sources for staffing.

Decisions in these areas are best made at a future time, nearer to PSCC implementation.

However, decisions concerning the management structure of the PSCC should be addressed in the immediate future. Lexington's existing pool of communications managers should be an integral part of the new PSCC management team. Accordingly, definition of a PSCC management structure will be one of the first planning elements of Implementation Plan (Section III).

Also known at this time is that:

About 61 persons will be needed to staff the center (shown in Exhibit II-1)

Most of the persons staffing the center will be drawn from existing communications personnel who are already qualified and therefore will require training only for new procedures/techniques associated with DATCS and combined dispatch.

Minimum staffing requirements for the PSCC are based on analysis of need, review of existing systems, and experience with similar systems. Personnel will be, in the long-term, the most significant ongoing cost item of Lexington's PSCC. The major personnel categories, including narrative description, are as follows:

- <u>Complaint Personnel</u> responsible for recording telephone complaints, routing complaints to the proper dispatcher and handling service calls not requiring public-safety agency response. Lexington will require an estimated 27 persons to staff complaint positions on a 24-hour, 7-daysa-week basis.
- Dispatch Personnel responsible for accurately disseminating complaint information to field units, and dispatching specialized equipment and personnel under emergency conditions. Eighteen persons are estimated to meet dispatch requirements on a 24-hour, 7-days-per-week basis.



EXHIBIT II-1 Proposed Lexington PSCC Personnel Requirements

CATEGORY			NUMBER
Complaint Reception			27
Dispatcher			18
Supervisors			5
Management		ý.	2
Administrative			4
Maintenance			5
			61

Supervisory Personnel - responsibile for supervising dayto-day operations of complaint, reception and dispatch operations. Five supervisory personnel are estimated, one per shift.

Management Personnel - responsible for overall PSCC operations by executing the directives and policies established by the Lexington Management Board or City Council, as appropriate. These responsibilities specifically include budgeting, staffing, adherence to procedures, maintenance, and productivity. Two persons are estimated to meet the management requirements.

Administrative Personnel - responsible for providing support services to management. Four persons are estimated.

Maintenance Personnel - responsible for the installation and maintenance of all DATCS and related electronic equipment for all city participants.

It is important to note that these staff requirements were derived for budgeting and therefore do not necessarily reflect an exhaustive list. Therefore, a key part of implementation will require that an accurate appraisal of job requirements be performed.

2.4 COST ESTIMATE

This section presents cost estimates associated with implementing a commonly owned digitally addressed trunked communications system (DATCS). Cost estimates for forming and operating a public safety communications center (PSCC) to meet the complaint reception and dispatch requirements associated with DATCS are also included.

These cost estimates present the best information available at this time. To arrive at these estimates a number of assumptions were formed, based on experience, discussions with the vendor community, and Lexington's present operations, staffing levels, and salaries. Many of the assumptions were provided by the Lexington Community Project Director (CPD).

These cost estimates will be adjusted as new information becomes available. Although hardware similar to DATCS is now commercially available, Lexington's requirements suggest that technically feasible modifications will be required. Therefore, pinpointing cost is difficult. Moreover, the vendor's degree of determination to pursue this market will, in the final analysis, determine the cost of DATCS equipment.

II-22

Costs are presented in two parts:

- Non-recuring costs one-time capital investment costs
- Recurring costs ongoing costs associated with operating and maintaining the DATCS system and operating the PSCC facility.

Radio Hardware

teria

Facility Construction

2.4.1.1 Radio Hardware

== 12 13

Fixed Site - one-time costs associated with the fixed site include: base station repeaters, tower structures, feedlines, antennas, transmitter combiners, receiver combiners, and system controller as well as installation costs. These costs are shown in Exhibit II-2.

.

A central downtown location will be selected, atop one of Lexington's tallest buildings, which will contain suffi-

cient indoor space to house all base station equipment.

Only mobiles will count toward channel loading, based on FCC criteria.

Seven hundred mobile units will be operational within five years of FCC licensing; twelve radio frequency pairs will therefore be developed.

Most equipment items are off-the-shelf, resulting in a high confidence level in cost.

Mobiles - one-time costs associated with mobile radio equipment include antennas, antenna feedlines, and installation, as well as the mobile radio itself and status package. These costs are shown in Exhibit II-3.

2.4.1 NON-RECURRING COSTS

Non-recurring costs are defined as one-time, initial costs associated with start-up. The estimated costs presented in this section are related to acquisition and installation of equipment and construction of the PSCC facility. Also included are one-time costs for maintenance equipment peculiar to the DATCS radio system.

Non-recurring costs are presented in terms of:

Fixed Site

Mobiles

Control

Conventional Portable Radio System Interface

Maintenance Equipment

The assumptions associated with Exhibit II-2 include:

A single site will be sufficient to provide reliable county-wide mobile radio coverage at 800 MHz.

Major Equipment Categories*	Initial Quantity	Cost (\$)
. Repeaters - \$9,500	12	114,000
. Tower Structure - \$5,000	2	10,000
. Antennas - \$600	4	2,400
. Transmitter Combiners - \$5,500	2	11,000
. Receiver Combiners - \$1,000	2	2,000
. System Controller - \$40,000	l	40,000
Total	N/A	\$179,400

1

T

Π

* Includes installation

1 -----

II-24

EXHIBIT II-2 Fixed Site Costs

 \mathcal{O}^{i}

[-----

D

		Initial Quantity of Mobiles				
	Agency	Status Reporting (\$300)	Supervisory (\$2300)	Non-Supervisory (\$2000)		
	 Public Works Engineering Sanitary Sewers Streets and Roads Sanitation Traffic Engineering Vehicle Maintenance 	0	15 10 5 5 10 9	1.0 20 20 23 10 0		
	 Lexington Police Fire Fire EMS Civil Defense 	425 0	445 90 6 0	0 0 0		
	. Sheriff	0	51	0		
25	. Jailer . Bluegrass Airport	0	0	0		
	. Fayette School Security	0	0	0 26		
	. Division of Housing Services	0	26	0		
	. Building Inspector	0	17	0		
	• Parks Department	0	11	39		
		425	706	148		



There will be two types of mobile units, which will be identical except with respect to the control head:

- Supervisor-Control Heads will provide for at least one dynamically reprogrammable group address and four fixed group addresses. All five group addresses will be operator selectable
- Non-Supervisory Control Heads will provide a single group address that can be dynamically reprogrammed.

The cost estimates are based on the present cost of off-theshelf trunked mobiles. This assumption seems valid because of the large quantity purchase and the favorable competitive environment. Regarding the status reporting equipment, the standard equipment offerings is suitable, independent of DATCS. Therefore, cost of status reporting is predictable.

Caution is urged in using the mobile radio costs in Exhibit II-3 because of the special nature of the public safety requirements. It is important to note therefore that the cost of each unit could be higher because of Lexington's special requirements, or could be lower because of the competition factor.

Control Equipment - one-time costs associated with control equipment are in two categories:

Control stations

Consoles for PSCC.

Control stations will be provided to agencies requiring access to their mobile radio units from a fixed location. These units, which are currently available off-the-shelf, will contain features comparable to the proposed supervisory-type mobile, only fixed in location. Each control station will have, as a minimum, access to all group addresses of responsibility and the system operator (to request change in group address assignments). In some cases, access will be provided to group addresses of common interest, dispite lack of direct responsibility.

Desk top, modular consoles will be provided the PSCC facility for Lexington Police, Fire, CD and EMS, as well as system control. Existing EMS consoles will be relocated at the new PSCC. However, the cost of relocating fire alarms to the PSCC if necessary, has not been estimated.

There will be two types of consoles: non-supervisory and supervisory (system operator). The non-supervisory console will provide for access of up to five group addresses, one of which will be dynamically reprogrammable. Interface of the conventional portable radio system with DATCS will be controlled by the appropriate dispatcher at a non-supervisory console. There will be one supervisory

TT-25

console/position which will have the capability to access all group addresses within the system, including those used by non-public safety. This position will be the only position capable of dynamically programming group addresses. This position will also receive alarm and system performance status information. Control equipment costs as well as management information system hardware and software costs are included in Exhibit II-4.

Conventional Portable Radio System Interface - non-recurring costs associated with equipment required to interface conventional portables with DATCS are shown in Exhibit II-5.

Conventional portable interfaces will initially be established for Lexington Police, Fire and Sheriff. For Police three channels will be interfaced: dispatch (operations), information, and detective bureau. For Fire and the Sheriff, one channel each will be interfaced.

Those Lexington agencies which are currently portable-only systems (Bluegrass Airport and Fayette School Security) will access the trunked systems through their own dispatcher; since no mobiles are initially planned for these agencies, a portable radio interface will suffice.

2.4.1.2 Maintenance

C. State

I

1

1

Contraction of the second s

There will be initial costs associated with developing a selfmaintenance capability, as shown in Exhibit II-6. Shown are the major cost items required to develop three maintenance bench positions.

2.4.1.3 PSCC Facility Construction

The costs shown in Exhibit II-7 reflect the minimum square footage requirements outlined in Section 2.3.1.1 - Required Facilities and Personnel, for the following operating areas in the PSCC:

Complaint Reception

Dispatch

Maintenance

Maintenance Storage Areas

Vehicle Bays

Administratative.

Construction industry figures for estimating the cost of these facilities per square foot, as shown in Exhibit II-7 were obtained from Lexington-area contractors and architects. The costs for the PSCC are the only major capital construction cost anticipated with DATCS implementation. A summary of non-recurring costs are shown in Exhibit II-8.

		Coi	ntrol Equipment	
Agency	Control Station \$4,000**	Non-Supervisory Control Console \$25,000***	Supervisory Control Console \$50,000***	Software, I Keyboard -
 Public Works Engineering Sanitary Sewers Streets and Roads Sanitation Traffic Engineering Vehicle Maintenance 	5 2 1 1 7 2			
. Cooperative Communi- cation Center*		5	1	1
. Sheriff . Jailer	1			
. Bluegrass Airport	2			
. Fayette School Security	1			
. Division of Housing Services	1			
. Building Inspector	1			
. Parks Department	3			
	28	5	1	1

.

* Lexington Public Safety Agencies Dispatch Police, Fire, EMS ** Includes antenna line and installation *** Includes installation

ł

0

II-28

Ð

 Γ

Ĥ



		Land Know High L				
	Numbe	r of Control Station	<u></u>			
Existing Radio System	Existing Conventional Channel	Interfaces - \$4,500 Pr Interface	Group Address	Total Cost		
Public Works	Operations	Not Required	N/A	N/A		
Police	Dispatch	1	. Dispatch	4,500		
9	Information	1	. Information	4,500		
	Administration	1	.Detective <u>Rureau</u> / Administration	4,500		
Fire	Dispatch	1	Dispatch or Fire Ground	4,500		
Sheriff/Jailer	Dispatch	1	Dispatch	4,500		
Bluegrass Airport	Dispatch Talk-Around	1	Dispatch Talk-Around Interagency Administration Maintenance	4,500		0
Fayette School/Security	Dispatch Talk-Around	3	Dispatch Talk-Around Interagency	13,500	EXH] Conventic Radio	
Services	Dispatch	Not Required	N/A	N/A U	BIT Int	
Building Inspection	Dispatch	2	Dispatch Interagency	9,000	II-5 Porta erface	
Parks & Recreation	Dispatch	1	Dispatch	4,500	e e b Le	
		12		\$54,000		

Q

Equipment Per Work Station	Quantity Needed	Total Cost (\$)
Service Monitor	3	\$21,000
SINAD Meter	1	186
Signal Generator	1	3,552
Audio Oscillator	1	306
AC Voltmeter	3	1,050
VOM	3	1,200
RF Multimeter	3	3,600
Scope	3	12,000
DC 12 Volt Power Supply	3	1,800
Wattmeter	3	300
Set of Elements	3	420
Dummy Load	3	390
Portable Test Set	3	1,500
Cable	3	180
Microwave Adapter	3	45
Universal Control Panel	3	750
Digital Test Set	1	4,275
Logic Probe	1	180
Spectrum Analyzer	1	15,000
Total		\$67,734

p---

21.44

thing -EXHIBIT II-6 Initial Radio Maintenance Equipment Costs

				f D		

 $\langle \hat{a} \rangle_{q}$

		The same set of the same set o	
Area Description	Square Feet (Cost per ft)	Square Feet	
Complaint Reception	55	560	
Dispatch	55	1120	
Maintenance	48	432	
Maintenance Storage	45	200	
Vehicle Bays	45	1215	
Administrative	55	1058	
Total			-

II-31

 \sim

\$

-

يودأ يعردهمك

Total Cost 30,800 61,600 20,736 9,000 54,675 58,192 235,003

EXHIBIT IT-7 Common Owned Com inications Center Facility Cost

, **6**2

 \odot

-



This section indicates total recurring costs related to operating the Lexington PSCC and DATCS mobile radio system. The recurring costs shown in Exhibit II-9 cannot be evaluated alone since many of these costs are already reflected in current operating budgets,

 \bigcirc

Dispatcher/complaint operator salaries

Maintenance Contracts

Recurring costs, shown in Exhibit II-9, are presented in the

2

Dispatchers

Complaint Operators

Maintenance Technicians

Management

Administration

In evaluating the recurring costs, it is important to consider the new capabilities gained and the improved ability to meet increasing demands for service as well as the elimination of redundant service. These cost estimates should be refined as more detailed information

II-33

 \odot

Major Recurring Costs	Quantity	Average Cost/ Unit*	Total Annual Cost	Current Cost	Par Cos
Personnel				0	-
 911 Operator Complaint Operators Maintenance Technicians Management Administration 	18 27 4 1 4	18,000 20,000 25,000 30,000 25,000	324,000 540,000 100,000 30,000 100,000		
. Support Staff Utilities	2	8,000	16,000		
. Telephone . Energy . Antenna Site		10,000 5,000	120,000 60,000 10,000		
Replacement Parts Overhead		10,000	120,000		
Total		· · · · · · · · · · · · · · · · · · ·	1,470,000	· · · · · · · · · · · · · · · · · · ·	

 \bigcirc

* Burdened

II-34

Station and the

2.1

ر از شرک rtial st New Cost X X X Х X X X Х х X ExHIBIT II-9 Estimated Annual Recurring х Cost Į.

	· · · · · · · · · · · · · · · · · · ·			
		₹ ¹		
		0		III. RE
				3] ТМ РТ. ЕМЕМПИЛИТО
	\mathcal{O}	a transmission of the second s		J.T. THE DEMERATIO
				The purpose o
				roles and responsi
			1586 7 67	— 1
				The managemen should be vested i
				advisory committee
				ority for planning
				Continuity be
				phase is desirable
				be appointed as sc
		a set a set of the set		a result of the pr
ϕ , where ϕ is the second se			3.	project will conti
	6		bar.au	
			and the second se	The project of tasks:
				. Work wit that def
	е Малария (р. 1997) Делания (р. 1997)			
			in the second	. Complete
			2	outline
				Define -
				resource
	4 	1		. Prepare towards
		<i>i</i>)		
				• Recommer
			f m	
				Staffing for
				project director of
				fixed hourly basis
				III I) chat need t
				. Finance
				budget n
				manageme
4				pudget.
\bullet			1200 200	
n <mark>E</mark> state of the			: bijai)	
			Same and the second	an a

EQUIRED IMPLEMENTATION ACTIVITIES

ON PROJECT MANAGEMENT

of this section is to discuss the actions Lexington nble an implementation management team, and the ibilities for members of the management team.

nt and direction of the implementation project in the project director, supported by staff and an e. The project director should have overall authg and directing the implementation management team.

etween the planning phase and the implementation e. Consequently, management staff should be drawn lved in the planning phase. These persons need to bon as possible so progress to date is not lost as roject being dormant for any length of time. Intments will also guarantee that momentum of the inue to build.

director will need to complete, at this time, five

th the DATCS advisory committee to draft a charter fines the members roles and responsibility

ely define the work that needs to be accomplished the implementation phase; this SIP will serve as an in defining the work

management responsibility and authority over the es needed to accomplish all tasks

time dependent schedules to ensure that progress implementation can be measured at all times

nd persons to fill positions in the implementation ent team.

the implementation team can be from Lexington's ol. All of the staff need not be assigned to the on a full time basis, but should be assigned on a s per week or month. There are six areas (Exhibit to be staffed:

- responsibilities include financial planning, ing a detailed schedule of financial milestones, management for on-going implementation activities, ent of grants and planning for the operations



	• Operation
	managemer
	responsil personnel
	Produceme
	package t
	vendor's vendor se
	. Systems I
	closely v
	DATCS Sys progress
	and accept
	Facilitie technica
	ments, bu
	3.2 AGENCY AGREEM
	The purpose of
	agency agreements a part of an agency a
	The At this time
	depending on Lexing
	mentation, funding these matters may i
	Commission regulat: which the shared or
	(Section (90.359).
	be part of the agree
	. Services
	J. Support j
	Facilitie
	to provid
	Funding of
	. Condition
	Authority
그는 것이 같은 것이 집에 가지 않는 것을 만들어 있었다. 것이 같은 것이 같은 것이 같은 것이 같은 것이 같은 것이 같이	

ns Planning - responsibilities will be to develop ment concept that will be expanded into a detailed nt plan as this plan will identify the roles, bilities, qualifications, and number of operations l needed

ent - responsibilities will be to develop a bid to be presented to the vendors, to evaluate proposals and make recommendations regarding election

Engineering - responsibilities include working with the technical consultant in designing the stem, preparing the FCC license monitoring the s of the installation and performing system test eptance

es - be responsible for working closely with the l consultant in identifying physical plant requireuilding design, and building security.

ENTS

of this section is to report the current status of and to recommend items that need to be included as agreement.

agency agreement have not been necessary and gton's approach in terms of participation, imple-, and management, individual agency agreements on not be needed. However, Federal Communications ions require a copy of a basic agreement under peration of a radio system will take place Therefore, a document that describes the roles es of users will be prepared. Items that should eement are:

to be provided to the participants

provided by the user agency to the system agency

es and/or equipment the system agency is obligated de user agencies

or cost sharing by agencies

ons under which the agreement can be modified

y of the project director.

C.3 FUNDING

The purpose of this section is to identify potential sources for funding, define the role/responsibility of management in securing funds and to outline a strategy of approaching funding sources.

The success of Lexington's implementation of DATCS will depend on the city's success in locating funds. These funds are needed for equipment, facilities, and on-going expenses.

Possible sources for one-time capital expenses are:

- State grants-in-aid Federal grants Grants from private sources Bond issues for capital expenses
- Gifts from private sources.

Possible funding sources for on-going costs are:

Taxes Shared revenue from state government Other municipal revenues.

The overall responsibility for securing funds will rest with the management team. This responsibility will involve the development of a strategy that will include:

Identifying sources of funds

Targeting key individuals in state and federal agencies, that will be approached for funds

Follow through on leads, commitments, and delivery.

Development of the funding strategy should be coordinated with the State Law Enforcement Communications Coordinator (LECC) as his active interest has been instrumental in the success Lexington has had in receiving funds.

3.4 OPERATING MANAGEMENT, PERSONNEL AND TRAINING

The purpose of this section is to identify the planning needed for system operation in terms of management, personnel resources, staffing, training and system manuals. It is not appropriate at this time to draft a detailed operations plan since many areas have not been defined, including the PSCC concept*. The level of staffing, staff qualifications and needed staff experience to implement PSCC concept of operation.

* This concept involves the combining of certain public safety communications features such as complaint reception, system maintenance, system management and collocating dispatch in such a way each division is in control of its respective operation.

II-38

To insure project continuity members of the implementation management team should be given first preference when staffing the operations management team.

Staffing of all positions will most likely be accomplished through the city's existing labor pool. Therefore, the staffing plan should be drawn up with the assistance of the city's personnel people. The role of the personnel people will be to assure that civil service requirements are met and city personnel procedures are followed.

The PSCC concept of operation will set the baseline for estimates of categories and quantity of personnel needed to staff the PSCC. This concept of operation will be followed up by a detailed personnel plan. The personnel plan will include organization and information flow charts that detail the categories and quantity of persons needed, as well as the qualifications needed for each staff position. The major source of personnel is the existing labor pool. It is desirable for the city to draw on these experienced people for staffing.

G

A training plan will need to be developed using the PSCC concept of operation, the DATCS system concept, and information provided by the vendor of choice for each piece of hardware. The training plan will be in two parts:

Operations Maintenance.

 \Box

L_)

These plans will address both initial training and refresher training. It is best to have the vendor develop training plans under the guidance of a member of the management team since the vendor is familiar with the equipment being supplied as well as the installation. The training manual will be the operating procedures manual, prepared by the vendor in accordance with the equipment purchase and the system installation with significant input from Lexingtons management team. It is expected that the procedures manual will be updated as the system is put through the test and acceptance phases. The manual will be further updated as hands on experience is gained by users and as the system configuration changes. Maintenance training also will be conducted by the vendor. Ideally, maintenance personnel should participate in regular service training courses given by the vendor factory representative. Training will be ongoing due to personnel replacement and the need for regular refresher courses.

N 11 +

The purpose of this section is to point out the progress to date on planning and implementation of the PSCC. At this time Lexington needs to develop a concept of operation and an implementation plan for a PSCC. There are three areas that need to be addressed:

> Facility "911"

3.5 PUBLIC SAFETY COMMUNICATIONS CENTER (PSCC)

Computer Aided Dispatch (CAD).

The Lexington Project Director will need to define the requirements, concepts, and implementation plan for the PSCC. Assistance from an outside profession 1 firm experienced in system and facility design will be desirable at this point. This assistance will assure Lexington that the requirements and concept of operations will precisely match the city's needs and industry offerings.

The time table for completion of the PSCC is uncertain at this time. The date of completion will be dependent on Lexington's start of development of the PSCC requirements, availability of funds, rate of hardware development, the completion of physical plant facilities, systems installation system test and acceptance.

3.6 FCC LICENSING

A signed contract defining the equipment to be purchased and the quantity of that equipment will provide sufficient information to file applications with the FCC for system licenses. Present FCC practice for licensing 800 MHz trunked system requires the following minimum submission:

Form 400 for each license requested (Section 90.357)

- Agreements with each user agency relative to control of the system (Section 90.179)
- . Agreements with each user agency relative to cost-sharing of installation and operation (Section 90.179)
- Financial plan showing the serivce will be provided to user agencies at cost (Section 90.359)
- System description and specification (Section 90.359, Section 90.371, Section 90.375, and Section 90.379)
- Eligibility statement for each user (Section 90.359)
- Evidence of purchase orders placed for the equipment.

It is probable that FCC waivers will have to be asked to extend the number of years for phased implementation of the system (Section 90.375).

Technical support for negotiating system licenses with the FCC will be requested for the engineering consultant and from the system contractor.

3.7 PROCUREMENT AND POST IMPLEMENTATION SUPPORT

The purpose of the section is to present a "theoretical" time table for procurement, discuss the current status of the Statement of Work (SOW) and articulate post implementation support activities. At this time there are still too many uncertainties to define an exact procurement time table. However, a time table has been prepared that shows a "wish list" for start and completion of each task necessary for procurement of an advanced communications system Exhibit III-2. Facility construction milestones, along with 911 and CAD milestones are shown on this time table to the extent that they effect implementation of DATCS (i.e., the construction of the PSCC must be completed before installation of the DATCS, 911 and CAD hardware begins).

 \square

63

A functional SOW has been prepared and is contained in Appendix C. The major elements of a high technology SOW are described therein.

During the implementation phase post implementation support activities will be defined in terms of continued funding, personnel needs, training and system evaluation. This role will fall to the implementation team members as appropriate.

Continued funding sources should also be defined. Recurring costs usually cannot be funded by State or Federal sources and the city will likely have to pay these costs from its operations budget. These costs will include: personnel facilities maintenance, equipment maintenance, equipment replacement, leases or rents, utilities, supplies, and training.

Post implementation evaluation is usually required for projects in which Federal grant funds are used. Evaluation is recommended for those systems not funded by federal grants. This evaluation plan should be prepared during implementation the plan will detail the data to be recorded, method of data collection, procedures for measuring data when the data is to be recorded, and evaluation report format and distribution.

The evaluation report should be results oriented. The report will include both system financial effectiveness and management effectiveness. This report, if thoughtfully constructed, can be an important management tool. In addition, the report can significantly enhance the city's position when applying for additional Federal funds.



EXHIBIT III-2.1

 \prod

 \square

4

 \square

Preceding page blank

 $\mathcal{L}_{\mathcal{L}}$

Task 1 - Appoint an advisory committee, appoint members of the management team, secure outside technical assistance, prepare agency agreements.

Task 2 - Develop and implement a funding strategy, have commitments for funding of the system.

Task 3 - Document and analyze, PSCC, 911 and CAD requirements, develop a concept of operation from the analysis and a system design concept for advisory submission to the FFCC.

Task 4 - Develop a complete statement of work for DATCS, PSCC, 911 and CAD.

Task 5 - Revise the final system concept to correspond with FCC regulations.

Task 6 - Solicit bids for the system, conduct bidders conference, receive bids.

Task 7 - Proposal evaluation and contract award.

Task 8 - Equipment ordering, system license, PSCC physical plant complete.

Task 9 - Equipment Delivery Installation of DATCS, 911, and CAD

Task 10 - System test, acceptance and personnel training

Task 11 - System cutover and evaluation

15

1.3

 \square

APPENDIX A

CURRENT SYSTEM INVENTORY

Lexington's present mobile radio communications systems consist of the following seven separate, individual operating systems and one system in development:

> Metro Police Fire Department Fayette County Sheriff's Department Public Works

Fayette County School Security

Bluegrass Airport

Emergency Medical Services (EMS) Lextran (proposed bus system) .

Exhibit A-1 summarizes the current situation of each radio system. Exhibit A-2 shows the current radio equipment distribution in Lexington.

II-45

:1

	System <u>Title</u>	Number of Channels	Number of Mobiles	Number of Portables	Number of Control Stations
	. Police	4	340	134	4
	. Fire	2		1 1 - 9 - ¹ 1	na de la construcción de la constru La construcción de la construcción d La construcción de la construcción d
	- Fire		84	65	15
	- EMS	*	6	0	
	- Civil Defense		0	õ	15
			90	65	30
	1. (1997) - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -			•	
	. Sheriff	3			
	- Sheriff's Department		44	16	• 1
	- Jailer		6	5	1
			50	$\frac{3}{21}$	<u>-</u>
					4
Н	. Public Works	1			
—	- Engineering		24	Ä	
46			24	U	5
	- Sanitary Sewers		22	8	2
	- Streets & Roads		24		1
	- Sanitation		28	V 0	· · 4
	- Traffic Engineering		16	2	7
	- Venicle Maintenance		5	0	1
	- Parks & Recreation		45	5	4
	- Housing Services		18	· 0 · · ·	1
	 Building Inspection 		_13	0	2
			195	16	27
	. Fayette School Security	7 2	0	28	1
1 (.					
147 ⁴ - 14	. Bluegrass Airport	2	n an an an Arthur An Anna Anna Anna Anna Anna Anna Anna A		
	- Police		2	5	0
	- Fire		3	1	1
	- Administration		0	1**	Û.
	- Maintenance		4	Õ	Õ
			9	$\frac{-3}{7}$	<u> </u>
	TOTAL	14			
	* Does not include number ** Vehicular mounting	er of EMS cha	nnels		
	Contraction of the second s				

 \wedge

Total Number of Units O P EXHIBIT A-1 Lexington's Present Mobile Radio Communications Systems


APPENDIX B

LEXINGTON DATCS SYSTEM REQUIREMENTS

FUTURE MOBILE RADIO REQUIREMENTS OF INDIVIDUAL AGENCIES

The information presented in this appendix is based on interviews with key department heads. The interviews focused on each agency's future mobile radio needs in terms of the question: What minimum capabilities and features of a mobile radio system are necessary to support your present and future business function and to what extent would additional capabilities enhance your department's effectiveness or safety of personnel?

In summarizing the finding of the interviews, the future mobile system* should:

Be adaptable to allow for growth in services as necessary to accommodate a rapidly expanding population (Exhibit B-

Be flexible since the exact growth in any specific agency is difficult to predict

Provide for a high degree of interagency communications to an extent not realized today

Immediately provide additional channels to relieve channel congestion

Serve as a management tool for monitoring the status of all resources and be able to deploy resources when necessary, thus enhancing departmental efficiency and, in the case of public-safety, also enahncing officer safety.

Exhibit B-2 depicits a summary of Lexington's future requirements in terms of number of channels, mobiles, and portables for each agency. The future requirement of each agency addressed in this section encompass immediate needs as well as future needs projected for Lexington-Fayette Urban County within the next five

The present Metro Police radio system is dependent upon a single channel to dispatch all city and county patrol units as well as to communicate instructions to non-patrol personnel (i.e., detective, juvenile, etc.). Because the number of mobile units on this channel is already large (340 mobiles and growing), the future Metro Police radio system should be capable of segmenting police communica-



	Agency	Number of Channels	Number of Mobiles	Number of Portables	Number of Control Stations	Total Number of Units			*
	- Police	5	445	284	5	734			
	. Fire								
	- Fire - EMS - Civil Defense	4 0 _0	90 6 0 96	55 0 0 55	17 1 <u>15</u> 33	162 7 <u>15</u> 184			- - - - - - - - - - - - - - - - - - -
	. Sheriff	2	51	16		68			
	. Jailer	<u>1</u> 3	<u>0</u> 51	<u>20**</u> 36	$\frac{1}{2}$	<u>21</u> 89			
	. Public Works								
II-51	 Engineering Sanitary Sewers Streets & Roads Sanitation Traffic Engineering Vehicle Maintenance 	1 1 1 1 1 <u>1</u>	25 30 25 28 20 9	4 2 0 7 2 0	5 2 1 1 7 2	34 34 26 36 29 11			
	. Fayette School Security	1	6	37	1	170	E Lexing Five Y		
	 Bluegrass Airport Police 	1	0	8		10	XHIBIT ton's ear Re		
	 Fire Administration Maintenance 	1	0 0 <u>0</u> 0	4 1 - <u>4</u> 17*	2	4 1 <u>4</u> 19	B-2 Projecta quiremen	0	
	 * To include vehicular ** 12 are vehicular uni 	charges ts	e	0			2 4 G 5 9 G		

• :

Agency	Number of Channels	Number of Mobiles	Number of Portables	Number of Control Stations
. Division of Housing Services	1	26	0	1
. Building Inspection	l	17	- 1	1
. Parks and Recreation	<u> </u>		5	
TOTAL	24	828	450	66

II-52

[]]

[]]

 \Box

 \Box

[]]

0

Total Number of Units	
27	
19	
58	
1344	

EXHIBIT B-2 (Cont'd) Lexington's Projected Five Year Requirements

 \square

C a

	a and a second sec	
		. <u>Three sec</u>
ρ		. <u>Non-patro</u> traffic h
		. <u>Tactical</u> single ev
		Presently, an (i.e., 10~7 10-8, e
		communications from system should have
		the status of each will become critica beyond the human at
		techniques. Presently, the
		approximately one p of concern for offi ency, there is an one with selected non-
		radios by Lexingtor it is reasonable to limiting factor on
		administration is t almost double maint ing this cost diffe
\mathcal{G}		At present, th bility between Metr agencies. However
	1	communicate direct
		. Fayette (Jailer . Airport H
		. Fayette (. Fire . Emergency
	and a second	The future mobile munications between
		determines that mut
		States, a management large-scale events plane crash, hostage
		communicate direct EMS, Airport Police for managing all re

ctor channels, one channel for each sector

and the second line and share the second statement with a

ol operations channel, to serve detective bureau, bureau, administrative, etc.

channel, to localize communications concerning a vent.

officer reports status information verbally etc.), often in competition for air time with more . To separate low priority status reporting m more urgent, priority dispatch, the future the capability to automate status reporting, utton. In this way, each dispatcher will realize unit through a visual indication. This capability ally important as Metro's fleet size expands bility to manage resources by conventional, manual

e Metro Police force has 130 portable radios, or portable for every three patrol officers. Because icer safety and demand for improved police efficion-duty patrol officer, and in many instances, patrol personnel. Growth in the use of portable n from 55 in 1975 to 130 in 1980 demonstrates that o expect 200 to 250 protable radios by 1985. The their complete adoption for patrol by police their 25 percent cost premium over a mobile and tenance cost, although the trend is toward lowererential.

here is no mobile-to-mobile communications caparo Police and other Urban County public-safety , there is a definite need for Metro Police to ly with:

County Sheriff

Public-safety County School Security

y Medical Services (EMS).

radio system should permit mobile-to-mobile comn Metro Police and these agencies when a dispatcher tual coordination is required.

municipal mobile radio systems in the United nt structure does not exist for dealing with involving multiple public-safety agencies (i.e., ge situation, etc.). Even if Metro police could ly with the Sheriff's Department, Fire Department, e, etc., a mechanism is not presently in existence esources at the scene. Accordingly, the future

system should posses a management structure which permits a single entity (i.e., Emergency Operating Center (EOC)) to coordinate all resources at the scene, without interfering with each agency's discretionary influence. Feasibility of establishing such a structure is partly technical (availability of a clear channel for coordination) and partly procedural (articulation of policies). If Lexington evolves toward an EOC-type arrnagement in the future, both technical and procedureal issues must be thoroughly considered during the planning process.

1.2 Fire Department

The present Fire Department system utilizes a single dispatch channel and a single fire ground channel, neither of which provides interagency communications with other public-safety agencies. The future communications system should also provide a single dispatch channel, but must be capable of several "fire ground" operations simultaneously. In this way, several fire events can be managed independently, without interference. While no firm requirement was expressed for a common coordination channel with Metro Police, the future communications system should provide the capability to directly communicate with Metro Police and other public-safety agencies when the need arises, as determined by a dispatcher.

In the same manner as Metro Police, the Fire Department is also trending toward increased portable communications for "fire ground" operations and the conduct of routine inspections. As a result, the future system must accomodate possibly 100 percent portable operations. High initial purchase and ongoing maintenance costs are still considered the primary limiting factors to increased adoption of portables.

1.3 Fayette County Sheriff and Jailer

The present Sheriff's Department radio system mutually serves the Sheriff's Department and the jailer via three channels, although dispatch and alerting are performed on only one of the three channels. The future radio system should provide separate channels for jailer and sheriff operations, but with the capability of switching between channels for interagency communications. Because of the need for frequency assistance, the sheriff and the jailer should also be equipped with the primary channels of the following agencies in order of priority: fire/EMS dispatch, Metro Police information, and Metro Police portable radios for the jailer, the future mobile radio system should accommodate multichannel portables.

1.4 Public Works

The Public Works system presently utilizes a single-channel radio system to support the services of nine separate divisions, six of which are divisions of Public Works. The future Public Works radio system will provide a radio channel dedicated for each division of Public Works will be accommodated separately in the future system. In this way, for example, communications among the Streets and Roads Division will not interfere with Traffic Engineering.

II-54

Because there is frequent need for divisions to intercommunicate at the supervisor level, selected units within each division should be capable of communicating directly with units in any division on an "as-needed" basis; non-supervisory units, on the other hand, should be capable of communicating with units only within their respective division.

Each division headquarters should be capable of communicating directly with not only their own respective units, but also with units/headquarters of all other divisions within Public Works.

The addition of several operating channels will have the clear effect of reducing the time required to access a channel, minimize user frustration, and encourage each user to be more attentive toward each radio message since most irrelevant communications will be eliminated.

To enhance channel discipline and insure accurate understanding of each unit's identification, every unit will automatically transmit the identification of the unit upon each transmission.

1.5 Airport Public-safety Agencies

The current airport radio system does not provide radio communications directly with Urban County public-safety agencies. Consequently, operations requiring mutual aid are hampered by the lack of coordination capability. The future system will provide radio communications between the airport and Lexington public-safety agencies when required, particularly Metro Police, Fayette County Sheriff, and the Fire Department.

Because of the nature of airport operations, portable radios will continue to be the dominant means of communications. The future radio system must provide multichannel portable radios if the airport is to participate fully. If portable radios are not available, then a public-safety radio interface must be established at the Airport Control Center.

1.6 Fayette County School Security

 \square

-

School Security is proposing an additional shift to meet the need for night security. Security officers added to the existing department will each require a portable radio. Presently, school security dispatch has direct radio communications with the Fire Department on Fire Channel to provide quick response to fire and first aid calls reported to School Security Dispatch; an interface with fire must be continued, referably to the extent that field security could report an incident directly to fire. The future system should also provide an interagency communications capability to field personnel on an "as-needed" basis; contact with EMS and Metro Police are the most critical.

1.7 Division of Housing Services

Presently, Housing Services shares the use of the Public Works radio system as a matter of convenience to Housing Services. The future radio system should provide communications not necessarily with Public Works, but rather with several other agencies with whom Housing Services have expressed a need to communicate directly:

- Division of Building Inspection
- Fire Inspection Division, Fire Protection Bureau.

A channel dedicated for Housing Services should be provided. Future plans to improve communications within Housing Services may have to be approved by Community Development, within Housing Urban Development (HUD) if HUD is to finance a prorated share of the conversion costs.

1.8 Division of Building Inspection

As with Housing Services, Building Inspection also shares the use of the Public Works radio system. The future radio system should provide communications with only selected divisions and a few public-safety agencies with which day-to-day interfaces exist, as listed below:

- Public Works, Division of:
 - Engineering
 - Traffic Engineering
 - Metro Police (dispatch)
- Fire (dispatch)
- Housing Services.

1.9 Parks and Recreation

Parks and Recreation also presently shares the Public Works radio system. The future system should provide a channel dedicated to Parks and Recreation with the additional capability to directly communicate with the following Divisions of Public Works on an "asneeded"basis:

. Streets and Roads	•	Streets	and	Roads
---------------------	---	---------	-----	-------

- Sanitary Sewers
- Sanitation.

5

The future operation is anticipated to continue to be assisted predominately by mobile radio.

2. FUTURE MOBILE RADIO SYSTEM REQUIREMENTS

The preceeding section described the mobile radio requirements of individual agencies. Requirements also exist at the system level. This section describes those system requirements which are critical to each agency's operation, but which have been developed based on a broad perspective of the overall operation of all agencies combined.

II-56

The following system requirements, which fall into several categories, should be second regardless of whether the city decides to refurbish existing equipment or develop a commonly owned, city-wide communications system:

Overall communications system management responsibility must be defined.

System management guidelines, controls, and operational policies should be established for the next generation system(s)

Configuration

Management

-

Addressing

e (

*

The system must be initially capable of providing at least nine dispatch points, geographically distributed, with several dispatch positions at some locations

The system must be expandable to include new user agencies

Land Area Coverage

The system must provide radio coverage throughout Fayette County, including all fringe areas

The system must be expandable to provide portable radio coverage, if required

The system must be capable of supporting a minimum of 2,000 mobile and portable units

The system must provide for at least 30 individual "channels"*

The system must be software adaptable to include or exclude specialized groups or discrete units or to recognize the group addresses

Performance

The time to access a channel by any user should not exceed 0.5 seconds under normal busy hour traffic conditions

The system must permit any group of mobile units to be transferred to any dispatch position within the system

Not necessarily ef channels.

Voice privacy should be enhanced to the extent practical

Interconnectivity

The system should be able to interconnect with the public telephone network

- Action

- The system should permit the use of conventional portable radios

Reliability

20

13

- Redundancy of key facilities will be provided
- Failed equipment will be automatically indicated at a central control facility.

The recommended elements of the Statement of Work are described as a major part of the System Procurement subsection in Section III of the SIP Guide. This appendix is intended as a topical outline of a typical Statement of Work (SOW) for quick reference by the planner or system procurement manager. The SIP Guide provides supporting material for this topical outline.

The following outline adheres to the recommended order of topics for a high technology system SOW. The planner or system procurement manager is cautioned to employ only those elements of the outline which relate to the specific procurement and which can be accurately evaluated or measured.

APPENDIX C

FORWARD

II-59

والمحاج المحاج والمحاج المحاج المحاج المحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج والمحاج	a da anti-anti-anti-anti-anti-anti-anti-anti-		
na sense de la constante de la Constante de la constante de la Constante de la constante de la	Q		
n en	i i i i i i i i i i i i i i i i i i i		I. <u>INTRODUCTION</u>
			The Introducti a general overview
			by the responders. in the following su
			1. Backgroun
			A brief h
			curement and t
			2. <u>Objective</u>
			A brief d achieved by th
			3. Scope of
		4 .	A basic a
			services that example:
			. Hard
			. Soft
			. Syst
			. Inst
			. Test
			. Trai
			. Main
			II. INSTRUCTIONS T
			This section o information needed
			1. Procureme
			The sched
			request will i
			, Time
			. Time
			and nave blank
			Algernme have

STATEMENT OF WORK

ion to the Statement of Work (SOW) should provide of the project to enable a better understanding The key elements of this appendix are outlined ubsections.

nd of the Project

historical description of the origin of the prothe constraints imposed on the design.

of the Procurement

discussion of the main system objectives to be he procurement.

the Procurement

and brief outline of the hardware, software, and are being purchased. These may include, for

lware

ware

em Design

allation

ing

ning

tenance.

O BIDDERS

of the SOW will provide the bidders with all the to respond correctly to the bid request.

nt Schedule

lule time and dates for responding to the bid .nclude:

and place of the pre-bid conference

and place bids and proposals are due

	. Bid and proposal opening time, date, and place	51	M	5. Proposal Eva
	. Proposed contract award date			Criteria sho
	. Expected date of contract completion.			Technik
	2. <u>Authorized Contracts</u>			. Technic
	The persons who are authorized to respond to bidders' requests will be identified prior to bid opening. These may be restricted to the:			. Cost p.
	. Procuring officer			These crite that each bidder
	. Technical consultant.	57	TI	requirements of
	The statement should be made regarding how the issues raised by these contacts will be reported to the other bidders.			6. <u>Contract Aw</u> The SOW wil
년 11월 - 11일 년 11	3. Bid and Proposal Requirements	8.		the successful b awarded without
and the second	Each of the items to be submitted as part of a responsive bid will be defined. Any limitations should be included in the			7. <u>Indemnifica</u>
	description. These items would include, for example: . Itemized cost proposal			The purchas damages imposed the contractor.
and the second secon	. Technical proposal		1	processes. All
	. Management proposal			fication from da employees or age
	. Provisions for alternate proposals			8. Titles and
	. Proposal limitations regarding page count, standardized forms (cost proposal), proposal size, unnecessary artwork, and number of copies to be submitted.			Rights of t clearly defined related to:
2.	4. Qualifications of the Bidders			. The tr
	The contractor for a high technology system must be qualified, technically and financially, to complete the required work. These qualifications should be expressed in the SOW by			. Rights part o
	Bidders' facilities and financial resources	A.R.		. Rights
- and the second	. Performance history in terms of references and a list	¥1		. The ri
	of similar and recent contract work	T		. The ri
ſ	. Support facilites such as field maintenance shops, training facilities, and spare parts replacement depots	u 1 1		. The ri
o construction of the second se	. The quality and availability of personnel for manage-	<u> </u>		and wording of t
	ment, supervision, and technical staff.			
9 	II-62	Ĥ	ŕ	
		L		

valuation Criteria

ould be established for the comparative evalua-

cal proposals

roposals

ement proposals.

eria should be clearly defined in the SOW so will understand the emphasis placed on the the RFP.

vard

Il identify the type of contract to be awarded bidder whether or not this contract will be pre-contract discussions.

ation

ser should require indemnification against through intentional or unintentional acts of High technology system procurements will need against the improper use of patented items or procurements which include services such as I test will be well advised to require indemniamages resulting from the actions of contractors' ents.

Rights

the purchaser and ownership of titles will be by the SOW. These items include statements

ransfer of title to equipment and software

s to data and software developed or provided as of the contract

s and title to the submitted proposals

ight to reject proposals

ight to negotiate with any bidder

ight to cancel negotiations or the contract.

sel should be consulted for the scope, content, this subsection of the SOW.

9. Bonds, Insurance, and Warranty

Each governmental entity will have its own limits for, and description of, the bonds, insurance and warranty provisions for contractor provided material and services. Frequently there will be state regulations which also must be adhered to. These requirements will provide for:

Performance bonds

Workman's compensation

Comprehensive liability insurance

Warranty on workmanship and materials.

The purchasing department of the major governmental agency included in the program will provide counsel as to how these requirements are to be included in the SOW.

10. Costs of Proposal Preparation

The SOW should state clearly that any costs related to the preparation of bids or proposals can not be included in the proposed cost of the project or billed back to the purchaser in any manner.

11. Contract Disputes

At the option of legal counsel, the SOW may indicate the means by which the governmental agency resolves contract disputes. This will be a function of the governmental body and will not be an option of project management. These provisions may include:

The procedure and method of resolving contract disputes

Termination of contracts for cause

The assessment of liquidating damages

The identity of the court of jurisdiction.

12. Applicable Federal/State Laws and Regulation

When State or Federal laws/regulations affect the system design, its operation, or allowable use, these laws or regulations should be identified in the SOW.

III. SCOPE OF WORK

This section of the SOW defines the items of work expected of the contractor; the specific tasks required by the contractor are described in detail.

II-64

Services to be Provided

1.

2.

3.

High technology systems will require professional services to be performed by the system contractor in addition to the delivery of hardware and software items. Typical services to be provided are:

System design which can include system engineering, radio propagation engineering, software development, and hardware development.

Installation and debugging of hardware and software.

Assistance in procuring required licenses and permits.

Developing and conducting performance tests and acceptance tests.

The preparation of sites and facilities which can include construction, HVAC, emergency power, lighting, installation of utility services, etc.

Deliverable Items

Each deliverable item should be identified in the SOW. Deliverable items include hardware, software, and services. The list will be unique to the complexity of the system to be procured. However, there will be some items common to all lists such as:

Each item of electronic or radio hardware

Each item of computer software

Maintenance support

Spare parts and spare module provisions

Towers, buildings and other structures

Manuals and documentation such as test plans, test procedures, maintenance manuals, operators manuals, user manuals, training aids, installation drawings, and test reports

Progress reports and deficiency reports.

Delivery Schedules

Each deliverable item should be supported by a delivery schedule. The SOW will require a formal and contractually obligated schedule for:

Each item of electronic or radio hardware

Each item of computer software

Each tower or building

The completion of all facility preparations

Delivery of all manuals and documents

The time and place of technical progress reviews

The delivery of reports including progress reports, test reports, discrepancy reports, and the final report or acceptance test report.

IV. SYSTEM SPECIFICATION

The system specification is a complete technical system description which includes the system performance, system interfaces, and system configuration. It places technical limits on these parameters and adds reliability and/or maintainability requirements where appropriate.

1. Hierarchy of Specification and Standards

Industry or regulatory specifications or standards frequently apply to the specification of a system. These should be identified. The hierarchy of these documents, as they apply to the system specification, should be defined so that conflicts between them can be resolved.

2. System Description

\$.

The system description includes the configuration of the system elements and their interconnection with each other. It is recommended that this be done graphically with supporting written clarification. The system description is derived from the system concept design.

3. System Interfaces and Supporting Services

The means by which the system interconnects with the world outside of the system should be fully described. These interfaces are relatively unique to each individual system, but some key interfaces will include:

> External communication systems such as mobile radios, private telephone systems, data processing, digital data systems, alarms, microwave transmission systems, and similar facilities

> Utility systems such as public access telephone systems commercial power systems, and water supplies

		'•	Emerg fire
			civil
	4.	Syste	em Sof
		The s	system
	tion requ:	of tl ired t	he ope to pro
	5.	Syste	em Ope
		The s	specif
	taild	ored t	to mee
	the s a cor	nplet	n conc e desc
		•	The f
			syste
		•	arrow
		•	The c
			ment.
	C		•
1.2	0.	Expai	nsion
	· b .	Expai Frequ	uently
. 9	o. preve	Expand Frequent th	uently ne ini
	preve	Expand Frequent th real:	uently ne ini izing
	preve from and p	Expan Frequent th real: perform	uently ne ini izing rmance
	preve from and p the s	Expand Frequent th real: perfor initia	uently ne ini izing rmance al sys n spec
	preve from and p the s 7.	Expan Frequent th real: verfor initia system System	uently ne ini izing rmance al sys n spec em Lic
	preve from and r the s 7.	Expand Frequent the real: verfor initia system System High	uently ne ini izing rmance al sys n spec em Lic
	preve from and p the the s 7.	Expan Frequent th real: perfor initia system System High	uently ne ini izing rmance al sys m spec em Lic techn r stat
	preve from and p the s 7. Feder dures	Expand Frequent the real: verfor initia system System High cal or s may	uently ne ini izing rmance al sys n spec em Lic techn r stat be ne
	preve from and r the s the s 7. Feder dures These	Expan Frequent th real: perfor initia system System High cal on s may e requ	aently ne ini izing rmance al sys m spec em Lic techn r stat be ne uireme
	preve from and p the s the s 7. Feder dures These speci confi	Expan Frequent th real: perfor initia system System High cal or s may e requent f; cat	al sys me ini izing rmance al sys m spec em Lic techn r stat be ne uireme tion.
v. <u>1</u>	preve from and r the s the s 7. Feder dures These speci confi	Expand Frequent the real: verfor nitia system System System High cal or s may e requent ficat	uently ne ini izing rmance al sys n spec em Lic techn r stat be ne uireme tion. tion a
v. j	preve from and p the s the s 7. Feder dures These spect confi EQUIPN	Expan Frequent th real: verfor nitia system System High cal or s may fical fical fical fical	uently ne ini izing rmance al sys m spec em Lic techn techn t stat be ne uireme tion. tion a SPECIF
V. j	preve from and r the s the s 7. Feder dures These speci confi EQUIPN The e	Expan Frequent th real: verfor nitia system System System High cal or s may e requi ficat dent s	uently he ini izing rmance al sys n spec em Lic techn r stat be ne uireme tion. tion a SPECIF nent s s of e
V. j chara the s	preve from and p the i the i the i the i f f dures These speci confi EQUIPN The e acteri system	Expandent Frequent the real: verfor nitia system System System System High cal or s may ficat figurat MENT S equiption	al sys mance al sys mance al sys m spec em Lic techn techn tion a spection. tion a SPECIF ment s s of e cifica

1

 \square

T

II-66

pency operations such as emergency power services, protection, emergency medical services, and disaster prevention operations.

 \bigcirc

tware

n specification will include a complete descriperations software and the applications software ovide the needed system performance.

eration and Performance

Eication of system operation and performance is et individual system requirements as developed in cept. Typically these specifications will include cription of:

Eunctional modes of system operation for routine am performance, emergency system performance and vable fail-safe or degraded levels of performance.

calculated levels of personnel staffing for the ating staff, maintenance staff, and system manage-

Requirements

y, limitations on funding or other resources will tial implementation of a sophisticated system the full potential of system operation, scope, when these expansion requirements are part of stem procurement, they must be fully described in cification.

censes and Approvals

nology systems may exceed the apparent limits of the licensing regulations and extraordinary procebeded to acquire the necessary permits or licenses. The should be clearly defined in the system They will be unique to each specific system and intended operation.

FICATIONS

The equipment specifications define the quality and performance characteristics of each item of hardware and software needed to meet the system specifications. The identity of each hardware and software item is derived from the system configuration diagram which was developed as part of the system concept.

1. General

•

 ≤ 5

This subsection of the equipment specifications accumulates all items of the individual equipment specifications requirements that are common among all items of equipment or software. It also defines the quality and quantity of each item of hardware and software that is required to meet the system configuration. Typical items that would appear under the general heading are:

- Applicable industry standards or publications and their hierarchy for this program
- Standard environmental conditions such as temperature, humidity, primary power, vibration, shock, dust, and dirt

All strategy

Deliverable equipment lists showing the quantity required of each item including spares

The required level of quality and workmanship for each hardware item.

2. Individual Item Specifications

Each equipment item will have an individual technical specification which reflects its construction and performance as required to meet the system specification. These are derived individually from the severe than those needed to meet the system performance specifications. Software specifications are typically combined with the equipment specification for the computer subsystem or the switching subsystem in which they are resident.





Submitted to:

Lt. Kent Rasmussen Director of Federal Projects Department of Public Safety P.O. Box 11415 Oklahoma City, Oklahoma 75136

SYSTEM IMPLEMENTATION PLAN FOR THE STATE OF OKLAHOMA DIGITALLY ADDRESSED TRUNKED COMMUNICATION SYSTEM

November 15, 1980

BOOZ · ALLEN & HAMILTON Inc.

776 SHREWSBURY AVENUE TINTON FALLS, NEW JERSEY 07724 747-9303 AREA CODE 201 TABLE OF CONTENTS

		Page
I.	INTRODUCTION	III-1
II.	SYSTEM CONCEPT	III-5
н 1	<pre>2.1 System Operation 2.2 System Description</pre>	III-5 III-9
III.	REQUIRED IMPLEMENTATION ACTIVITIES	III-19
	 3.1 State Agency Approvals 3.2 Preliminary Licensing Considerations 3.3 Implementation Schedule 3.4 System Management and Operation 3.5 Procurement 3.6 Licensing 3.7 Facilities Acquisition and Preparation 3.8 Personnel and Training 3.9 Post Implementation Support 	III-19 III-21 III-21 III-22 III-23 III-25 III-25 III-26 III-26
APPEI	NDICES: A - SYSTEM REQUIREMENTS B - SUGGESTED AGENCY AGREEMENTS C - STATEMENT OF WORK	III-29 III-37 III-43

This system implementation plan for a digitally addressed trunked communication system (DATCS) for the State of Oklahoma was prepared by Booz, Allen and Hamilton and the Associated Public-Safety Communication Officers (APCO) in response to LEAA Grant No. 79-SS-AX-0013.

BACKGROUND AND OBJECTIVES

1.

The Second Report and Order in the proceedings related to Docket No. 18262, published by the Federal Communication Commission (FCC) in 1974, promised relief of the increasingly more difficult problems of frequency congestion in the public-safety community. These proceedings allocated 200 channels in the 800 MHz portion of the spectrum for "trunked" systems. In February 1977, LEAA recognized that many technical, economic, and management questions about the applications of this newly available spectrum to public-safety operations must be answered before the potential benefits inherent in trunked systems could be made available. Under LEAA Grant No. 77-SS-6009 APCO was requested to analyze these problems, make appropriate recommendations for future actions, and describe a program to demonstrate the potential capabilities of these new concepts.

A series of programs under APCO Project 16 were initiated. The first of these (Project 16) resulted in an analysis of the technical and regulatory factors affecting the applicability of the 800 MHz portion of the radio spectrum to law-enforcement communication system problems. A second program (Project 16A) identified the specific operational capabilities and functional requirements that should be incorporated into a public-safety trunked communication system. The third program (Project 16B) provided specific planning assistance to two cities (Lexington, KY and Phoenix, AZ), two counties (Bucks Co., PA and Salt Lake County, UT), and the State of Oklahoma for the feasibility study and implementation plan for DATCS demonstration systems.

achieved are:

1

To determine the feasibility of DATCS to relieve statewide radio channel congestion

To establish the potential of DATCS to accommodate future statewide radio communications needs

To address the solution of the complex command/ control problems confronting the public-safety agencies

To develop a common DATCS to support all public-safety and/or other governmental communications requirements within the State of Oklahoma

To provide a solution to interagency radio connectivity between governmental agencies appropriate to their mission. responsibilities.

ii

I. INTRODUCTION

Since February 1980, Oklahoma has been participating in Project 16B as one of the demonstration areas. The key objectives to be

TTI-1

	Other advantages of operational flexibility are inherent in a dynamically programmed digitally addressed communication system. These have been evaluated as DATCS capabilities determining the system requirements.		ور با		been voluntary. the State of Ok
	2. PROJECT DIRECTION AND PARTICIPANTS				bodies.
	The project was under the direction of Lt. Kent Rasmussen, Director of Federal Projects, Oklahoma Department of Public-Safety.				
-	The potential DATCS participants in the State of Oklahoma include the following:				
	. Oklahoma Highway Patrol	A SALE OF THE OWNER			
•	. Office of the Chief Medical Examiner				
	. State Fire Marshal	()		ŭ.))	
	. Oklahoma Alcoholic Beverage Control				
	. U.S. Drug Enforcement Agency*	8			
	. Federal Bureau of Investigation*	Community of			
	. U.S. Secret Service*				
	U.S. Department of Alcohol, Tabacco & Firearms*			4_)	
	. Oklahoma State Board of Investigations	51			
	. Oklahoma Department of Health**	L		7 ····]	
	. Oklahoma State Bureau of Narcotics*	1			
	. Department of Wildlife Conservation	53			
	. Department of Agriculture, Forestry Division	and a second			
	. Department of Corrections*	and the second se			
	. State Civil Defense.*	(,)		The state of the s	
	Each of the above listed public-safety and public service				
	to the interagency and intra-agency communications analysis and the	(``)			
	requirements analysis for the DAICS concept design.				
	develop this System Implementation Plan (SIP) with the technical assistance of Booz, Allen as engineering consultants to APCO. The cooperation of all participating agencies and their personnel has	and an and a second sec	an share and a set of the second s		
	Cooperation of arr participation of the oblehers Wighney Datrol		Second and the second second		
	* Participation limited to maintaining interface with Oklahoma Highway Fattor.	12	A STREET		
	interface study.				
	III-2				

and the second

.

•

. There have been no binding commitments required of clahoma or any of the other participating governmental

feasibility. access. the mobile equipment. 2.1 SYSTEM OPERATION Preceding page blank

II. SYSTEM CONCEPT

It is the objective of this section to document the proposed system concept of operation. This has been accomplished by describing. in a conceptual sense, how each performance requirement (previously documented) will be satisfied. It is important to note that the proposed system concept of operation documents the thinking to date and, as a result, there remain a few elements of the system concept still unresolved. These unresolved elements are expected to be removed through the course of approving the system concept.

The basic DATCS land mobile radio system for Oklahoma has been conceived to replicate current state radio system operations as closely as possible. The difference in function and response between conventional radio systems now in use and DATCS radio system should be nearly transparent to the mobile operator except cochannel sharing will appear to the operator to have been eliminated.

This philosophy directs that operations which require a unique skill of the mobile operator be minimized. It also directs the avoidance of subsystem interfaces in which trunked technology or switching requirements might degrade the operation of existing communications systems when used in conjunction with DATCS. For example, the recommendation to interface portable radio operations at the dispatch console instead of automatic cross-banding with DATCS is based on this philosophy rather than the lack of technical

The DATCS offers features and options related to digital control functions and programming that enable operational flexibility not previously available. Some of these features include:

Privacy of communication with or between field units

Dynamic reallocation of field forces

Interagency communication

Automatic identification of individual field units

Automatic quegeing of units requesting system

Implementation of these have been concentrated at the regional control points and/or dispatch points to minimize the complexity of

The basic principle of DATCS operation is the shared use of a common base station network and a pool of frequencies (numbering from five to twenty, in present systems) by a number of independent radio system users. Under trunked system operation, the number of independent users can exceed the number of available frequencies.

Each system user (independent agency) is assigned an operating "channel*" for its routine use. Communications among mobile units of any given agency can be conducted on any one of the system frequencies; choice of the particular frequency for a communication is made by the system hardware, independent of operator control, and the selection is made from among the unused frequencies. In this manner, more efficient use of frequency spectrum is accomplished, as channel assignments are based on demand and the availability of channels.

The remainder of this section presents the proposed operation of Oklahoma's future trunked mobile radio system. It is proposed that:

- Current complaint reception and dispatch methods be largely unaltered in the new system
- DATCS will be transparent to user agencies, except that additional capabilities and system improvements will be readily apparent.
- Independent agencies or departments within agencies will share channels not based on the constraints of conventional radio systems, but rather based on the need to intercommunicate
- Network control will be established only to the degree necessary to preclude mutual interference and maintain overall system performance.

These elements of system operation are described in greater detail in the following paragraphs.

2.1.1 Philosophy Of Operation

The proposed statewide trunked mobile radio system for Oklahoma is intended to replicate current system operations, except where basic improvements are appropriate. The first report* identified certain major deficiencies in present mobile radio systems and indicated that DATCS offers an opportunity to overcome many of these deficiencies. It is the philosophy of the proposed operation to suggest only those changes in the present operations that will overcome the previously noted deficiencies.

Oklahoma's major mobile radio deficiencies are a statewide lack of channelization, a trend toward increased fragmentation in state systems and insufficient radio coverage for many systems. Mobile radio operations will be changed, therefore, only in that fewer

* "Channel" in the sense used in DATCS is not equivalent to a channel in a conventional mobile radio system. In a DATCS, the number of "channels" exceeds the number of frequency (pairs), which is impossible in a conventional system.

* "Documentation of Oklahoma Mobile Radio Systems and Requirements", July 3, 1980.

III-6

design stages.

.

.

DATCS also offers increased capabilities for accomodating communications among independent agencies which occasionally may need to respond to events of mutual concern. These capabilities are provided through dynamic channel flexibility (reassignment of channels from one or more fixed locations) and commonality of equipment. For example, during an event involving the Oklahoma Highway Patrol and the FBI, selected units from both agencies could be assigned to a common channel for the duration of the event. Afterwards, each unit would return to its previously assigned channel. As a result, both agencies could intercommunicate only to the extent necessary for the event and routine operations would remain unaffected.

Similar flexibility is applicable to all other DATCS users, as all mobile radio units in the system would be compatible.

2.1.2 Datcs Network Control

The concept of network control for a commonly owned and operated statewide communication system represents the only significant deviation from present operations. Network control should take the form of a two-tier management structure:

> An advisory group should be established, responsible for policy and planning issues

An existing organization should be identified to take responsibility for day-to-day operation of the new system.

The advisory group would plan system implementation, identify sources of funding, generate operating policies and plan ongoing system expansion consistent with increased participation by local agencies and changes in operations of existing users. The organization responsible for day-to-day operations should be responsible for the following:

> Responding to requests for change in group address. Individual agencies requiring non-routine communications with other agencies would indicate their needs to network control, which would have the capability of temporarily reassigning individual mobile units to other groups and forming (or dissolving) special groups, as needed.

> Monitoring system performance. The failure of base station network components and degradation of performance in any part of the common network will be noted by the day-to-day system management organization.

Coordination of Maintenance. The responsibility for system maintenance includes planning routine maintenance and coordinating service for failed or degraded system components.

users will be assigned to each channel (enhancing intercommunications within and among agencies), a statewide uniform level of communications capability will be provided and radio coverage will be enhanced statewide by attention to engineering detail during system

The day-to-day network control could be conducted at both a regional and state level:

Regional Level - monitor system performance and respond to requests for temporary changes in group address

State Level - plan and coordinate maintenance of common equipment.

The Department of Public Safety (DPS) is best suited to function as the day-to-day network control organization. The DPS is a statewide organization which currently functions on a state and regional level, and it has established regional facilities in all areas throughout Oklahoma.

2.1.3 Principles Of Operation

The principles described in this section are based on the assumption that complaint reception and dispatch for all participating agencies will be conducted from existing dispatch locations. The configuration of a statewide DATCS is such that all users can maintain their independent dispatch facilities although the base station network is common to all users.

The principles of operation are expressed in terms of:

- Group addresses or "channels"
 - Dynamic assignment of primary group address
 - Accessibility to all appropriate channels by each dispatch location.
- Portable Radio Interface
- Priority Calling
- . Network Control.

2.1.3.1 Group Address or "Channels"

Each agency will be provided one or more unique "channels" or group addresses according to its specific requirements. These group addresses will be recognized according to instructions provided by network control. The mobile radios will operate in manner similar to today, but fewer mobile units will be assigned to each "channel." Only those mobiles which associate routinely will share a particular channel.

Each mobile will contain from one to five "channels", with at least one channel capable of being dynamically programmed from network control. Each agency and dispatch point will have access to all group addresses of its associated mobiles and network control to request changes in group address.

III-8

Portable Radio Interface

Conventional portable radios will be used in two possible ways: maintain routine operation of existing portable radio systems independent of DATCS and interface conventional portables via a single cross-band vehicular repeater (for OHP mobile command post operations only). The use of portable radios in the Oklahoma DATCS is further detailed in Section III - System Description.

Priority Calling

2.1.3.2

2.1.3.3

2.1.3.4

•

2.2.1

The capability of priority calling will not be provided in the initial system implementation. This feature, which represents a modification of existing trunked hardware is not a major requirement at this time. The software of the initial system implementation, however will allow for future implementation of priority calling.

Network Control

The function of network control is to execute the policies established by the advisory group, including:

Managing and executing requested changes in mobile group addresses

Ensuring that changes are with the expressed approval of all agencies involved

Ensuring that dispatchers will not be overburdened with requests for changes in group address.

2.2 System Description

This section describes the future DATCS mobile radio system based upon Oklahoma's requirements. Specifically, addressed are the performance requirements, external interfaces, control functions, radio spectrum and FCC licensing implications, and ancillary items associated with fixed facilities. These are discussed individually in the following paragraphs.

Performance Requirements

This section specifies the performance requirements that must be satisfied to implement a DATCS in Oklahoma. Specific performance requirements that are discussed include:

DATCS capacity

Intra- and inter-agency communications

Mobile control head characteristics

Two-part digital address

Dynamically assignable group address

Talk-around capability

Interface with conventional portable radios

Radio coverage

Management information

System protection measures.

2.2.1.1 DATCS Capacity

The DATCS system design must be able to support eventually 2,500 to 3,000 state mobile units. This number can be expected to increase if local government agencies participate in the future. Initially, the DATCS must be able to support 1,200 units. The system must also be capable of supporting 150 unique group addresses with as few as one to as many as 100 individual units assigned to a specific group address.

The specific number of frequency pairs that will be initially planned will be based upon the channel loading criteria established by the FCC. Present loading criteria is based solely on single site trunked system implementations, which will not be representative of a trunked implementation for Oklahoma. It is uncertain at this time what loading criteria the FCC might apply to a statewide trunked system, as the specific method of achieving statewide coverage is as yet undetermined and will likely influence the FCC's decision. Further, it is not clear at this time to what extent the interfacing of conventional portables will be considered legitimate channel loading.

2.2.1.2 Intra- and Inter-agency Communications

Intra- and inter-agency communications will be accomplished, depending upon specific circumstances, by one of three methods:

> Limited Selection of Group Addresses - Supervisory level mobile radio users will be equipped with the group address of other users, in instances where access to other group addresses would be routinely required. This method provides selected users the option of operating on other group addresses without dispatcher initiative. It is expected that user-initiated access to other group addresses will be limited to supervisors and administrative personnel of selected agencies.

> Cross Patching - Will be provided to most dispatchers. The dispatcher will be able to cross-patch two or more group addresses on an as needed basis. In this way, the dispatcher can make or break a cross-patch arrangement, thereby maintaining channel discipline. Patching will be conducted generally among only those channels available to a particular agency; cross patching between different public-safety agencies is not proposed.

> > III-10

ways:

Ż.

Occasionally the user may not be able to immediately access the desired channel because all channels will be temporarily allocated in which case, a "channel busy" indication will be provided. The user will also automatically enter a queue set up by the controller to seize the next available channel, in the order of requests; when a channel becomes available to a mobile in the queue, the mobile operator will be notified by an audible and/or visual indication.

 \mathcal{O} .

U

1:

As a minimum, each unit will both be uniquely identified (unalterable) and identified as a member of a group address (alterable). The unit identification will be an integral part of the mobile unit, not readily changeable. On the other hand, the affiliated group address will be dynamically programmable to be able to satisfy changing needs.

Reassignment of Group Addresses - This method of interagency communications will be initiated only by network control personnel (system operator) when users need to temporarily become part of another group. It is anticipated that reassignment of group addresses will be used to respond to large-scale events of mutual responsibility and therefore, will probably occur infrequently. The reassignment of group addresses will be accomplished by one of two

Change in group address from a regularly assigned to an already existing group address of another user group

Creation of a "phantom" group whereby individual units would be assigned to an entirely new group address created to meet a temporary, one time need.

2.2.1.3 Mobile Control Head Characteristics

The mobile control head will be provided a variety of operational features, as determined by the specific needs of each agency. There will be two classes of mobile control heads:

> Supervisory Control Head - will provide a mobile radio user with up to five unique group addresses for interagency communications. Access to these group addresses will be user selectable. The primary group address, at the very least, will be dynamically reassignable, whereas the four other group addresses will be essentially fixed (but reprogrammable by maintenance technicians, if necessary).

Non-Supervisory Control Head - will provide a mobile radio user a single group address, dynamically assigned by the user's agency. If a user requires access to another group address, the user's dispatcher will request a change-ingroup-address status via the control operator or will cross-patch the user to the appropriate group address.

2.2.1.4 Two-part Digital Address

The Oklahoma DATCS must have a capacity eventually for 3,500 individual mobiles and 150 unique group addresses. Although current equipment offerings typically provide three levels of address (i.e., unit, subfleet, and fleet), no need currently exists to segment Oklahoma's operations beyond the unit and subfleet level.

2.2.1.5 Dynamically Assignable Group Address

The capability will be provided to dynamically assign and change at least one group address of the mobile units from a central or regional location; each agency will individually decide which group address (among field of five) will be dynamically assignable. However, this capability will only reside at fixed locations (regional network control). In this way, effective network control can be exercised.

The regional system operators, who perform the function of network control, will have available two options to dynamically assign a group address:

> Modify a particular unit's regularly assigned group address to an already existing group address (i.e., Department of Corrections to OHP etc.)

> Create a "phantom" group, not regularly assigned, which will be dissolved when the need for its continued existance disappears; units will then return to the regularly assigned group address.

Changes in group addresses will follow firmly established operational procedures to maintain strict control; these procedures are best established jointly by the advisory group (composed of representatives of the user agencies (i.e., OHP, Department of Corrections, etc.)

2.2.1.6 Talk-around Capability

Mobile units require communications independent of the base station facilities when beyond radio range or when the dispatcher is not involved in a message between close mobiles. This talk-around capability should be provided through a non-DATCS channel specifically licensed and set aside for the purpose of direct simplex communications (talk-around).

Talk-around will be mobile-user selectable as currently exists with the present radio systems. However, unlike the present systems, all units desiring to use talk-around must agree in advance to switch to talk-around and must secure dispatcher permission; if only one unit switches communications will not be possible. While DATCS could provide automatic switching to talk-around under specific conditions (i.e., sense of "out-of-range" condition), manual switching is preferred to maintain predictable communications.

III-12

2.2.1.7

 \square

Although industry estimates vary for the introduction of a trunked portable radio (3-6 years), there are no assurances that trunked portable radios will ever be available at an affordable cost or compatible with DATCS. Also, there is concensus that any type of trunked portables will not be available sooner than three years. For these reasons, a method is therefore required which will provide Oklahoma continued access to portable radio operations.

To provide for portable operations in the interim there are two methods which are proposed for Oklahoma:

There are currently only two potential DATCS candidates whose operations are portable intensive: the Department of Corrections (DOC) and the Oklahoma Bureau of Narcotics and Dangerous Drugs (OBNDD). Both these agencies presently use portable radios for onscene communications (communications within a correctional facility compound for DOC and local surveillance for OBNND) and do not present a driving requirement for conventional portable interface with DATCS. OHP, however, does require conventional portable interface with DATCS to maintain their present level of mobile command post communications.

The interface of mobile command post portables with DATCS will be accomplished with a single vehicular repeater which will be part of the command post and utilized only for the duration of tactical operations. The vehicular repeater will receive portable transmissions and will retransmit them over the DATCS "channel" or group address associated with the mobile command post. Because the possibility exists that all DATCS channels will be busy when a portable attempts to communicate over the DATCS, unique operational procedures are required of the portable operator, and current portable operations cannot be replicated with DATCS at this time (the operational implications of crossbanding conventional portables with DATCS are detailed in "Analysis of Problems and Requirements Associated with Implementation of a Statewide DATCS for Oklahoma," September 30, 1980. Vehicular repeaters for portable interface are proposed only for OHP mobile command post operations and are not recommended for day-to-day operations.

The eventual adoption of a trunked portable for use in Oklahoma will require enhanced radio coverage through those areas of the state where portables are required. The enhanced coverage will likely be provided by satellite receiver systems similar to those used to support conventional portable radio systems, and the Oklahoma DATCS must be capable of integrating a satellite receiver system in the future.

Interface with Conventional Portable Radios

Maintain existing portable systems for on-scene, tactical operations independent of DATCS

Interface through a single vehicular repeater for command post operations (OHP only).

		ſ	7 í	4	
					. <u>Separate Tr</u>
	2.2.1.8 Mobile Radio Coverage	F i		田	the trunked location.
	Reliable mobile radio coverage will be provided throughout Oklahoma, at least to the same or greater reliability than currently realized by the existing statewide mobile radio sustance. Our statewide	L)			cate over t properly us
	mobile radio coverage reliability is proposed to exceed 90-97 percent, which is considered very good by industry standards.				. <u>Hybrid Trun</u> Trunked Sys
	Several options are available to achieve wide-area coverage for Oklahoma, including:				of the stat because of
	. Simultaneous broadcast of identical information on common channels		5-1-		discussions required.
	. Simultaneous broadcast of identical information on different channels		J.		2.2.1.9 <u>Management 1</u> Management infor
	. Separate trunked systems for each site	5			will be the primary t the DATCS system is m performance data will
	. Hybrid trunked/conventional system.	(***)			following forms:
	No single method of achieving wide-area coverage stands out as a preferred choice over the others, and it is proposed that each	L			- User (
4	vendor outline his preferred method in the bid response submitted to Oklahoma. The choice of specific sites that will be necessary to achieve the required radio coverage throughout the state will be the				- Time (
	vendor's responsibility. The successful vendor will also be responsible for:	F			- Sample sample
	. Indicating the need for the remote sites, and the justifi- cation for their selection			5	. User profil
	. Specifying the characteristics of each site	L			- Agency
	. Assisting Oklahoma procure the remote sites (i.e., buy, lease, etc.), if necessary.				- Group - Indivi
	Since each vendor's approach to state-wide coverage may vary no				- Site/k
	until time of bid.				. Status of s use," curre
	It is appropriate, however, to describe alternative approaches to state-wide coverage:	(. Status of k
	. <u>Simulcasting</u> (both on identical and separate channels) - mobiles receive identical radio communications from one or				- Base s
	more sites simultaneously. This technique has the advantage that a mobile is assured of receiving a communication				- Site d
	regardless of its location in the jurisdictional area. Conversely, there are disadvantages of high cost, technical	()			2.2.1.10 System Prot
	complexity, and conspicuous use of frequency spectrum.				One major intent a mobile communicatio availability* which v
	III-14				* Percentage of time the
		F P			
		Balancia di Santa di Santa di Santa	1		

unked Systems - mobiles would communicate with system associated with the mobile's specific This approach is simpler and less costly than ng, but special measures are required to communithe entire jurisdiction simultaneously and to se the control channels.

ked/Conventional System - same as the "Separate stems" except a conventional radio system(s) sed to cover remote, infrequently traveled areas e. However, FCC licensing eligibility is uncertain an absence of well-defined regulations, although with the FCC indicate a hybrid approach may be

Information

mation in the form of system performance data cool to determine, on an ongoing basis, how well meeting Oklahoma's mobile radio needs. System be provided to management of the DATCS in the

rintout

(group address and individual identifier)

(initial access time and duration)

e utilization of common facilities (adjustable e rate)

Le, according to category

address

idual identifier

base station

specific group addresses, on command (i.e., "in ently idle, etc.).

key components to pinpoint failure

stations

controllers

tection

t of developing a DATCS for Oklahoma is to offer on system which exhibits a very high degree of would be significantly improved over the present

system is functional at full capacity.

collection of individual systems. For example, loss of a single DATCS base station will only slightly reduce a mobile's chance for obtaining a channel, on request. Whereas today, loss of a single base station (or land lines, tower, antenna, etc.) completely inhibits further communications until repair.

In the event of the loss of a major component (i.e., controller or equivalent component), the DATCS system will revert to a conventional mode in which mobiles would be directed to pre-assigned channels. To accomplish this, the SOW will impose a number of system protection measures that must be satisfied to adequately insure Oklahoma against the consequences of catastrophic system failure. Vendors will be obligated to submit a specific plan on their approach to dealing with failed site components (plan subject to approval by Oklahoma).

Regarding alarm features, network control (central or regional) will be automatically altered of subsystem and somponent failure. In this way, immediate action can be taken by the system operator, probably even before the field units notice any discernable change in performance.

Time-out timers will be provided on the mobile to preclude inadvertent capture of the system for extended periods as well as on the base stations since unnecessarily long-duration conversations tend to deny channel access to other DATCS users. Long-duration conversations could potentially be a problem since the mobile radio user could falsely perceive that DATCS is a completely private, always clear-channel radio system which is unlikely to deny channel access to other users because of one's own usage. The use of timeout timers on base station equipment however, will present operational problems to potential EMS DATCS users which transmit telemetry. A means of accommodating EMS and at the same time affording the system protection against prolonged channel usage has yet to be determined.

2.2.2 External Interfaces

The section describes the interface between DATCS and the systems external to DATCS:

Telephone

Conventional portable radio

An interface between DATCS and the switched public telephone network will not be established at this time. The APCO Project 16A document did however, include this requirement for a DATCS. If an interface in the future, it must confirm to FCC rules and regulations in effect at that time.

Conventional portable radios are to be interfaced with the proposed DATCS system to enable portable radio users to communicate with DATCS mobiles, but only for OHP mobile command post operations. A crossband connect between the DATCS and conventional repeaters, under the control of the mobile command post operator, is proposed

III-16

to accomplish this function. This solution is the least disruptive approach to dealing with a lack of trunked disruptive approach to dealing with a lack of trunked portable radios, for the interim. The remaining agencies will maintain their existing portable radio systems and will interface with DATCS through dispatcher-controlled cross-patching, where necessary.

2.2.3

Network control represents the only other significant change in operations. The establishment of a network control function is mandatory for DATCS. This is because monitoring system performance and managing a commonly owned, statewide mobile communication system must be centrally performed and cannot be distributed among several departments.

The responsibilities of a network control function will be divided into two categories:

> Policy - responsible for: establishing guide-lines for the use of DATCS, accommodating demands from various independent agencies impartially, and ensuring that day-to-day operations are appropriately performed

> System Operations - responsible for the day-to-day operations of the network. The new or existing (system operations) agency would respond to the policies and guidelines established by the parent group. For example, system operations would handle requests for group address changes, but would not dictate under what circumstances a request for change in group address may be honored, since that is matter of policy.

For further study, it is preliminaryily proposed that policy be established by an all-new Advisory Group created to serve this specific function, comprised of representatives of the user agencies. System Operations, on the other hand, could be vested in an existing agency or developed independently. These choices warrant the closest attention.

2.2.4 Dispatching

.

Each agency that presently performs a dispatching function for itself or other agencies will be provided a dispatch capability under DATCS. As a minimum, individual agencies will be provided a desk-top type console containing all group addresses of responsibility and, in some cases, group addresses of mutual interest with other agencies/departments. These consoles will be capable of: radio patching among the channels of access; selecting the desired area coverage; communications with network control to request changes in group address; and receiving all messages from mobiles of responsibility. In addition, communications will be possible with at least two group addresses simultaneously; come agencies may have the capability to communicate with three or more group addresses simultaneously.

Network Control Functions

2.2.5 Radio Spectrum and FCC Licensing

Presently, DATCS is only feasible at 800 MHz, which previously was a television-only band, but recently reallocated for mobile radio. For the foreseeable future, at least, sufficient 800 MHz spectrum will be available in Oklahoma to support DATCS.

Another favorable aspect is the current FCC demeanor which regards these kind of mobile radio systems with favor and something to be promoted. Therefore, if any negotiations are required with the FCC in the future to resolve regulatory issues (i.e., multiple base station sites, etc.) favorable rulemaking can be anticipated.

It is advisable, however, to begin establishing rapport with appropriate FCC staff to help "smooth the path" to successful licensing,

2.2.6 Ancillary Items

Ancillary items for the support of the Oklahoma System Concept of Operation include the following:

Emergency Power

Maintenance.

Emergency power will be necessary for the mobile sites throughout the state and the regional network control centers. Depending on the inherent emergency power capabilities of the individual network control sites, emergency power may not be required. However, emergency power will be necessary for the remote sites. Hydrocarbon-type fuel generators are proposed because of the substantial power requirements of each base station (1000 watts continous each).

The maintenance strategy for the DATCS system in Oklahoma is two fold:

> Initial Contract Maintenance - Initially maintain the DATCS system under a maintenance contract with a commercial shop, at least until a self-maintenance capability is established and sufficient confidence in the DATCS system has been achieved. This period will provide the primary basis for Oklahoma's maintenance staff to acquire the appropriate new skills required.

> Future Self-maintenance - Anticipated only after adequate training, reallocation of staff resources, and development of a management structure. Development of a competent self-maintenance capability for DATCS is a large task and therefore will be a major component of the System Implementation Plan (SIP).

> > • #

III-18

Implementation of the system concept will proceed only after thorough review by the potential user agencies and their department heads. After this has been achieved and the implementation plan has been officially sanctioned, another critical milestone must be met before procurement activities can be initiated.

Federal Communications Commission (FCC) licensing requirements for large scale 800 MHz trunked radio systems in the local government radio service have not been clearly defined or tested at this time. The initial implementation activity following agency approval must be directed toward satisfying all of the elements needed to secure reasonable assurance from the FCC that the trunked radio system, as planned and approved, can be licensed.

3.1. State Agency Approvals

State agency approvals will be achieved in three steps, the securing of user agency agreements, the development of funding agreements to implement and operate the system, and the legally binding commitment of the state agencies to participate in and support the system.

3.1.1

•

Signed agreements by each agency planning to participate in the trunked radio system must be acquired. These agreements will conform with FCC requirements (Section 90.359 and 90.179). As a minimum they will include:

> Agreements regarding the supervision and control of the trunked system operation

Method of cost-sharing and pro-rating of costs among all cooperating parties

Responsibility for maintaining system operation records and filing reports with the FCC (Section 90.391).

In addition to the items needed for FCC licensing, the agency agreements will also include provisions for:

Management authority over the system staff and equipment

Centralized maintenance of system components

Property, equipment and facilities provided by the user agencies

Modification and/or termination of the agreement as subject to FCC Rules and Regulations (Section 90.391).

Agency agreements will be the key element in securing the funding agreements and government approvals.

III. REQUIRED IMPLEMENTATION ACTIVITIES

Agency Agreements

3.1.2 Funding Agreements

2

The acquisition of agency agreements will provide dimensions for revised cost estimates needed to establish funding requirements. These requirements will include estimates of non-recurring implementation costs and recurring operations costs. The following elements of cost will be identified:

Implementation Costs

- Planning Activities
- · Professional Services
- Hardware/Software
- Facilities
- Installation/Test/Acceptance
- Personnel Staffing and Training
- Operating Costs
 - Personnel Salary and Overhead
- Supplies
- Maintenance of Hardware/Software
- Equipment Replacement
- Continued Training

The distribution of these costs among the funding sources will be decided and firm commitments and/or agreements acquired. These funding sources will include:

- Federal and State Grants Funds
- In-kind Services, Equipment or Facilities
- User Agency Cost Sharing Assessments
- Grants from Private Foundations
- Gifts from private sources.

The funding agencies must approve and be committed to the funding plan before final and binding commitments will be sought from the state of Oklahoma and participating agencies.

III-20

3.1.3 Government Approvals

 \prod

cial schedule.

Based on the agency agreements and funding agreements reached, firm commitments will be required from the State government to assure continued project support and to respond to Federal and State grants requests. These approvals will be requested in the form of resolutions or other legally binding documents.

3.2 Preliminary Licensing Considerations

The licensing of multiple base station wide-area trunked radio systems in state and local government service and public safety service is not clearly defined by present FCC Rules and Regulations. Waivers to the present FCC Rules and Regulations may be required, particularly Section 90.375 which defines time limits for phased implementation. It is probable that waivers to Section 90.379 which limits power and antenna height will also be needed.

Before hardware and software commitments are made, preliminary discussions will be held with the FCC. An expression of assurance will be sought indicating that the trunked system, as planned and approved by the user agencies and their governmental authorities will receive favorable licensing consideration from the FCC when formal applications for licenses are submitted.

3.3 Implementation Schedule

The detailed implementation schedule will be developed following assurances from the FCC that the trunked radio concept will be licensable under FCC Rules and Regulations and that any revisions to the concept required by FCC evaluation of the system design has met the approval of the user agencies and the state government. The implementation schedule will consist of two key interrelated parts; a system implementation schedule and a corresponding finan-

3.3.1 System Implementation Schedule

The system implementation schedule will be a time-dependent schedule. The project will be defined in terms of separate tasks with clearly defined milestones indicating the completion of each task and definitive measurable milestones leading to the completion of each task or project element. Major tasks include:

Project Planning and Reporting

Engineering Services

Preparation of Specifications

Preparation and Approval of Statement of Work

Solicitation of Bids

Contract Award

- Facility Preparation
- System Licensing
- Equipment Delivery/Installation
- Test and Acceptance
- Personnel Training
 - System Cut-over
- System Evaluation.

Each major task will be supported by task break-down schedules which define the subtasks needed to achieve the completion of each major task.

Financial Schedule 3.3.2

The financial schedule will be coordinated with the system implementation schedule. This schedule will be based upon the funding agreements and will include the following major elements:

- Milestones showing each major commitment of funds
- The date and amount funds will be made available to the project from all sources
- The schedule dates for the submission of financial reports.

The submission and distribution of financial reports will be elements of the financial agreements and governmental approvals.

3.4 System Management and Operation

The management and direction of the implementation program will be the responsibility of the Program Director. The Program Director will be supported by a full-time and part-time staff representing each key function of the implementation program. These functions include:

- System Engineering this staff function will be assisted by consulting engineers and the engineering staff of the successful system contractor
- Procurement this staff function will be assisted by the purchasing and procurement personnel of the state of Oklahoma

Finance - this staff function will be a part-time service provided by the State of Oklahoma

III-22

Faci.	Lit
tion	ar
equip	ome
part-	-+ i

Personnel - the project staff personnel are under the direct supervision of the Program Director; personnel, activities such as job descriptions, salary determination, and benefits administration will be under the direction of the State of Oklahoma unless the person is on loan from another agency under the state government.

In addition to the project staff reporting to the Program Director, two advisory groups will be responsible for overall project policy with the Project Director serving as chairperson for each advisory group. These groups consist of:

> User Representatives - this group is composed of appointed representatives of each user agency; they provide direct liason between the project organization and their respective agencies

Government Representatives - this group is composed of appointed representatives of each state government section which has an operational or financial interest in the program.

It is the intention that this management group be carried over in form and function to the operation of the trunked radio system after it has been installed, tested, and accepted. Additional personnel needed to operate the system from a central location will be added to the Facility function of the project staff. At that time the Facility function will become a full-time staff position under the Program Director.

3.5 Procurement

N.

The procurement of the trunked system will proceed in two sequential activities; system engineering and purchasing.

3.5.1 System Engineering

Technical assistance will be required to develop the engineering design of the trunked radio system based on the system concept. Key questions to be resolved before procurement specifications are prepared include:

> Radio Propagation Analysis - ergineering analysis of 800 MHz radio propagation from candidate base station sites will be conducted to determine radio coverage of the state and to establish antenna pattern characteristics. FCC Rules and Regulations (Section 90.379) clearly limit the effective radiated power and antenna elevation at each proposed site and engineering analysis will be required if waivers are needed from these FCC rules and regulations

ties - this staff function related to the acquisind preparation of special facilities for the ent and personnel of the project will be provided time by the State of Oklahoma

Switching Logic - the switching logic for the master controller, site controller, mobile stations, and control station will be engineered to be compatible with available vendor hardware offerings

Software Management - the system software for the master controller and site controller will be engineered to allow efficient software management from a master terminal

Based upon the results of these technical activities, system engineering will prepare detailed system specifications and equipment specifications for competitive procurements.

Purchasing 3.5.2

The purchasing function will consist of three key activities; developing a Statement of Work (SOW), soliciting competitive bids, and awarding a system contract.

- Statement of Work (SOW) the SOW is a precise description of the materials and services required of the contractor and the time domain in which delivery is to be made. The key elements of the SOW will be:
 - Background and Objectives of the project
 - Required qualifications of successful bidders
 - Services and Support to be provided
 - Quantities of each equipment to be provided
 - Schedule for all deliverable items hardware, software, services, and support items

Soliciting Competitive Bids - competitive bids from all qualified vendors will be solicited by means of a procurement package which will include:

- State legal requirements
- Statement of Work
- System Specification
- Equipement Specifications.

A one-step solicitation is anticipated in which each bidder will provide a technical proposal and a firm fixed price cost proposal for the complete installation of an operating digitally addressed trunked radio system.

III-24

• •	Form 4
•	Agreent the sy
•	Agreen sharin
•	Financ user a
•	Syster Sectio
•	Eligii
•	Evider

3.6 Licensing

It is probable that FCC waivers will have to be requested to extend the number of years for phased implementation of the system (Section 90.375) and for the limitations on effective power and antenna height.

Technical support for negotiating system licenses with the FCC will be requested from the engineering consultant and from the system contractor.

It is the responsibility of the Program Director to provide adequate facilities for the equipment and staff of the trunked radio system. Subject to revision by the engineering study, the following facilities will be required:

Award the System Contract - the combined technical proposal and cost proposal must advantageous to the State of Oklahoma and associated user agencies will identify the successful bidder. A firm fixed price contract without cost escalation is anticipated.

The signed contract defining the equipment to be purchased and the quantity of that equipment will provide sufficient information to file applications with the FCC for system licenses. Present FCC practice for licensing 800 MHz trunked system requires the following minimum submission:

400 for each license requested (Section 90.357)

ments with each user agency relative to control of ystem (Section 90.179)

ments with each user agency relative to costng of installation and operation (Section 90.179)

cial plan showing the service will be provided to agencies at cost (Section 90.359)

m description and specification (Section 90.359, on 90.371, Section 90.375, and Section 90.379)

bility statement for each user (Section 90.359)

nce of purchase orders placed for the equipment.

3.7 Facilities Acquisition and Preparation

Base Station Sites equipment shelters for five to twenty 800 MHz repeaters, trunked switching logic, ancillary equipment for the antenna subsystem, emergency power facilities, climatic control equipment, and antenna support structure for two to six antennas, as required.

Network Control Center equipment shelter for master control terminal, alarm equipment, recorders, printers, microwave terminations and other items of equipment associated with the central control function. Staff facilities for a Program Director, Facilities Manager, Engineering (Software/Hardware Manager, and programmer/ operators) will also be required

Regional Network Control Centers equipment shelters for regional control terminal, alarm equipment, recorders, printers, microwave terminals and other items associated with the regional control function. Facilities for terminal operations will also be required.

Centralized Maintenance will also require special facilities which must be provided by the State of Oklahoma.

It will be the responsibility of the Facilities Manager on the staff of the Program Director to obtain these facilities and prepare them in a timely manner.

3.8 Personnel and Training

Personnel for the management and operation of the trunked radio system will be identified and prepared for their assignments in coordination with equipment installation and test. It is intended that the Project Director and staff will continue their roles after the trunked radio system has been installed and accepted. The additional people required to operate the system will be recruited and trained during the system installation test so that the experience gained can serve as on-the-job training before the system becomes operational.

Job descriptions for staff positions will be developed by the Program Director and the personnel offices of the State of Oklahoma, Recruiting will be by public announcement of available job openings. It is anticipated that the positions will be filled by transfer of presently employed and skilled personnel.

The training and certification of all operating personnel, maintenance technicians, and management personnel will be a contractual obligation of the system contractor.

3.9 Post Implementation Support

The post implementation support program will address the funding and staffing needed for sustained operation of the trunked

3.9.1 Post Implementation Funding

The recurring costs of continued operation include the following major items:

III-26

Maintenance - maintenance of the facilities and system hardware/software after the warranty and first-year maintenance contract

Supplies - all expendable materials used in operating and managing the system

Training - refresher training and replacement training

Utilities - power, telephone, and other services provided by an outside agency

Equipment Replacement - periodic replacement of worn or obsolete equipment; equipment that can no longer be repaired economically.

These costs will have been identified and submitted to the FCC as part of the licensing requirements. They will be reviewed periodically. These will be billed to the user agencies in accordance with a formula developed and reported in each agency agreement.

3.9.2

The replacement of personnel and the up-dating of staff personnel will be a recurring problem. There will be labor turnover. Recruiting of personnel will be by suitable public announcement of job openings and the testing of applicants. Probationary employees will be trained by the technical staff.

3.9.3 Support Staff

The system will have the full time services of equipment maintenance personnel and software system personnel. At least one senior system programmer/analyst will be required. It is anticipated that a full time staff of specially trained maintenance technicians will be required and qualified control operators will also be required to provide 24 hour service to the user agencies.

Personnel - salary and benefits

Personnel Replacement and Training

	I ITTI	
i de la companya de l	1 111	
	and the second s	
	-	
	1 11 1	
	1	
		The system req
	f filment i fi	data) present some
		trunked system is t
	I III	considerations which
		more reutinely Th
	i thomas i	more routinery. In
	1	subsections.
	1 17	
		1. MAJOR SYSTEM P
	() U	
		The fellering
	1 57	The LOLIOWING
		presented as being
	E Land	trunked radio syste
		offered reflect sta
	1 11	not seriously viola
		offerings that can
		tive programment do
	<u> </u>	rive procurement do
	1 17	
		(l) <u>Wide</u> Area
	1 11-1	· · · · · · · · · · · · · · · · · · ·
	H	Present v
	l m	aurront ECC 1
	[] U_U	trunked system
	f	requirement fo
	I TT	unique hardwar
		of a multi-sit
	U_1	
	1	(2) Operation
	ll ann	(2) <u>Operation</u>
	K L.I	The avail
		upon the succe
,我们就是你们的你,你们就是你们的你,你们就是你们的你,你们就是你们的你们,你们就是你们的你们,你们就是你们的你们。""你们,你们就是你们的你们,你们不是你们,你	1 1	frequency synt
		legic orch of
	1 . Und	TOGIC Each of
		These circuits
	(and	subassemblies,
		permit at leas
이 같은 것이 같	L Lund	made available
		Estimatos by i
	PT-TH	
		or such a unit
		radios are ava
		will include o
,我们就是你们的你们,你们就是你们的你们,你们们就是你们的你们,你们就是你们的你们,你们们就是你们的你们,你们就是你们的你们。""你们,你们们不是你们,你们不是你	(n n	stations.
		Chatari da
	11	BLALEWIUE
	Inn	portable radio
		upon hand held
	비니	existing porta
		existing equip
,我们就是你们的你们,你们就是你们的你?""你们,你们就是你们的你们,你们就是你们的你们,你们就是你们的你们,你们就是你们的你们。""你们,你们就是你们的你们,你	Inn	trunkod radio
	11 LLL	very minimum c
	ll g n	
	1 7 7	Dresoding name blank
		Liecennig hade mauk
		en en la companya de la companya de La companya de la comp
\sim 2. The second s		
수가 있는 것이 가지 않는 것이 같이 많이 많이 많이 많이 많이 많이 많이 하는 것 같은 것이 같이 많이	and the second second	
,我们就是你们的,你们就是你们的,你们就是你们的,你们就是你们的,你们还是你们的你们,你们还不是你们的你们,你们不是你们的?""你们,你们不是你们的你们,你们不能		

APPENDIX A

SYSTEM REQUIREMENTS

quirements (as reported through the system survey major system problems which must be resolved if a to be considered feasible. There are other system ch, although of serious concern, can be accommodated nese are addressed individually in the following

ROBLEMS

problem areas and their proposed solutions are critical to the successful implementation of a em in the State of Oklahoma. The potential solutions ate-of-the-art system technology and hopefully do ate the state-of-the-art hardware and software be made available in the near future when competiocumentation has been developed.

 $\left| \left(\frac{1}{2} - \frac{1}{2} + \frac{1}{2}$

Coverage

vendor offerings in trunked system hardware and icensing procedures provide only for single-site m implementations. The statewide radio coverage or Oklahoma's trunked radio system will present ce and regulatory considerations for implementation te system.

with Portable Radios

Lability of trunked portable radios will depend essful development of a technically acceptable chesizer and associated control system switching small size and with small power requirements. s together with the usual 800 MHz portable radio , and a battery pack with sufficient capacity to st 8 hours of high performance service, must be e in a small hand held portable radio case. Industry sources place the earliest availability c in the 1983-85 time frame. Until such portable ailable, trunked radio system equipment configurations only base stations, control stations and mobile

e radio systems in Oklahoma are not generally o intensive. However, a few are heavily dependent d portable radios. This significant investment in able radio equipment must be protected. These oments must be permitted to operate with the system with no modification or, at the most, a of modification.

One alternative would be to have these conventional portables interface with the trunked radio system through a cross-band repeater. This repeater would have a two frequency half duplex VHF channel on one side and a frequency synthesized trunked radio control station on the other side. In the trunked system logic the cross-band repeater would have a single unit address for a cluster of portable radios all sharing the same VHF channel.

Trunked systems can not pass the system control information through to a cross banded portable. Therefore, the portable would not be able to detect an "all channels busy" signal and would not know why a channel request has been denied. To correct this the trunked control station side of the cross band repeater can cause an audible tone to be sent on the receive channel of the cluster of portables when trunked channel access has been denied. The denied request for channel access from a portable radio should not be placed in the queue for the trunked system (if a waiting queue is possible) even if a "channel available" signal is also transmitted to the cluster of portables. Because the cross-band repeater can not individually address each portable in the cluster, the portable radio queue would have to be a contention queue. Since all portables are in a contention queue for trunked channel access, the portable radio would have to manually repeat the request for a channel whether or not a "channel available" signal is provided.

The users of portable radios cross-banding into a trunked system would also have to learn to "quick call" the system for a channel request and wait for the request to be acknowledged before they start to speak. Also the "hold time" of the repeater/ base station combination will require special adjustment to prevent dropping the trunked channel during normal press-totalk switching through the repeater. These techniques have proved troublesome for field personnel to master.

A preferred solution to VHF/UHF portable operation is to continue using the portable radios on conventional VHF/UHF radio channels and manually cross patch these channels into the trunked system at the dispatcher's console when cross band coordination is required. Until trunked portable radios become available and have been field tested, the recommended solution for portable interface is to keep the conventional VHF and/or UHF frequencies for portable radio operation and use the trunked system for the mobile radios. The Oklahoma Highway Patrol, however, has expressed a requirement for cross-band portable interface with DATCS for mobile command post use.

(3) Satellite Voting Receivers

Radio propagation experience with 800 MHz trunked systems indicates that base-to-mobile and mobile-to-base coverage has been similar. The effective radiated power of the two types of stations is very much the same when the installation of both has been well engineered. Therefore, a base station/ control

III-30

station/mobile trunked system without trunked portables would likely need no satellite receivers or possibly need only one or two remote satellite receiver sites to correct a unique trouble spot and to vote with the base station receivers in a transmitter steering configuration. It will be the obligation of the system contractor to conduct radio propagation tests to minimize the need for any satellite voting receivers.

If trunked portable radios are eventually added to the system, the need for satellite receiver installations throughout Oklahoma will have to be reexamined.

(4) Multiple Dispatch Points

The State of Oklahoma land mobile radio system, with participation by all appropriate agencies, will require a large number of dispatch points to maintain the existing level of command over the field units. The assignment of an individual mobile unit to a specific dispatch point will be information that resides only in the system control equipment and can not be changed or initiated from the mobile unit.

In concept, the mobile unit in transmitting a request to access the trunked system would only transmit its own identity and possibly its assigned group address. From this address code, the controller can determine from its data file which agency the mobile belongs to, the dispatch point the mobile reports to at that particular time, and the group (or subgroup) the mobile will be permitted to include in its call-up. It is also conceived that some mobiles will have the limited capability of modifying the group request; this feature should be permitted for the singular purpose of reducing the dispatcher work load.

To respond to an incident which calls for a mobile command post and a mix of public safety/public service agencies responsible to that command post, the system controller can create a phantom group and the mobile units can be attached to that group by software instructions based on the mobile address codes.

(5) System Control Failure

An individual base station repeater is installed for each channel licensed to the trunked system at each appropriate base station site. Radio coverage problems may require the installations of repeaters for all channels at all base station sites. These individual repeaters are interconnected and controlled by a controller. If the controller fails completely, the repeaters can operate independently as conventional repeaters. They can be accessed by mobile stations and control stations that are operating on the assigned frequency for each channel of the trunked system. The mobile stations can be preprogrammed to home on an assigned channel if there is a loss of control signal or idle tone (depending upon the trunked system selected);

the available conventional mode 800 MHz channels will be distributed among the mobile fleet units according to need and coordination requirements. Access to the repeater from the dispatch point can be achieved by control station, wireline, or microwave.

Conversion to conventional mode of operation from a trunked operation can however result in a severe shortage of frequency resources needed to conduct routine public safety operations.

Failure of the control system in any mode other than a total catastrophic failure should result in a graceful degradation of the system capability. The failure should be limited to the loss of control over individual channels. This limited failure mode should be transparent to the system users. The loss of one or more channels should not effect the operation of the remaining channels.

(6) Time-out-Timers

A repeater configured land mobile radio system such as a trunked system can be paralyzed by accidental or intentional prolonged mobile or control station transmissions. To protect the trunked system from this failure mode, all transmitters of the base stations, control stations, crossband repeaters, and mobile stations must be equipped with time-out timers similar to the timers used in conventional repeater type systems. If Oklahoma's EMS systems are incorporated into DATCS, however, the use of time-out-timers on base station equipment must be carefully considered. Time-out-timer operation may interfere with transmission of the telemetry used in advanced lifesupport systems. Two categories of timing circuits should be specified; the first permitting an absolute maximum time to hold the channel with or without modulation, the second should provide a limited channel hold time without modulation.

(7) Distributed Logic Failure

The inboard logic functions assigned to control stations and mobile stations should be minimized; critical system logic and data files must be concentrated at the fixed site locations. Failure of the logic functions or RF functions at the mobile or control station should effect only that individual unit; such failures need not be alarmed to a control point.

(8) System Testing and Alarming

The complexity of a trunked radio system that will provide adequate communication servies for the State of Oklahoma will require a built-in capability to evaluate the individual channel quality and alarm back to a control point any degradation in channel quality below established threshold limits or other significant system degradations. This requirement is not unlike the remote testing and alarm capability used routinely in private microwave systems.

III-32

(9) Channel Quality Evaluation

To minimize catastophic channel failures, individual channel quality should be continuously monitored by one of the following methods:

Idle Channel noise threshold (no signal)

Busy Channel S+N threshold (signal or idle tone).

Departure from a predetermined threshold limit will be alarmed to a control point.

(10) System Test

į

1

The ability of the trunked system to respond to mobile/control stations and to process their message content should be tested periodically. It is recommended that programmable frequency synthesized test transmitters with known radiation characteristics and test message content (multifrequency tone and/or digital message) be positioned throughout the State at locations that provide maximum 800 MHz coverage to the high density population areas. These test transmitters can be programmed to transmit test signals on each trunked radio channel sequentially and the received test signal at each fixed receiver location in the area can be compared by quality and message content with a standard. The standard can be established at the time the test transmitter system is installe. Any degradation below established limits will be alarmed to a control point. It is preferable that this test system be peripheral to the trunked radio system.

(11) Test Alarms

The alarm system that alerts to any element of degraded performance should also provide preliminary diagnostic information. This level of diagnosis may be limited to the identity of the offending unit of equipment and the kind of offense being observed. This requirement should be part of the system procurement specification and its characteristics proposed by the system contractor.

The trunked radio system operating in a public safety environment should have the future capability of providing prompt channel access to mobile units exposed to life threatening situations or other incidents which require the immediate response of emergency services.

The general approaches to emergency priority capability for a trunked radio system are discussed in the following paragraphs. The conditions under which initiation of any of these priority functions would be authorized include all of the following:

(12) Emergency Priority/Ruthless Preemption

A routine request for channel has been denied by the trunked system because all channels are in use and

There has been a waiting queue established by emergency priority level for the next available channel (dedicated control channel systems only) and

The activation of the priority function has been a conscionable decision; i.e., there has been a manual switch closure by the user requesting a priority.

These conditions presume the implementation of a dedicated control channel type of trunked radio system. If all channels are in use in a distributed logic system, the mobile will be in a frequency search mode and will have no capability of addressing the controller; there will be no capability of requesting emergency priority considerations and there will be no capability of developing a waiting queue with or without priority. Therefore, the following three basic emergency priority alternatives of Ruthless Preemption, Dispatcher Alert, and Waiting Queue Priority refer only to a dedicated control channel trunked system.

Ruthless Preemption will result in the seizure of any radio channel (except the Control Channel) from its current user. For a reasonable and automated decision to be made, the assumptions must be that:

> The controller has a position indication that the seized channel was not carrying equally urgent emergency traffic at the time it was seized

The controller is able to select a channel carrying non-critical traffic.

There is no way of assuring that these assumptions can be realized. The normal mobile/control station identifier or preamble will contain only the unit address. There is no information in the identifier that would signify the urgency of a give communication even among those mobile units who, by their identifier, are known by the controller to have preemption capability.

To permit the ruthless preemption of an in-use channel appears to be a very difficult equipment problem, programming problem, and operational problem. This approach to emergency priority is not recommended.

The activation of a unique and manually initiated emergency priority request can be recongnized by the controller. An alert alarm message will be directed to the dispatch point assigned to the mobile unit making the request; this message will also contain the identity of the mobile unit. It is concievable that this dispatch point will have one or more of the system channels in use at that time the priority alert is received and can release one of these channels to the mobile that requested the priority. This assumes that:

ing traf

The dispatcher can "freeze" the channel long enough so that the mobile can be switched into it before the system drops the channel (otherwise the next mobile in the waiting queue will capture the idle channel).

It is possible to meet those requirements through hardware and software development. The complexity and estimated cost of the development constrains the recommendation of this method of enabling an emergency priority.

A more easily implemented approach to emergency priority is to move the priority request for channel access to the top position in the waiting queue without attempting to seize a channel. The success of this method depends upon the following assumptions:

seconds).

This type of emergency priority is inexpensive to implement in a dedicated control channel trunked system when compared to the other alternatives. Channel access to a well managed trunked system will normally be made within a few seconds even when the first call has been blocked. The Waiting Queue Priority system will insure that additional lost time is not incurred due to the size of the waiting queue. The assumption is that an emergency priority request from an agency will move that request to the first position in the queue unless there is already an emergency priority request in the queue. This method of emergency priority is recommended for the Oklahoma trunked system.

III-34

The dispatcher has information relative to the identity of the trunked radio channel being used by his agency and can identify this channel to the Controller so that the mobile can be instructed to move to that specific radio channel and

The trunked system is properly engineered and, in complying with FCC licensing criteria, the probability of blockage at peak busy hours in minimal

Channel discipline is such that channel "hold time" for the average communication transaction is no greater than a few seconds (hopefully less than 5

I. INTRODUCTION counsel should be consulted. 1. 2. between the parties. Federal regulation. II. SUGGESTED MODEL OUTLINE

APPENDIX B

SUGGESTED AGENCY AGREEMENTS

The agreement by two or more local government public-safety . agencies to enter into a joint enterprise which provides a mutual service of value to the participants is best documented by written agreements between the parties. By their nature, these agreements are highly individual, tailored to specific requirements of each program in each locality. This appendix to the System Implementation Planning Guide (SIP) is intended to suggest only a general outline of such agreements. The details and the order of presentation will depend upon each kind of enterprise contemplated, its complexity, and the immediate cooperative environment. Legal

Purposes of the Agreement

The purpose of a written agreement is to describe the reason for the cooperative effort and the results that are expected to be achieved through mutual participation. By reducing this to a clear and unambiguous written statement, the involved parties will fully understand their mutual participation. Equally important, the governing bodies, of which the participants are a part, will be able to determine the true nature of the program and the expected results and thereby provide their support.

Scope of the Agreement

The agreement should define the dimensions within which the terms of the agreement apply. The responsibilities and liabilities of each of the participants should be clearly documented. There should be no unwritten understandings

The following model outline of an agreement form provides a series of topical considerations to be evaluated during the development of an interagency agreement. The precise form of a specific agreement, and the language to be used in each case, should be provided by legal counsel. In many cases, standard legal form and substance is required by State or

The following topics and items are intended to delineate key considerations in the preparation of an interagency agreement.

General Terms and Legal Base 1.

The general terms will include the following as a minimum:

- The identification of each participating agency
- The requirements to be imposed on those agencies that may wish to participate at a later date

The concise purpose of the joint enterprise and the mutually advantageous results that are expected to be achieved

A statement of the legal authority under which the public agencies are able to contract with each other for the stated purpose; this may be a state law, a local government act, a series of resolutions passed by the local governmental authorities, or similar authorization.

2. Services to be Provided

The services provided by each agency and to each agency will be defined. These include the following considerations:

> Definition of the services to be provided by each agency to the mutual system; these services could include, for example, equipment maintenance, equipment, facilities, operating manpower, support services, and similar items of value to the system.

> Definition of services the system will provide to each participating agency; such services could include the dispatching of field forces, management reports, statistical data, and similar items of value to the agency.

The definition of the quantity or extent of each service to be provided.

The responsibility each agency has for delivering its service and the responsibility the mutual operation has for delivering its services to each agency.

The extent of any back-up or redundancy to assure that services are delivered in a timely manner throughout the life of the agreement.

3. Liability

A key element of the agreement between two or more public agencies is a clear definition of the liability each of the agencies has for the actions of other parties to the

III-38

	•	1 1 p: 0
4.	Am	ount
of t port part must will	The he ion icij be be	e pro serv: of patin spec def:
	•	Tl of
	•	Tł
	•	Th se ar th
	•	Tł
	•	Th or
5.	Fis	cal
tion of th in th follo	Sou of le p le a win	nd f publ rovi genc g el

tions.

2 -

1

2.1

agreement. This is particularly true if one or more of the agencies provides public services on behalf of the other agencies. Agreements to limit the liability exposure of participating agencies would include the following considera-

> Responsibility for defending the provider agency in potential lawsuits.

Indemnification of the providing agency against liability judgements against a participating agency.

Indemnification of the participating agency against legal actions or liability judgements against the provider agency arising from the normal provision of agreed-to services.

and Manner of Payment

oviding agency must be reimbursed for the expenses ice it provides to the participating agencies. A this obligation will fall to the individual ing agencies. The agreement between the agencies cific regarding this individual responsibility and initive of the following considerations:

he specific services to be provided and the cost f each service

he method by which the cost and value of each ervice is calculated

he means of payment for example: cash, in-kind ervices, cash equivalent in material or equipment, nd any other valuable consideration agreeable to he parties to the agreement

he time and manner that payments are to be made

he use of outside funding sources such as grants r gifts.

Procedures

fiscal planning is key to the successful implementaic agency cooperative programs. The responsibility ider agency for financial integrity must be defined cy agreements. The agreement will specify the ements:

The type of financial records and how they are maintained

- The contents of the periodic financial reports and the schedule for submitting these reports to higher authority and to participating agencies
- Access to the financial records by each participating agency
- Certification and/or audit of the records and financial reports
- Terms and conditions relative to the periodic reassessment of rates charged for each service provided.

6. Administration

The administrative structure of the cooperative program should be defined in the agency agreements. The authority and responsibility of each agency for participation in policy administration and the authority and responsibility for dayto-day operation should be included. Some of the key elements in the agreement are:

- . The identification of the person(s) authorized to sign the agreement on behalf of each party to the agreement
- The identity of the person(s) who will represent the participating agency and providing agency in policy matters
- The responsibility and authority of the providing agency to supervise project personnel and establish personnel policy
- . The responsibility and authority of the providing agency over joint-use facilities and equipment
- The retention by participating agencies of control, responsibility, and authority over their personnel, facilities and equipment that may be used in the cooperative project
- Identity and authority of persons responsible for the review of the agreements and for the determination of compliance.

These and other administrative elements suitable to the unique environment and nature of the proposed project should be negotiated between the parties and made part of the written agreement to participate.

III-40

Real Property and Equipment

7.

8.

tions are:

T

CHES

1

P

The cost-effective implementation of a multi-agency cooperative program should make maximum use of facilities, property and equipment available from the participating agencies. The agreement should identify these properties and materials; it should establish the rules under which the properties and materials are to be used. Some key considera-

> Identify each item of property and equipment that is to be transferred to the use of the cooperative system

Establish the value of the property and/or equipment and show the method used to establish the value

Provide for the return of the property and equipment upon termination or cancellation of the agreement

Define the method to be used to establish the value of property and material to be returned to the original supplier upon the termination, modification, or cancellation of the agreement.

Duration, Termination, and Amendments

The agency agreement must provide for modification to the agreement during its term. It must also define the duration of the agreement, make provisions for cancellation of the agreement, and disclose how the agreement is to be terminated. The following items should be included:

The term of the agreement

The circumstances under which the agreement can be terminated during its intended duration

The terms under which the agency can withdraw from the agreement

The circumstances under which either party can cancel the agreement

Procedures for amending the agreement during its intended term

Disposition of property and equipment upon termination or cancellation.

IJI-41

9. Other Provisions

.

.

.

•

g =

The agency agreement is a custom document generated to reflect the unique needs of the specific circumstance and the environment in which the mutual agreement is being generated. Each agreement will include terms and conditions of an individual nature. Key among these are the following suggestions:

- Method of resolving disputes; for example, arbitration, application of appropriate State or Federal laws, etc.
- Identity of the court of jurisdiction for adjudicating disputes.

R.

hand

0

Section States and States

Definition of terms used in the agreement.

Statutory status and filing of the agreement.

Interrelationship of agreements.

The drawing of interlocal contracts and agreements to provide public services or hold public property for the mutual advantage of multiple governmental organizations may be subject to local, state, or Federal regulations and/or laws. These documents should be drawn up under the advice of legal counsel to insure their status as legal instruments.

III-42

APPENDIX C

STATEMENT OF WORK

III-43
· · · · · · · · · · · · · · · · · · ·		م المحمد ال مس ور بالمحمد المالية الم	n - Arren and Arren a	$\omega = \omega_{0} \omega_{0} + \omega_{0} + \omega_{0} + \omega_{0} + \omega_{0} + \omega_{0}$	• • • • • • • • • • • • • • • • • • •		and an and a second the second s	
			an a					
		h						
						аны 4 ал		
								I. INTRODUCTION
					and a second		teeta	
								The Introducti a general overview
		$\varphi_{i} = \varphi_{i} + \varphi_{i$			and the second sec			by the responders.
							1000	in the following su
	1						1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1	
			1					1. Backgroun
				· · · · · · · · · · · · · · · · · · ·				A brief h
				2 1				procurement an
					-			2. ODJECLIVE
					an a			A brief d
					and a second			achieved by th
	e e e e e e e e e e e e e e e e e e e							3 Scope of
								3. <u>Scope or</u>
								A basic a
								services that
	1						μ	example:
						•	π	. Hard
					i ₩ · ·			• Soft
								. Svst
						. Vř		
								. Inst
				•				Tiest
					2° - 63		l	
					940). 	(1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,1,		, Trai
· .								Main
								• Mall
	· · · · · · · · · · · · · · · · · · ·				a da ante a compositor a compositor de la c			II. INSTRUCTIONS I
				•				
								This section c
•								
								1. Procureme
		n an				$\frac{1}{2} \left(\frac{1}{2} + \frac{1}{2} \right) = \frac{1}{2} \left(\frac{1}{2} + \frac{1}{2} + \frac{1}{2} \right) \left(\frac{1}{2} + \frac{1}{2} + \frac{1}{2} + \frac{1}{2} \right) \left(\frac{1}{2} + \frac{1}{2} + \frac{1}{2} + \frac{1}{2} \right) \left(\frac{1}{2} + \frac{1}{2} + \frac{1}{2} + \frac{1}{2} + \frac{1}{2} \right) \left(\frac{1}{2} + 1$		The sched
					5			request will i
- - -					9	an de la companya de Persoa de la companya		
	an a		2					. Time
							□	. Time
		a da ang kanalan na sa			الم الم المحمد الله المحمد الله المحمد ال المحمد المحمد		. Г	Dreading note hlank
								rieceuiig hage main
L				•		* *	•	
				ا میں ایک ایک میں میں میں میں ایک ایک میں ا 4 ایک ایک ایک میں میں ایک میں ایک ایک ایک ایک ایک ایک ایک ایک ایک م 4 ایک				$\mu_{\rm eff}$ is a set of the set

APPENDIX C

STATEMENT OF WORK

ion to the Statement of Work (SOW) should provide of the project to enable a better understanding The key elements of this appendix are outlined ubsections.

nd of the Project

nistorical description of the origin of the nd the constraints imposed on the design.

e of the Procurement

discussion of the main system objectives to be he procurement.

the Procurement

and brief outline of the hardware, software, and are being purchased. These may include, for

dware

tware

tem Design

tallation

ting

ining

ntenance.

TO BIDDERS

of the SOW will provide the bidders with all the to respond correctly to the bid request.

ent Schedule

dule time and dates for responding to the bid include:

e and place of the pre-bid conference 🕤

e and place bids and proposals are due

				778750	
			موند محمد روانور مدر این م		መከረ
			de la constante		. Ine man
•	Bid and proposal opening time, date, and place		مناسبهم والمراسم		5. Proposal
	Proposed contract award date	<u> </u>	an , a can see the set of the	m	
•	Expected date of contract completion.		and the second		evaluation of
2. <u>A</u>	authorized Contracts	Section Section			. Tec
T reques may be	the persons who are authorized to respond to bidders' ts will be identified prior to bid opening. These restricted to the:				. Cos . Man
•	Procuring officer		aa provinsi kun fiyoto, Aner		These cr
	Technical consultant.				that each bid requirements
T raised	the statement should be made regarding how the issues by these contacts will be reported to the other bidders.				6. <u>Contract</u>
3. <u>B</u>	and Proposal Requirements				the successfu awarded witho
E bid wi the de	ach of the items to be submitted as part of a responsive Il be defined. Any limitations should be included in scription. These items would include, for example:				7. Indemnif
•	Itemized cost proposal				The purc damages impos
	Technical proposal				the contractoned indemnif
•	Management proposal	T. I			items or proc such as insta
	Provisions for alternate proposals				indemnificati contractors'
	Proposal limitations regarding page count, standard- ized forms (cost proposal), proposal size, unneces- sary artwork, and number of copies to be submitted.	t ggi Cilibanana ya			8. <u>Titles a</u> Rights c
4. <u>c</u>	ualifications of the Bidders	Sound "Sound State	тС,		clearly defin related to:
T qualif requir the SC	The contractor for a high technology system must be ied, technically and financially, to complete the ed work. These qualifications should be expressed in W by requiring reasonable responses to limits imposed				. The . Rig as
on:					Pie
•	Bidders' facilities and financial resources				• KIQ
	Performance history in terms of references and a list of similar and recent contract work				• The
	Support facilites such as field maintenance shops,	in state of the st			. The
	training facilities, and spare parts replacement depots		1997 - 1997 1997 - 1997 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1 1997 -		Legal co
	III-46				and wording c

e quality and availability of personnel for nagement, supervision, and technical staff.

l Evaluation Criteria

a should be established for the comparative of:

chnical proposals

st propusals

nagement proposals.

riteria should be clearly defined in the SOW so dder will understand the emphasis placed on the s of the RFP.

t Award

will identify the type of contract to be awarded bidder whether or not this contract will be out pre-the ct discussions.

fication

chaser should require indemnification against osed through intentional or unintentional acts of or. High technology system procurements will fication against the improper use of patented ocesses. All procurements which include services callation and test will be well advised to require tion from damages resulting from the actions of employees or agents.

and Rights

of the purchaser and ownership of titles will be ned by the SOW. These items include statements

he transfer of title to equipment and software

Ights to data and software developed or provided s part of the contract

ghts and title to the submitted proposals

he right to reject proposals

he right to negotiate with any bidder

he right to cancel negotiations or the contract.

counsel should be consulted for the scope, content, of this subsection of the SOW.

9. Bonds, Insurance, and Warranty

Each governmental entity will have its own limits for, and description of, the bonds, insurance and warranty provisions for contractor provided material and services. Frequently there will be state regulations which also must be adhered to. These requirements will provide for:

- Performance bonds
- Workman's compensation
- Comprehensive liability insurance
- . Warranty on workmanship and materials.

The purchasing department of the major governmental agency included in the program will provide counsel as to how these requirements are to be included in the SOW.

10. Costs of Proposal Preparation

The SOW should state clearly that any costs related to the preparation of bids or proposals can not be included in the proposed cost of the project or billed back to the purchaser in any manner.

11. Contract Disputes

At the option of legal counsel, the SOW may indicate the means by which the governmental agency resolves contract disputes. This will be a function of the governmental body and will not be an option of project management. These provisions may include:

The procedure and method of resolving contract disputes

Termination of contracts for cause

The assessment of liquidating damages

The identity of the court of jurisdiction.

12. Applicable Federal/State Laws and Regulation

When State or Federal laws/regulations affect the system design, its operation, or allowable use, these laws or regulations should be identified in the SOW.

III-48

III. SCOPE OF WORK

1.

2.

Contraction of

in and the second second

This section of the SOW defines the items of work expected of the contractor; the specific tasks required by the contractor are described in detail.

Services to be Provided

High technology systems will require professional services to be performed by the system contractor in addition to the delivery of hardware and software items. Typical services to be provided are:

> System design which can include system engineering, radio propagation engineering, software development, and hardware development.

> Installation and debugging of hardware and software.

Assistance in procuring required licenses and permits.

Developing and conducting performance tests and acceptance tests.

The preparation of sites and facilities which can include construction, HVAC, emergency power, lighting, installation of utility services, etc.

Deliverable Items

Each deliverable item should be identified in the SOW. Deliverable items include hardware, software, and services. The list will be unique to the complexity of the system to be procured. However, there will be some items common to all lists such as:

Each item of electronic or radio hardware

Each item of computer software

Maintenance support

Spare parts and spare module provisions

Towers, buildings and other structures

Manuals and documentation such as test plans, test procedures, maintenance manuals, operators manuals, user manuals, training aids, installation drawings, and test reports

Progress reports and deficiency reports.

3. Delivery Schedules

Each deliverable item should be supported by a delivery schedule. The SOW will require a formal and contractually obligated schedule for:

Each item of electronic or radio hardware

Each item of computer software

Each tower or building

The completion of all facility preparations

Delivery of all manuals and documents

The time and place of technical progress reviews

The delivery of reports including progress reports, test reports, discrepancy reports, and the final report or acceptance test report.

IV. SYSTEM SPECIFICATION

The system specification is a complete technical system description which includes the system performance, system interfaces, and system configuration. It places technical limits on these parameters and adds reliability and/or maintainability requirements where appropriate.

1. Hierarchy of Specification and Standards

Industry or regulatory specifications or standards frequently apply to the specification of a system. These should be identified. The hierarchy of these documents, as they apply to the system specification, should be defined so that conflicts between them can be resolved.

2. System Description

The system description includes the configuration of the system elements and their interconnection with each other. It is recommended that this be done graphically with supporting written clarification. The system description is derived from the system concept design.

3. System Interfaces and Supporting Services

The means by which the system interconnects with the world outside of the system should be fully described. These interfaces are relatively unique to each individual system, but some key interfaces will include:

III-50

External communication systems such as mobile radios, private telephone systems, data processing, digital data systems, alarms, microwave transmission systems, and similar facilities

Utility systems such as public access telephone systems commercial power systems, and water supplies

Emergency operations such as emergency power services, fire protection, emergency medical services, and civil disaster prevention operations.

System Software

4.

5.

6.

7.

The system specification will include a complete description of the operations software and the applications software required to provide the needed system performance.

System Operation and Performance

The specification of system operation and performance is tailored to meet individual system requirements as developed in the system concept. Typically these specifications will include a complete description of:

> The functional modes of system operation for routine system performance, emergency system performance and allowable fail-safe or degraded levels of performance.

The calculated levels of personnel staffing for the operating staff, maintenance staff, and system management.

Expansion Requirements

 $\langle \cdot \rangle$

Frequently, limitations on funding or other resources will prevent the initial implementation of a sophisticated system from realizing the full potential of system operation, scope, and performance. When these expansion requirements are part of the initial system procurement, they must be fully described in the system specification.

System Licenses and Approvals

High technology systems may exceed the apparent limits of Federal or state licensing regulations and extraordinary procedures may be needed to acquire the necessary permits or licenses. These requirements should be clearly defined in the system specification. They will be unique to each specific system configuration and intended operation.

V. EQUIPMENT SPECIFICATIONS

The equipment specifications define the quality and performance characteristics of each item of hardware and software needed to meet the system specifications. The identity of each hardware and software item is derived from the system configuration diagram which was developed as part of the system concept.

1. General

.

.

This subsection of the equipment specifications accumulates all items of the individual equipment specifications requirements that are common among all items of equipment or software. It also defines the quality and quantity of each item of hardware and software that is required to meet the system configuration. Typical items that would appear under the general heading are:

Applicable industry standards or publications and their hierarchy for this program

Standard environmental conditions such as temperature, humidity, primary power, vibration, shock, dust, and dirt 51

1

[]] [

 \square

153

Deliverable equipment lists showing the quantity required of each item including spares

The required level of quality and workmanship for each hardware item.

2. Individual Item Specifications

Each equipment item will have an individual technical specification which reflects its construction and performance as required to meet the system specification. These are derived individually from the severe than those needed to meet the system performance specifications. Software specifications are typically combined with the equipment specification for the computer subsystem or the switching subsystem in which they are resident.

III-52

IV. PHOENIX, ARIZONA

System Implementation Plan _ Draft

Submitted to:

ł.,,

1.5

1.1

L.

L

Mr. John E. Simmons Project Director Phoenix Fire Department 620 West Washington St. Phoenix, Arizona 85003

DIGITALLY ADDRESSED TRUNKED COMMUNICATION SYSTEM

્રં ડ્

SYSTEM IMPLEMENTATION PLAN

PHOENIX, ARIZONA

November 15, 1980

 Ω

÷., .

BOOZ · ALLEN & HAMILTON Inc.

776 SHREWSBURY AVENUE TINTON FALLS, NEW JERSEY 07724 74799303 AREA CODE 201

	-			
I. 1. 2.	INT Bac Pro	ROI kgr jec	ouc ou t	T In D
II. 1. 2. 3. 4. 5. 6. 7. 8. 9. 10. 11. 12.	SYS Par Non Adj Bas Fre Add Con Sys Fac Est	TEN -Pa ace e gue ile tro ten ili ima	f C iptent Start Stars (C) () () () () () () () () () () () () ()	loai tycopped
III. 1. 2. 3. 4. 5. 6. 7. 8. 9.	REQ Lic Pre Imp Sys Pro Lic Fac Per Pos	UII al len ten cun ens ili sor t	RED GC nir ner ner sir Lti Imp) vaita ia ia ia ia ia ia ia ia ia ia ia ia ia
APPEN	NDIX	A		S
APPEN	DIX	В	-	D
APPEN	NDIX	С	• • •	С
APPEN	1DIX	D	_	S

T

1

1

TABLE OF CONTENTS

	Page
ION	IV-1
l and Objectives	IV-1
rection and Participants	IV-2
TETETIPATTON	TV-5
ing Agencies	TV-5
insting Phoonix Agencies	TV-6
Community Public Service Agencies	TV-6
Community Public Safety Agencies	TV-7
on Configuration	TV-7
Dlan	TV-11
	TV-13
trol Stations	TV-13
vinte	IV-14
pration	IV-16
and Derconnel	TV-22
Costs	TV-24
COSCS	
MPLEMENTATION ACTIVITIES	IV-31
ernment Approvals	IV-31
v Licensing Considerations	IV-33
ation Schedule	IV-33
agement and Operation	IV-34
nt	IV-35
	IV-37
Acquisition and Preparation	IV-38
and Training	IV-38
ementation Support	IV-39
aggested Agency Agreements	IV-41
ATCS System Requirements	IV-49
irrent System Inventory	IV-59
catement of Work	IV-97

ii

program and and a second second second	
	This system i
	trunked communicat was prepared by Bo safety Communicati
	1. BACKGROUND AN
	The Second Re Docket No. 18262, cation (FCC) in 19
	difficult problems community. These portion of the spe
	questions about th public-safety oper benefits inherent
	Under LEAA Grant N Communications Off problems, make app
	describe a program these new concepts
	first of these (Pr and regulatory fac portion of the rad
	system problems. operational capabi incorporated into
	tance to two citie (Bucks, PA and Sal a feasibility stud
	stration systems. Since Novembe
	16B as one of the achieved are:
	. To estab
	municipa metropol
	. To addre problems
	and/or o within t
	and a second

I. INTRODUCTION

implementation plan for a digitally addressed tion system (DATCS) for the Phoenix, Arizona area boz, Allen and Hamilton and the Associated Publicions Officers, Inc. (APCO) in response to LEAA K-0013.

ND OBJECTIVES

eport and Order in the proceedings related to published by the Federal Communications Communi-74, promised relief for the increasingly more s of frequency congestion in the public-safety proceedings allocated 200 channels in the 800 MHZ ectrum for "trunked" systems. In February 1977, nat many technical, economic, and management he application of this newly available spectrum to rations must be answered before the potential in trunked systems could be made available. No. 77-SS-99-6009, the Associated Public-safety ficers, Inc. (APCO) was requested to analyze these propriate recommendations for future actions, and a to demonstrate the potential capabilities of

programs under APCO Project 16 was initiated. The coject 16) resulted in an analysis of the technical ctors affecting the applicability of the 800 MHZ dio spectrum to law-enforcement communication A second program (Project 16A) identified specific ilities and functional requirements that should be a public-safety trunked communication system. (Project 16B) provided specific planning assisbe (Lexington, KY and Phoenix, AZ), two counties It Lake County, UT), and the State of Oklahoma for dy and basic implementation plan for DATCS demon-

er 1979, Phoenix has been participating in Project demonstration cities. The key objectives to be

cmine if DATCS can relieve municipal radio channel ion in the Phoenix area

olish the potential of DATCS to accomodate future al radio communications needs of a rapidly growing litan area

ess the solution of the complex command/ control s confronting the public-safety agencies

lop a common DATCS to support all public safety other governmental communications requirements the Phoenix metropolitan area

			Each of the above
	between governmental agencies appropriate to their mission responsibilities.		agencies have been act to the interagency and also the requirements
Othe dynamical	er advantages of operational flexibility are inherent in a ly programmed digitally addressed communication system.		It has been the I develop this System II
system re	equirement.		assistance of Booz, A
2. <u>PRO</u>	VECT DIRECTION AND PARTICIPANTS		been voluntary. There
The	project was under the direction of:		bodies.
•	John E. Simmons, Project Director Phoenix Fire Department		
•	Paul W. Salter, Deputy Project Director Equipment Management Division, City of Phoenix		
The	participating public-safety agencies include the following:	(Free 1)	
• • •	Phoenix Police Department, Chief Seth Allen Mesa Police Department, Mr. Robert Gates		
•	Glendale Police Department, Capt. Jerry Eldridge Scottsdale Policy Department, Mr. William F. Jordan Tempe Policy Department, Mr. Bill Pardee		
•	Phoenix Fire Department, Chief Thomas F. Sawer Mesa Fire Department, Mr. Robert Gates Tempe Fire Department, Chief R. E. Stayner		
	Glendale Fire Department, Chief Bob Gilbreath Special Emergency (EMS), Chief Thomas F. Sawer	F	
Othe and servi	er participating governmental and public service departments .ces include:		
•	Phoenix Building Safety, Mr. Max P. Wold Phoenix Civic Plaza, Mr. Dave Schupback		
•	Phoenix Human Resources Department, Mr. Marvin Boles Phoenix Aviation Department, Mr. Robert Erickson Phoenix Public Transit Administration, Mr. Ed Colby		
•	Phoenix Engineering, Mr. William S. Wood Phoenix Parks and Recreation, Mr. Cecil P. Mason Phoenix Traffic Engineering, Mr. Bill Bain		
•	Phoenix Traffic Signals, Mr. Walt Cicioni Phoenix Facilities Maintenance, Mr. Norman Smith Phoenix Streets Maintenance, Mr. Dick Shine		
	Phoenix Sanitation Department, Mr. Harry I. Kelman Phoenix Equipment Management, Mr. Sam Collett Phoenix Sewers Department, Mr. Robert B. Steytler		
•	Phoenix Water Distribution, Mr. Lyle E. Orr Phoenix Water Customer Service, Mr. Jack Blaha	67	
	Mesa Public Works, Mr. Robert Gates Glendale General Services. Mr. Dave Hall		
•	Tempe Public Works, Mr. Bill Pardee Tempe Sanitation/Refuse, Mr. Bill Pardee		
	IV-2		
an - Same The action of the second state of the se			

ve listed public-safety and public service ctive participants in the program, contributing nd intra-agency communications analysis and s analysis for the DATCS concept design.

responsibility of the Project Director to Implementation Plan (SIP) with the technical Allen as engineering consultants to APCO. The articipating agencies and their personnel has are have been no binding commitments required of or any of the other participating governmental

U_U
()
The basic DAT
has been conceived
response between t
use and the DATCS
been eliminated.
This philosop
trained skill on t
the system concept interfaces in which
might result in a
(_) systems.
The configura
part on information
systems. It is no
have not been prev
form. Further stu hardware and softw
the system configu
1. PARTICIPATING
Condidate age
who are mobile rad
are portable radio
communities adjace
resolved. The fol.
I) Phoenix
The follo
by their method
Eng
Par
Tra
Tra
Fac.
Hum
Equ
Preceding page blank

II. SYSTEM CONFIGURATION

CS land mobile radio system for the Phoenix area to replicate current governmental radio system ely as possible. The difference in function and the conventional radio system configuration now in radio system should be nearly transparent to the cept that co-channel sharing will appear to have

by directs that operations which require a unique, the part of the mobile operator be minimized in . It also directs the avoidance of subsystem th trunked technology or switching requirements degraded performance of existing communications

tion of the Phoenix area DATCS is based in part irements discussed in the previous reports and in on provided by existing suppliers of trunked radio ted that the equipments described as part of this uration are not standard production items and iously provided by the vendors in this definitized dy by the vendor community may modify the final are offerings. This could result in changes to tration.

AGENCIES

ncies for participation in DATCS include those io intensive (full participation) and those who intensive (partial participation). At this o a question regarding the participation of the int to Phoenix - a question that has not been lowing subsections will discuss these groups

Public Service Agencies

owing list includes those City of Phoenix governthat are mobile radio intensive and who would, od of operation, profit most from DATCS features:

ineering

ks and Recreation

ffic Engineering

ffic Signals

ility Maintenance

an Resources

ipment Management

Sanitation

Streets Maintenance

Job Stimulus Program

Building Inspection

Sewers

Water Production

Water Distribution

Water Customer Service.

Together, these 15 agencies report having 1,087 mobile units.

2) Phoenix Public Safety Agencies

The Phoenix Police Department and the Phoenix Fire Department are portable radio intensive and plan to expand the use of mobile data terminals (MDT at UHF). This restricts the use of DATCS to those police and fire mobiles engaged in field operations or command operations in which the portable radio is not the primary means of communications. It is estimated that DATCS mobile radios could be installed in approximately 25 percent of the police and fire vehicles - a potential total of 230 vehicles.

2. NON-PARTICIPATING PHOENIX AGENCIES

in DATCS because of unique requirements or limitations imposed on their radio systems:

Aviation Department

Public Transit Administration

Special Emergency (EMS)

Civic Plaza.

3. ADJACENT COMMUNITY PUBLIC SERVICE AGENCIES

Thefollowing agencies are mobile radio intensive and, by their method of operation, would profit most from DATCS features. However, at this time the question of their participation with Phoenix is unresolved.

Glendale General Service

Mesa Public Works

Mesa Public Sanitation

IV-6

•						
				Теп	ipe	Ρι
				Пот	~	c.
				Ten	ipe	50
	unit	To s.	get	her	, t	:he
	4.	ADJ	ACE	NT	CON	íM
	radi in D	The .o in DATC	e p nte S.	ubl nsi Th	ic ve, ese	Sa , a a a
				Gle	nda	ale
				Gle	nda	10
				Mes	a I	20.
				Mes	a I	?i)
				Sco	tts	sđa
				Tem	pe	Po
				Tem	pe	F:
	in t more appr	If the a that that that	in abo an mat	ter ve 10 ely	age pul pei 2(end ol: cce
	5.	BA	SE	STA	TIC	DN
	adja base Moun MHZ	The icent stain tain tes	e p t c ati n i ts	orop omm on s d hav	ose uni loc riv e y	ed ti zati zei zei
	rece are cont it.	The ive show rol De	e s co wn ch tai	ite ntr in ann 1s	s a ol Fic el of	t cl ju in tl
	capa tion simu cont othe	The ible is. iltan rol er co	e s of Th nec ch ont	ite op usl ann rol	era wo y c el cł	ond co un nan

ublic Works

anitation and Refuse.

ese agencies report having 623 mobile

UNITY PUBLIC SAFETY AGENCIES

afety agencies adjacent to Phoenix are portable and it is assumed unlikely that they will participate agencies include:

e Police

e Fire

lice

re

- -

ale Police

olice

ire.

cy coordination required the installation of DATCS ic safety mobile units, it is estimated that no ent of the public safety vehicles would be included mobiles, maximum.

CONFIGURATION

DATCS base station configuration for Phoenix and ies is shown in Figure II-1. The selection of tions at North Mountain, Thompson Peak, and South in by radio propagation experience at 450 MHZ; 800 it to be initiated.

North Mountain and Thompson Peak transmit and hannel information. Details of these installations re II-2. The site at South Mountain receives nformation from the mobiles but does not transmit his installation are shown in Figure II-3.

troller at each site is fully programmable and ing independently based on control channel instrucontrol channels transmit the same information different frequencies. Mobiles will home on one intil the signal is lost then seek and lock-on the nnel.







face switching as required. control operation center. unit. FREQUENCY PLAN 6. (Part 90, Section 90.375).

The site controllers are connected through modems and an RS-232 interface to the master controller and master control terminal at DATCS control operation center. This terminal can make program changes in the site controllers, selectively turn off or turn on individual channels, key up individual transmitters, change the address code responses, and perform interagency coordination interface switching as required.

Alarm functions generated by the site controller are connected through modems and an RS-232 interface to the alarm board in DATCS control operation center.

Receiver voting and transmitter steering information from each DATCS voice channel at each site are brought to the voter selector equipment at the DATCS control operation center. The control channels are not voted; voting is suspended from the back-up control channels when they are selected by the site controller.

Audio signals into and out of the control channel receiver and transmitter at each site are interconnected at the control channel audio unit in the DATCS control operation center. Any bridging, amplification, attenuation, or balancing required is done by this

The DATCS control operation center coordinates and controls the operation of the three base station sites and provides an interface for the restricted control terminals located at user agency locations. The master controller provides logic and switching for the multiple drop circuit to the agency terminals, and routes management information generated by site controllers to the journal recorder. The master controller is commanded by the master control terminal which can address the programming and coding of the master controller and the site controllers.

The journal recorder is a magnetic tape transport that records all management information generated by site controllers, alarm board, and master control terminal. It prints out selective reports on its printer on command of the master controller. It is intended that most management reports, except for spot reports, will be generated off-line from the journal recorder tapes.

A proposed frequency plan for a 20-channel area DATCS for Phoenix is shown in Figure II-4. The reported number of public service mobile units in Phoenix and adjacent communities are more than enough for 70 percent loading of a 20-channel mixed service group trunked radio system under present FCC Rules and Regulations (Part 90, Section 90.375).

The distribution of channels for North Mountain, Thompson Peak, and South Mountain is shown in Figure II-4. Channel 3 is the back-up control channel for North Mountain, and channel 4 is the back-up control channel for Thompson Peak. If a control channel failure (or deactivation) will require the use of channel 3 or 4 as a control channel, the transmitters and receivers for the back-up channel selected will be exempt from voting and transmitter steering.



ŝ

 $2 \lambda_{2}$

 \mathcal{O}_{c}

			7. ADDRESS CODES
		G 	The digital ac units has two parts address. The comb site controllers and and mobiles to inc mode operation, al
		3	dispatch point) with mobile communication type conventional required such as in mobile communication code enables unique
			instruction to the control station can
			Radio access
			is not contemplated tional repeater typ
			1) <u>Mobile S</u>
			The conf: in Figure II- reception is of the licensed logic decodes
			activating the address assign has a half-duy a duplexer.
			Mobile d
			Hand rec
α. 			. Ind cha
			. Ind . Ind
			Con The mobi
			address code However, the switchable fr
	3		

address of all mobile units and control station ts, an individual unique address and a group bination of these two address codes informs the and the master controller which control stations clude in the request for a channel. Under normal 11 mobiles assigned to a control station (group ill be included in any group call. Mobile-toion within the group is achieved as in any repeater radio system. If temporary modification is interagency coordination, intergroup mobile-toion, the individual address portion of the address ue routing by agency terminal or master terminal e controllers. Similarly, an individual mobile or an be denied access to DATCS.

OL STATIONS

to DATCS is accomplished by mobile stations and Direct wireline control to the site controller ed; the mode of operation is similar to a convenype radio system.

Stations

figuration of all mobile stations is shown symbolically -1. The frequency selection for transmission and determined by frequency synthesis constrained to channels by a programming plug or PROM. Switching s the information provided by the control channel, he channel change switching and displays if the gned to the unit is recognized. The mobile unit uplex configuration, using a T/R switch instead of

displays include the following:

ndshake indicating a request for channel has been ceived by the system

dication that the request has been accepted and a annel has been made available

dication that the request has been placed in queue

dication that the mobile unit has gone beyond the ntrol channel range.

ile radio has no external control over its programmed and has no capability to add additional codes. mobile can have multiple program plugs or PROMs rom the control center.

2) Control Stations

Control stations are the means by which control points and dispatch points access DATCS. There are no wireline or microwave control links to the base station site controllers. The configuration of control stations is shown symbolically in Figure III-1; it is identical to the configuration of a mobile station.

The displays at the control station differ from those of the mobile station. The control station displays include the mobile station displays in addition to the following:

- Code identity of the mobile unit in communication with the control station
- Code identity of units in the "calls waiting" queue for that control station
- Identity of any mobile in the "calls waiting" queue which has exercised an emergency priority
- Status data display of units assigned to the control station which has a status encoder.

The control station may have the capacity to change its group address or to receive traffic from more than one group. This can be achieved by switching program plugs or PROMs. By exercising this option, a control station can change dispatch responsibilities to accommodate after-hour coverage.

9. CONTROL POINTS

The configuration for DATCS results in two types of control points: the master system control at DATCS control operation center and the control point at each agency.

1) Master Control

The master control is located at DATCS control operation center as shown in Figure III-1. The function of the master control is to monitor system operation as reported by the equipment status and alarm data provided by site controllers and to respond to alarm status with corrective action.

By means of the master control terminal, the programming of the master controller and site controllers can be modified to achieve required changes to system operation. A journal recorder and printer delivers hard copy of system spot status and changes initiated by master control.

votin alarm	ıg s ı bo	sel bar	.ec	t
	Th:	is	pc	s
2)	Age	enc	y	Ċ
	Age	enc	y	c
and,	ın	SC	me ∽	e (
LO LI	le i	lias	ite	ι. ΈΓ
contr	OT	pc) I I	
polic	e a	and	l 1	:1]
a còn	tro	51	st	a
servi	.ce	aq	rer	C:
locat	io.	າຮ້	re	ef:

Those agencies that would probably have control stations and the estimated number of control stations that each would have is shown in the following listing:

Agencies

Phoenix:		
	EOC	7
	Police	6
	Fire	4
	Engineering	- 4
	Traffic Engineering	1
	Traffic Signals	1
	Facility Maintenance	2
	Human Resources	1
	Equipment Management	1
	Sanitation	4
	Street Maintenance	2
	Job Stimulus Program	1
	Building Inspection	2
	Sewers	2
	Water Distribution	2
	Water Customer. Service	· 1
	Parks and Recreation	2
Glen	dale:	
	Fire	1
	Police	2
	General Service	2
Mesa		
	Police/Fire	3
	Public Works	6
	Public Sanitation	4
Scot	tsdale:	
	Police	1
Temp	e:	
	Police	2
	Fire	2
	Public Works	3
	Sanitation/Refuse	1
	TV-15	

The alarm board will alert the master control to faulty operation of any major unit at any of the three base station sites. Faulty operation of the master controller and the ors on each voice channel is also shown on the

ition is viewed as being manned 24 hours a day.

ontrol Points

ontrol points are identified as control stations cases, control terminals on a drop-line circuit controller. Some agencies may have multiple s colocated at a single dispatch center. The re dispatch centers in Phoenix may have access to tion at each dispatch position. Other public ries may have separate control stations at several flecting the multiple control point configuration of their command and control operation.

Number of Control Stations

The agency data terminals are restricted to those agencies that have the need to change fleet assignments frequently and have the technical skills to operate a computer terminal. Each terminal would be restricted by software protection to the activities of the agency in which it is installed. Candidate agencies for terminals include:

Phoenix Police Phoenix Fire Phoenix EOC Phoenix Engineering Phoenix Streets Maintenance Phoenix Water Production Phoenix Water Distribution Mesa Police/Fire Mesa Public Works Scottsdale Police Glendale Police Tempe Police Tempe Public Works

10. SYSTEM OPERATION

The philosphy of system operation requires that the current command and control procedures of the participating agencies be largely unaltered in DATCS operation and, to the degree possible, the functions of DATCS will be transparent to the mobile radio operators. The system concept and operation are based on that philosophy except to resolve existing deficiencies or to employ digital system technology where it is found to be desirable. The following subsections will address DATCS operation only and will not review the existing operating procedures of the candidate agencies.

Control Station/Mobile Station Operation 1)

The half-duplex configuration of control stations and mobile stations is symbolized in Figure II-1. The main difference is the degree of control each can exercise and differences in the output displays. Each responds to control channel commands and each accesses the control channel(s) in the same way. Both have frequency synthesizers which span the entire 800 MHz trunked radio spectrum and are constrained to the licensed channels by a replaceable code plug or PROM. The code plug or PROM will also determine the unit address and the group address.

Control Stations

The control station is the means by which the dispatch center or control point accesses DATCS for communication with its mobile fleet and possibly other control stations. There is no need for wireline or microwave control links between the dispatch points and the base station sites.

IV-16

_

 \bigcirc

The control stations will have more control flexibility than the mobile stations. This flexibility includes manual switching of code plugs or PROMs to effect changes in fleet configuration or for interagency coordination. The public safety control stations will also have an emergency priority request capability.

The control station idles on the system control channel. When initiating a call, the control station will be switched to an unoccupied channel together with all mobile units or control stations having the same group address. A mobile unit or control station in the same group initiating a call will cause the control station to switch to the same unoccupied voice channel assigned by the site controller.

The displays of a control station will include:

- Handshake indicating a request for channel has been received by the system
- Indication that a requested channel has been assigned
- Indication that a channel request has been placed in the waiting queue
- Code identity of the mobile unit in communication with the control station
- Indication that there are units in the "calls waiting" queue
- Indication that a unit in the "calls waiting" queue has exercised an emergency priority request.

If the control station is part of a dispatch center for public safety operations, the following displays should be added:

- Status display for those mobile units provided with a status encoder
- Code identity of the units in a "call waiting" queue
- Code identity of a unit which is exercising an emergency priority request and is either in the "call waiting" queue or in communication with the control station.

Mobile Stations

Mobile stations are similar in configuration and function to control stations. Displays are limited to the following:

- Handshake indicating a request for channel has ---been received by the system
- Indication that a requested channel has been assigned
- Indication that a channel request has been placed in the waiting queue
- Indication that the mobile is beyond the range of a system base station.

Mobile controls and switching capability are limited. The mobile will not be able to change its address codes or group assignment. Mobiles can be provided with multiple code plugs or PROMs which can be switched by the master controller. Mobiles can also be provided with a "repeater talk-around" switch for use outside the Phoenix area; public safety mobiles can be provided with an emergency priority switch. Those agencies with a central dispatch and/or CAD can also add a unit status reporting encoder to their mobiles to drive a status display at the dispatch center and update the CAD status displays.

2) Base Station Operation

Base station operation is similar to conventional radio repeater type operation except for the site controller functions. There is a conventional repeater installation for each DATCS voice channel at each base station site. There is a conventional repeater installation for the control channel at North Mountain and Thompson Peak. A digital site controller manages the DATCS functions at each site. The basic functions of the site controller are:

Recover and decode inbound control channel requests

Select and assign unoccupied voice channels as requested

Generate and encode outbound control channel signals directing system users to assigned channels

Monitor voice channel activity

Generate subaudible or digital code information on voice channels to unmute authorized receivers

Control base station repeaters

IV-18

Monitor and report to alarm panels all base station abnormal conditions

Provide system management interface for the master controller and master control terminal.

The remote master controller and master control terminal will have a wide range capability to modify the applications program in the site controller.

3)

T

۲. o

. .

Mobile-to-mobile communication is inherent in a repeater type operation. Within the group address, all mobiles will hear and be in communication with all other mobiles in the same address group. Intergroup mobile-to-mobile communication can be initiated by the dispatcher or master control terminal using the individual unit address. These one-time intermobile links can be set up by keyboard entry or by cross-channel patching.

4)

The ability of the mobile to talk around the repeater car-to-car in a half-duplex operation (single frequency simplex in both directions) is a manually selected option of the mobile control head. The option is automatically disabled within the range of a control channel signal. Outside the range of a control channel signal, the option (if selected manually) will place the mobile in a single frequency halfduplex mode on a frequency previously programmed from among the frequencies available through the program code plug or PROM. The intent of this option is to enable the possibility of minimal communications between Phoenix area mobiles while they are out of range of the Phoenix DATCS.

5)

Estimates by industry sources place the very earliest availability of a technically acceptable frequency synthesized trunked portable radio in the 1983-85 time frame. Until such portable radios are available, trunked radio system equipment configurations will include only base stations, control stations, and mobile stations.

Public safety radio systems in Phoenix, Glendale, Mesa, and Tempe are portable radio intensive. The significant investment in this mix of VHF high band/UHF portable radio equipment must be protected. These existing equipments must be permitted to operate without modification with the trunked radio system.

Identify all base station repeaters in accordance with FCC regulations for transmitting station identification

Mobile-to-Mobile Operation

Mobile Talk-Around

Portable Radio Interface

The public safety agencies will continue using the portable radios on conventional VHF/UHF radio channels and manually cross patch these channels into the trunked system at the dispatcher's console when crossband coordination is required. Conventional VHF/UHF frequencies will be retained for portable radio operation and the trunked system will be used for the mobile radios.

6) MDT/CAD Interface

Public safety agencies in Phoenix and adjacent communities plan a continuing expansion of mobile data terminals (MDT) and expanded computer aided dispatch (CAD) operations. In-depth technical studies will be needed to investigate the effects of digital communications transmis sion characteristics of 800 MHz on MDT operations, particularly as these relate to data error rate limitations, permissible data rates, error detection and correction codes, and message retransmission limitations. These questions are unresolved at this time, and a judgement has been made to continue MDT operations on UHF until more definitive technical studies can be provided.

Manually activiated status encoders can be installed in public safety vehicles and interfaced with the DATCS mobile radio. This status information can be decoded at the public safety dispatch conters, displayed on a status board, and passed on to the CAD computer to update the CAD status displays.

Interagency Communication 7)

.

Interagency communication in the DATCS concept can be accomplished by the following key methods:

> Radio channel cross patching or phone patching at the dispatch consoles.

The pattern of interagency coordination connectivity can be programmed into the controllers and activated by terminals installed at key dispatch centers of agencies where flexibility in fleet programming is required (see Figure II-1).

Routine and frequently activated cross-agency communication routes can be programmed into the control station code plug or PROM, and these code plugs or PROMs can be manually switched.

The provisions for mobile radio interconnect to the Phoenix Centrex telephone system will enable any mobile so equipped to communicate with any municipal agency which is on the Centrex system.

Any individual method or combination can be configured into the vendor's proposal as best suited to his hardware and software offerings.

8)

the second second

9)

The DATCS system redundancy offered by the three-base station site configuration makes it unlikely that the Phoenix area DATCS will ever revert to a conventional radio channel operation due to controller failure. In general, the results of the key catastrophic system element failures indicate some coverage loss if North Mountain or Thompson Peak are shut down and some loss of talk-back capability if South Mountain is shut down. Only an extreme catastrophic failure of both North Mountain and Thompson Peak simultaneously will prevent the system from operating in a trunked mode.

The two types of intersystem operations to be considered are operations with other conventional radio systems and operations with other trunked radio systems.

Conventional radio system operations within the Phoenix area can be interfaced with DATCS by cross patching at the control consoles. Automatic cross hand repeaters will not be used for interconnection.

Conventional radio system operations from outside the Phoenix area will not be interfaced with DATCS unless they are both terminated at the same dispatch center in the Phoenix area. Mutual aid police and fire channels at 155.475 MHz and 154.280 MHz could be included in this category, but it is unlikely that mobile units from outside Phoenix would require interconnection with Phoenix DATCS.

There is no requirement placed on the Phoenix area DATCS for interconnectibility with any other planned or anticipated DATCS system outside the area. At this time, the hardware and software offerings of the key vendors of trunked radio systems are incompatible with each other. The signaling and control methodology and protocols are incomprehensible among the competitors' equipment. Mobile equipment supplied by the same vendor for a system outside

Phantom Groups

Temporary one-time needs for multi-agency response or intra-group limited response can be satisfied by creating a software "phantom group" to which individual vehicles can be assigned by their unit address code. This assignment will not alter the fixed unit address code or group address code of the involved mobile. Cancellation of the "phantom group" will return all mobile units to their original group.

Degraded Operation

10) Intersystem Considerations

Conventional Radio Systems

Trunked Radio Systems

of the Phoenix area would be unable to access the Phoenix control channel unless it was by chance licensed for the same group of 800 MHz frequencies as Phoenix. The control channel search range would also have to include the Phoenix control channel frequencies.

Unless Federal standards for control methodology and protocols are established, an attempt to achieve regional or statewide standardization to permit Arizona intersystem connectivity appears to be well beyond the scope or expectations of this project.

11. FAACILITIES AND PERSONNEL

In keeping with the philosophy of minimum change in operations and procedures to implement DATCS, there will be no change in the dispatch or control point locations now used by any participating agency. Control stations will replace the need for a telephone line or microwave channel between the control points and the base station sites. Some dispatch console modifications will be required to accommodate the DATCS control functions along with the conventional radio channel controls currently installed. Some modification of CAD software and hardware may be needed to accept status reporting codes. Facilities and personnel requirements unique to DATCS are discussed in the following paragraphs.

1) Facilities

Unique facilities at the base station sites and at the DATCS control operations center will be a consideration. Specialized maintenance facilities will be required.

Base Station Sites

It is intended to use existing equipment shelters and power systems at North Mountain, Thompson Peak, and South Mountain. Tower space and intermodulation studies may require the erection of a separate antenna support structure at each of the three locations.

DATCS Control Operations Center

Space must be provided for the DATCS control operations center. This will require a computer environment for housing the master controller and the peripheral alarm board, journal recorder, printer, and master control terminal. The voter selector installation and the control channel audio integration equipment are also a part of the center but need not be colocated with the computer controller. The area must include personnel space for manning the center 24 hours a day and should also provide office space for the system manager.

Special maintenance facilities for the DATCS should be included. A full- time maintenance program is anticipated to keep all of the DATCS system elements functioning at a level of quality needed to minimize performance differences between the multiple channels. This facility must store special parts/modules and will feature special tools and test equipment.

Personnel

2)

Personnel considerations include management personnel, system personnel, operating personnel, and maintenance personnel. Personnel staffing will be critical to the success or failure of the DATCS concept.

Management

It is suggested that DATCS be managed by an organization separate from the user organizations. The high technology requirements of DATCS calls for specialized telecommunications management which can not usually be spared from the individual user Organizations. Auser group should be formed to set policy and review operations but day-to-day management should be vested in a dedicated manager. The existing pool of communications managers in the Phoenix area should provide the resources for selecting the DATCS manager.

Software maintenance and hardware maintenance will be a full-time occupation to keep the quality of this high technology system to an expected standard. Thorough understanding of the system operation, operation software, and applications software will be required, suggesting the need for a systems analyst who can maintain the software and provide needed modifications.

Hardware maintenance will require special skills to obtain and retain 800 MHz integrity. Maintenance personnel skilled in computer operations, terminals, and displays will be required. It is assumed that providing training for these skills will be part of the vendor obligations. The existing pool of systems personnel and hardware maintenance personnel responsible for the CAD and MDT systems should be an integral part of this staff.

Operating Personnel

The DATCS control operation center is conceived as a 24-hour-a-day operation. One terminal operator/ controller per shift would appear to be adequate. This could be a shared responsibility once DATCS has become operational and debugged. This would require approximately five specially trained persons, probably drawn from the existing CAD/MDT staff.

Maintenance Facilities

System Maintenance Personnel

Support services such as purchasing, financial control, audits, benefits administration, facility maintenance, and the like are assumed to be provided by the City of Phoenix using existing municipal facilities.

12. ESTIMATED COSTS

This section presents engineering cost estimates associated with implementing a digitally addressed trunked communications system (DATCS) in the Phoenix area. The configuration of the DATCS is described in the system concept report dated September 15, 1980. Costs are presented in two parts:

Non-Recurring Costs - One-time capital investment cost

Recurring Costs - Ongoing costs associated with owning, operating, and maintaining the system

These cost estimates represent the best information available at this time. Much of the hardware and software has not been produced in the specific form and function required by the system concept. Therefore, a number of assumptions were required based on engineering experience, discussions with the vendor community, and representative cost experience of commercial trunked system owners.

The cost estimates will also need adjustment as new information becomes available. No attempt has been made to factor in the inflationary pressures or to extend the value of these funds into some future date when implementation of DATCS is more probable. The degree of determination on the part of the vendor community to pursue the trunked system market and the competitive posture of that market at the time of implementation will be the final arbiter of these estimated costs.

1) Summation of Estimated Non-recurring Cost

The following summation includes the estimated initial onetime cost for fixed installations and mobile installations for Phoenix area DATCS. The break-down of these estimated costs is shown in the following subsection.

Fixed Installations

North Mountain	S	322.000
Thompson Peak	т	345,000
South Mountain		312,000
Control Operations		299,200
Maintenance Equipment		77,000
Control Stations (Phoenix)		246,500
Control Stations (Adjacent Cities)	· . •	121,500
SUBTOTAL - FIXED INSTALLATIONS	\$1,	723,200

IV-24

Adjacent C Service Adjacent C Safety SUBTOTAL -2)

Repeaters \$9,5 Multicouplers/ Duplexers \$600 Antennas/Trans Site Controlle Modems \$300 Tower 50' X \$1 Microwave Inte 22 Channel MUX Installation (

Thompson Peak

Repeaters \$9,5 Multicouplers/ Duplexers \$600 Antennas/Trans Site Controlle Modems \$300 Tower 50' X \$1 Microwave Inte 22 Channel MUX 6 GH Microwave Installation (

Mobile Installations

Phoenix Public Service Phoenix Public Safety	\$2,486,300
Adjacent Community Public Service	1,432,900
Adjacent Community Public Safety	46,000
SUBTOTAL - MOBILE INSTALLATION	\$4,494,200
• Engineering Services	\$ 125,000

TOTAL ESTIMATED NON-RECURRING COSTS \$6,342,400

Cost Detail - Estimated Non-recurring Cost

The following cost estimates assume that the 12 GHz microwave system at North Mountain and South Mountain has sufficient channel capacity to accommodate the DATCS requirements for centralized control. If this is not the case, additional microwave terminals will have to be added to the estimated cost of North Mountain, South Mountain, and Control Operations. It is also assumed the Maricopa County microwave system to Thompson Peak does not have the spare channel capacity to accommodate DATCS. Therefore 6 GHz microwave terminals are added to the estimated costs for Thompson Peak and Control Operations. The exact configuration of the DATCS transmission system will be an output of the required engineering study.

North Mountain Estimated Costs

500 'Combiners \$10,600)	qty. 19 qty. 4 gty. 4	\$ 180,500 42,400 2 400
mission Lines \$800 er \$45,000	qty. 4 qty. 1	3,200 45,000
10/ft prface Modules \$300	qty. 2 qty. 1 atv 45	5,500
(\$615/ch (5%)	qty. 45 qty, 1	13,500
		\$ 322,000
Estimated Costs		•
00 (Combiners \$10,600 mission Lines \$800 r \$45,000	qty. 19 qty. 4 qty. 4 qty. 4 qty. 1 qty. 1 qty. 2	\$ 180,500 42,400 2,400 3,200 45,000 600
10/ft rface Modules \$300 \$615/ch Terminal \$22,000 5%)	qty. 1 qty. 45 qty. 1 qty. 1	5,500 13,500 13,500 22,000
₩ ₩ 2		\$ 345,000

South Mountain Estimated Costs					
Repeaters \$9,500	qty.	18	\$	171,000	
Receives \$750	qty.	2		1,500	
Multicouplers/Combiners \$10,600	qty.	4		42,400	
Duplexers \$600	aty.	4		2,400	
Antennas/Transmission Lines \$800	aty.	4		3,200	
Site Controller \$45,000	atv.	1		45,000	
Modenis \$300	atv.	2		600	
Tower 50' X \$110/ft	atv.	1		5,500	
Microwave Interface Modules \$300	atv.	42		12,600	
21 Ch MUX \$615/ch	atv.	1		12,900	
Installation (5%)	7-7-			14,900	
			s	312,000	
			Ŧ	5127000	
Control Operations Estimated Cost					
Master Controller \$40,000	atv.	1		40,000	
Alarm Board \$12,000	atv.	1		12.000	
Voter Selectors \$2,200	atv.	18		39,600	
Journal Recorder \$17,000	atv.	1		17.000	
Master Terminal \$3,500	atv.	1		3,500	
Agency Terminals \$2,700	atv.	9		24,300	
Printer \$4,500	atv.	1		4.500	
Audio Subassembly \$1.500	qey.	ן. ן		1,500	
Migrowave Interface Modules \$200	atv.	132		26.400	
65 ch MUX \$615/ch	qey.	1		40,000	
6 GH Microwaye Terminal \$22,000	aty.	1		22 000	
Modems \$300	atv	6	Ś	1 800	
Applications Software	drà.	U .	. Y	30,000	
Telco Installation \$125/line atv	200	lines	.	25 000	
Radio Installation (5%)	. 200	TTHE	•	11 600	
Maro motariation (58)			\$	299 200	
			, Y	299,200	
Maintenance Equipment/Spares Estimated (Cost				
Digital Test Sets \$1.450	atv.	2	Ś	2,900	
Portable Digital Analyzer \$1.500	atv.	4	т	6,000	
Logic Probes \$100	atv.	4		400	
Spectrum Analyzer \$5,000	atv.	2		10.000	
800 MHz Signal Generator \$1,200	atv.	2		2 400	
Spare Repeaters \$9,500	dty.	2		28 500	
Microwave Interface Modules \$300	quy.	12		3 600	
Spare Multicouplers/Combiners \$10 606	4-y.	те : 1		10 600	
Spare Modems \$300	4-y.			70,000	
Spare Voter/Selectors \$7 200	ч-у. а+т	1		2:200	
Spare Terminal Keyboards \$1 100	4-y.	- 1		1 100	
Spare Boards and Modules (59)	4-y.	<u> </u>			
Spare Dourds and Hoadtes (28)	•		¢	77 000	
			4	11,000	

Control Stations (Phoenix) Estimated Costs

Control station costs are estimated at \$4,500 each, installed. The following tabulation estimates the number of control stations needed, their distribution, and the cost distribution.

Console Modi Fire Console Modi Engineering Traffic Engi Traffic Sign Facility Mai Human Resour Equipment Ma Sanitation Streets Main Job Stimulus Building Ins Sewers Water Product Water Distri Water Custome Parks and Red Control Stati Glendale Fire Glendale Poli Glendale Gene Mesa Police/F Mesa Public W Mesa Public S

EOC Police

1

 \square

)

to the second

. 🛃

. . .

Mobile station costs are estimated at \$2300 each, installed. The tabulations in this subsection and the following subsections related to mobile stations reflect the quantity of mobiles reported during the data surveys of February and April 1980. The number of mobile units listed for public safety agencies will also relect the limitations of DATCS due to portable radio operations and the interface problems with MDT users.

IV-26

	1997 - 1997 -		qty.	7	\$	31,500
: .			qty.	6		27,000
lileations			qty.	14		14,000
· # 2			qty.	4		18,000
lications			qty.	12	i. K	12,000
			qty.	4		18,000
ineering			qty.	1		4,500
lais	1. 19		qty.	1		4,500
ntenance	н. 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 - 1917 -		qty.	2		9,000
ces			qty.	1		4,500
inagement			qty.	1		4,500
			qty.	4		18,000
tenance			qty.	2		9,000
			qty.	1		4,500
pection			qty.	2		9,000
المتعادية		56. T	qty.	2		9,000
Clon			qty.	6		27,000
Ducion			qty.	2		9,000
er service			qty.	1		4,500
creation			qty.	2		9,000
	• 1				\$ 2	46,500

Concroi Stations (Adjacent Cities)	Estimated	Cost	
Glendale Fire			· · · · ·
Glendale Doliga	qty.	Τ.	\$ 4,500
STORATE FOILGE	atv.	2	9.000
Glendale General Services	Comboo de la comboo de	n	5,000
Mesa Police /Fire	quy.	2	9,000
Mara Dillerice	qty.	3	13.500
Mesa Public Works	atv	6	27 000
Mesa Public Sanitation	Y •		27,000
Contradal Tal:	qty.	4	18,000
SCOLLSUA & POIICE	atv.	1	4 500
Tempe Police		~	4,505
Tempe Fire	qcy.	2	9,000
tembe titte	aty.	2 .	9.000
Tempe Public Works		- -	70,000
Tempe Sanitation /Defense	qty.	3	13,500
rempe panitation/Refuse	qty.	1	4,500
			\$ 121,500

Mobile Stations (Phoenix Public Service) Estimated Cost

			1	n ne angan ng mang na ang na
		1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1		
$h = \frac{1}{2} \left(\frac{1}{2} - \frac{1}{2} \right)^2 \left(\frac$				
			1 1900	
		$\int \int $		Radio propagation testing and analysis
Engineering	aty, 119 \$ 273,700			
Parks and Recreation	g_{1} , g_{2} , g_{2} , g_{1} , g_{2} , g_{2} , g_{1} , g_{2} , g_{1} , g_{2} , g_{2} , g_{2} , g_{1} , g_{2} , g			• Technical evaluation of vendor proposals
Traffic Engineering	$q_{1}q_{2}$, $q_{2}q_{2}$, $q_{1}q_{2}$, $q_{2}q_{2}$, $q_{1}q_{2}$, $q_{2}q_{2}$, $q_{1}q_{2}$, $q_{2}q_{2}$,	r -1	(*****	
Traffic Signals	a_{12} a_{12} a_{12} a_{12} a_{13}			Monitoring DATCS hardware/software implementation
Facility Maintenance	$q_{2}, 12$ 27,000 $q_{2}, 12$ 119,600		hundred -	
Human Resources	$q_{2}, 32$ $115,000$			Evaluation of implementation debugging tests
Equipment Management	q_{2y} , 52 , $75,000$			
Sanitation	q_{cy} , g_{1} , g_{1} , g_{1} , g_{1} , g_{2} , g_{1} , g_{1} , g_{2} , g_{2} , g_{1} , g_{2} , g_{1} , g_{2} , g_{1} , g_{2} , g_{2} , g_{1} , g_{2} , g_{2} , g_{1} , g_{2} , g_{2} , g_{2} , g_{1} , g_{2} ,		types, lett	Analysis of acceptance test data
Streets Maintenance	q_{cy} , y_{2} $211,000$			
Job Stimulus	$q_{2}y_{1}^{2}$ $25 \qquad 204,700$		(TT)	Adequate engineering support for the above is estimated at \$125.0
Building Inspection	q_{2y} . 20 $39,000$			incequate engineering support for the above is estimated at \$12570
Samers	q_{cy} , 72 105,000		*uni	3) Estimate Recurring Costs
Water Production	$q_{\rm Ly}$, 74 $170,200$	and the second se	add sega	5) <u>IDCLIMACC ACOULTING CODED</u>
Water Distribution	q_{Ly} 114 202,200 a_{Ly} 150 345 000			The estimate of recurring costs does not include the
Water Customer Seriyoo	q_{Ly} . 100 343,000			of operating the existing communications system by each not
Water, Customer Berryce	q_{LY} . 101 <u>232,300</u>			nating agency. It is assumed that this cost would not cha
	\$2,486,300		T	pacing agency. It is assumed that this cost would not the $\Delta T = 0$
Mobile Stations (Phaenix Bublis	Coffeter) Retimeted Cost			a resure of imprementing DATCS as a equivarent transmission
MODILE SCALLONS (PHOEHIX PUBLIC)	Salety) Estimated Cost	n in the second se	• •••i.towi	the accorted operating procedures and practices of the part
Dhanning Dolige (25%)		F	07-18	the accepted operating procedures and practices of the para
Phoenix Poince (25%)	dty. 170 \$ 391,000			patting user organizations.
Phoenix File (25%)	qty. 60 - 138,000	L.J.		
	\$ 529,000			The estimate of regurring gost doog inglude the extra
Mobile Ctations (Adiasant Commun			1 50	the estimate of recurring cost does include the extra
Mobile Stations (Adjacent Commun.	ity Public Service)			statting and factificies needed to centralize the control of more in no of
ESCIMALED COSC			4	Black configuration similar to the configuration and func-
		17	IT1	DATES COnfiguration similar to the configuration and func-
Giendale General Service	qty. 1.30 \$ 299,000			required of this proposed system. Merefore, there is no
Mesa Public Works	qty. 163 374,900			cost experience to reference and assumptions are made base
Mesa Public Sanitation	qty. 60 138,000			experience with other digitally controlled, computer operation
Tempe Public Works	qty. 230 529,000			telecommunications systems. These estimated costs are su
Tempe Sanitation/Refuse	qty. 40 92,000			marized as follows:
	\$1,432,900			
				Maintenance Technicians \$22,500= qty. 4 \$ 90,000
Mobile Stations (Adjacent Commun	ity Public Safety)			Control Operator (24 hrs) \$25,000* qty. 4 100,000*
Estimated Cost			une d'	Administrator $$31,250*$ qty. 1 $31,250*$
		P ⁻ - a	January 1	SUBTOTAL - ESTIMATED LABOR \$221,250*
Glendale Police (10%)	qty. 8 \$ 18,400	· · · · · · · · · · · · · · · · · · ·		
Glendale Fire (10%)	qty. 3 6,900			Maintenance Parts Replacement \$110,000
Mesa Police (10%)	qty. 4 9,200			Maintenance Training, 3,000
Tempe Police (10%)	qty. 2 4,600	r	17	EDP Services (Management Reports) 3,900
Tempe Fire (10%)	qty. 2 4,600			Equipment Up-Dating 180,000
Scottsdale Police (10%)	qty. 1 2,300			Utilities (Telephone/power) 12,000
	\$ 46,000	F	77-1	SUBTOTAL - SERVICE COST \$ 308,900
Engineering Services Estimated Co	ost	\$,1	J 4.2	TOTAL - ESTIMATED ANNUAL RECURRING COST \$ 530,150
The estimated cost for engineering	ng services is based upon the need	for J	1 17	
technical assistance as anticipated a	t this time. As better informatio	n (1 - 1 - 1		
becomes available, this need will chan	nge. The following items of techn	ical		
assistance are considered vital to the	e implementation of DATCS:		1	* These estimated costs include full burden.

 \square

IV-28

0

000.

ne cost partici-nange as Lon nge to artici-

ca and existing ctions o reliable sed on cated am-

These estimated costs assume that Phoenix or an equivalent governmental organization will take full responsibility for system operation and maintenance for the benefit of all DATCS users. These costs also assume that early life failures and basic debugging of the software are covered by warranty.

	-		T.
	±	- T	
2	1mp) tei	men

ттт

-

tation of the system concept will proceed only after The thorough review by the potential user agencies, their department heads, and the public safety director of each participating municipality. After this has been achieved and the implementation plan has been officially sanctioned by each participating municipality, the satisfaction of licensing requirements is another critical milestone that must be met before procurement activities can be initiated.

Federal Communications Commission (FCC) licensing requirements for large-scale 800 MHx trunked radio systems in local government service have not been clearly defined or tested at this time. Following local government approval, the initial implementation activity must be directed toward satisfying all of the elements needed to secure reasonable assurance from the FCC that the trunked radio system, as planned and approved, can be licensed.

° 1. LOCAL GOVERNMENT APPROVALS

Local government approvals will be achieved in three steps: securing of user agency agreements, development of funding agreements to implement and operate the system, and the legally binding commitment of the municipalities to participate in and support the system.

Agency Agreements 1)

Signed agreements by each agency planning to participate in the trunked radio system must be acquired. These agreements will conform with FCC requirements (Sections 90.359 and 90.179). As a minimum they will include:

Responsibility for maintaining system operation records and filing reports with the FCC (Section 90.391).

In addition to the items needed for FCC licensing, the agency agreements will also include provisions for:

IV-30

È)

REQUIRED IMPLEMENTATION ACTIVITIES

Agreements regarding the supervision and control of the trunked system operation

Method of cost sharing and prorating of costs among all cooperating parties

Management authority over the system staff and equipment

Centralized maintenance of system components

Property, equipment and facilities provided by the user agencies and municipalities

Modification and/or termination of the agreement as subject to FCC Rules and Regulations (Section (90.391).



Agency agreements will be the key element in securing the funding agreements and municipal government approvals.

2) Funding Agreements

The acquisition of agency agreements will provide dimensions for revised cost estimates needed to establish funding requirements. These requirements will include estimates of nonrecurring implementation costs and recurring operations costs. The following elements of cost will be identified:

Implementation Costs

- Planning activities Professional services
- Hardware/software
- Facilities
- Installation/test/acceptance
- Personnel staffing and training
- Operating Costs
- Personnel salary and overhead
- Supplies
- Maintenance of hardware/software
- Equipment replacement
- Continued training.

The distribution of these costs among the funding sources will be decided, and firm commitments and/or agreements acquired. These funding sources will include:

Federal and state grants funds

Local municipal funds

In-kind services, equipment, or facilities

User agency cost sharing assessments

Grants from private foundations

Gifts from private sources.

The funding agencies must approve and be committed to the funding plan before final and binding commitments will be sought from the participating municipalities.

3)

6

Based on the agency agreements and funding agreements achieved, firm commitments will be required from the municipal governments to assure continued project support and to respond to Federal and state grants requests. These approvals will be required in the form of resolutions or other legally binding documents.

2. PRELIMINARY LICENSING CONSIDERATIONS

The licensing of multiple base station wide-area trunked radio systems in local government service and public-safety service is not clearly defined by present FCC rules and regulations. Waivers to the present FCC rules and regulations may be required, particularly to Section 90.389 which restricts interconnection with telephone wire facilities and Section 90.375 which defines time limits for phased implementation.

Before hardware and software commitments are made, preliminary discussions will be held with the FCC. An expression of assurance will be sought indicating that the trunked system, as planned and approved by the user agencies and their municipal governments, will receive favorable licensing consideration from the FCC when formal applications for licenses are submitted.

IMPLEMENTATION SCHEDULE 3.

The detailed implementation schedule will be developed following assurances that the trunked radio concept will be licensable under FCC rules and regulations and that any revisions to the concept required by FCC evaluation of the system design has met the approval of the user agencies and the participating municipal governments. The implementation schedule will consist of two key interrelated parts, a system implementation schedule and a corresponding financial schedule.

1)

The system implementation schedule will be a time-dependent schedule. The project will be defined in terms of separate tasks with clearly defined milestones indicating the completion of each task and definitive measurable milestones leading to the completion of each task or project element. Major tasks include:

Project planning and reporting

Engineering services

Preparation of specifications

Preparation and approval of statement of work

IV-32

Municipal Government Approvals

System Implementation Schedule

- Solicitation of bids
- Contract award
- Facility preparation
- System licensing
- Equipment delivery/installation
- Test and acceptance
- Personnel training
- System cut-over
- System evaluation.

Each major task will be supported by task breakdown schedules which define the subtasks needed to achieve the completion of each major task.

2) Financial Schedule

The financial schedule will be coordinated with the system implementation schedule. This schedule will be based upon the funding agreements and will include the following major elements:

Milestones showing each major commitment of funds

- The amount of funds from all sources and the date of their availability to the project
- The schedule dates for the submission of financial reports.

The submission and distribution of financial reports will be elements of the financial agreements and governmental approvals.

4. SYSTEM MANAGEMENT AND OPERATION

The management and direction of the implementation program will be the responsibility of the program director. The program director will be supported by a full-time and part-time staff representing each key function of this implementation program. These functions include:

> System Engineering - this staff function will be assisted by consulting engineers and the engineering staff of the successful system contractor

Procurement - this staff function will be assisted by the purchasing and procurement personnel of the City of Phoenix

			· · · ·	
		•	Fina	nce -
		•	Budg Faci	et litie
			and pers basi	prepa onnel s by
		•	Pers dire acti	onnel ct su vitie
			and the anot	benef City her m
	tor, polic	In ac two a cy wit	dditi advis th th	on to ory g e pro
	advi	sory o	group	. The
		•	repro liai tive	esent son b agen
		•	Gove: appo coun	rnmen inted ty go
		It is	fina s into	ncial ended
	form it ha neede	and f as bee ed to	funct: en in opera	ion to stallo ate ti
	facil progr	Lity f	funct.	ion w or.
	5.	PROCU	JREME	NT
	seque	The pential	act:	remen
LB		1)	Syste	em End
Ū		neeri	Techi ing de	nical esign The l
		speci	ifica	tions

- this staff function will be a part-time service d by the City of Phoenix, Division of Management and

ies - this staff function related to the acquisition paration of special facilities for the equipment and el of the project will be provided on a part-time y the City of Phoenix

el - the project staff personnel are under the supervision of the program director; personnel ies such as job descriptions, salary determination, efits administration will be under the direction of y of Phoenix unless the person is on loan from municipal jurisdiction.

to the project staff reporting to the program direcgroups will be responsible for overall project roject director serving as chairperson for each These groups consist of:

presentatives - this group is composed of appointed ntatives of each user agency; they provide direct between the project organization and their respecencies.

ent Representatives - this group is composed of ed representatives of each municipality and state or government sections which have an operational or al interest in the program.

ed that this management group be carried over in to the operation of the trunked radio system after lled, tested, and accepted. Additional personnel, the system from a central location, will be added unction of the project staff. At that time, the will become a full-time staff position under the

ent of the trunked system will proceed in two ties, system engineering and purchasing.

Engineering

al assistance will be required to develop the engign of the trunked radio system based on the system a key issues to be resolved before the procurement hs can be prepared include:

Radio Propagation Analysis - engineering analysis of 800 MHz radio propagation from the three candidate base station sites will be conducted to determine radio coverage of the Phoenix area and to establish antenna patter characteristics. FCC Rules and Regulations (Section 90.379) clearly limit the effective radiated power and antenna elevation at each proposed site.

and the second

l j

Contraction of

6.

LICENSING

<u>Switching Logic</u> - the switching logic for the master controller, site controller, mobile stations, and control stations will be engineered to be compatible with expected vendor hardware offerings.

Software - the system software for the master controller and site controller will be engineered to enable effecient software management from a master terminal.

Digital Data - an engineering study will be conducted to establish the technical feasability of integrating the existing mobile data terminals (MDT) into the 800 MHz trunked radio system without degradation of the present MDT capability.

Based upon the results of these technical activities, systems engineering will prepare detailed system specifications and equipment specifications for competitive procurement.

2) Purchasing

The purchasing function will consist of three key activities: developing a statement of work (SOW), soliciting competitive bids, and awarding a system contract.

- Statement of Work (SOW) the SOW is a precise description of the materials and services required of the contractor and the time domain in which delivery is to be made. The key elements of the SOW will be:
 - Background and objectives of the project
 - Required qualifications of successful bidders
 - Services and support to be provided

IV-36

Quantities of each equipment to be provided

Schedule for all deliverable items - hardware, software, services, and support items

Soliciting Competitive Bids - competitive bids from all qualified vendors will be solicited by means of a procurement package which will include:

State and city legal requirements

Statement of work

System specification

Equipment Specifications.

A one-step solicitation is anticipated in which each bidder will provide a technical proposal and a firm, fixed-price cost proposal for the complete installation of an operating digitally addressed trunked radio system.

Awarding the System Contract - the combined technical proposal and cost proposal most advantageous to the City of Phoenix and associated municipalities will identify the successful bidder. A firm, fixed-price contract without cost escalation is anticipated.

The signed contract defining the equipment to be purchased and the quantity of that equipment will provide sufficient information to file applications with the FCC for system licenses. Present FCC practice for licensing 800 MHz trunked system requires the following minimum submission:

Form 400 for each license requested (Section 90.357)

Agreements with each user agency relative to control of the system (Section 90.179)

Agreements with each user agency relative to cost sharing of installation and operation (Section 90.179)

Financial plan showing the service will be provided to user agencies at cost (Section 90.359)

System description and specification (Sections 90.359, 90.371, 90.375, and 90.379)

Eligibility statement for each user (Section 90.359)

Evidence of purchase orders placed for the equipment.

It is probable that FCC waivers will have to be asked to extend the number of years for phased implementation of the system (Section 90.375) and for the provisions for interconnecting with wire line telephone systems (Section 90.389).

Technical support for negotiating system licenses with the FCC will be requested of the engineering consultant and from the system contractor.

FACILITIES ACQUISITION AND PREPARATION 7.

It is the responsibility of the program director to provide adequate facilities for the equipment and staff of the trunked radio system. Subject to revision by the engineering study, the following facilities will be required:

> North Mountain - equipment shelter for twenty 800 MHz repeaters, a site controller, ancillary equipment for the antenna subsystem, antenna support structure for four antennas, and appropriate power/climatic control equipment

South Mountain - equipment shelter similar to North Mountain

Thompson Peak - equipment shelter similar to North Mountain

Phoenix Pbulic Safety Building - equipment shelter for the master controller/terminal, alarm equipment, recorders, printers, microwave terminations and other items of equipment associated with the central control function. Staff facilities for a program director, facilities manager, engineering, software/hardware manager, and programmer/ operators will also be required.

Controlled maintenance will also require special facilities which are intended to be part of the Communications Section, Equipment Management Division, City of Phoenix. It will be the responsibility of the facilities manager on the staff of the program director to obtain these facilities and prepare them in a timely manner.

PERSONNEL AND TRAINING 8.

Personnel for the management and operation of the trunked radio system will be identified and prepared for their assignments in coordination with equipment installation and test. It is intended that the project director and staff will continue in their positions after the trunked radio system has been installed and accepted. The additional people required to operate the system will be recruited and trained during the system installation test so that their services can be used as much needed test personnel, and the experience gained can serve as on-the-job training before the system becomes operational. It is anticipated that the following additional personnel will be required:

> Maintenance technicians Qty. 4 Control operators Qty. 4

> > IV-38

POST IMPLEMENTATION SUPPORT system. 1) folloiwng major items:

9.

8

1

These costs will have been identified and submitted to the FCC as part of the licensing requirements; they will be reviewed periodically. The costs will be billed to the user agencies in accordance with a formal developed and reported in each agency agreement.

2)

The replacement of personnel and the updating of staff personnel will be a recurring problem because there will be labor turnover. Recruiting of personnel will be by suitable public announcement of job openings and the testing of applicants. Probationary employees will be trained by the technical staff.

System programmer/analyst

Facilities administrator Qty. 1

Job descriptions for these positions will be developed by the program director and the personnel divisions of the City of Phoenix. Recruiting will be by public announcement of available job openings. It is anticipated that the positions will be filled by transfering presently employed, skilled personnel of the municipalities.

The training and certification of all operating personnel, maintenance technicians, and management personnel will be a contractual obligation of the system contractor,

The post implementation support program will address the funding and staffing needed for sustained operation of the trunked radio

Post Implementation Funding

The recurring costs of continued operation include the

Personnel - salary and benefits

Maintenance - maintenance of the facilities and system hardware/software after the warranty and first-year maintenance contract

Supplies - all expendable materials used in operating and managing the system

Training - refresher training and replacement training

Utilities - power, telephone, and other services provided by an outside agency

Equipment Replacement - periodic replacement of worn or obsolete equipment, i.e., equipment that can no longer be repaired economically.

Personnel Replacement and Training

3) Support Staff

The system will have the full-time services of equipment maintenance personnel and software system personnel. At least one senior system programmer/ analyst will be required. It is anticipated that a full-time staff of four specially trained maintenance technicians will be required. Four qualified control operators will also be required to provide 24-hour service to the user agencies.

IV-40

APPENDIX A

(continue)

Lille

们

 \square

 \square

 \square

0.1

 \square

IJ

(__)

[]

1.1

6.1

*

1.1

 $\left(\right)$

 \square

•

SUGGESTED AGENCY AGREEMENTS

I. INTRODUCTION consulted. 1. support. 2. Scope of the Agreement II. SUGGESTED MODEL OUTLINE 1. Preceding page blank

APPENDIX A

SUGGESTED AGENCY AGREEMENTS

The agreement by two or more local government public safety agencies to enter into a joint enterprise which provides a mutual service of value to the participants is best documented by written agreements between the parties. By their nature, these agreements are highly individual tailored to the specific requirements of each program in each locality. This appendix to the System Implementation Planning Guide (SIP) is intended to suggest only a general outline of such agreements. The details and the order of presentation will depend upon each kind of enterprise contemplated, its complexity, and the immediate cooperative environment. Legal counsel should be

Purposes of the Agreement

The purpose of a written agreement is to describe the reason for the cooperative effort and the results that are expected to be achieved through mutual participation. By reducing this to a clear and unambiguous written statement, the involved parties will fully understand their mutual participation. Equally important, the governing bodies of which the participants are a part will be able to determine the true nature of the program and the expected results and thereby provide their

The agreement should define the dimensions within which the terms of the agreement apply. The responsibilities and liabilities of each of the participants should be clearly documented. There should be no unwritten understandings between the parties.

The following model outline of an agreement form provides a series of topical considerations to be evaluated during the development of an interagency agreement. The precise form of a specific agreement and the language to be used in each case should be provided by legal counsel. In many cases, standard legal form and substance is required by state or federal regulation.

The following topics and items are intended to delineate key considerations in the preparation of an interagency agreement.

General Terms and Legal Base

The general terms will include the following as a minimum: The identification of each participating agency

The requirements to be imposed on those agencies that may wish to participate at a later date

- The concise purpose of the joint enterprise and the mutually advantageous results that are expected to be achieved
- A statement of the legal authority under which the public agencies are able to contract with each other for the stated purpose; this may be a state law, a local government act, a series of resolutions passed by the local governmental authorities, or similar authorization.

2. Services to be Provided

The services provided by each agency and to each agency will be defined. These include the following considerations:

> Definition of the services to be provided by each agency to the mutual system; these services could include, for example, equipment maintenance, equipment, facilities, operating manpower, support services, and similar items of value to the system..

Definition of the services the system will provide to each participating agency; such services could include the dispatching of field forces, management reports, statistical data, and similar items of value to the agency

The definition of the quantity or extent of each service to be provided

The responsibility each agency has for delivering its service and the responsibility the mutual operation has for delivering its services to each agency

The extent of any back-up or redundancy to assure that services are delivered in a timely manner throughout the life of the agreement.

3. Liability

•

.

•

A key element of the agreement between two or more public agencies is a clear definition of the liability each of the agencies has for the actions of other parties to the agreement. This is particularly true if one or more of the agencies provides public services on behalf of the other agencies. Agreements to limit the liability exposure of participating agencies would include the following considerations.

IV-44

Responsibility for defending the provider agency in potential lawsuits

Indemnification of the providing agency against liability judgements against a participating agency

Indemnification of the participating agency against legal actions or liability judgements against the provider agency arising from the normal provision of agreed-to services.

Amount and Manner of Payment

-11

4.

.

5.

1

The providing agency must be reimbursed for the expenses of the service it provides to the participating agencies. A portion of this obligation will fall to the individual participating agencies. The agreement between the agencies must be specific regarding this individual responsibility and will be definitive of the following considerations:

The specific services to be provided and the cost of each service

The method by which the cost and value of each service is calculated

The means of payment: for example, cash, in-kind services, cash equivalent in material or equipment, and any other valuable consideration agreeable to the parties to the agreement

Time and manner payments are to be made

Use of outside funding sources such as grants or gifts.

Fiscal Procedures

agency

Sound fiscal planning is key to the successful implementation of public agency cooperative programs. The responsibility of the provider agency for financial integrity must be defined in the agency agreements. The agreement will specify the following elements:

The type of financial records and how they are maintained

The contents of the periodic financial reports and the schedule for submitting these reports to higher authority and to participating agencies

Access to the financial records by each participating

Certification and/or audit of the records and financial reports

Terms and conditions relative to the periodic reassessment of rates charged for each service provided.

6. Administration

The administrative structure of the cooperative program should be defined in the agency agreements. The authority and responsibility of each agency for participation in policy administration and the authority and responsibility for day-to-day operation should be included. Some of the key elements in the agreement are:

> The identification of the person(s) authorized to sign the agreement on behalf of each party to the agreement

The identity of the person(s) who will represent the participating agency and providing agency in policy matters

The responsibility and authority of the providing agency to supervise project personnel and establish personnel policy

The responsibility and authority of the providing agency over joint-use facilities and equipment

The retention by participating agencies of control, responsibility, and authority over their personnel, facilities and equipment that may be used in the cooperative project

Identity and authority of persons responsible for the review of the agreements and for the determination of compliance.

These and other administrative elements suitable to the unique environment and nature of the proposed project should be negotiated between the parties and made part of the written agreement to participate.

7. Real Property and Equipment

The cost-effective implementation of a multi-agency cooperative program should make maximum use of facilities, property and equipment available from the participating agencies. The agreement should identify these properties and materials; it should establish the rules under which the properties and materials are to be used. Some key considerations are:

> Identify each item of property and equipment that is to be transferred to the use of the cooperative system

	•	Establ	
		and sh	(
		Provid	
	•	during	
		or can	(
	•	Define	
		proper	í
		suppli	e
		cellat	
8.	Dura	ation, T	e
agre the and lowin	ement agree disc ng it	ement, ma lose how tems show	
	•	The ter	Ł
	•	The ci	Ľ
		termina	5
		The ter	r
		agreeme	-
	_	The ci	
	•	the ag	Ľ
		Data at 7	
	•	intende	1
			-
		Disposi	i

The age	ency	7
reflect the	uni	Ĺg
environment	in	Ŵ
The individu	ıal	t
inglude the	£ _ 1	1.1

9.

 \mathbb{R}

etc.

disputes

IV-46

ish the value of the property and/or equipment ow the method used to establish the value

le for the return of the property and equipment the term of the agreement and upon termination cellation of the agreement

the method to be used to establish the value of ty and material to be returned to the original er upon the termination, modification, or canion of the agreement.

ermination, And Amendments

agreement must provide for modification to the its term. It must also define the duration of ake provisions for cancellation of the agreement, the agreement is to be terminated. The foluld be included:

rm of the agreement

rcumstances under which the agreement can be ated during its intended duration

rms under which the agency can withdraw from the ent

rcumstances under which either party can cancel reement

ares for amending the agreement during its ed term

ition of property and equipment upon termination or cancellation.

Other Provisions

agreement is a custom document generated to we needs of the specific circumstance and the which the mutual agreement is being generated. erms and conditions of each agreement should include the following:

> Method of resolving disputes, for example: arbitration, application of appropriate state or Federal law,

> Identity of the court of jurisdiction for adjudicating

Definition of terms used in the agreement

Statutory status and filing of the agreement

Interrelationship of agreements.

The drawing of inter-local contracts and agreements to provide public services or hold public property for the mutual advantage of multiple governmental organizations may be subject to local, state, or Federal regulations and/or laws. These documents should be drawn up under the advice of legal counsel to insure their status as legal instruments.

I

T

 \square

ľ

APPENDIX B

DATCS SYSTEM REQUIREMENTS

This appendix has been derived from the review of the Task 6 Report "Analysis of Problems and Requirements" by Phoenix project staff. The contents of this appendix are also contained in the Task 7 Report "DATCS Design Concept" dated September 19, 1980.

 \mathbb{C}

 $\sum_{i=1}^{n}$

IV-49

APPENDIX B

SYSTEM REQUIREMENTS

The system requirements have been developed from the review of the Task 6 report "Analysis of Problems and Requirements" dated June 15, 1980. The summation of this review was developed during a meeting of Phoenix on July 23, 1980 and is reflected in the following subsections.

1. WIDE AREA COVERAGE

Radio coverage experience at UHF in the Phoenix area precludes any reasonable expectation that a single 800 MHz base station site will provide adequate coverage.

(1) City of Phoenix

The growth of Phoenix north to the Deer Valley area and northeast to Paradise Vally indicates North Mountain as a primary base station site covering both Phoenix and the northern communities.

North Mountain has notable shadow effects from Squaw Peak, Phoenix Mountain, and Camelback which indicates a second base station at Thompson Peak would be required.

A probable third base station site at South Mountain is indicated to ensure reliable talk-back coverage in areas where North Mountain has proved to be less than fully reliable on talk-back.

(2) Suburban Participation

The coverage of Glendale should be adequate from North Mountain and South Mountain.

Control channel coverage of Scottsdale, Tempe, and Mesa should be adequate from North Mountain and Thompson Peak. Talkback coverage should be improved from Tempe and Mesa from the South Mountain site.

Separate control channels at North Mountain and Thompson Peak will be selected by each mobile station according to the readability of each base station site control channel at the mobile receiver. All DATCS voice channels at the three base station locations will be voted for signal quality from the mobile and the transmitters will be steered by the voter selectors for each trunked channel. There will be no simulcasting of multiple transmitters on the same channel

IV-50

MULTIPLE DISPATCH POINTS

 \prod

State of the second

2.

The existing local government radio services in Phoenix and adjacent communities consists largely of independent radio channels, each agency having its own command and control requirements over its own fleet of mobiles. In its operation, CATCS will be replica of this configuration. Therefore, the DATCS concept will not only provide for a multitude of fleet and/or subfleet addresses but must also be programmed to direct these subfleets to their individual dispatch points by means of their address.

The number of individual dispatch points for the City of Phoenix could include the following:

Phoenix Police (12 positions)

Phoenix Fire/Ambulance (12 positions in new system)

Engineering

Parks and Recreation

Traffic Engineering

Traffic Signals

Facility Maintenance

Human Resources

Equipment Management

Sanitation

Sewers

Streets Maintenance

Job Stimulus Program

Building Inspection

Water Production

Water Distribution

Water Customer Service.

This list assumes that the Phoenix Bus/Transportation, Aviation, Civic Plaza, and EMS remain outside of DATCS due to unique requirements of their communications systems and also functions such as the Radio Shops and Purchasing can be accommodated.
The adjacent communities, if they participate in DATCS, have the potential of adding at least the following dispatch points:

Glendale Police

Glendale Fire

Glendale General Services

Mesa Police/Fire

Mesa Public Works

Mesa Public Sanitation

Scottsdale Police

Tempe Police

Tempe Fire

Tempe Public Works

Tempe Sanitation/Refuse.

In the final configuration, each dispatch point must be independent of any other dispatch point without incidental interaction between them except on command of the central controller. The mobile will not have the capability to modify his address code and thereby change his dispatch point.

To respond to an incident that calls for a mobile command post and a mix of public safety/public service agencies responsible to the command post, the system controller can create a phantom group, and the mobile units can be attached to that group control point by software instructions based on the mobile address codes. There would be no change needed in the mobile logic to create the phantom group or to return the units to their original group and control point assignment.

INTERAGENCY COORDINATION 3.

The survey of Phoenix area agencies* resulted in a pattern of interagency coordination connectivity though by the agencies to be appropriate to their operating needs. This pattern of permissible connectivity between the agencies will be programmed into the controller software and provisions made for it to be modified by program instruction as needed.

IV-52

The need for interagency communication coordination other than police and fire mutual aid channels, was not expressed by the local government agencies of the communities adjacent to Phoenix. If these communities elect to participate in DATCS, the programming of the controller will accommodate their interagency conductivity requirements as needed.

It is expected that regional public safety and public service interagency coordination plans will include the regional disaster assistance plan developed in accordance with Federal Act PK93-288 (Disaster Relief Act of 1974) as amended.

PUBLIC SAFETY INTERFACES 4.

桶

WT.

間

 $\{\Pi\}$

R

In accordance with the system philosophy adopted for DATCS, public safety communications systems, as now configured in the Phoenix area, would not fully participate in the DATCS concept. The key exceptions are described in the following subparagraphs.

Public safety radio systems in Phoenix are portable radio intensive, and present plans are to increase the number of portable radios used by police and fire. This policy is also endorsed by police and fire in Glendale, Mesa, and Tempe. The Scottsdale police also plan to increase their use of portable radios. There is a significant investment in relatively new VHF high-band and UHF portable radio equipment which must be protected. There is also a question of the availability and cost of trunked radio portables. A judgement has been made to continue the operation of police and fire portable radios on conventional radio system channels where they now operate and provide any required coordination with the DATCS by cross patching at the appropriate police and fire dispatch console.

(2) CAD/MDT Interface

Mobile data terminals (MDT) at UHF are now installed in Phoenix police and Mesa public works vehicles. An expansion in the use of MDT is being implemented by Phoenix police and the installation of MDT equipment is being planned by Phoenix Fire and the fire departments of Glendale, Mesa, and Tempe. The police of Glendale, Mesa, and Scottsdale also intend to install MDT equipment.

Questions relative to the compatability of the digital switching architecture of DATCS and MDT systems include consideration of:

(1) Portable Radio Interfaces

Data rate error limitations

Error detection, error correction, retransmission codes

Allowable tandem switching delays

Modifications to CAD to recognize DATCS address codes. The judgement at

The judgement at this time is to continue the operation and planning of MDT as a conventional UHF system. Coordination between MDT and DATCS will occur at the appropriate dispatch centers.

5. EMERGENCY ACCESS PRIORITY

The capability for prompt access to DATCS under conditions which justify the exercise of channel access priority when the DATCS system is fully occupied would be limited to public safety mobiles and possibly certain selected public officials. The exercise of a priority access function when DATCS is fully occupied requires a system frequency plan which designates at least one DATCS radio channel for control information or dedicates one radio channel for emergency signaling. The dedicated control channel configuration is recommended.

The waiting queue priority method has been selected as the means of emergency access to the fully occupied DATCS. When denied access to DATCS, those mobiles with emergency access priority capability can activate this function manually and thereby move their request for channel access to the top of the waiting queue, but not ahead of an emergency request already in the queue. The activiation of this emergency function will be recorded by the central controller along with the identity of the mobile unit making the request and this information will also be displayed at the dispatch points. The central controller will cause the first available DATCS channel to be assigned to the priority request that is first in the queue.

MOBILE-TO-MOBILE COMMUNICATION 6.

Mobile-to-mobile communication is inherent in the repeater configured trunked radio system. Each mobile is assigned an individual address and a group address. The controller software determines what group code or codes will be accessed by each mobile on an initial call-up. A mobile is programmed to communicate with any other mobile in the group through the repeater. By means of the individual address assigned to each mobile, the controller can also be instructed to activate individual units within the group and exclude all other units with the same group address.

If intergroup or interagency flexibility is needed, the dispatcher or terminal operator can, through software instruction to the controller, enable the mobile unit to call up any individual address, group address, or subgroup address within the total capability of the trunked system. Certain paths of connectivity may be specifically prohibited by software programming.

тv-54

TELEPHONE INTERFACE

I

U.E

7.

The survey of existing radio system facilities in the greater Phoenix area* identifies 152 desk top remote control units throughout the total area excluding any remote control units used by police and fire. There are 82 of these units within the City of Phoenix used by city public service agencies, and 70 reported in the communities adjacent to Phoenix.

The reported use of desk top remote control units is mainly for person-to-person communication between an individual mobile unit and the supervisor at a fixed-office location. These office locations in Phoenix also have telephones which are part of the Centrex telephone system serving the Phoenix local government. A DATCS/Centrex interconnect will remove the need for a majority of the 82 City of Phoenix desk top remote radio control units. The interconnect will allow a mobile to access the Centrex telephone system and, by keying the four digit Centrex number on a touch tone pad, the supervisor can access the mobile either by telephone patch at a control point or by a direct radio access touch tone code followed by the mobile unit address code keyed on the telephone touch tone pad.

8. SYSTEM FAILURE MODES

A DATCS must inherently protect itself against common failure modes. The following paragraphs address system features required for system failure protection.

(1) Central Control Failure

The total failure of the master controller at the central location or any of its terminals should not be apparent to field operations. The switching and control logic is resident in the site controllers. The central controller monitors the site controllers and initiates any changes or corrections. In the event of central controller failure, the journal recorder should be switched to the output of the site controllers and any voter selector interfaces with the master controller should be deactivated.

Site Controller Failure (2)

The failure of a site controller should permit full DATCS operations in a degraded mode using the remaining two base station sites. There should be no need to switch the DATCS to conventional radio channel operation or "fail soft" operation. A failure of the

site controller at North Mountain or Thompson Peak will be evident to the users as a temporary reduction in coverage due to shadow effect; the failure of the site controller at South Mountain will probably create some talk-back problems in Phoenix. The failed base station site should also be muted to prevent interference on the voice channels.

(3) Voting Selector Failures

The failure of a voting selector should not be apparent to the field operations; one voice channel will be disabled and its traffic will be distributed automatically among the remaining DATCS voice channels (control channels are not voted). A total failure of the voter selector system affecting all voice channels should enable the DATCS to function in a degraded mode using one base station site, North Mountain or Thompson Peak, whichever proves to have the greater area coverage. The remaining two sites should be muted.

(4) Base Site Repeater Failure

The failure of the repeater at any base site should be transparent to field operations. The failed repeater will not be voted in the DATCS sequence of operation until it is repaired and returned to service. It should not be necessary to take any DATCS voice channel out of service due to the failure of any one of its three repeaters.

(5) Antenna Subsystem Failure

The antenna subsystem at each base station site shall be configured so that a catastrophic failure of any one antenna, combiner, filter, or multicoupler will have a minimum effect on DATCS operations in the Phoenix area. To the degree possible, these antenna subsystem elements will be standardized so that repairs can be made by replacement or interchanging elements.

(6) Control Channel Failure

The DATCS concept includes two separate control channels, one on North Mountain and one on Thompson Peak. The failure of either channel will cause the site controller to assign a back-up voice channel at the site as a substitute control channel. The mobiles or control stations will not require programming modification to accept this substitution. If the loss of a voice channel is not desired due to traffic loads, a degraded mode of operation can be tolerated using a single control channel from either North Mountain or Thompson Peak. The selection of the mode of operation can be made at the master terminal.

Failure or degradation of individual control stations or mobile stations will not degrade DATCS system performance.

IV-56

SYSTEM TESTING AND ALARMS

9.

subsections.

{]}

An alarm panel will be installed at the master control terminal location. This alarm panel shall provide failure alarms (performance below acceptable thresholds) for at least the following:

Excessive VSWR in any antenna subsystem transmission line

Loss or reduction is output power of any transmitter

Loss or reduction in sensitivity of any receiver

Unauthorized carrier on any channel

Loss of site controller integrity

Degradation in repeater receiver interface

Degradation in repeater transmitter interface

Failure of degradation of a voting selector.

The alarm system that alerts to any element of degraded performance should also provide preliminary diagnostic information. This level of diagnosis may be limited to the identity of the offending unit of equipment and the kind of offense being observed.

10. SYSTEM MANAGEMENT AND CONTROL

The high technology features of DATCS are centralized and required integrated control. This implies central management and control for sustaining operations on behalf of the user organizations. Key management requirements are discussed in the following

(1) Journal Recorder/Printer

System operations, as reported by the site controllers and the voting selector, are recorded on a magnetic tape transport. This information will include:

System activity by each user agency control station

System activity by each mobile station

Total channel activity time

Channel use profile by hour, day, week, month

Use of special features such as emergency priority

Use of telephone interconnect

System queue population and holding time in the queue by hour, day, week, month

All alarms by hour, day, week, month.

Additional information entered manually by the master control terminal can include:

User complaint reports

Channel down time for routine maintenance

Channel down time due to failure

Maintenance reports-time/cost.

Selected reports from the journal file can be programmed by the master controller and printed out periodically on the system printer. Other reports can be programmed and printed out off-line by off-line batch process EDP operations.

(2) Centralized System Control

Overall DATCS operations on behalf of the multiple users will be controlled from a master control terminal at a central location. This control point will be responsible for sustaining the quality of system operation, providing limited access to the address code structure from user agency terminals, making programming changes, and implementing temporary program modifications such as interagency coordination and setting up phantom address groups.

(3) Centralized Maintenance

The maintenance of the DATCS must be uniform and standardized. The performance quality of each voice channel should be uniform with every other voice channel regardless of which of the three base station sites is used for each transaction. The skill and training required to achieve this calls for a highly skilled, and centrally managed maintenance staff.

(4) Management

.

The recurring costs of operating, maintaining and expanding DATCS over a phased implementation program, and the high level of performance required of the system required a dedicated and centralized management organization. This management should be at the level of a Division or Department in order to exercise the needed authority. The following data is derived from the Task 4 Report "Documentation of Existing Radio Systems and Reported System Requirements" initially prepared January 10, 1980 and reissued on July 15, 1980, The revised report added Glendale, Mesa, Scottsdale, and Tempe to the data acquired from Phoenix.

APPENDIX C

CURRENT SYSTEM INVENTORY

FORWARD

1V-59

This reproducts with provide wi		2014 	an an an airte an
This real complete with the dear rep conclusion of survey data at survey data at			
This reproducts the second sec			17 17 18
This propulse with promite with provide and in a constraint of reviews and in a constraint of reviews and in a constraint of in			
This series promises with the data reproducts and conclusion of artises and conclusion of messa series Program. 1 . <u>purpose</u> , the 1 . <u>purpose</u> 1 . <u>purpos</u>			
aL complies with phoening, Ati, The data record coviews and record survey data) all 1. gurvey data) mess, scotts program. []] 1. gurvey data) mess, scotts data (1) []] 1. gurvey data) mess, scotts []] . []] <td></td> <td></td> <td>This repo</td>			This repo
The data room conclusion of reviews and 1 Buryoy data 1 Mesa, scottas Trogan.			complies with Phoenix Arizo
Contents and Function of the present			The data report
Survey data 3 Program. 1. PuRPOSE The ult: information 2 DUTOSS for 1 DUTOSS, the Con Systems. Section of the present Section of the present Section of the present Section of the present Section of the present Section The doministic of I. Definition Section Section of the present Section Section of the present Section Secti			reviews and ir
Program. 1. <u>PURPOSE</u> The ult: information 4 in effective (DATCS) for 1 in the information 4 information 4 informatio		0	survey data re Mesa, Scottsda
1. PURPORE The ult: Information a Dards) for purpose, the Information a Dards) for purpose, the Information a Particle Information a Particle Information a Particle Information a Particle Information a Information a <t< td=""><td></td><td>-</td><td>Program.</td></t<>		-	Program.
The ult: information a an effective (DATCS) for (purpose, the . Con . Co			1. PURPOSE C
L			The ultin
Image: construction of the present of the the great of the present of the presen			an effective of
L Cor Sr Sr L Cor Sr L Cor Sr L Cor Sr L Cor Sr L Cor Sr L Cor Section Of the preser Section Of the preser Section I Cor Section I Cor I Cor			purpose, the r
Image: Systems of the present of			. Conf
4.1 . The . Pla . Pla . Section of the presensition of the presensition of the presensition of the presensition of the greater of th			syst
Pla . Pla . DAT Section of the presense systems. See . also identified <			. The
 Pla DA' Section of the present systems. Set will be required in the required in the rest of the great is identified. 2. APPROACT The cent resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i new period. Determine the rest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of interest of the resources emplayed approach to i consisted of the resources emplayed app		g we could be	. Defi
DA Section of the presen systems. Sec will be requi- munity of usi- also identif: to the great 2. <u>APPROACI</u> The cent resources em approach to consisted of . Det . Thi Preceding page blank			. Plar
Section of the presen- systems. Sec will be require munity of use also identif: to the greate () 2. <u>APPROACH</u> The cent approach to a consisted of . Bet . In per . In per . In per . In per 		an a	. DATC
of the press systems. Set will be required munity of us also identifi to the greate 2. APPROACT The cent approach to a consisted of . <t< td=""><td></td><td></td><td>Section 1</td></t<>			Section 1
will be required in the required in the required in the greater in the design of the greater in the		to any other sector	of the present systems. Sect
Also identify to the greate 2. <u>APPROACT</u> The cent resources em approach to consisted of			will be requir munity of user
2. APPROACH The cent resources em approach to consisted of . Det . Rev . Int per Preceting page blank			also identifie to the greater
The cent resources emp consisted of . Det . Rev . Int per Preceding page blank		A CLARK	2. APPROACH
Interesting approach to consisted of interesting approach to consisted approach		A series of the	The centu
L approach to f consisted of			resources empl
De Ren In per Preceding page blan			consisted of t
Re- In- pe: I Preceding page blank			. Dete
L Interpretended in the second s	$^{\circ}$ and $^{\circ}$. Rev
pe:			. Inte
L Preceding page blant		m	pers
L The second sec		and provide a second seco	
			Preceding nage blank
			I I AAAAm9 haba marini
			an a

I. INTRODUCTION

ort, prepared by Booz, Allen under contract to APCO, Task 4 of the "Support Agreement and Work Plan for ona Under APCO Project 16B" dated January 10, 1980. Inted herein is a direct result of the successful Task 2 (Data Collection Guide) and Task 3 (document Interviews). This revision includes the additional esulting from the decision by the cities of Glendale, ale, and Tempe to join with Phoenix in the Project 16B

OF THE REPORT

mate purpose of this report is to provide a body of and data upon which to develop the system concept for digitally addressed trunked communications system and City of Phoenix and its suburbs. To achieve this report includes the following:

figuration of existing municipal and suburban radio tems

present operation and interfaces of the radio systems

ciencies of the existing systems

n for system improvement

CS functional requirements.

II of the report discusses and illustrates the status t radio systems including problems with existing tion III of the report addresses the functions which red to satisfy the communications needs of the comrs as discussed during the interviews. Section III es those DATCS functions which appear most appropriate r Phoenix area DATCS concept.

TO DATA ACQUISITION

ralized management and maintenance of radio system loyed by the City of Phoenix permitted an efficient he collection of the required data. This process the following steps:

ermination of the kind of data to be acquired

iew of available radio system documentation

erviews with department and division communications sonnel.

The type and elements of data needed was decided by mutual agreement with the project director for Phoenix and Booz, Allen project personnel. These decisions were transformed into a written data collection guide which was used to record the needed data during the review of documents and during the personal interviews.

The review of appropriate documents was confined to three key documents:

A City of Phoenix radio system configuration manual created and maintained by the Communications Project Section of the Equipment Management Division

.

- The computerized inventory of radio equipment owned by the City of Phoenix
- The records of annual billing to the City of Phoenix municipal departments and divisions for radio equipment maintenance and replacement.

These records enabled the determination of the City of Phoenix radio equipment inventory, equipment locations, radio system control points, and radio system configurations prior to the interviews.

Personal interviews were then conducted with all principal City of Phoenix department and division communications personnel. The one-hour interviews were conducted to:

- . Confirm previously recorded equipment inventory equipment locations and system configurations
- Define the mode of operation for each of the radio systems
- . Identify present radio system problems and alleged defects
- . Identify plans for improving individual radio systems
- Identify DATCS functions that could improve agency radio operations.

The data from the document reviews and the personal interviews was consolidated and summarily recorded in data collection guides for each agency. One set of these guides is held by Booz, Allen, and a copy is held by the deputy project director in Phoenix.

program. tives of	Personathe fol.
•	Mesa Po
•	Mesa Pu
•	Scotts
•	Glendal
	Glendal
•	Glendal
•	Tempe I
•	Tempe P
Data compiled data colle by Booz, director	from th for the ection g Allen, a in Phoen

(-)

同

- Listen

T

IV-62

At a later date, the suburban communities of Mesa, Scottsdale, Glendale, and Tempe elected to participate in the Project 16B program. Personal interviews were then conducted with representatives of the following agencies:

olice and Fire

ublic Works

dale Police

le Police

le Fire

le General Services

Police and Fire

Public Works and Sanitation/Refuse.

hese interviews was consolidated with the data City of Phoenix. The input data is recorded in guides for each agency, one copy of which is held and a duplicate copy is held by the deputy project hix.

	••••••••••	
	T	
	time 1.1	
	144	II.
\int		The status of
	1.5 Y	lowing paragraphs.
		accurate. It is s
\sim	(T**)	sents the status a reconfirmed with t
		planning.
	n 1	1. RADIO SYSTEMS
		The survey da
	TI	pairs. In additio
		dedicated radio sy aviation, and reso
		of radio channels
	11-	Phoeni
	<u>1</u> .1	No. Agencies F
		21 S
	¥2, «	C
		5 D
	11 ¹ 3	
	17	
	F	
		Mha Dhaoniw r
	السل	the total radio fr
		and the fire depar frequency resource
	the st	are considered tog
	64 15	
	A Contraction of the second seco	
	MA	
		Preceding page blank
\mathbf{L} , where \mathbf{r} is the second		na na mana ana ana amin'ny fisiana amin'ny fisiana amin'ny fisiana amin'ny fisiana amin'ny fisiana amin'ny fis Na fisiana amin'ny fisiana amin'ny fisiana amin'ny fisiana amin'ny fisiana amin'ny fisiana amin'ny fisiana amin'

STATUS OF PRESENT SYSTEM

the present radio system is described in the fol-This information evolves from the data base terviews and from computerized data believed to be subject to inevitable change and although it repreat the time the data was acquired, the data should be the originators before being used for subsequent

ata shows 21 municipal activities in the City of ix VHF high-band channels and four UHF frequency on to these 10 shared radio systems, there are five stems serving the police, fire, city transit buses, cue/EMS functions. Table II-1 shows the distribution among the 26 City of Phoenix activities.

TABLE II-1

ix Radio Channel Distribution

Facility	VHF	UHF	Special	1
Shared Channels	6	4		
Dedicated Channels				
Police	9	8		•
Fire	7	1		
Rescue/EMS		10		
Transporta- tion (Bus)		2	• 1	
Aviation	1		6 VHF Aviat	ion AM

police department uses approximately 35 percent of requency resouces available to the City of Phoenix, rtment uses approximately 38 percent of the radio es, assuming rescue/EMS channels and fire channels gether as fire assignment channels.

The suburban Phoenix area radio systems are predominately VHF high-band and tend toward dedicated channel operation with each agency operating on its own frequency with the exception of mutual aid channels. The channels configuration is mainly dual frequency repeater operation. The survey data shows that among the 16 departments of the 11 participating suburban agencies the following is the radio channel distribution.

TABLE II-2

S	uburban	Phoenix	Radic	Chann	nel	Distr	ibution		
		Dua	l Frec VHF	luency		Sir	ngle Frequ VHF	lency	UHF
Police			8				7		4
Fire			2				1		0
Public W	lorks		5		1		2		2

This table does not include area mutual aid operational frequencies or repeater talk-around channels. It shows a need for 40 individual VHF high-band frequencies and six UHF pairs to serve the 16 user organizations. Of these, the police departments use approximately 58 percent of the VHF resources and 67 percent of the UHF resources; the fire departments use approximately 12 percent of the VHF resources and no UHF resources.

(1) Radio Channels and Assignments

The VHF high-band radio channel frequencies and their activity assignments are shown in Tables II-3 and II-4. The UHF radio channel frequencies and their activity assignments are shown in Tables II-5 and II-6. It is noted that in Table II-3 the Streets #2 radio channel had no activity assignment at the time the data was taken. Earlier use of this frequency resulted in adjacent channel interference which is believed to have been corrected. However, tests are still being run by simulcasting Streets #1 and Streets #2 channels, and Streets #2 channel has remained unassigned until the test results are conclusive.

Table II-3 (City of Phoenix VHF Assignments) does not include the six VHF Aviation Services frequencies used by the Phoenix Aviation Department in addition to their local government radio channel. These frequencies are:

121.70 MHz Litchfield ground control

121.80 MHz Deer Valley ground control

IV-66

ACTIVITY
Equipment Management Sanitation Streets Maintenance Purchasing
Building Maintenance Crisis Intervention Job Stimulus Planning City Attorney Building Inspection
Water Production Sewers
Water Distribution
Water Customer Service
Aviation
Fire
Police
(Unassigned)
L

TABLE II-3 CITY OF PHOENIX VHF ASSIGNMENTS

	· · · · · · · · · · · · · · · · · · ·		and a second
RADIO CHANNEL	FREQUENCY	MOBILES	PORTABLES
STREETS #1	157.290 MHz	48 92 89 <u>6</u> 235	0 0 0 0 0
BUILDING SAFETY	155.145 MHz	1 26 - - <u>-</u> 72 99	1 - - - - 1
WATER #1	158.130 MHz	114 <u>74</u> 188	0 _0 0
WATER #2	158.775 MHz	150	0
WATER #3	153.680 MHz	101	0
AVIATION	155.760 MHz	115	29
Fire I Fire II Fire III	154.190 MHz 154.250 MHz 154.070 MHz	230	
Fire IV Fire V Fire VI Fire VII	154.280 MHz 153.830 MHz 154.310 MHz 154.145 MHz	0	147
		230	147
Chase100Information200Detectives300South400Central Talk500NorthNorth600East700West800Interagency	154.890 MHz 154.755 MHz 155.790 MHz 155.370 MHz 155.070 MHz 155.640 MHz 155.520 MHz 155.430 MHz 155.430 MHz	636	977
STREETS #2	156.180 MHz	0	0

TABLE II-4-1 PHOENIX SUBURBAN VHF ASSIGNMENTS

.

- r

the second secon

Activity	RADIO CHANNEL	FREQUENCY	MOBILES	PORTABLES
Mesa Police	Main Outgoing	155.010 MHz	27	126
	Main Incoming	155.130 MHz		
		159.150 MHz		
	Detectives	154.815 MHz	I.	
		158.850 MHz		
Mesa Fire	Main Fire	154 340 MHz	15	14
	main 1110	154.010 MHz	- -	ана стана 1 та
	Mutual Aid	154.280 MHz		
	Civil Defense	47.660 MHz		
Mesa Public Works	Utility Ops	155.820 MHz	163	37
		153.920 MHz		
	Streets/Parks	156.015 MHz		
		153.815 MHz		
	Electrical Dept.	153.470 MHz		
Scottsdale Police	Operations	155.850 MHz	80	61
		154.845 MHz		
	Back-Up Ops	155.550 MHz	1	
	•	155.725 MHz		
	Talk-around	155.550 MHz		
	Mobile-Inter	155.475 MHz		
	Mobile Only	156.030 MHZ		
	Local Govt	153.750 MHz		:
•				· · ·
Glendale Police	Operations	<u>156.210 MHz</u>	0	60
		154.830 MHz		
	Talk-around	156.210 MHz	1 	
	Talk-Dack	155.670 MHZ		
Glendale Fire	Main Operations	154.310 MHz	30	30
	Mutual Aid	154.280 Mhz		
Glendale General	Operations	158.830 MHz	130	0
Services		153.785 MHz		
	Dial-a-Ride	154.995 MHz		
Tempe Public Works	Operations	154.965 MHz	230	20
· · · · · · · · · · · · · · · · · · ·		155.880 MHz		
	Talk-around	154.965 MHz		
Tempe Sanitation/	Operations	158,865 MHz	40	30
Refuse	operations	153.740 MHz	1	30

ACTIVITY Tempe Police \square Tempe Fire []1 11 ₫.). f 8 TV (). []U

J.

 \mathbf{M}

 \square

IV-68

TABLE II-4-2 PHOENIX SUBURBAN VHF ASSIGNMENTS (Continued)

RADIO CHANNEL	FREQUENCY	MOBILES	POPTARIES
			TORIABLES
Operations	155.835 MHz	5	110
Operations	154.800 MHz		
operacions	$\frac{155.685 \text{ MHz}}{154.710 \text{ MHz}}$		
Investigation	155.685 MHz		
State NARC	155.445 MHz		
Public Works	155.475 MHz		
	154.965 MHz		
		· · · ·	
Operations	154.145 MHz	10	
	153.770 MHz	10	25
Fire-Ground	154.145 MHz		
Phoenix Fire	154.280 MHz		
Phoenix Fire	154.250 MHz		
Phoenix Aid	154.070 MHz		
mesa Fire	154.010 MHz	4	
Mesa Talk-around	154.340 MHz		

CHIVITY	RADIO CHANNEL	FREQUENCY	MOBILES	PORTABLES	f 6 ·
Indipering		453 500 MHz	119	6	
Parks and Recreation	LOCAL GOV'T.#1	458.500 MHz	64	9	
Traffic Engineering			27	0	
			210	15	
Traffic Signals	LOCAL GOV'T.#2	453.950 MHz	12	4	T
Facility Maintenance		458.950 MHz	52	2	H,
			64	6	Fe
Human Resources	HUMAN RESOURCES	453.875 MHz			
		458.875 MHz	32 🎉	1	بر جنه کا
Communications Maint.		453.625 MHz	9	12	Constant Constant
Civic Plaza	COMM. MAINT.	458.625 MHz	0		R
			9	37	R i
	CITY BUS #1	452.625 MHz			
		457.625 MHz			₹ 5 8
Transportation			253	2	7
	CITY BUS #2	452.725 MHz			1944 - 947 - 14
		457.725 MHz			· · · ·
Fire	FIRE UHF	460.575 MHz			
		465.575 MHz	15	33	
Police	450-1	453.100/458.100			
	450-2	453.200/458.200			
•	450-3	453.450/458.450			
	450-4	453.600/458.600	27	246	ţ,
	450-5	453.800/458.800		-	
	Interagency	460.375/465.375			
	MDT-1	460.525/465.525			ſ
	MDT-2	460.550/465.550			
Rescue - EMS	Med-1 State	463.000/468.000			1 5 - 14
	Med-2 State	463.025/468.025		a - 19	
	Med-3 Phoenix	463.050/468.050			Ł. J
	Med-4 State	463.075/468.075			
	Med-5 State	463.100/468.100	10	17	
	Med-6 Phoenix	463.125/468.125			
	Med-7 Phoenix	463.150/468.150			
	Med-8 Phoenix	463.175/468.175			
	Med-9 Phoenix	462.950/467.950			
	Med-10 State	462.975/467.975			9 - 19 - 19 - 19 - 19 - 19 - 19 - 19 -
	Mobile Repeater 1	460.550/468.775	(1)		P. Stand
	Mobile Repeater 2	460.550/468.175	(1)		

TABLE II-5

Mesa Public Works

ACTIVITY

Mesa Police

- March

-

- 8 -

 \square

T

Ħ.

19.10

IV-70

TABLE II-6 PHOENIX SUBURBAN UHF ASSIGNMENTS

RADTO CHANNEL	FPROTENCY	MODITING	
	TREQUENCI	MOBILES	PORTABLES
	a.		
Detectives Detectives	453.325 MHz 458.325 MHz 453.375 MHz 458.375 MHz		42 not yet in service
Administration	453.750 MHz		
MDT	458.750 MHz 453.775 MHz 458.775 MHz	8	•
Sanitation	453.350 MHz 458.350 MHz	60	
Utility Data (MDT)	453.900 MHz 458.900 MHz	27 87	

121.90 MHz Sky Harbor ground control

122.95 MHz Unicom - Sky Harbor/Deer Valley

123.00 MHz Litchfield Unicom

123.50 MHz Instructional.

Due to the FCC restrictions related to permissable communications on these frequencies and the type of radio service in which they are employed, it is assumed that these channels would not be integrated with any other municipal radio service. Therefore, these radio channels have not been included in the tables of frequencies.

For similar reasons, any General Mobile Radio Service or Personal Radio Services (Citizens Band Radio Service) channel or frequency that may now be used by a governmental activity has not been included in these tables.

City-wide paging within the City of Phoenix is also assigned to the Phoenix building safety radio channel (155.145 MHz). The data base indicates approximately 125 paging receivers may be in service on this channel.

(2) Mobiles and Portables

The distribution of mobiles and portables among the participating governmental activities and radio channels is also shown in Tables II-3 and II-4 for VHF high-band and in Tables II-5 and II-6 for UHF. The data base indicates the following summation:

	Mobiles	Portables
Phoenix VHF hi-band	1,754	1,154
Suburb VHF hi-band	738	513
Phoenix UHF	620	357
Suburb UHF	95	42
	3,207	2,066

Of the total number of mobile radios operated by the surveyed agencies, the police use approximately 25 percent, and the fire departments (including rescue units) use approximately 10 percent. Of the total number of hand-held portables (not including CB units) operated by the surveyed agencies, the police use approximately 79 percent, and the fire departments use approximately 13 percent. The City of Phoenix accounts for approximately 74 percent of all mobile and portable municipal radio equipment in use in the surveyed area.

ALC: N

Most primary base station installations for the City of Phoenix radio systems are located on North Mountain and South Mountain. Only the Phoenix transit radio system (bus communication) does not have a base station on North Mountain, but plans indicate that this installation will be made in the near future.

In addition to the primary base stations, there are 38 fixed stations at selected locations in Phoenix. These stations serve as stand-by facilities or operational facilities for local use. There are five UHF repeaters included in the 38 fixed stations which account for an additional nine fixed control stations.

The distribution of City of Phoenix base stations, control stations, and other fixed stations is shown in Table II-7.

The suburban Phoenix area communities have a wider distribution of base station/repeater locations. The number of these in each community, as reported during the survey, is shown in Table II-8. The location of each base station as reported by each community is shown in Table II-9.

(4) Satellite Receivers

Satellite voting receivers are used extensively by the City of Phoenix and the four participating suburban communities for police operations because of the hand-held portable configuration of police communications. It is planned to expand the use of satellite voting receivers for City of Phoenix fire and rescue (EMS) communications. Nearly all of this City of Phoenix expansion is in the process of being completed.

Satellite voting receivers are also extensively used by the police in the four participating suburban Phoenix communities. Glendale, Mesa, and Tempe report plans to expand Satellite receiver coverage for their public safety operations. Tempe also reports plans to extend Satellite receiver coverage for Public Works and Sanitation.

A summation of all planned satellite receiver locations in the City of Phoenix and the radio channels they serve is shown in Table II-10. The table shows there are 295 receivers at 29 locations which vote 10 police channels, three fire channels, and five EMS radio channels. At the present time, there are 165 receivers installed and operating; 92 percent of these are used on the 10 police channels, and eight percent of these are used in fire channels, mainly Fire III.

(3) Base Stations and Fixed Stations

ACTIVITY Engineering Parks and Recreation Traffic Engineering Traffic Signals Facilities Maintenance Equipment Management Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	DISPATCH CENTER	A REMOTE CONTROLS 5 3 7 2 1 3 5 2 1 3 5 2 2 4 4 2 4 3	CCESS CONTROL STATIONS 2 1 2 1 2 1 3	EOC REMOTE X X	FI: NORTH MT. XR XR XR	XED STATT SOUTH MT. XR XR XR	ONS OTHER IR IR 2			
ACTIVITY Engineering Parks and Recreation Traffic Engineering Traffic Signals Facilities Maintenance Equipment Management Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	DISPATCH CENTER	REMOTE CONTROLS 5 3 7 2 1 3 5 2 2 4 4 2 4 3	CONTROL STATIONS 2 1 2 1 2 1 3	EOC REMOTE X X	NORTH MT. XR XR X	SOUTH MT. XR XR XR	OTHER IR IR 2		و من من المراجع المراجع والمراجع المراجع المراجع والمراجع المراجع	
Engineering Parks and Recreation Traffic Engineering Traffic Signals Facilities Maintenance Equipment Management Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	5 3 7 2 1 3 5 2 4 4 2 4 3	2 1 2 1	x x x	XR XR X	XR XR X	1R 1R 2			
Parks and Recreation Traffic Engineering Traffic Signals Facilities Maintenance Equipment Management Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	3 7 2 1 3 5 2 4 2 4 3	2 1 2 1	x	XR	XR	1R 1R 2		الموادية الم الموادية الموادية الم الموادية الموادية الم	
Traffic Engineering Traffic Signals Facilities Maintenance Equipment Management Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Severs	1	3 7 2 1 3 5 2 2 4 2 4 3	1 1 1 3	x	XR	XR X	1R 2		می دود. و میکند از میکند میکند و میکنوند. از میکنوند و میکند از میکنوند و میکنوند. همچنین میکنوند و میکند میکند و کارمیکی میکنوند. و میکنوند و میکنوند و میکنوند و میکنوند و میکنوند و میکنوند و م	
Traffic Signals Facilities Maintenance Equipment Management Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	2 1 3 5 2 4 4 2 4 3	2 1 3	x	XR	XR X	1R 2		المحمد بالمحمد الأمانية معالم المحمد الم	
Facilities Maintenance Equipment Management Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	1 3 5 2 4 2 4 3	3	x	X	X	1R 2			
Equipment Management Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	1 3 5 2 4 2 4 3	3	x	x	x	2			
Sanitation Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 1 City Bus 2 Water Production Sewers	1	3 5 2 4 2 4 3	3	x	x	X	2		مر می مرکز می اور این از مرکز می اور این اور ای مرکز می اور این اور این مرکز می اور این	(Tii)
Street Maintenance Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	2 2 4 2 4 3	3	x	x	X	2			
Purchasing Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	4 2 4 3	3	X	•	A				111
Building Maintanance Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	4 2 4 3	3	x						1711
Job Stimulus (JSP) Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	4 2 4 3	3	x					1 311	the state
Building Safety Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	4 2 4 3	3	x	1			fam t		
Human Resources Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	2 4 3	3			x	4			
Communications Maint Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	4	3		XR	XB	18	- San I		12
Civic Plaza Transportation City Bus 1 City Bus 2 Water Production Sewers	1	3			XR		<u>+</u> K	-		
Transportation City Bus 1 City Bus 2 Water Production Sewers	1	3		<u>}</u> .				TT		TT
City Bus 1 City Bus 2 Water Production Sewers		1 3						╡∦≬		
City Bus 1 City Bus 2 Water Production Sewers			1				סו	5-21 4		<u>11</u>
Water Production Severs						-	10	17 th 1		
Sewers	1 1	14	┥╤╾╌╴	×		· · · · ·	5			11.7
Devers		7.4			•	· A	J	- ((¹		
Wator Dictribution			<u> </u>	v		v	2	-		
Water Distinguide			+	<u> </u>	v v	× ·	<u>2</u>	- 63		17-1
Ariation	7	12	+	<u> </u>		(X)	6	+ $($		
Sky Harbor	-	<u>т</u>	10.00			(A)	0	1		ال _{سم} ا
Litchfield	· · · · · ·						7	R*3		the start
Deer Waller		•	1. A. S.		1					11
Police				<u> </u>	 		_		H	L.)
Chace 100	2				v i	Y	1			•
Information 200			1. Sec. 1.		Y I	X V		()	E.	£7*** }
Detectives 200						N V)) (
South 400				v	i v	A I				
Control Malk 500				A	v	N N		a n		
North 600		1. T		v	N N	A V	L. 			
North 800			1			A V	1	Kana		
East 700		1				A V	- I - J			
West 800						•	2	\$ 1		(F)
AFO 1						v				
450-1		5				A V		Non-scal		L)
450-2				47		A V		197-1		
450-5						A V				$\langle \rangle$
450-4		· · · ·				A V	and Alexandre	(L.)		
						•				
Interagency UHF								r		T
MDT-1				. .	- A	· A		1 X		
MD1-2					Rec	A	·····	-		8/
Fire T		8 a.		v v		v	· · ·			
fire i Fire I			4		v −	v	1			57
FIRE II				A A		A V		C 3		
fire 111						A V	L .			
rire IV				1 .	•	•		51		()
Fire V			а. 	. · · · ·			7			
fire Vi							1	\$. I		£_}
rire VII			1.1				:	AT SHITTL		1
FIRE UHF		<u> </u>	<u></u>		+	· · · · · · · · · · · · · · · · · · ·				C 7
Rescue						v	No-1 C			
Med-1 Thru Med-10	L .	ΤŪ	1		A Dee	X Doc	mea-9			, 94an, 44
MODILE Repeaters	ļ	<u> </u>	<u> </u>	4	Kec	Rec	4			Economic
		I	TT_ 7 A					10 H	1 11	1 1 N

	· PHOEN	IX SUBURB (COMMAND AND	CONTROL		TABLE .	11-8
		ACCI	ESS	· · ·	FIXE	D STATIONS	
ACTIVITY	DISPATCH CENTER	REMOTE CONTROLS	CONTROL STATIONS	EOC REMOTE	REPEAT- ER	BASE STATION	OTHER
Mesa Police Mesa Fire Mesa CD Mesa Public Works	1 - - 1	0 0 0 57	0 0 0 0	0 0 - 0	7 2 0 4	0 0 1 1	0 0 0
Scottsdale Police	1	2			2	2	0
Glendale Police Glendale Fire Glendale General Svc	1 1 2	0 0 3	0 0 0	1 1 1	2 0 1	2 1 1	0 0 0
Tempe Police Tempe Fire Tempe Public Works Tempe Sanitation	1 1 0 0	0 0 8 1	0 4 0 0	0 0 0 0	3 1 1 1	0 0 0	0 4 0 0

25

TABLE II-9 SUBURB BASE STATION LOCATIONS

42	Sector Se		N	
	MESA	SCOTTSDALE	GLENDALE	TE
LOCATION	Police #1 Police #2 Police #3 Fire #1 Civil Defense Utilities Streets/Parks Electric Sanitation	Police #1 Police #2 Local Govt	Police #1 Police #2 Fire Gen Svcs Dial-A-Ride	Police #1 Police #2
MESA 300 E. 6th 130 N. Robson 340 E. 6th 20 E. Main 13 W. 1st St. 425 S. Belleview	XXX XXX X X X X X X X X X X			
SCOTTSDALE 3739 Civic Plaza City Yard		x		
GLENDALE 63rd/Northern 59th/Grand		•	X X X X X	
TEMPE Tempe Butte				xxx

IV-76



SATELLITE RECEIVER LOCATIONS-PHOENIX FIRE, POLICE, AMBULANCE

TABLE II-10

			FIRE I	FIRE II	FIRE III	MED-3 TELEM	MED-6 TELEM	MED-7 TELEM	MED-8 TELEM	MED-9 COORDINATION		CHASE 100-POLICE	INFO 200-POLICE	POLICE 300	POLICE 400-SOUTH	POLICE 500-CENTRAL	POLICE 600-NORTH	POLICE 700-EAST	FOLICE 800-WEST	POLICE 450 - 1	POLICE 450 - 2	
:	FIRE STA 4				х		·					X	X	X	X	х	-	х	X	X	X	
	FIRE STA 8		X	X	X	X	X	X	Х	Х												
	FIRE STA 12		X	Х	х	Х	X	Х	X	X		X	X	X		X		X		X	Х	
	FIRE STA 13																	X				
-	FIRE STA 14	-	<u>X</u>	<u>x</u>	<u>X</u>							X	<u>X</u>			X		-	X	X	<u>X</u>	
	FIRE STA 18		X	X	х							X	X	X		X	X	X		X	X	
	FIRE SWA 19	- 1		.,				!!							X	X						
	FIRE STA 20	, n.	X	X	X	v	v	v	· v	v		X V	X	X	v	v	х	х	i.	X	•Х •	
	FIRE STA 21		A V	. A.	A V	A	X	X		. .		X	A V	X V	X. V	X V			X	Ň	X V	•
-	FIRE SIA 22		A V	- <u>-</u>	A V	v	v	v	v	4		<u>^</u>	 v	- <u>A</u>	 V	N V						
	FIRE STA 23 FIRE STA 27		A V	Y	A Y	A V	л У	A Y	A V	v v		A Y	A Y	A Y	Δ	▲.	v			A V	Y	
	FIRE STA 31		A Y	N N		x	Y.	x ·	X	A X		A X	x	л У			X			x X	X	
	FTRE STA 34		X	x	x	A	*	42	. *			X	x	x	x		~		X	x	x	
	FTRE STA 42	.	X	x	X	x	x	x	x	x		x	x	x			x		**		••	
. •	COUNTY HOSP	-1	x	x	x	x	x	X	x	x		x	X	x	X	x		x		x	x	
	MUNI BLDG		X	х	X	X	х	х	X	х		Χ.	х	Х	х	x		-	х	X	X	
	CIVIL DEFENSE		х	х	х	x	х	X	X	X		X	X	X	х	Х		х		X	X	
	MT. BELL ROOF		X	х	х	х	х	х	X	х		X	Х	х		X		X		X	Х	
	MARYVALE BRIEFING STA				•		•	-				X	Х	X	:	Х	-		Х	X	Х	
	NORTH BRIEFING STA		X	Х	Х	Х	X	X	X	X	1	X	Х	Х		-	X		-	х	X	
	PUMP STA-75th AVE		X	х	х	X	X	X	X	X		X	Х	х		X			X	Х	X	
	DEER VALLEY WATER		X	Х	Х	X	Х	X	X	Х		·X	Х	X						X	Х	
	SQUAW PEAK WATER PLT		Х	X	Х	X	Х	X	X	X		X	X	X	•		Х	X		X	х	
	SKY HARBOR BR STA	_							· .						X						:	
	S. MT. RESERVOIR														X	`				2		
	CAPT. GALLEY REST	·														X						
	EAST WING BAGGAGE-SKY HARE	3.											. •			X		24 i				
	NORTH W. BRIEFING STA		X	Х	X	X	Х	X	Х	Х												
		1					7												•. •			

IV-77

(····

 \bigcirc



A summation of all reported satellite receiver locations for Mesa, Scottsdale, Glendale, and Tempe is shown in Table II-11. There are 36 receivers at 13 locations which vote nine police channels, two fire channels, and a civil defense radio channel.

6)

2. SYSTEM OPERATION

The following discussion of system operations is derived from the interview data and from available documentation. It is believed that these discussions fairly represent system operations as they are known to the radio system users at this time.

(1) Interagency Communications

Interagency communication is crucial to the coordination of municipal services and to the coordination of public safety emergency services. The present interagency communications links, as reported by the City of Phoenix personnel interviews are shown in Table II-12.

The City of Phoenix interagency coordination matrix shown in Table II-12 is in part by factors not necessarily related to the need for functional coordination. Incidental coordination is forced on some City of Phoenix activities by their being cochannel users. Other activities achieve coordination through equipment and controls installed for after-hours monitoring of their working radio channel by other municipal services. The City of Phoenix responders to the interview questions set forth a revised set of interagency communication links that they felt to be more appropriate to their needs. This set of interagency communications links is shown in Table II-13.

Interagency communication in Mesa, Scottsdale, Glendale, and Tempe is mainly limited to the regional police mutual aid mobile radio channel and the regional fire mutual aid mobile radio channel. Mesa dispatches fire and police from the same dispatch center which provides automatic coordination of these two agencies. Tempe also has direct mobile radio communications on Mesa fire and Phoenix fire frequencies. Tempe police have a remote control for the Tempe Public Works base station, and Mesa Police/Fire dispatch center has a remote control for the Civil Defense base station. Any other suburban interagency communication is limited to the use of telephones according to the survey responses.

(2) Dedicated Systems

The radio systems that are dedicated to the service of one City of Phoenix municipal agency include police, fire, transit, aviation, and rescue/EMS. The need for dedicated service in the City of Phoenix is determined by the requirements of public safety emergency operations and by high radio channel loading. The characteristics of these City of Phoenix radio systems are

(----

IV-79

TABLE II-11 SUBURB SATELLITE RECEIVER LOCATIONS

	MESA	SCOTTSDALE	GLENDALE	TEMPE
LOCATION	Police #1 Police #2 Police #3 Fire #1 Civil Defense	Police #1 Police #2	Police #1 Police #2 Fire #1	Police #1 Police #2
<u>MESA</u> 8300 E. Main 300 E. 6th 1400 S. Dobson 20 E. Main	X X X X X X X X X X X X X X X X X			
<u>SCOTTSDALE</u> Airport Miller/Kellips		x x x x		
GLENDALE 67th and Bell 115th Ave 2040 W. Camelback 7601 N. 63rd 32nd and Greenway 69th and Bethany			X X X X X X X X X X X X X X X X X X X	
TEMPE South Station				xx



RESENT INTERAGENC	Y COMMUNICATION			,				, ,			RE	POR	TED	CO	MMU	NIC	ATI	ON .	LIN	к Т	<u>_</u>	<u> </u>
INKS - PHOENIX, AN ABLE II-12 REPO	RIZONA RTING AGENCY	90		lance	Projects	c Plaza	. Safety	n Resources	tion	ic Transit	neering .	5	fic Eng.	littes Maint.	ets Maint.	cation	pment Mgt.	r Prod.	S	c pist.	c Svc.	
GENCY	RADIO SYSTEM	Polic	Fire	Ambu	Comm	Civi	Blåg	Humai	Avia	Publ	Engi	Park	Trafi	Faci	Stree	Sanit	Equil	Wateı	Sewei	Water	Water	
OLICE	POLICE	x	x	x																		T
IRE	FIRE	x	x	x	x				x	ъ.,												
MBULANCE	FIRE/EMS	x	x	x					x								•					
COMM. PROJECTS	RADIO SHOP		x		х	x	[
IVIC PLAZA	RADIO SHOP				x	х				.:							-					
UILDING/SAFETY	BLDG/SAFETY						x	x			Ċ											T
UMAN RESOURCES	HUMAN RESOURCES							x			•									•		
VIATION	AVIATION								x							•						
UBLIC TRANSIT	CITY BUS	x	X,	x		÷				x		· .	х		x			 .			(⁻ .	
NGINEERING	LG #1			1.1							x	x	x		Ķ	x						
PARKS/RECREATION	LG #1	·									X	x	х				-					Ì
RAFFIC ENG.	LG #1	•] .		- 11				•	x	x	x				•				ł	
RAFFIC ENG (Signals)	LG #2					•	-				x	x	x									
ACILITIES MAINT.	LG #2	5				•						1	x	x	x					1 .	·	l
STREETS MAINT.	STREETS #1			1											X	х	X					T
SANITATION	STREETS #1											ł			х	X	x					
EQUIP. MGT.	STREETS #1		1	. *		-							- 24		x	x	x		}			
ATER PROD.	WATER #1	-																x	x	x	x	
SEWERS	WATER #1									. *								x	X			
ATER DISTR.	WATER #2			-														x	x	x		I
ATER CUST.SVC.	WATER #3							ŀ			н. н. ,							x	x	x	x	
ITY ATTY.	BLDG. SAFETY	1					X	x											1			
LANNING DEPT.	RLDG. SAFETY	•					x	x					.			.						
				ļ	-					-	Ļ			L		L	L	Į	<u>ي</u>	1		

IV-80

Γ

 $\hat{\omega}$

 $\langle \rangle$



DESIRED INTERAGENCY COMMUNICATION LINKS

REPORTED COMMUNICATION LINK

																_		-		
	TABLE II-13					ts			ces		it			×	faint.	lt.		Ŀ.		
	REPC	RTING AGENCY	υ		ance	Projec	Plaza	Safety	Resour	ion	c Trans	eering		ic Eng.	ities M	ts Mair	ation	ment Mg	Prod.	и И
•	AGENCY	RADIO SYSTEM	Polic	Fire	Ambul	Comm.	Civic	Bldg.	Human	Aviat	Publi	Engin	Parks	Traff	Facil	Stres	Sanit	Equip	Water	Sewer
	POLICE FIRE	POLICE FIRE	x x	x x	x x					x										
	AMBULANCE COMM. PROJECTS	FIRE/EMS RADIO SHOP	x	x x	х	х·												•		
	CIVIC PLAZA	RADIO SHOP	x	x			x													
	BUILDING/SAFETY	BLDG/SAFETY						x	•											
	HUMAN RESOURCES	HUMAN RESOURCES	x	x	x]		x	0		а. Р								
	AVIATION	AVIATION	x	x						x	· .	X				х				
	PUBLIC TRANSIT	CITY BUS	x	х	х			· .			х	ľ		·x		X			.,	
	ENGINEERING	LG #1								х		x				X	х	· · · · ·		
	PARKS/RECREATION	LG #1	x	x	1							х	х	X						
	TRAFFIC ENG.	LG #1	X											х		x				
	TRAFFIC ENG (Signals)	LG #2	X	x	а.								•	x		X				
	FACILITIES MAINT.	LG #2	x						÷.						x	X				
	STREETS MAINT.	STREETS #1	х	x	1								•	<u>e</u>		X				0
	SANITÀTION	STREETS #1	x			1	•	. .				x		x		x	х	x		
	EQUIP. MGT.	STREETS #1					1						х		х		•	x	x	x
	WATER PROD.	WATER #1		Į											- 				x	
	SEWERS	WATER #1											÷					•	1	X
	WATER DISTR.	WATER #2	·	x			ľ											-	x	X
	WATER CUST.SVC.	WATER #3						ľ							} •				x	x
	CITY ATTY.	BLDG.SAFETY									т. 1					1		.		· ·
	PLANNING DEPT.	BLDG.SAFETY			-								-							
	المرجوبية المسيخيين ومستحصين وبنبة المعينيوسة			3		المستنسدة		L	· · · · · · · · · · · · · · · · · · ·				ومعرجت ما		S	8. e		a 1	است ا	å

53

IV-81

(

County Sheriff DPS Public Utilities Civil Defense -Regional Police Regional Fire Airports Hosp Dist. Regional Regional Water Ma x x x x x X Cx x X x x x X XX X X X X X X х Х. X x x X xx

shown in Table II-14. The radio frequencies and channel designations are shown in Table II3 and Table II-5.

The City of Phoenix transit radio system is a unique system specifically developed for city bus operations. It is vendor engineered, vendor supplied, and vendor maintained.

The Phoenix aviation radio system includes three airports: Sky Harbor, Deer Valley, and Litchfield.

The Phoenix rescue/EMS radio system is operated by the City of Phoenix Fire Department from the fire radio dispatch center.

The municipal radio systems for the communities of Mesa, Scottsdale, Glendale, and Tempe are all dedicated radio channel systems. The summation of these system characteristics is shown in Table II-15.

(3) Shared Systems

The radio systems for the City of Phoenix that are shared by more than one user include streets, building safety, water, local government, human resources, and communications. These systems are generally characterized by having no central dispatch facility relying instead on a multiple of desk top type remote controls for access to their radio system. The characteristics of these radio frequencies, co-channel users, and general configurations are shown in Table II-3, Table II-5, and Table II-7.

The City of Phoenix Water radio system has three dedicated channels: Water #1, Water #2, and Water #3. Because of the different characteristics of these three channels, they are discussed separately in Table II-16.

The City of Phoenix Building Safety radio system is unique in that it also provides city-wide paging.

(4) Operating Costs

The radio equipment used by the City of Phoenix activities is controlled by the Equipment Management Division. Each City of Phoenix division and department is billed periodically for the operation and maintenance of the equipment each uses according to the computerized inventory report. An additional charge is also made periodically for the replacement of the radio equipment of each division and department.

Table II-17 shows the anticipated 1980 annual charges based upon the extension of billed 1979 charges (approximately 10 percent increase).

Q

ê

G

 \bigcirc

	$\sum_{\substack{m \in \mathbb{N} \\ m \in \mathbb{N}}} f_{m} \leq f_{m} < f_{m} < f_{m} < f_{m} < f_{m} < f_{m} < f_{m} < f_{m} $		TT 1/ 1		
	CITY OF P	TABLE HOENIX DEDICATED SYST	LI-14-1 EMS-OPERATING CHARACTER	ISTICS	
					. 6
RADIO SYSTEM	CITIZEN ACCESS	COMMAND AND CONTROL	DATA SYSTEM ACCESS	CHANNEL ACTIVITY	PLANNED GROWTH
POLICE RADIO	-Telephone; 7 digit emergency number- -25 Lines; 10 more in reserve -8 complaint takers -CAD input	-24 hour dispatch center at PPSB -Backup dispatch at North Mt. -Remotes on 450-1 only	-CAPRI -ACIC -NCIC -NLETS	-Monthly activity report available on all channels -All VHF channels heavily loaded -Peak load 4PM to	-2 new police districts by 1982 -MDT growth from 27 units to 505 units -New tactical UHF
		-CAD assisted		midnight.	channel needed
	-Telephone; 7 digit emergency number	-24 hour dispatch center at FAHQ	-No data system access at this,	-Not recorded; random 100 call samples	-New dispatch cen- ter with advanced
FIRE RADIO	dispatchers	and Rescue/EMS	time	-Peak load 4PM to midnight	ented PPSB -Experimenting with portable replacement for
	-No routine citizen	-16Hr/day; 6 days	-No data system	-Not recorded	mobile radios -New dispatch
	access	a week dispatch center at termi-	access	-Peak load reported at rush hours	center planned for Bank Bldg.
RADIO		nal -Bus radios have failure alarms		AM and PM	-New base station on North Mt. -CAD/AVL being
		and silent alarms			considered
$\mathscr{T}_{\mathcal{T}}$	-No routine citizen access	-24Hr dispatch center at Sky Harbor security	-Security data system only	-Not recorded -Reported over- loaded by fuel	-New base station on North Mt. -Separate VHF
RADIO		rols in Sky Harbor operations		trucks operations during daylight hours	trucks
		<pre>& maintenance -Dispatch Fuel Trucks,Airport Security, & Airport Fire/</pre>		random	-Add radio tech.
		Crash/Rescue			

		TA	BLE II-	-14-2		
CITY OF	PHOENIX	DEDICATED	SYSTEMS	-OPERATING	CHARACTERISTIC	28

			Net to the second se		· · · · · · · · · · · · · · · · · · ·	· · · · · · · · · · · · · · · · · · ·
	RADIO	CITIZEN	COMMAND AND	/ DATA SYSTEM	CHANNEL	PLANNED
_ [SYSTEM	ACCESS	CONTROL	ACCESS	ACTIVITY	GROWTH
- [-Telephone; 7 digit	-24 hr. dispatch	-No data system	-Not recorded; random	-New dispatch
1	RESCUE	emergency number	center at FAHQ	access at this	100 call sample	center with
1	EMS	-6 trunks direct	-Dispatch Fire &	time	taken each month	advanced CAD
	RADIO	to EMS dispatcher	Rescue/EMS		-Peak load 4PM to	being implement-
- 1					midnight	ed PPSB
						-Experimenting
						with vehicle
	and the second second					portables to
						replace mobile
1						radios

IV-84

0.

.0

-

D

		SUBURB DEDICATED SYSTEM	S-OPERATING CHARACTER	ISTICS	
			4 · · · · · · · · · · · · · · · · · · ·		
RADIO SYSTEM	CITIZEN ACCESS	COMMAND AND CONTROL	DATA SYSTEM ACCESS	CHANNEL ACTIVITY	PLANNED GROWTH
MESA Police/Fire	-Telephone; 7 digit emergency numbers -12 Lines police; 5 lines to fire -2 complaint takers per shift	-24 hr dispatch center; police/fire -No remotes -Dispatch Fire, Police, Rescue	-NCIC -ACIC -City Utility Files	-Not recorded -Sample taken each shift every 6 mo -Peak 5pm to llpm -65-75% EMS/police	-Plan CAD with MDT -New firegroun frequency -More Satelli receivers
MESA Public Works	-Telephone; 7 digit number -3 lines rotary -6 dispatchers/ complaint takers	-2 shift operation -Police dispatches on 3rd shift -57 Remote Controls	-City Utility Files	-Not recorded	-Plan for sat ellite recei -Plan for MDT
SCOTTSDALE Police	-Telephone-7digit emergency number -8 lines rotary -4 dispatchers/ complaint takers per shift	-24 hr dispatch Center-Civic Plaza -2 Remotes in PDHQ	-NCIC -CAPRI	-Total activity from counters -Peak load 4pm to midnight	-Plan to use Phoenix MDT -More satelli receivers
GLENDALE Police	 Telephone; 7 digit emergency number plus Centrex 5 emergency lines on rotary & Centrex 2 Complaint takers per shift 	-24 hr dispatch at PDHQ -1 Remote to EOC -Simulcast on General Service Channel	-NCIC -CAPRI -COM 10 -NLETS	-Not recorded -Peak load 4pm to 8pm and 10pm to 12 pm	-Plan to use Phoenix MDT -More satelli receivers
GLENDALE Fire	-Telephone-7 digit emergency number -4 emergency lines on rotary; 2 more in Alarm Room -1 dispatch/complaint taken per shift	-24 hr dispatch at Alarm HQ -1 Remote to EOC -Remote Control of Public Works base Station	-None	-Not routinely recorded -Peak load 6pm to llpm	-Plan integra with Phoenix Fire CAD/MDT -Plan separat fire ground channel

RADIO SYSTEM	CITIZEN ACCESS	COMMAND AND CONTROL	DATA SYSTEM ACCESS	CHANNEL ACTIVITY	PLANNED GROWTH
GLENDALE General Services	-Telephone; 7 digit No. -4 lines; not rotary -1 complaint taker and 2 back-up clerks per shift	-One shift; 7am to 5pm (5 days/wk) -Dispatch Center 6210 W. Myrtle -3 remotes W. Myrtle -1 remote, EOC -Fire dispatches after hours	-None	-Not recorded -Peak load 7am to 9am and 3:30pm to 5:30 pm	-None reported
TEMPE Police	-Telephone; 7 digit emergency number -5 lines on rotary -2 to 3 complaint takers per shift	-24 hr dispatch at PDHQ -Remote control of Public Works base for off-hours	-NCIC -ACIC -CAPRI	-Calls per shift recorded from counters -Peak load 4pm to midnight -Peak channel overload	-Plan for CAD in future -Need additional radio channel
TEMPE Fire	-Telephone; 7 digit emergency number -4 lines on rotary -1 or 2 dispatcher/ complaint takers per shift	-24 hr dispatch at Fire HQ -Control Stations can dispatch from 2 Fire Houses	-None	-Not recorded -Peak load 4pm to midnight -Fireground traffic overloads radio channel	-Plan integratio with Phoenix Fire CAD/MDT -Plan new channe for fireground
TEMPE Public Works	-Telephone; routine 7 digit numbers -Approx 20 lines to 7 remote control points	-No Central Dispatch -7 Remote Controls -Operates 7am to 4 pm -After hours dispatch by Tempe P.D.	-None	-Not Recorded -Peak load 7am to 9am and 3pm to 4:30pm	-Plan satellite receivers

TABLE II-15-2 SUBURB DEDICATED SYSTEMS-OPERATING CHARACTERISTICS

\$

1. S. C. C.

I.

							1	
		TABLE SUBURB DEDICATED SYSTEMS	II-15-3 S-OPERATING CHARACTED	RISTICS				1000 - 1000 1000 - 1000 10000 10000 - 1000 1000 - 1000 1000 - 1000 1000 - 1000 1000 - 1000
RADIO SYSTEM	CITIZEN ACCESS	COMMAND AND CONTROL	DATA SYSTEM ACCESS	CHANNEL ACTIVITY	PLANNED GROWTH			ð Í
TEMPE Sanitation/ Refuse	-Telephone; routine 7 digit number -1 dispatcher/ complaint taker per shift	-No Central Dispatch -One shift operation 6am to 4pm 5 days/wk	-None	-Not recorded -Peak load 6am to 8:30am and 3pm to 4:30pm	-None reported			
			•					
IV-87								
							12 m ⁶	
		and a second	ار میدهای در می معنی می ایند این میزود. اینمایی والی ا		and a second s	en e		

			TA	BLE	II-16-1	
CITY	OF	PHOENIX	SHARED	SYST	EMS-OPERATING	CHARACTERISTICS

£

				· · · · · · · · · · · · · · · · · · ·		
	RADIO	CITIZEN	COMMAND AND	DATA SYSTEM	CHANNEL	PLANNED
	SYSTEM	ACCESS	CONTROL	ACCESS	ACTIVITY	GROWTH
	STREETS #1	-Streets Maint. has	-No central dispatch	-No data system	-Not recorded	-No growth plan
	STREETS #2	11 lines for access	-Distribution of	access	-Channel overload	reported
	RADIO	-Sanitation has 10	local remote		at peak hours	
	(2 channels)	lines for access	controls shown in		6AM to 10AM	
		-Equip.Maint. has	Table II-7		-Least daytime	
		no routine citizen	-All operation from		load 2PM to 6PM	
		access	5AM to 6PM		- -	
		-Bldg.Insp. has	-No central dispatch	-Utility company	-Not recorded	-No growth plan
	BUILDING	6 tele. lines &	-4 local remote	(SAP/APS) tele-	-Channel reported	reported
	SAFETY	Code-A-Phone for	controls used 8AM	type system	saturateá 8AM to	
-	RADIO	24 hr coverage	to 5PM		10:30AM by Bldg.	
		-Bldg.Maint. has	-Bldg.Maint. is		Insp.	
		no routine	24 hr operation		-Approx. 125 paging	
		citizen access			receivers on	
VI VI					channel	
ľ	· · ·	-Water Prod. has no	-No central dispatch	-No data system	-Not recorded	-No growth plan
ő	WATER #1	routine citizen	-Distribution of	access	-Channel load	reported
	RADIO	access	local remote cont-		reported to be	
		-Sewer has 2 lines	rols shown in		moderate	
		rotary 24 hrs per	Table II.		-Peak load 8AM to	
		day	-Water Prod.operates		9:30 AM	
			24 hrs; sewers			
		Matan Diat 1as	8 hrs	No. John maker	Nationa and a d	No sussifik wilson
	113 mm #0	-water Dist. has	-No central dispatch	-No data system	-Not recorded	-No growth plan
	WATER #2	10 tere. lines on	-4 remote control	access	-Channel reported	reported
-	KADTO	10 and an amountain	units operate 24		Jord	
-		TO are all emergen-	IILS Drowide 24 br		LOAU Dook lood RAM to	
i		Cy number /	-FLOVIDE 24 IIL		-reak toau oni to	
			41 and WATER		TOUL	
			TI ANU WAIDE TO			

A

 $\langle \cdot \rangle$

TABLE II-16-2 CITY OF PHOENIX SHARED SYSTEMS-OPERATING CHARACTERISTICS

	RADIO	CITIZEN	COMMAND AND	DATA SYSTEM	CHANNEL	PLANNED
.		-10 lines for quet-	-No contral dignatch	-No data system	-Not recorded	-No growth plan
	113mmn #3	-it ines for cust-	-No central disputen	-No daca system		-NO GLOWCH Plan
1	WATER #3	omers, 3 lines for	-4 remote control	access	-Channel reported	reportea
{	RADIO	field admin; 13	stations		saturated at peak	
- {		complaint takers	-8hr service is		load hours	
· {		-Operate 8AM to 5PM	normal; backed by		-Peak load 8:30AM	
		•	Water Dist. 24		to 10AM; 1:30PM	
			hrs		to 3PM	
I		-No routine citizen	-No central dispatch	-No data system	-Not recorded	-No growth plan
	LOCAL	access	-Numerous remote	access	-Channel reported	reported
i	GOVT. #1	· · · ·	control stations		saturated at peak	
	RADIO		shown in Table II-7		load hours	
			-Normally operated		-Peak load 7AM to	
			6:30AM to 5:30PM		8:30AM; 1PM to	
1					2:30PM	
ы 1		-No routine citizen	-No central dispatch	-Facilities Maint.	-Not recorded	-No growth plan
$\langle \rangle$	LOCAL	access	-3 remote control	accesses ASO	-Channel load	reported
8	GOVT. #2		stations operate	Energy Conserva-	reported as moder-	
ω I	RADIO		8 hrs a day	tion Office	ate	
				Computer System	-Peak load at AM	
	•				and PM rush hours	
		-8 tele lines and	-No central dispatch	-No data system	-Not recorded	-No growth plan
	HIMAN	3 "hot lines" for	-2 remote control	access	-Channel load	reported
	DESOURCES	direct citizen	stations	decess	reported as light	reported
	NBOOM DD	inout	-Suston configured		and random	
	MILLO	-Operator 21M to	for mobilo-to-		and random	and the second
- 1		Spw	TOT MODILE-CO-			
	•	JPM .				
			repeaters (no			
			control stations)	No. 2010		<u></u>
		-No routine citizen	-No central dispatch	-NO data system	-Not recorded	-Civic Plaza 25
	COMMUNICATIONS	access	-4 remote control	access	-Channel load	portable system
	RADIO		units; 3 control	a ser a start a ser a	reported light	is planned for
			stations		-Load randomly	this channel
			-Repeater configured		distributed	
			for mobile-to-			
			mobile service 24			
			hrs			
				L	أفوي ومحمد ومنابع ومحمد ومحمد والمنافق والمحمد	

\Box	\square		

E

Municipal Activity
Planning
Engineering
Facilities Mgt.
Equipment Mgt.
Equipment Mgt.
Police
Fire
Bldg. Safety
Streets
Traffic Eng.
Airport
Sanitation
Sewers
Parks/Recreation
Civic Plaza
Water-Cust.Svc.
Water Production
Water Distribution
Human Resources
JSP

 \mathbb{C}

~j).

0

00

 Ω

E) "

TABLE II-17

1980 Radio Rental Rates and Estimated Operation/Maintenance Charges (Projected from 1979 Rates)

 \square

 \square

[]

1

63

F

 \square

CITY OF PHOENIX

Account	Operation and Maintenance	Replacement
555XX	\$ 924.00	\$ 884.00
56XXX	7,194.00	8,606.00
572XX	5,702.00	7,036.00
5743X	647.00	607.00
5744X	2,112.00	2,059.00
60XXX	438,293.00	385,836.00
615XX	123,697.00	113,084.00
62XXX	4,554.00	4,554.00
65XXX	6,138.00	6,165.00
66XXX	2,257.00	2,930.00
67XXX	15,748.00	12,936.00
70xxx	5,927.00	5,887.00
71xxx	4,356.00	4,356.00
73XXX	5,504.00	6,798.00
75xxx	990.00	1,333.00
794XX	5,478.00	5,478.00
795XX	5,478.00	5,439.00
796XX	5,478.00	5,320.00.
90XXX	2,191.00	1,518.00
92XXX	1,637.00	1,505.00
4 	\$644,305.00	\$582,331.00

ý 62°

0

TABLE II-18 SUMMATION OF REPORTED COMMUNICATIONS PROBLEMS-CITY OF PHOENIX

RA	DIO	С	OVERAGE PROBLEM	CHAN	NEL OVERLOAD	MORE	CHANNELS	IN	TERFERENCE
SYS	TEM	YES	WHERE/WHY	YES	WHEN	YES	WHY	YES	WHAT
POLICE		х	WEST	x	4PM-Midnite	x X X	2 new Police Districts New MDT (500) Tactical Oper- ation		
FIRE		X	PORTABLES	X	4PM-Midnite	X	Fireground	X	Hospital IM
RESCUE/EMS		x	PORTABLES						
CIVIC PLAZA		x	INSIDE COMPLEX						
BLDG. SAFETY		X	NE					x	Paging Opera- tor
v-91								X X	Police IM-SM EMS-North Side
AVIATION						X	Separate Fuel Truck	x	Fuel Trucks
CITY BUS		x	N/NE	x	Rush Hours	x	Bus Expansion		
LOCAL GOVT #	1	x	NE/N	X	8AM-3PM			x	Telco IM
LOCAL GOVT #	2	X	NE						
STREETS #1		X	(See Notes)	x	6AM-10AM				
WATER #1		X	NE				стана (стана) (Колдону (стана) (X	IM
WATER #2	· · · · · · · · · · · · · · · · · · ·	X	NE	X	8AM-10AM			1	
WATER #3		X	NE/N	x	8:30-10 AM 1:30-3PM				

т,		

The maintenance costs for the City of Phoenix are influenced by the following:

Police Department - the police department performans in-house maintenance on communications equipment contained within the PPSB headquarters including Computer Assisted Dispatch (CAD) terminals and Mobile Data Terminal (MDT) equipment. The cost is not reported in the survey data.

Fire Department - the fire department maintenance shop performs much of its own radio maintenance; cost is reported to be \$252,000 per year for fire maintenance, and \$125,000 per year for EMS maintenance.

Aviation - the aviation department performs all of its own radio maintenance in its own radio/electical shop. The cost is not reported in the survey data.

<u>City Bus</u> - the city bus radio system is a vendorinstalled and vendor-supported system. The maintenance cost was reported as \$28,568 for 1978; more recent costs were not reported.

The cost of maintaining and replacing radio equipment has been controlled through centralized maintenance and management. The present trend in the City of Phoenix is toward decentralized maintenance.

The City of Mesa and the City of Tempe operate their own municipal radio shops. The total costs reported for radio maintenance in 1979 are:

Mesa - \$15,000

Tempe - \$70,000

Scottsdale Police and the City of Glendale contract for radio maintenance. Their reported costs for contract radio maintenance in 1979 are:

Scottsdale Police - \$18,000

Glendale - \$9,000

These are estimates of maintenance costs and are not otherwise confirmed.

3. REPORTED PROBLEM AREAS

The summation of problem areas that were reported for each City of Phoenix municipal radio system is shown in Table II-18. Appropriate notes related to these problem areas are as follows:

IV-92

Police - Additional MDT units are expected to reduce some of the channel loading problems

Fire - Installation of the new satellite voting receivers will help solve the portable radio coverage problem. Some of the channel overload problem will be reduced by implementation of the new CAD system

<u>Rescue/EMS</u> - Installation of the new satellite voting receivers will help solve the portable radio coverage problem

<u>Civic Plaza</u> - Portable radios are planned to replace unsatisfactory coverage by paging receivers now used inside the Phoenix Civic Plaza complex. The portable radios will operate in the otherwise lightly loaded communication UHF repeater channel

Building Safety - The city paging operation causes interference because the paging operator does not monitor the channel before keying. Police calls are heard from South Mountain; ambulance calls are heard from North Mountain. Building Safety reports the cost of the present radio system is too high for the service it offers

Aviation - Fuel truck operations at Sky Harbor are very active and break up airport operations traffic. Airport operations at Sky Harbor should be separated from airport operations at Deer Valley and Litchfield

<u>City Bus</u> - A new base repeater is planned for North Mountain to correct coverage problems; a new dispatch center is planned to correct security problems

Local Government #1 - Mobile telephone traffic is heard on the channel in the morning and again in the evening - may want to vote all fixed station receivers to improve the coverage problem

<u>Streets #1</u> - A report by Sanitation that some mobiles have trouble receiving North Mountain and South Mountain is not supported by other co-channel users; an equipment problem is implied. A successful test of Streets #2 channel could help reduce the channel load on Streets #1

Water #2 - The separation of mobile-to-mobile radio traffic from base-to-mobile radio traffic has been suggested as a means to reduce channel loading

<u>Water #3</u> - The coverage problem may be improved by voting the existing six fixed station receivers. General comments regarding the reported system problems include the following:

Lack of radio discipline and failure to use efficient radio procedures have been cited as major causes of excessive channel air-time

Selective voting of multiple fixed station receivers should be investigated as a means of improving reported coverage problems

The communications problem reported by the participating suburban Phoenix agencies are summarized in the following paragraphs:

> <u>Mesa Police/Fire</u> - A new fireground frequency is needed to relieve the congestion on the main fire frequency; there are portable radio talk-back coverage problems that are expected to be corrected by additional satellite receiver installations

Mesa Public Works - Radio coverage will be needed southeast of Mesa toward Florence; there is interference from Florence Prison Farm mobiles operating near Mesa

Scottsdale Police - There is interference from unidentified co-channel user; intend to change to CTCSS to see if this will help

<u>Glendale Police</u> - There is co-channel interference from the University of Arizona; portable talk-back coverage problems are expected to be solved by installing additional satellite receivers

<u>Glendale Fire</u> - Intermode problems with mobile telephones and co-channel user interference with Riverside California indicate the need for a separate fireground channel to separate fireground and operations traffic

<u>Glendale General Services</u> - There are coverage problems on the north side of Glendale due to co-channel user interference with City of Peoria

Tempe Police - Co-channel user interference is evident; channel load at peak traffic hours indicates the need for another police radio channel

Tempe Fire - Additional satellite receivers will be needed to correct portable talk-back coverage south of Tempe; a fireground channel is needed to separate fireground and operations traffic

Tempe Public Works - Portable radio talk-back coverage problems indicate the need for satellite voting receivers

IV-94

Ð

Tempe Sanitation/Refuse - Portable radio talk-back coverage problems indicate the need for satellite voting receivers.

The majority of reported radio system problems for the City of Phoenix and the four participating suburban communities appear to be generic to radio system operations at any frequency and do not seem unique to VHF, UHF, or potential 800 MHz operations.

APPENDIX D

STATEMENT OF WORK

FORWARD

64

This appendix is intended as a typical outline for a Statement of Work (SOW) for quick reference by planners and system procurement personnel. The planner is cautioned to employ only those elements of the outline which relate to the specific procurement and which can be accurately evaluated and measured.



Preceding page blank

 \square

I.

INTRODUCTION

1.

2.

3.

1

ŀ

II.

1.

 \sim

APPENDIX D

STATEMENT OF WORK

The introduction to the Statement of Work (SOW) should provide a general overview of the project to enable a better understanding by the responders. Key elements of this section are outlined in the following subsections of the Introduction.

Background of the Project

A brief historic description of the origin of the procurement and the constraints imposed on the design.

Objective of the Procurement

A brief discussion of the main system objectives to be achieved by the procurement.

Scope of the Procurement

A basic and brief outline of the hardware, software, and services that are being purchased. These may include for example:

Hardware

Software

System Design

Installation

Testing

Training

Maintenance.

INSTRUCTIONS TO BIDDERS

This section of the SOW will provide the bidders with all the information needed to respond correctly to the bid request.

Procurement Schedule

 \bigcirc

The schedule time and dates for responding to the bid request will include: $~^\circ$

Time and place of the pre-bid conference

Time and place bids and proposals are due

Bid and proposal opening time, date, and place

Proposed contract award date

Expected date of contract completion.

2. Authorized Contracts

The persons will be identified who are authorized to respond to bidders requests prior to bid opening. These may be restricted to the:

Procuring officer

Technical consultant.

The statement should be made regarding how the issues raised by these contacts will be reported to the other bidders.

Bid and Proposal Requirements 3.

Each of the items to be submitted as part of a responsive bid will be defined. Any limitations should be included in the description. These items would include for example:

Itemized cost proposal

Technical proposal

Management proposal

Provisions for alternate proposals

Proposal limitation such as page count, standardized forms (cost proposal), proposal size, unnecessary artwork, and number of copies to be submitted.

4. Bidders Qualifications

The contractor for a high technology system must be qualified, technically and financially, to complete the required work. These qualifications should be expressed in the SOW by requiring reasonable responses to limits imposed on:

Bidders facilities and financial resources

Performance history in terms of references and a list of similar and recent contract work

Support facilites such as field maintenance shops, training facilities, and spare parts replacement depots

The quality and availability of personnel for management, supervision, and technical staff.

IV-100

	ation	of:	
f_{i}^{i} .		•	Tec
		•	Cos
		•	Mai
	each ments	These bidde of t	e ci er v :he
	6.	Conti	cact
	succe withc	The Sessful out pr	SOW L b: ce-a
	7.	Inder	nni
	impos tract ficat All F test resul	The p ed th or. ion a procum will ting	buro Hio aga: ceme be fro
	8.	<u>Title</u>	es a
	clear relat	Right ly de ed to	cs d efin o:
		•	The
		•	Rig

5.

•

Proposal Evaluation Criteria

Criteria should be established for the comparative evalu-

chnical proposals

st proposals

nagement proposals.

riteria should be clearly defined in the SOW so that will understand the emphasis placed on the require-RFP.

t Award

will identify the type of contract to be awarded the idder whether or not this contract will be awarded contract discussions.

fication

chaser should require indemnification against damages ugh intentional or unintentional acts of the conah technology system procurements will need indemniinst the improper use of patented items or processes. ents which include services such as installation and well advised to require indemnification from damages om the actions of contractors' employees or agents.

and Rights

of the purchaser and ownership of titles will be ned by the SOW. These items include statements

e transfer of title to equipment and software

ghts to data and software developed or provided as part of the contract

Rights and title to the submitted proposals

The right to reject proposals

The right to negotiate with any bidder

The right to cancel negotiations or the contract.

Legal counsel should be consulted for the scope, content, and wording of this subsection of the SOW.

9. Bonds, Insurance, and Warranty

Each governmental entity will have its own limits for, and description of, the bonds, insurance and warranty provisions for contractor provided material and services. Frequently, there will be state regulations which also must be adhered to. These requirements will provide for:

Performance bonds

Workman's compensation

Comprehensive liability insurance

Warranty on workmanship and materials.

The purchasing department of the major governmental agency included in the program will provide counsel as to how these requirements are to be included in the SOW.

10. Costs of Proposal Preparation

The SOW should state clearly that any costs related to the preparation of bids or proposals can not be included in the proposed cost of the project or billed back to the purchaser in any manner.

11. Contract Disputes

At the option of legal counsel, the SOW may indicate the means by which the governmental agency resolves contract disputes. This will be a function of the governmental body and will not be an option of project management. These provisions may include:

The procedure and method of resolving contract disputes

Termination of contracts for cause

The assessment of liquidating damages

The identity of the court of jurisdiction.

12. Applicable Federal/State Laws and Regulations

When state or federal laws/regulations affect the system design, its operation, or allowable use, these laws or regulations should be identified in the SOW.

III. SCOPE OF WORK

This section of the SOW defines the items of work expected of the contractor; the specific tasks required by the contractor are described in detail.

IV-102

Services to be Provided

1.

2.

3.

following:

High technology systems will require professional services to be performed by the system contractor in addition to the delivery of hardware and software items. Typical services to be

> System design which can include system engineering, radio propagation engineering, software development, and hardware development

Installation and debugging of hardware and software

Assistance in procuring required licenses and permits

Developing and conducting performance tests and acceptance tests

The preparation of sites and facilities which can include construction, HVAC, emergency power, lighting, installation of utility services, etc.

Deliverable Items

Each deliverable item should be identified in the SOW. Deliverable items include hardware, software, and services. The list will be unique to the complexity of the system to be procured. However, some common items on all lists will include the

Each item of electronic or radio hardware

Each item of computer software

Maintenance support

Spare parts and spare module provisions

Towers, buildings and other structures

Manuals and documentation such as test plans, test procedures, maintenance manuals, operators' manuals, user manuals, training aids, installation drawings, and test reports

Progress reports and deficiency reports.

Delivery Schedules

Each deliverable item should be supported by a delivery schedule. The SOW will require a formal and contractually obligated schedule for:

Each item of electronic or radio hardware

Each item of computer software

Each tower or building

The completion of all facility preparations

Delivery of all manuals and documents

The time and place of technical progress reviews

The delivery of reports including progress reports, test reports, discrepancy reports, and the final report or acceptance test report.

IV. SYSTEM SPECIFICATION

The system specification is a complete technical system description which includes the system performance, system interfaces, and system configuration. It places technical limits on these parameters and adds reliability and/or maintainability requirements where appropriate.

1. Hierarchy of Specification and Standards

Industry or regulatory specifications or standards frequently apply to the specification of a system. These should be identified. The hierarchy of these documents, as they apply to the system specification, should be defined so that conflicts between them can be resolved.

2. System Description

The system description includes the configuration of the system elements and their interconnection with each other. It is recommended that this be done graphically with supporting written clarification. The system description is derived from the system concept design.

System Interfaces and Supporting Services 3.

The means by which the system interconnects with the world outside of the system should be fully described. These interfaces are relatively unique to each individual system, but some key interfaces will include:

> External communication systems such as mobile radios, private telephone systems, data processing, digital data systems, alarms, microwave transmission systems, and similar facilities

> > IV-104

4.

Frequently, limitations on funding or other resources will prevent the initial implementation of a sophistigated system from realizing the full potential of system operation, scope, and performance. When these expansion requirements are part of the initial system procurement, they must be fully described in the system specification.

7.

High technology systems may exceed the apparent limits of federal or state licensing regulations, and extraordinary procedures may be needed to acquire the necessary permits or licenses. These requirements should be clearly defined in the system specification. They will be unique to each specific system configuration and intended operation.

Utility systems such as public access telephone systems, commercial power systems, water supplies

Emergency operations such as emergency power services, fire protection, emergency medical services, and civil disaster prevention operations.

System Software

The system specification will include a complete description of the operations software and the applications software required to provide the needed system performance.

System Operation and Performance

The specification of system operation and performance is tailored to meet individual system requirements as developed in the system concept. Typically, these specifications will include

> The functional modes of system operation for routine system performance, emergency system performance, and allowable fail-soft or degraded levels of performance

The calculated levels of personnel staffing for the operating staff, maintenance staff, and system manage-

System Licenses and Approvals

V. EQUIPMENT SPECIFICATIONS

The equipment specifications define the quality and performance characteristics of each item of hardware and software needed to meet the system specifications. The identity of each hardware and software item is derived from the system configuration diagram which was developed as part of the system concept.

1. General

.

This subsection of the equipment specifications accumulates all items of the individual equipment specifications requirements that are common among all items of equipment or software. It also defines the quality and quantity of each item of hardware and software that is required to meet the system configuration. Typical items that would appear under the general heading are:

Applicable industry standards or publications and their hierarchy for this program

Processos -

T

Standard environmental conditions such as temperature, humidity, primary power, vibration, shock, dust, and dirt

Deliverable equipment lists showing the quantity required of each item including spares

The required level of quality and workmanship for each hardware item.

2. Individual Item Specifications

Each equipment item will have an individual technical specification that reflects its construction and performance as required to meet the system specifications. These are derived individually from the system performance requirements and should be no more severe than that needed to meet the system performance specifications. Software specifications are typically combined with the equipment specifications for the computer subsystem or the switching subsystem in which they are resident.

V. SALT LAKE COUNTY, UTAH

System Implementation Plan - Draft



Lt. Rex L. Vance Project Director Salt Lake County Sheriff's Office 6278 Glen Oaks Salt Lake City, Utah 84107

DIGITALLY ADDRESSED TRUNKED COMMUNICATION SYSTEM

SYSTEM IMPLEMENTATION PLAN

Ľ

1

SALT LAKE COUNTY, UTAH

November 15, 1980

BOOZ · ALLEN & HAMILTON Inc.

776 SHREWSBURY AVENUE TINTON FALLS, NEW JERSEY 07724 747-9303 AREA CODE 201

	9		
I.	INTI	RODI	JCT
	1. 2.	Bac Pro	ckg oje
II.	SYST	EM	CO
	1. 2. 3. 4. 5. 6. 7. 8.	Pai Bas Fre Ado Mol Con Ope Bas	rti se equ lre oil ntr ntr era
	9. 10. 11. 12. 13. 14. 15. 16. 17.	Mol Poi Pha Dec Int Fac Pei Enc	oil ter gra ter cil rso gin
III.	REQU.	[REI	DI
	1. 2. 3. 4. 5. 6. 7. 8. 9.	Loc Pre Imp Sys Pro Lic Fac Pei Pos	cal ole ste ocu cen cil rso st
APPEN	DICES	: 2 1 (1	A - 3 - 2 - 0 -

TABLE OF CONTENTS

مىرىمىيىتىنى ئېرىپارىيە تەرە مەرمەر ئارىيەلەن بوچىنا تەرە

	Page
DUCTION	V-1
ckground and Objectives oject Direction and Participants	V-1 V-2
CONFIGURATION	V-5
articipating Agencies se Station Configuration requency Plan dress Codes obile/Control Stations ontrol Points ontrol Station/Mobile Station	V-5 V-10 V-10 V-15 V-16 V-17
peration use Station Operation obile-to-Mobile Operation ortable Radio Interface steragency Communication antom Groups ograded Operation stersystem Considerations acilities ersonnel ogineering Cost Estimate	V-19 V-20 V-20 V-20 V-21 V-21 V-21 V-22 V-23 V-24
D IMPLEMENTATION ACTIVITIES	V-33
ocal Government Approvals eliminary Licensing Considerations oplementation Schedule stem Management and Operation cocurement censing cilities Acquisition and Preparation ersonnel and Training ost Implementation Support	V-33 V-35 V-35 V-37 V-38 V-39 V-40 V-41 V-41 V-41
A - SUGGESTED AGENCY AGREEMENTS B - SYSTEM REQUIREMENTS C - CURRENT SYSTEM INVENTORY D - STATEMENT OF WORK	V-43 V-51 V-63 V-87

1. a.
| | This system in
trunked communication |
|---|---|
| | Associated Public-S
response to LEAA Gra |
| $egin{aligned} & & & & & & & & & & & & & & & & & & &$ | 1. BACKGROUND AND
The Second Rep |
| | Docket No. 18262, r
mission (FCC) in 19
difficult problems
community. These p |
| | MHZ portion of the
1977, LEAA recognize
ment questions about
spectrum to publice |
| | potential benefits
available. Under 1
Public-Safety Commu |
| | future actions, an
potential capabiliti |
| | The first of these
technical and regul
the 800 MHZ portic
communication system |
| | <pre>Identified specific
requirements that is
trunked communicati
provided specific p
KY and Phoenix, AZ)
County, UT), and th
and basic implement</pre> |
| | Since November
in Project 16B as or
objectives to be ach |
| | . To determ
channel co
area |
| | . To establ
future pub |
| | To address problems of |
| | |
| | |

I. INTRODUCTION

nplementation plan for a digitally addressed on system (DATCS) for the Salt Lake County, pared by Booz, Allen and Hamilton and the Safety Communication's Officers (APCO) in ant No. 79-SS-AX-0013.

OBJECTIVES

bort and Order in the proceedings related to published by the Federal Communication's Com-974, promised relief of the increasingly more of frequency congestion in the public-safety proceedings allocated 200 channels in the 800 spectrum for "trunked" systems. In February ed that many technical, economic, and manageut the application of this newly available safety operations must be answered before the inherent in trunked systems could be made LEAA Grant No. 77-SS-99-6009 the Associated unications Officers (APCO) was requested to olems, make appropriate recommendations for nd describe a program to demonstrate the ies of these new concepts.

rograms under APCO Project 16 were initiated. (Project 16) resulted in an analysis of the latory factors affecting the applicability of on of the radio spectrum to law-enforcement em problems. A second program (Project 16A) c operational capabilities and functional should be incorporated into a public-safety on system. The third program (Project 16B) clanning assistance to two cities (Lexington,), two counties (Bucks Co., PA and Salt Lake he State of Oklahoma for a feasibility study tation plan for DATCS demonstration systems.

1979 Salt Lake County has been participating ne of the demonstration participants. The key hieved are:

mine if DATCS can relieve municipal radio ongestion in the Salt Lake County/ Utah County

Lish the potential of DATCS to accommodate blic safety radio communications needs

s the solution of the complex command/ control confronting the public-safety agencies

To develop a common DATCS to support multi-agency public safety and/or other governmental communications requirements within the Salt Lake County, Utah County area

to provide a solution to interagency radio connectivity between governmental agencies appropriate to their mission responsibilities.

Other advantages of operational flexibility are inherent in a dynamically programmed digitally addressed communication system. These have been evaluated as DATCS capabilities in determining the system requirement.

2. PROJECT DIRECTION AND PARTICIPANTS

The project was under the direction of:

Lt. Rex L. Vance, Project Director Salt Lake County Sheriff's Department

Craig M. Jorgenson, Deputy Project Director State of Utah Department of Transportation and Public Safety

The participating public-safety agencies include the following:

Utah Highway Patrol, Rex Hale,/Col. Robert Reid

Salt Lake County Sheriff's Office, Lt. Rex L. Vance Utah County Sheriff's Control Dispatch, Sheriff Mack Holley

Pleasant Grove Police Department, Chief Mike Ferre West Jordan Polic Department, Chief Robert Stockwell

University of Utah, Public-Safety Division, Mr. Wayne Dee Shephard

Brigham Young University, Security/Police, Mr. J. Wesley Sherwood

Salt Lake County Fire Department, Chief David A. Barrett

Salt Lake City Fire Department, Mr. Harold C. Newman

Salt Lake County Attorney, Mr. Ted Cannon

v-2

Salt Lai
Salt Lak Ray Emer
Salt Lak
Universit
Each of the agencies have been ing to the interaction of the assistance of Booz
The cooperation of the seen voluntary

 \square

Contraction of the local division of the loc

1

Other participating governmental and public service departments and services include:

Utah Department of Transportation, Mr. Graig M. Jorgennsen

Salt Lake County Fleet Management, Mr. Frank L. Lear Salt Lake County Emergency Services Department, Mr. J. Ray Emery

Salt Lake County Public Works, Mr. C. W. Braoly

Salt Lake County Surveyor, Mr. Douglas H. Brammer

Salt Lake County Division For The Aging, Mr. Monte K.

Salt Lake County Building Inspector, Mr. Rex Bronson Salt Lake County Security, Mr. W. O. Cowden

University of Utah Media Services, Mr. Milton L. Davis

Each of the above listed public-safety and public service agencies have been active participants in the program, contributing to the interagency and intra agency communications analysis and also the requirements for the DATCS concept design.

It has been the responsibility of the Project Director to develop this System Implementation Plan (SIP) with the technical assistance of Booz, Allen as engineering consultants to APCO. The cooperation of all participating agoncies and their personnel has been voluntary. There have been no binding commitments required of Salt Lake County, the State of Utah or any of the other participating agencies.

J



Can J communications systems. 1. 2. 1È Snowbird (Hidden Peak). Preceding page blank

SYSTEM CONFIGURATION II.

The basic DATCS land mobile radio system for the Salt Lake County area has been conceived to replicate current radio system operations as closely as possible. The difference in function and response between the conventional radio system configuration now in use and the DATCS radio system should be nearly transparent to the mobile operator except co-channel sharing will appear to the operator to have been eliminated.

This philosophy directs that operations which require a unique trained skill on the part of the mobile operator be mini-mized in the system concept. It also directs the avoidance of subsystem interfaces in which trunked technology or switching requirements might result in a degraded performance of existing

The configuration of the Salt Lake County/Utah County DATCS is based in part on the system requirements and in part on information provided by existing suppliers of trunked radio systems. It is noted that the equipments described as part of this DATCS radio configuration are not standard production items and have not been previously provided by the vendors in this definitized form. Further study by the vendor community may modify their final hardware and software offerings. This could result in changes to the system configuration.

PARTICIPATING AGENCIES

Candidate agencies for participation in DATCS include those agencies who participated in the Task 4 survey activities and provided data for the Task 4 report.* These agencies are identi-fied in Section I, Subsection 2. From Task 4 report* it is estimated that these participants can load the DATCS with 1082 mobiles. This is adequate loading for a 15 channel mixed service group under the present FCC Rules and Regulations**.

BASE STATION CONFIGURATION

The proposed DATCS base station configuration for Salt Lake County/Utah County is shown in Figure II-1. The selection of base station locations at Lake Mountain, Mt. Vision, and Snowbird (Hidden Peak) is driven by radio propagation experience at these locations, and previous 800 MHz propagation tests.

The sites at Lake Mountain and Mt. Vision transmit and receive control channel information. The details of a conceptual installation at these sites is shown in Figure II-2. The site at Snowbird (Hidden Peak) receives control channel information but does not transmit it. The details pff a conceptual installation at this site is shown in Figure II-3. Radio propagation tests at 800 MHz may indicate a reversal in the roles of Mr. Vision and





[



The site controller at each site is fully programmable and capable of operating independently based on control channel instructions. The two control channels transmit the same information simultaneously on different frequencies. Mobiles will home on one control channel until the signal is lost then seek and lock on the other control channel. The site controllers are connected through modems and an RS-232 interface to the master controller and master control terminal at DATCS control operation center. This terminal can make program changes in the site controllers, selectively turn off or turn on individual channels, key up individual transmitters, change the address code responses, and perform interagency coordination interface switching as required. T Alarm functions generated by the site controller are con-22.2 nected through modems and an RS-232 interface to the alarm board in DATCS control operation center. Receiver voting and transmitter steering information from each DATCS voice channel at each site are brought to the voter selector equipment at the DATCS control operation center. The control channels are not voted; voting is suspended from the back-up control channels if they are selected by the site controller. Audio signals into and out of the control channel receiver and transmitter at each site are interconnected at the control channel audio unit in DATCS control operation center. Any bridg-ing, amplification, attenuation, or balancing required is done at this unit. The DATCS control operation center coordinates and controls the operation of the three base station sites and provides an interface for the restricted control terminals located at user agency locations. The master controller provides logic and switching for the multiple drop circuit to the agency terminals, and routes management information generated by site controllers to the journal recorder. This information is listed in Appendix B, Subsection 10. The master controller is commanded by the master control terminal which can address the programming and coding of the master controller and the site controllers. The journal recorder is a magnetic tape transport that records all management information generated by site controllers, alarm board, and master control terminal. It prints out selective reports on its printer on command of the master controller. It is intended that most management reports, except for spot reports, will be generated off-line from the journal recorder tapes. v-9 1

3. FREQUENCY PLAN

The frequency plan for a 20 channel area DATCS for Salt Lake County/Utah County is shown in Figure II-4. A count of public service and public safety mobile units, as reported by the participants in Task 4, are more than enough for 70 percent loading of a 15 channel mixed service group trunked radio system under present FCC Rules and Regulations (Part 90, Section 90.375).

Overhead burden of two control channels is approximately 13 percent of a 15 channel system. The distribution of channels for Lake Mountain, Mt. Vision, and Snowbird (Hidden Peak) is shown in Figure II-4. Channel 3 is the back-up control channel for Lake Mountain and Channel 4 is the back-up control channel for Mt. Vision. If a control channel failure (or deactivation) will require the use of channels 3 or 4 as control channels, the transmitters and receivers for the back-up channel selected will be exempt from voting and transmitter steering. Engineering analysis of system reliability based on known hardware and software components may reduce the need for back-up Control channels.

A suggested frequency plan for the statewide base station locations, talk-back station locations and intermediate repeater locations, listed in Appendix B, is also shown in Figure II-4. It is noted that the control channel frequencies are reused throughout the state so that no more overhead channels are charged against a statewide system. Operations (voice) frequencies are spread beyond the first six channels in the Salt Lake County/Utah County system to minimize possible interference on the most used channels in the State Capitol area. A suggested base station site configuration is shown in Figure II-5; the talk-back station configuration is shown in Figure II-6. A suggested intermediate repeater site configuration is shown in Figure II-7.

4. ADDRESS CODES

The digital address of all mobile units and control station units has two parts; an individual unique address and a group address. The combination of these two address codes informs the site controllers and the master controller which control station and mobiles to include in the request for a channel. Under normal mode operation all mobiles assigned to a control station (group dispatch point) will be included in any group call. Mobile-to-mobile communication within the group is achieved as in any repeater type conventional radio system. If temporary modification is required such as interagency coordination or intergroup mobile-to-mobile communication, the individual address portion of the address code enables unique routing by terminal instruction to the controllers. Similarly an individual mobile or control station can be denied access to DATCS.

		المتحدية من المحدية ال المحديث المحدية						L						<u>1</u>										
	-				CHAN	NEL	ASSI	GNME	NT-S	TATE	VIDE	(C=Co	ntrol	Chan	nel)	<u></u>					-	The first state of the second		
TATION LOCATION	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16	17	18	19	20				
Nasatch Front Lake Mt. Mt. Vision Hidden Pk. Little Mt.	C	С	•(C) • •	•(C) •	•	•	•	•		•	•		•	•								وی این این این این این این این این این ای		
Southwest Frisco Pk. Beaver Mt.	С	с				-		-								9 9 9	•	•	•					
Northeast Asphalt Ridge	с					÷						•	•	•	•	2	-		-	- -				
Southeast Bald Mesa Adajo Pk.	С	с	•	•	•	•											•					na pirana kata (na mina da mina na mina da min		
Intermediate Tabby Mt. Ford Ridge Cedar Mt. Dutch John Red Spur Delle Barney Top Moccasin Pk. Monroe Pk. Levan Pk.							•									•			•	•	FIGURE II DATCS Frequenc	the fight pointing of the spectra and the state of the spectra and		

 Γ







	There is a second se	
		5. MOBILE/CONTRO
		Radio access control stations.
		is not contemplate ventional repeater
		1) <u>Mobile St</u>
	And the second sec	The con symbolically
		transmission thesis contra
\sim		plug or PROM vided by the
		switching and is recognize figuration, u
		Mobile c
		. Han cha
		. Inc
	and a second	a C Inc
		. Ind que
		. Inc the
		The mob
	y Killinger	programmed address codes.
		(2) <u>Control</u>
	a series and a second s	Control
		and dispatch microwave co
		ically in Fig
		The dis
		of the Mobile the mobile st
		. Coc
	() (TT)	wit
		un like ang manana manangan ng manana ang mang m
	To any appropriation	
이 아이들은 것은 것은 것은 것은 것은 것은 것은 것을 해야 한다. 것은		

OL STATIONS

to DATCS is accomplished by mobile stations and Direct wireline control to the site controller ted; the mode of operation is similar to a conr type radio system.

tations

nfiguration of all mobile stations is shown in Figure II-1. The frequency selection for and reception is determined by frequency synainted to the licensed channels by a programming I. Switching logic decodes the information procontrol channel; activating the channel change and displays if the address assigned to the unit ed. The mobile unit has a half-duplex* conusing a T/R switch instead of a duplexer.

displays include the following:

ndshake indication indicating a request for annel has been received by the system

dication that the request has been accepted and channel has been made available

dication that the request has been placed in eue

dication that the mobile unit has gone beyond a control channel range.

bile radio has no external control over its as code and has no capability to add additional

Stations

stations are the means by which control points h points access DATCS. There is no wire or ontrol links to the base station site controlonfiguration of control stations is shown symbolgure II-1; it is identical to the configuration station.

splays at the control station differ from those e station. The control station displays include tation displays in addition to:

de identity of the mobile unit in communication th the control station

Code identity of units in the "calls waiting" queue for that control station

- Identity of any mobile in the "calls waiting" queue who has exercised an emergency priority
- Status data display of units assigned to the control station who have a status encoder.

The control station may have the capacity to change its group address or to recieve traffic from more than one group. This can be achieved by switching program plugs or PROMs. By exercising this option a control station can change dispatch responsibilities to accomodate afterhour coverage.

CONTROL POINTS 6.

The configuration for DATCS results in two types of control points; the master system control at DATCS control operation center and the control point at each agency.

1) Master Control

The master control is located at DATCS control operation center. This is shown in Figures II-1, II-5, and II-6. The function of the master control is to monitor system operation as reported by the equipment status and alarm data provided by site controllers and to respond to alarm status with corrective action.

By means of the master control terminal, the programming of the master controller and site controllers can be modified to achieve required changes to system operation. A journal recorder and printer delivers hard copy of system spot status and changes initiated by master control.

The alarm board will alert the master control to faulty operation of any major unit at any of the three base station sites. Faulty operation of the master controller and the voting selectors on each voice channel is also shown on the alarm board.

This position is reviewed as being manned 24 hours a day.

2) Agency Control Points

Agency control points are identified as control stations and, in some cases, control terminals on a drop line circuit to the master controller. Some agencies may have a multiple of control points co-located in a dispatch center. Other public service agencies may have a separate control station at several locations reflecting the multiple control point configuration of their command and control operation.

The agency control terminals are restricted to those agencies that have need to change fleet assignments frequently and have the technical skills to operate computer terminals. Each terminal is restricted by security code to the activities of the agency in which it is installed.

CONTROL STATION/MOBILE STATION OPERATION

The half-duplex configuration of control stations and mobile stations is symbolized in Figure II-1. The main difference is the degree of control each can exercise and differences in the output displays. Each responds to control channel commands and each accesses the control channel(s) in the same way. Both have frequency synthesizers which span the entire 800 MHz trunked radio spectrum and are constrained to the licensed channels by a replaceable code plug or PROM. The code plug or PROM will also determine the unit address and the group address.

1) Control Stations

7.

The control station is the means by which the dispatch center or control point accesses DATCS for communication with its mobile fleet and possibly other control stations. There is no need for wireline control links or microwave control links to the base station sites.

The control station idles on the system control channel and when initiating a call will be switched to an unoccupied voice channel together with all mobile units or control stations having the same group address. A mobile unit or control station in the same group initiating a call will cause the control station to switch to the same unoccuppied voice channel assigned by the site controller.

The displays of a control station will include:

Handshake indicating a request for channel has been received by the system

V-16

Indication that a requested channel has been assigned

Indication that a channel request has been placed in the waiting gueue

- Code identity of the mobile unit in communication with the control station
- Indication of units in the "calls waiting" queue
- Indication if any unit in the "calls waiting" queue has exercised an emergency priority request at any level priority.

If the control station is part of a dispatch center for public safety operations the following displays should be added:

- Status display for those mobile units provided with a status encoder Code identity of the units in a "call waiting" queue
- Code identity of a unit exercising an emergency priority request who is either in the "call waiting" queue or in communication with the control station.

The control stations will have more control head flexibility than mobile stations. This feature includes manual switching of code plugs (or PROMs) to effect changes in fleet configuration or for interagency coordination. The public safety control stations will also have an emergency priority capability.

Mobile Stations 2)

Mobile stations are similar in configuration and function to control stations. Displays are limited to the following:

- Handshake indicating a request for channel has been received by the system
- Indication that a requested channel has been assigned
- Indication that a channel request has been placed in the waiting queue
- Indication that the mobile is beyond the range of a system base station.

	The	Mobi	le d	СС С
	ther vide	eby d wi	chang th	ge
	with unit	i a c	entr	a
	stat mult	us di iple	spla PRON	y YIs
8.	BASE	STAT	ION	0
repe	ater	type	oper	a
voic	e is e _. cha	a c nnel	at e	n a
repe and	ater Mt.	inst. Visio	allat on.	ci
func ler	tions at ea	at e ch si	ach te a	s: r
	• * *	Reco	ver	ĉ
1	•	Sele ques	ct ted	ar
	•	Gene dire	rate ctin	g
	•	Moni	tor	v
•	•	Gene voic	rate e ch	aı
	•	Cont	rol I	ba
		Iden FCC	tify regu	a]
	•	Moni abno	tor rmal	a
	•	Prov cont	ide rolle	s
will progr	The have am in	remot e a n the	te m wide site	a:
9.	MOBI	LE-TO	-MOB	EI
type and	Mobi oper be i	le-to ation n com	-mobi W	

 \square

we we are an in the set of the set

V-18

ontrols and switching capability are limited. ll not be able to change its address codes and je its group assignment. Mobiles can be pro-n emergency priority switch. Those agencies al dispatch status display decoder can add a eporting encoder to their mobiles to drive a at the dispatch center. The mobiles can have s switchable from the central control point.

PERATION

operation is similar to conventional radio tion except for the site controller functions. tional repeater installation for each DATCS ich base station site. There is a conventional on for the control channel at Lake Mountain A digital site controller manages the DATCS ite. The basic functions of the site controle:

and decode inbound control channel requests nd assign unoccupied voice channels as re-

and encode outbound control channel signals system users to assigned channels

oice channel activity

subaudible or digital code information on nnels to unmute authorized receivers

ase station repeaters

all base station repeaters in accordance with ltions for transmitting station identification

and report to alarm panels all base station conditions

system management interface for the master r and master control terminal.

ster controller and master control terminal range capability to modify the applications controller.

LE OPERATION

le communication is inherent in a repeater thin the group address all mobiles will hear, cation with, all other mobiles in the same address group. Intergroup mobile-to-mobile communication can be initiated by the dispatcher or master control terminal using the individual unit address. These one-time intermobile links can be set up by keyboard entry or by cross-channel patching.

10. PORTABLE RADIO INTERFACE

Estimates by industry sources place the very earliest availability of a technically acceptable frequency synthesized trunked portable radios in the 1983-85 time frame. Until such portable radios are available, trunked radio system equipment configurations will include only base stations, control stations and mobile stations.

Public safety radio systems are portable radio intensive. The significant investment in this mix of VHF high band (and/or UHF) portable radio equipment must be protected. These existing equipments must be permitted to operate with the trunked radio system without modification.

The public safety VHF portable operation is to continue using the portable radios on conventional VHF radio channels and manually cross patching these channels into the trunked system at the dispatcher's console when crossband coordination is required. The recommended solution for the Salt Lake County/Utah County area is to keep the conventional VHF frequencies for portable radio operation and trunked system will be used for the mobile radios and their control stations.

11. INTERAGENCY COMMUNICATION

Interagency communication in the DATCS concept can be accomplished by the following key methods:

Radio channel cross patching or phone patching at the dispatch consoles

The pattern of interagency coordination connectivity can be programmed into the controllers and activated by terminals installed at key dispatch centers of agencies where flexibility in fleet programming is required (see Figure II-1)

Routine and frequently activated cross-agency communication routes can be programmed into the control station code plug (or PROM) and these code plugs (or PROMs) manually switched at control stations.

Any of these, or a combination of them, can be configured into the vendor's proposal as best suited to his hardware and software offerings.

12. PHANTOM GROUPS

Temporary one-time needs for multi-agency response or intra group limited response can be satisfied by creating a software "phantom group" to which individual vehicles can be assigned by their unit address code. This assignment will not alter the fixed unit address code or group address code of the involved mobile. Cancellation of the "Phantom group" will return all mobile units automatically to their original group.

V-20

13. DEGRADED OPERATION

The DATCS system redundancy offered by the three base station site configuration makes it unlikely that the Salt Lake County/Utah County DATCS will revert to a conventional radio channel operation due to contoller failure. Some key failure modes have been discussed Appendix B. In general the results of key catastrophic system element failures indicate some coverage loss if Lake Mountain or Mt. Vision are shut down and some loss of talk-back capability in the canyons if Snowbird (Hidden Peak) is shut down. Only an extreme catastrophic failure of both Lake Mountain and Mt. Vision simultaneously will prevent the system from operating at some level of efficiency in a trunked mode. In this case, automatic conversion to a planned conventional mode operation will be programmed.

14. INTERSYSTEM CONSIDERATIONS

ł.

The two types of intersystem operations to be considered are operations with other conventional radio systems and operations with other trunked radio systems.

1) Conventional Radio Systems

Conventional radio system operations within the Salt Lake County/Utah County area can be interface with DATCS at the control consoles. Automatic crossband repeaters will not be used for interconnection.

Conventional radio system operations from outside the Salt Lake County/Utah County area will not be able to interface with DATCS unless they are both terminated at the same dispatch center in the Salt Lake County/ County area. Mutual aid police and fire channels could be included in this category but it is unlikely that mobile units from outside Salt Lake County/Utah County area would require interconnection with Salt Lake County/Utah County DATCS.

2) Trunked Radio Systems in Utah

There is no immediate requirement placed on the Salt Lake County/Utah County area DATCS for interconnectibility with any other DATCS system outside the area. At this time the hardware and software offerings of the key vendors of trunked radio systems are incompatible with each other. The signaling and control methodology and protocols are incomprehensible between the competitor's equipment. To expand the Salt Lake County/Utah County DATCS to a statewide system will require the State of Utah to standardize the DATCS frequency plan (particularly the control channel frequencies), address coding, site controller functions, signaling, and control methods or protocols. This standardization is implied in this reports. The mobile radio configuration of the Salt Lake County/Utah County DATCS would then be compatible with every other DATCS implemented in Utah as long as these standards were imposed.

Assuming the standardization of signaling and control protocols in Utah, a DATCS mobile leaving the service area of a Utah DATCS will lose the control channel signal. It will automatically search for another control channel and, failing to find one, will automatically search for a "fail soft" signal. A "fail soft" signal converts the DATCS mobile to conventional mode 800MHz operation on the channel the "fail soft" signal is found. This "Fail soft" signal can be assigned to an intermediate site repeater which serves the agency to which the mobile is also assigned (separate repeaters and frequencies for Utah Highway Patrol, Utah DOT, etc.). By locating the intermediate sites at terminal points of the existing 2GHz backbone microwave system, the repeaters at the intermediate sites can be controlled by control centers for each agency thereby providing commend and control capability in addition to agency mobile-to-mobile communications within the area of the intermediate site.

When the mobile is again within the range of a Utah DATCS, the priority search capability of the Utah DATCS mobile will find a control signal and switch the mobile from conventional mode to the Utah trunked system configuration. All features of the Utah DATCS are then restored to the mobile unit.

Centralized control of a multi-cell DATCS configuration can be achieved by remote control of a Control Station in each DATCS cell. This can be accomplished through the 2GHz backbone microwave system; control from Salt Lake City can be exercised by participating state agencies if desired.

3) Trunked Radio System in Adjacent States

Outside of Utah the probability of Utah DATCS mobiles being able to communicate with any other DATCS is remote due to the lack of Federal standards and industry standards. Similarly trunked mobile equipment from outside Utah would be unable to operate in the Utah DATCS unless its frequencies and control channel protocols where compatible.

15. FACILITIES

Unique facilities at the base station sites and at the DATCS control operations center will be a consideration. Specialized maintenance facilities will also be required.

Base Station Sites in Salt Lake County/Utah County 1)

It is intended to use existing equipment shelters and power systems at Lake Mountain, Mt. Vision, and Snowbird (Hidden Peak). It is assumed that tower space and intermodulation studies will require the erection of a separate antenna support structure at each of the three locations.

Space must be provided for the DATCS control operations center. This will require a computer environment for housing the master controller and the peripheral alarm board, journal recorder, printer, and master control terminal. The voter selector installation and the control channel audio integration equipment are also a part of the center but need not be colocated with the computer controller. The area must include personnel space for manning the center 24 hours a day and ahould also provide office space for the system manager.

3)

-

Special maintenance facilities for the DATCS should be included. A full time maintenance program is anticipated to keep all of the DATCS system elements functioning at a level of quality needed to minimize performance differences between the multiple channels. This facility must store special parts/modules and will feature special tools and test equipment. 16. PERSONNEL

Personnel considerations include management personnel, system personnel, operating personnel, and maintenance personnel. Personnel staffing will be critical to the success or failure of the DATCS concept.

1) Management

It is suggested that DATCS be managed by an organization separate from but supported by the user organizations. The high technology requirements of DATCS calls for specialized telecommunications management; resources that can not usually be spared from the user organizations. A user group should be formed to set policy and review operations but day-to-day management should be vested in a dedicated manager. The existing pool of communications managers in the area should provide the resources for selecting the DATCS manager.

Software maintenance and hardware maintenance will be a full-time occupation to keep the quality of this high technology system to an expected standard. Thorough understanding of the system operation, operation software, and applications software will be required suggesting the need for a data system analyst who can maintain the software and provide needed modifications.

2) DATCS Control Operations Center

Maintenance Facilities

2) System Maintenance Personnel

Hardware maintenance will require special skills to obtain and retain 800 MHz integrity. Maintenance personnel skilled in computer operations, terminals and displays will also be required. It is assumed that developing these skills in Utah personnel will be part of the vendor obligations for training.

3) Operating Personnel

The DATCS control operation center is conceived as a 24 hour a day operation. One terminal operator/ controller per shift would appear to be adequate. This could be a shared responsibility after DATCS has become operational and debugged. This would require approximately 5 specially trained persons with unique operations training; one would be the Supervisor.

Support service such as purchasing, financial control, audits, benefits administration, facility maintenance, and the like are assumed to be provided by the participating city, county or state agencies.

17. ENGINEERING COST ESTIMATE

This section presents engineering cost estimates associated with implementing a digitally addressed trunked communication system (DATCS) in Salt Lake County and Utah County. Costs are presented in two parts:

Non-recurring costs - one-time capital investment costs

<u>Recurring costs</u> - ongoing costs associated with owning, operating and maintaining the DATCS in the Salt Lake County area.

This cost estimate also presents the probable cost of the components of a statewide expansion of the Salt Lake County/Utah County DATCS.

These cost estimates present the best information available at this time. Much of the hardware and software has not been produced in the form and function required by the system concept. Therefore, a number of assumptions were required based on engineering experience, discussions with the vendor community, and representative cost experience of the commercial trunked system users.

These costs estimates will also need to be adjusted as new information becomes available. No attempt has been made to accommodate price escalation due to inflationary pressures or to extend the value of these funds into some future date when the implementation of DATCS is more probable. The degree of determination on the part of the vendors to pursue the market and the competitive posture of the market at the time of implementation will be the final arbiter of these estimated costs. L

(1) Non-Recurring Costs - Salt Lake County/Utah County

Non-recurring costs are the one-time, initial acquisition costs and the costs of start-up. These costs do not include the first year maintenance costs although, at the option of the purchaser, these costs may be included in the initial purchase contract.

The following summation includes the initial one-time cost for fixed installations and mobile installations for Salt Lake County/Utah County DATCS. The break-down of these costs is shown in the subparagraphs to follow:

Fixed Installations

t Vision	\$ 257,000
Mountain	257,000
pird (Hidden Peak)	247,400
col Operations	253,400
enance Equipment	33,100
col Stations	162,000
l Station Subtotal	\$1,209,900
e Installations	\$2,488,600
Total Estimated Non-	40,000
recurring Costs	\$3,738,500
on Estimated Costs	

\$9,500	atv. 14	\$ 133,000
ers/Combiners \$10,600	atv. 3	31,800
\$600	gty. 3	1.800
ransmission Lines \$800	qty. 3	2,400
oller \$45,00	qty. 1	45,000
0	qty. 2	800
X \$110 ft.	qty. 1	5,500
Interface Modules \$200	qty. 32	6,400
owave Terminals \$18,000	qty. 1	18,000
on (5%)	qty. 1	12,300
		\$ 257,000

(3) Lake Mountain Estimated Costs

	Repeaters \$9,500 Multicouplers/Combiners \$10,600 Duplexers \$600 Antennas/Transmission Lines \$800 Site Controller \$45,000 Modems \$400 Tower 50' X \$110 ft. Microwave Interface Modules \$200 2 GHz Microwave Terminal \$18,000 Installation (5%)	qty. 14 qty. 3 qty. 3 qty. 3 qty. 1 qty. 2 qty. 1 qty. 32 qty. 1 qty. 1 qty. 1	$ $ 133,000 31,800 1,800 2,400 45,000 800 5,500 6,400 18,000 12,300 $ 257,000 } $
(4)	Snowbird (Hidden Peak) Estimated Costs		-
(5)	Repeaters \$9,500 Receiver \$600 Multicouplers/Combiners \$10,600 Duplexers \$600 Antennas/Transmission Lines \$800 Site Controller \$45,000 Modems \$400 Tower 50' X \$110 ft. Microwave Interface Modules \$200 2 GHz Microwave Terminal \$18,000 Installation (5%) Control Operations Estimated Costs	qty. 13 qty. 1 qty. 3 qty. 3 qty. 3 qty. 1 qty. 2 qty. 1 qty. 31 qty. 1 qty. 1 qty. 1	$\begin{array}{c} \$ 123,500 \\ 600 \\ 31,800 \\ 1,800 \\ 2,400 \\ 45,000 \\ 800 \\ 5,500 \\ 6,200 \\ 18,000 \\ 11,800 \\ \$ 247,400 \end{array}$
	Master Controller \$40,000 Alarm Board \$12,000 Voter Selectors \$2,200 Journal Tape Transport \$17,000 Master Terminal \$3,500 Agency Terminals \$2,700 Printer \$4,500 Audio Control Assembly \$1,500 Microwave Interface Modules \$200 Modems \$400 2 GHz Microwave Terminals \$18,000 Telco Installation \$125/line Applications Software Installation (5%)	qty. 1 qty. 1 qty. 13 qty. 1 qty. 1 qty. 1 qty. 1 qty. 1 qty. 1 qty. 1 qty. 25 qty. 7 qty. 3 qty. 20 qty. 1	<pre>\$ 40,000 12,000 28,600 17,000 3,500 27,000 4,500 1,500 19,000 2,800 54,000 2,500 30,000 11,000 \$ 253,400</pre>

(6) Maintenance Spe Digital Tes Special Log Spectrum Ana 800 MHz Sigr Portable Dig Transmission (7) Control Stat Salt Lake Co Utah County Pleasant Gro Pleasant Grow Utah Highway Salt Lake Cit Salt Lake Con Utah Departm Salt Lake Co Brigham Youn Utah Univers Salt Lake Co (8) Mobile Stati Salt Lake Co Utah County Pleasant Grov Utah Highway Salt Lake Cit Salt Lake Cou Salt Lake Cit Salt Lake Cou Salt Lake Cou Salt Lake Con Salt Lake Con

 \square

T

T

Port Annual State

Contraction of the local data

And Andrewson and

(· ·) Transferrant,

A CONTRACTOR OF A CONTRACT

The second s

Contraction of the second

Print Print

V-26

naintenance special loois/Test Equipmen	nt Es	timat	ed	Costs	5
Digital Test Sets \$1,450 Special Logic Probes \$150 Spectrum Analyzer \$6,000 800 MHz Signal Generator \$1,500	qty qty qty qty	· 2 · 4 · 2 · 2	\$	2, 12, 3,	,900 600 ,000
Portable Digital Analyzer \$1,500 Transmission Test Set \$4,300	qty qty	4		6, _8,	,000 ,600
Control Stations Estimated Cost - (\$4,5	500 Ir	nstal.	\$ led]	33,)	100
Salt Lake County Sheriff Utah County Central Dispatch Pleasant Grove Police Utah Highway Patrol Salt Lake City Fire Department Salt Lake County Fire Department Salt Lake County Fire Department Salt Lake City/County Health Department Salt Lake County Public Works Salt Lake County Public Works Salt Lake County Surveyor Salt Lake County Surveyor Salt Lake County Building Inspector Salt Lake County Security Salt Lake County Aging Transportation Utah Department of Transportation Salt Lake County Fleet Management Brigham Young University Police Utah University Police Salt Lake County Attorney	qty. qty. <t< td=""><td>4 6 1 4 2 2 1 4 1 2 1 1 2 1 1 1</td><td>\$</td><td>18, 27, 4, 18, 9, 4, 18, 4, 9, 9, 4, 4, 4,</td><td>000 000 500 000 500 500 500 500 500 500</td></t<>	4 6 1 4 2 2 1 4 1 2 1 1 2 1 1 1	\$	18, 27, 4, 18, 9, 4, 18, 4, 9, 9, 4, 4, 4,	000 000 500 000 500 500 500 500 500 500
Salt Lake County Actorney	<u>qty.</u> 36	1	\$	<u>4,</u> 162,	500 000
Mobile Station Estimated Cost (\$2,300 I	nstal	<u>led)</u> *			
Salt Lake County Sheriff Utah County Central Dispatch Pleasant Grove Police Utah Highway Patrol	qty. qty. qty.	250 73 19	\$	575 167 43	,000 ,900 ,700
Salt Lake City Fire Department Salt Lake County Fire Department Salt Lake City/County Health Department	qty. qty. qty.	125 75 75		287, 172, 172,	,500 ,500 ,500
Salt Lake County Public Works Salt Lake County Surveyor Salt Lake County Building Inconstant	qty. qty. qty.	21 140 25		48, 322, 57,	,300 ,000 ,500
Salt Lake County Security Utah Department of Transportation	qty. qty. qty.	20 1 200		46, 2, 460,	,000 ,300 ,000
Brigham Young University Police Utah University Police	qty. qty. qty.	16 13 23		36, 29, 52,	800 900 900
Salt Lake County Attorney	<u>qty.</u> 1082	<u>6</u>	\$2,	<u>13,</u> 488,	<u>800</u> 600

(2) Recurring Costs-Salt Lake County/Utah County

The recurring cost estimate does not include the cost of operating the existing communication system by each of the participating agencies. The DATCS concept has not changed the existing operating procedures or control locations for any participating agency and the assumption is made that no change will be required in their operating staff or facilities other than the installation of new DATCS equipment. The recurring cost estimates include staffing and facilities needed to centralize the control and maintenance of unique DATCS equipment. There is no existing DATCS configuration similar to the configuration and functions required of this proposed system. Therefore, there is no reliable cost experience to reference and assumptions are made based on past experience with other digitally controlled, computer operated telecommunications systems. These estimated costs are summarized below: Maintenance Technicians \$22,500* att A \$ 90 000*

Maintenance Technicians \$22,500*qty. 4Control Operator (24 hrs) \$25,000*qty. 4Administrator \$31,250*qty. 1	\$ 90,000* 100,000* 31,250*	
Total Estimated Annual Labor	\$221,250*	Server a
Maintenance Parts (Fixed/Mobile) Maintenance Training (2 men) (Updating and Replacement Personnel)	\$110,000 3,000	
EDP Services (Management Reports) (260 hrs) Replacement Equipment (Up-dating) Utilities (Telephone/Power) Total Estimated Annual Service Cost	3,900 179,400 <u>12,000</u> \$308,300	
Total Annual Estimated Recurring Cost Annual Labor	\$221,250*	Contraction of the second
Annual Service	<u>308,300</u> \$529,550	
		Contraction of the second seco
nese estimated costs include full burden.		
V-28		

These estimated costs assume that Salt Lake County/Utah County, or an equivalent governmental agency, will take full responsibility for system maintenance following the expiration of any warrantee provisions. The estimated costs are assumed to be average costs and do not account for early life failures of equipments or components covered by warrantee.

PC PS

En'n

The following smmation of estimated non-recurring costs assume that DATCS for Salt Lake County/Utah County will be implemented initially and that this system will than be expanded statewide by adding DATCS base stations with intermediate repeaters. The estimated non-recurring costs for all statewide subsystems are presented in the following summation; a later section of this appendix will provide cost breakdown details for each subsystem.

WASATCH FRON County)Lake Mo Mt. Vis: Hidden Control Mainten Control Engineer Mobile SOUTHWEST

	Frisco	\mathbf{P}
	Beaver	M
	Cedar (Ci
	Barney	T
	Moccasi	ln
	Levan H	?ea
	Monroe	P
	Enginee	er
NORTH	IEAST	
	Asphalt	. T

Vernal C Tabby Mor Ford Ride Dutch Jo Engineer

* T

(3) Estimated Non-Recurring Costs-Statewide Summation

 $\left\{ \cdot \right\}$

\underline{T} (see Section A.1	S	alt	Lake	County/Uta	h
untain ion Peak Operations ance Equipment Stations ring Services Installations (1082) Total Subsystem	\$	257 257 247 253 33 162 40 2,488 3,738	,000 ,400 ,400 ,100 ,000 ,000 ,600 ,500		
Peak Mountain Ity Control Top Repeater Peak Repeater Peak Repeater Peak Repeater Total Subsystem	Ş	132 132 136 50, 50, 50, 30, 634,	,700 ,550 ,600 ,600 ,600 ,600 ,600 ,000 ,350		
Ridge Control untain Repeater ge Repeater hn Repeater ing Services Total Subsystem	\$ \$	132, 136, 50, 50, 50, 30, 451,	700 550 600 600 600 000 050		

SOUTHEAST

	Bald Mesa Adajo Peak Moab Control Cedar Mountain Repeater Engineering Services Total Subsystem	\$	132,700 132,700 136,550 50,600 30,000
NORTHWEST			
	Little Mountain	\$	247,400
	Expansion		30,400
	Delle Repeater		50,600
	Engineering Services Total Subsystem	\$	404,000
	TOTAL ESTIMATED STATEWIDE DATCS COST	\$5	,779,450

It is again noted that additional mobile stations have not been included in this statewide estimates.

(4) Statewide Non-Recurring Cost Detail

The implementation of DATCS statewide is assumed to follow the base station/intermediate repeater configuration described in the system concept The following estimated costs are based on that configuration and the DATCS frequency plan. The number of mobile stations which may be licensed in areas other than the Salt Lake County/Utah County is unknown, therefore, these cost estimates will include only the fixed base and fixed control station equipment.

Frisco Peak Estimated Costs

Repeaters \$9,500 Multicoupler/Combiner \$10,600 Duplexer \$600 Antenna/Transmission Line \$800 Site Controller \$40,000 Modems \$400 Microwave Interface Module \$200 2 GHz Microwave Terminal \$18,000 Tower 50' X \$110 ft. Installation (5%)

\$47,500 qty. 5 10,600 qty. 1 600 qty. 1 800 qty. 1 40,000 qty. 1 800 gty. 2 2,400 qty. 12 18,000 qty. 1 5,500 qty. 1 6,500 qty. 1

482,550

Master Con Alarm Board Voter Sele Journal Tay Master Ter Printer \$4 Audio Cont Microwave Modems \$40 2 GHz Micr Telco Inst Application Control St Installati Asphalt Ri (See Frisc Vernal Con (See Cedar Bald Mesa (See Frisc Adajo Peak (See Frisc Moab Contr (See Cedar

 $\left[\right]$

P

tent thereas

tarburnetari

Little Mou (See Snowb Additional Voter Micro Modem

Cedar City Control Estimated Costs

troller \$25,000 d \$10,000 ctors \$2,200 pe Transport \$11,000 minal \$3,500 ,500 rol Assembly \$1,200 Interface Modules \$200 0 owave Terminals \$18,100 allation \$125/Line ns Software (Transfer) ations \$4,500 on (5%)	qty. 1 qty. 1 qty. 4 qty. 1 qty. 1 qty. 1 qty. 1 qty. 24 qty. 4 qty. 2 qty. 10 qty. 1 qty. 1 qty. 3 qty. 1	$\begin{array}{c} \$ 25,000 \\ 10,000 \\ \$,800 \\ 11,000 \\ 3,500 \\ 4,500 \\ 1,200 \\ 4,800 \\ 1,600 \\ 36,000 \\ 1,250 \\ 10,000 \\ 13,500 \\ 5,400 \\ \$136,550 \end{array}$
dge Estimated Costs		
o Peak)		\$132,700
trol Estimated Costs		
City Control)		\$136,550
Estimated Costs		
o Peak)		\$132,700
Estimated Costs	: 	
o Peak)		\$132,700
ol Estimated Costs		
Gity Control)		\$136 550
CILY CONCION)		φ 1 30,330
intain Estimated Costs oird-Hidden Peak)		\$247,400
Selector Modules \$400 wave Interface Modules \$200 s \$400	qty. 13 qty. 32 qty. 2	5,200 6,400 800
Microwave Terminals \$18,000	gty. 1	18,000

2 GHz Microwave Terminals \$18,000 gty. 1 Total Estimated Cost to Add Little Mountain

V-31

\$277,800

Tabby Mountain Repeater Estimated Costs		
Repeater \$9,300 Multicoupler/Combiner \$10,600 Duplexer \$600 Antenna/Transmission Line \$800 Repeater Interface \$500 Tower 50' X \$110 ft. Microwave Interface \$200 Installation (5%)	qty. 3 qty. 1 qty. 1 qty. 1 qty. 3 qty. 1 qty. 6 qty. 1	27,900 10,600 600 800 1,500 5,500 1,200 2,500 50,600
Barney Top Repeater Estimated Costs		
(See Tabby Mountain Repeater)		\$ 50,600
Moccasin Peak Repeater Estimated Costs		
(See Tabby Mountain Repeater)		\$ 50,600
Monroe Peak Repeater Estimated Costs		
(See Tabby Mountain Repeater)		\$ 50,600

The implementation of the system concept will proceed only after thorough review by the potential user agencies, their department heads, and the Public Safety Director of each participating municipality. After this has been achieved and the implementation plan has been officially sanctioned, another critical milestone must be met before procurement activities can be initiated. Federal Communications Commission (FCC) licensing requirements for large scale 800 MHz trunked radio systems in local government service have not been clearly defined or tested at this time. Following local government approval the initial implementation activity must be directed toward satisfying all of the elements needed to secure reasonable assurance from the FCC that the trunked radio system, as planned and approved, can be licensed. LOCAL GOVERNMENT APPROVALS 1. Local government approvals will be achieved in three steps, securing of user agency agreements, development of funding agreements to implement and operate the system, and the legally binding commitment of the municipalities to participate in and support the system. (1) Agency Agreements Signed agreements by each agency planning to participate in the trunked radio system must be acquired. These agreements will conform with FCC requirements (Section 90.359 and 90.179). As a minimum they will include: Agreements regarding the supervision and control of the trunked system operation Method of cost-sharing and pro-rating of costs among all cooperating parties Responsibility for maintaining system operation records and filing reports with the FCC (Section 90.391).

(1)

15

6.

×....

-

1

Management authority over the system staff and equipment Centralized maintenance of system components

v-32

III. REQUIRED IMPLEMENTATION ACTIVITIES

In addition to the items needed for FCC licensing, the agency agreements will also include provisions for:

Property, equipment and facilities provided by the user agencies and municipalities

Modification and/or termination of the agreement as subject to FCC Rules and Regulations (Section 90.391).

Agency agreements will be the key element in securing the funding agreements and municipal government approvals.

(2) Funding Agreements

The acquisition of agency agreements will provide dimensions for revised cost estimates needed to establish funding requirements. These requirements will include estimates of non-recurring implementation costs and recurring operations costs. The following elements of costs will be identified:

Implementation Costs

- Planning Activities
- Professional Services
- Hardware/Software
- Facilities
- Installation/Test/Acceptance
- Personnel Staffing and Training
- Operating Costs
 - Personnel Salary and Overhead
 - Supplies
 - Maintenance of Hardware/Software
 - Equipment Replacement
 - Continued Training

The distribution of these costs among the funding sources will be decided and firm commitments and/or agreements acquired. These funding sources will include:

Federal and State grants funds

Local municipal funds

In-kind services, equipment, or facilities

V-34

•

 \mathbb{E}

圆

2

(E)

1

2.

The funding agencies must approve and be committed to the funding plan before final and binding commitments will be south from the State of Utah, county governments, and municipalities.

Based on the agency agreements and funding agreements achieved, firm commitments will be required from State, County and municipal governments to assure continued project support and to respond to Federal and State grants requests. These approvals will be required in the form of resolutions or other legally binding documents.

PRELIMINARY LICENSING CONSIDERATIONS

The licensing of multiple base station wide-area trunked radio systems in state and local government service and public-safety service is not clearly defined by present FCC Rules and Regulations. Waivers to the present FCC Rules and Regulations may be required, particularly Section 90.375 which defines time limits for phased implementation. It is probable that waivers to Section 90.739 which limits power and antenna height wll also be needed.

Before hardware and software commitments are made, preliminary discussions will be held with the FCC. An expression of assurance will be sought indicating that the trunked system, as planned and approved by the user agencies and governmental authorities, will receive favorable licensing consideration from the FCC when formal applications for licenses are submitted.

3. IMPLEMENTATION SCHEDULE

The detailed implementation schedule will be developed following assurances from FCC that the trunked radio concept will be licensable under FCC Rules and Regulations and that any revisions to the concept required by FCC evaluation of the system design has met the approval of the user agencies and the participating municipal governments. The implementation schedule will consist of two key interrelated parts; a system implementation schedule and a corresponding financial schedule.

User agency cost sharing assessments

Grants from private foundations

Gifts from private sources.

(3) Municipal Government Approvals

(1) System Implementation Schedule

The system implementation schedule will be a timedependent schedule. The project will be defined in terms of separate tasks with clearly defined milestones indicating the completion of each task and definitive measureable milestones leading to the completion of each task or project element. Major tasks include:

Project Planning and Reporting

Engineering Services

Preparation of Specifications

Preparation and Approval of Statement of Work

Solicitation of Bids

Contract Award

Facility Preparation

System Licensing

Equipment Delivery/Installation

Test and Acceptance

Personnel Training

System Cut-over

System Evaluation.

Each major task will be supported by task breakdown schedules which define the subtasks needed to achieve the completion of each major task.

(2) Financial Schedule

.

The financial schedule will be coordinated with the system implementation schedule. This schedule will be based upon the funding agreements and will include the following major elements:

Milestones showing each major commitment of funds

The date and amount funds will be made available to the project from all sources

The schedule dates for the submission of financial reports.

V-36

	ment	al	app	r
4.	SYSI	ΈM	MAN	AC
will Dire repr Thes	The be t ctor esent e fun	mar the wil ing icti	nage res ll l ea ons	
	•	<u>Sy</u> as st	ste sis aff	m te
	•	Pr th Co	ocu e p unt	re u y
	•	<u>Fi</u> se	nan rvi	
	•	Fa si eg pa	cil tio uip rt-	i m t
	•	Pe di ac	rso rec	n t i

0

mel - the project staff personnel are under the supervision of the Program Director; personnel ties such as job descriptions, salary determination, and benefits administration will be under the direction of Salt Lake County unless the person is on loan from another municipal jurisdiction.

In addition to the project staff reporting to the Program Director, two advisory groups will be responsible for overall project policy with the Project Director serving as chairperson for each advisory group. These groups consist of:

> User Representatives - this group is composed of appointed representatives of each user agency; they provide direct liason between the project organization and their respective agencies

> Government Representatives - this group is composed of appointed representatives of each municipality and state or county government sections which have an operational or financial interest in the program.

The submission and distribution of financial reports will be elements of the financial agreements and governovals.

GEMENT AND OPERATION

ent and direction of the implementation program ponsibility of the Program Director. The Program e supported by a full-time and part-time staff ch key function of the implementation program. include:

Engineering - this staff function will be ed by consulting engineers and the engineering of the successful system contractor

ement - This staff function will be assisted by irchasing and procurement personnel of Salt Lake

e - this staff function will be a part-time e provided by Salt Lake County

ties - this staff function related to the acquiand preparation of special facilities for the nent and personnel of the project will be provided ime by the State of Utah and Salt Lake County

It is the intention that this management group be carried over in form and function to the operation of the trunked radio system after it has been installed, tested, and accepted. Additional personnel needed to operate the system from a central location will be added to the Facility function of the project staff. At that time the Facility function will become a fulltime staff position under the Program Director.

5. PROCUREMENT

The procurement of the trunked system will proceed in two sequential activities; system engineering and purchasing.

(1) System Engineering

Technical assistance will be required to develop the engineering design of the trunked radio system based on the system concept. Key questions to be resolved before procurement specifications are prepared include:

> Radio Propagation Analysis - engineering analysis of 800 MHz radio propagation from candidate base station sites will be conducted to determine radio coverage of the Salt Lake County area and to establish antenna pattern characteristics. FCC Rules and Regulations (Section 90.379) clearly limit the effective radiated power and antenna elevation at each proposed site and engineering analysis will be required if waivers from these FCC Rules and Regulations are needed

Switching Logic - the switching logic for the master controller, site controller, mobile stations, and control stations will be engineered to be compatible with expected vendor hardware offerings

Software - the system software for the master controller and site controller will be engineered to enable efficient software management from a master terminal

Based upon the results of these technical activities, systems engineering will prepare detailed system specifications and equipment specifications for competitive procurement.

(2) Purchasing

The purchasing function will consist of three key activities; developing a Statement of Work (SOW), soliciting competitive bids, and awarding a system contract.

Statement of Work (SOW) - the SOW is a precise description of the materials and services required of the contractor and the time domain in which delivery is to be made. The key elements of the SOW will be:

Background and Objectives of the project Required qualifications of successful bidders Services and Support to be provided

Quantities of each equipment to be provided

Schedule for all deliverable items - hardware, software, serivces, and support items

Soliciting Competitive Bids - competitive bids from all qualified vendors will be solicited by means of a procurement package which will include:

State and County legal requirements

Statement of Work

system

LICENSING

6.

 \square

12

K

and Mines .

System Specification

Equipment Specifications.

A one-step solicitation is anticipated in which each bidder will provide a technical proposal and a firm fixed price cost proposal for the complete installation of an operating digitally addressed trunked radio

Award the System Contract - the combined technical proposal and cost proposal most advantageous to Salt Lake County and associated municipalities will identify the successful bidder. A firm fixed price contract without cost escalation is anticipated.

The signed contract defining the equipment to be purchased and the quantity of that equipment will provide sufficient information to file application with the FCC for system licenses. Present FCC practice for licensing 800 MHz trunked system requires the following minimum submission:

> Form 400 for each license requested (Section 90.357) Agreements with each user agency relative to control of the system (Section 90.179)

> > v-39

Agreements with each user agency relative to costsharing of installation and operation (Section 90.179)

Financial plan showing the service will be provided to user agencies at cost (Section 90.359)

System description and specification (Section 90.359, Section 90.371, Section 90.375, and Section 90.379)

Eligibility statement for each user (Section 90.359)

Evidence of purchase orders placed for the equipment.

It is probable that FCC waivers will have to be asked to extend the number of years for phased implementation of the system (Section 90.375) and for the limits on power and antenna height (Section 90.379).

Technical support for negotiating system licenses with the FCC will be requested of the engineering consultant and from the system contractor.

FACILITIES ACQUISTION AND PREPARATION 7.

It is the responsibility of the Program Director to provide adequate facilities for the equipment and staff of the trunked radio system. Subject to revision by the engineering study, the following facilities will be required:

> Mt. Vision - equipment shelter for fifteen 800 MHz repeaters, a site controller, ancillary equipment for the antenna subsystem, antenna support structure for four antennas, and appropriate power/climatic control equipment

> Lake Mountain - equipment shelter similar to Mt. Vision

Hidden Peak - equipment shelter similar to Mt. Vision

Control Operations - equipment shelter for the master controller/terminal, alarm equipment, recorders, printers, microwave terminations and other items of equipment associated with the central control function. Staff facilities for a Program Director, Facilities Manager, Engineering, Software/Hardware Manager, and Programmer/Operators will also be required.

Centralized maintenance will also require special facilities which must be provided by the State of Utah or Salt Lake County.

It will be the responsibility of the Facilities Manager on the staff of the Program Director to obtain these above facilities and prepare them in a timely manner.

V-40

required:

1

Contraction of the second

 \square

I

Constant Server

8.

Maintena Control System P: • Faciliti .

The training and certification of all operating personnel, maintenance technicians, and management personnel will be a contractual obligation of the system contractor.

POST IMPLEMENTATION SUPPORT 9.

The post implementation support program will address the funding and staffing needed for sustained operation of the trunked radio system.

The recurring costs of continued operation include the following major items:

Supplies - all expendable materials used in operating and managing the system

PERSONNEL AND TRAINING

Personnel for the management and operation of the trunked radio system will be identified and prepared for their assignments in coordination with equipment installation and test. It is intended that the Project Director and staff will continue in their positions after the trunked radio system has been installed and accepted. The additional people required to operate the system will be recruited and trained during the system installation test so that their services can be used as much needed test personnel and the experience gained by them can serve as on-the-job training before the system becomes operational. It is anticipated that the following additional personnel will be

nce Technicians	Quantity 4
Operators	Quantity 4
rogrammer/Analyst	Quantity 1
es Administrator	Quantity 1

Job descriptions for these positions will be developed by the Program Director and the personnel offices of Salt Lake County. Recruiting will be by public announcement of available job openings. It is anticipated that the positions will be filled by transfer of presently employed and skilled personnel of Salt Lake County and Salt Lake City.

(1) Post Implementation Funding

Personnel - salary and benefits

Maintenance - maintenance of the facilities and system hardware/software after the warranty and first-year maintenance contract

<u>Training</u> - refresher training and replacement training

<u>Utilities</u> - power, telephone, and other services provided by an outside agency

Equipment Replacement - periodic replacement of worn or obsolete equipment; equipment that can no longer be repaired economically.

These costs will have been identified and submitted to the FCC as part of the licensing requirements. They will be reviewed periodically. These will be filled to the user agencies in accordance with a formula developed and reported in each agency agreement.

(2) Personnel Replacement and Training

The replacement of personnel and the up-dating of staff personnel will be a recurring problem. There will be labor turnover. Recruiting of personnel will be by suitable public announcement of job openings and the testing of applicants. Probationary employees will be trained by the technical staff.

(3) Support Staff

•

The system will have the full time services of equipment maintenance personnel and software system personnel. At least one senior system programmer/ analyst will be required. It is anticipated that a full time staff of four specially trained maintenance technicians will be required and four qualified control operators will also be required to provide 24 hour service to the user agencies. Legally binding agency agreements are needed to support FCC license applications for 800 MHz Multi-agency trunked systems and to define correctly the role each agency will plan in the administration and use of the cooperative system. This Appendix will offer suggestions as to the scope and content of each set of agreements. Furthur discussion will be found in the System Implementation Planning Guidelines, a National APCO publication prepared by Booz Allen and Hamilton, dated November 15, 1980.

APPENDIX A

SUGGESTED AGENCY AGREEMENTS

FORWORD

	SUG
	The agreement
	agencies to enter i
	service of value to
	agreements are hig
	ments of each pro-
	System Implementat
	gest only a general
	nrise contemplated
	environment. Legal
	1. <u>Purposes</u>
	The nurne
	reason for th
	expected to b
	reducing this
	narticipation
	which the part
	the true natu
	thereby provid
	2 Scope of
	The agree
	the terms of
	liabilities of documented
	between the pa
	The follo
	vides a serie
	precise form of
	used in each
	many cases, st
	state or Feder
	II. SUGGESTED MODE
	The following
	considerations in
	the name high
	Alecennik has him
医马克氏 化二乙基 化乙基乙基 化乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基乙基	

APPENDIX A

GGESTED AGENCY AGREEMENTS

by two or more local government public safety into a joint enterprise which provides a mutual to the participants is best documented by writetween the parties. By their nature, these ghly individual; tailored to specific requireogram in each locality. This appendix to the tion Planning Guide (SIP) is intended to sugl outline of such agreements. The details, and entation, will depend upon each kind of enter-, its complexity, and the immediate cooperative l counsel should be consulted.

of the Agreement

pose of a written agreement is to describe the he cooperative effort and the results that are be achieved through mutual participation. By s to a clear and unambiguous written statement, parties will fully understand their mutual a. Equally important, the governing bodies, of ticipants are a part, will be able to determine ture of the program and the expected results; de their support.

the Agreement

ement should define the dimensions within which the agreement apply. The responsibilities and of each of the participants should be clearly There should be no unwritten understandings parties.

lowing model outline of an agreement form proes of topical considerations to be evaluated development of an interagency agreement. The of a specific agreement, and the language to be case, should be provided by legal counsel. In tandard legal form and substance is required by eral regulation.

EL OUTLINE

topics and items are intended to delineate key the preparation of an interagency agreement.

1. General Terms and Legal Base

The general terms will include the following as a minimum:

The dentification of each participating agency

The requirements to be imposed on those agencies who may wish to participate at a later date

The concise purpose of the joint enterprise and the mutually advantageous results that are expected to be achieved

A statement of the legal authority under which the public agencies are able to contract with each other for the stated purpose; this may be a state law, a local government act, a series of resolutions passed by the local governmental authorities, or similar authorization.

2. Services to be Provided

.

•

٠

The services provided by each agency and to each agency will be defined. These include the following considerations:

> Definition of the services to be provided by each agency to the mutual system; these services could include, for example, equipment maintenance, equipment, facilities, operating manpower, support services, and similar items of value to the system

> Definition of services the system will provide to each participating agency; such services could include the dispatching of field forces, management reports, statistical data, and similar items of value to the agency

> The definition of the quantity or extent of each service to be provided

The responsibility each agency has for delivering its service and the responsibility the mutual operation has for delivering its services to each agency

The extent of any back-up or redundancy to assure that services are delivered in a timely manner throughout the life of the agreement. ations. agency 4. considerations:

Liability

3.

Î

Time

Use of outside funding sources such as grants or gifts.

A key element of the agreement between two or more public agencies is a clear definition of the liability each of the agencies has for the actions of other parties to the agreement. This is particularly true if one or more of the agencies provides public services on behalf of the other agencies. Agreements to limit the liability exposure of participating agencies would include the following consider-

Responsibility for defending the provider agency in potential lawsuits

Indemnification of the providing agency against liability judgements against a participating agency

Indemnification of the participating agency against legal actions or liability judgements against the provider agency arising from the normal provision of agreed-to services.

Amount and Manner of Payment

The providing agency must be reimbursed for the expenses of the service it provides to the participating agencies. A portion of this obligation will fall to the individual participating agencies. The agreement between the agencies must be specific regarding this individual responsibility and will be definitive of the following

The specific services to be provided and the cost of each service

The method by which the cost and value of each service is calculated

The means of payment; for example, cash, in-kind services, cash equivalent in material or equipment, and any other valuable consideration agreeable to the parties to the agreement

Time and manner payments are to be made.

5. Fiscal Procedures

Sound fiscal planning is key to the successful implementation of public agency cooperative programs. The responsibility of the provider agency for financial integrity must be defined in the agency agreements. The agreement will specify the following elements:

- The type of financial records and how they are maintained
- The contents of the periodic financial reports and the schedule for submitting these reports to higher authority and to participating agencies
- Access to the financial records by each participating agency
- Certification and/or audit of the records and financial reports
- Terms and conditions relative to the periodic reassessment of rates charged for each service provided.

Administration 6.

The administrative structure of the cooperative program should be defined in the agency agreements. The authority and responsibility of each agency for participation in policy administration and the authority and responsibility for day-to-day operation should be included. Some of the key elements in the agreement are:

- The identification of the person(s) authorized to sign the agreement on behalf of each party to the agreement
- The identity of the person(s) who will represent the participating agency and providing agency in policy matters
- The responsibility and authority of the providing agency to supervise project personnel and establish personnel policy
- The responsibility and authority of the providing agency over joint-use facilities and equipment

The retention by participating agencies of control, responsibility, and authority over their personnel, facilities and equipment that may be used in the cooperative project

Identity and authority of persons responsible for the review of the agreements and for the determination of compliance

These and other administrative elements suitable to the unique environment and nature of the proposed project should be negotiated between the parties and made part of the written agreement to participate.

Real Property and Equipment

T

LaD)

Constant.

.

7.

8.

tions are:

The cost-effective implementation of a multi-agency cooperative program should make maximum use of facilities, property and equipment available from the participating agencies. The agreement should identify these properties and materials; it should establish the rules under which the properties and materials are to be used. Some key considera-

Identify each item of property and equipment that is to be transferred to the use of the cooperative system

Establish the value of the property and/or equipment and show the method used to establish the value

Provide for the return of the property and equipment during the term of the agreement and upon termination or cancellation of the agreement

Define the method to be used to establish the value of property and material to be returned to the original supplier upon the termination, modification, or cancellation of the agreement.

Duration, Termination, and Amendments

The agency agreement must provide for modification to the agreement during its term. It must also define the duration of the agreement, make provisions for cancellation of the agreement, and disclose how the agreement is to be terminated. The following items should be included:

The term of the agreement.

The circumstances under which the agreement can be terminated during its intended duration

The terms under which the agency can withdraw from the agreement

The circumstances under which either party can cancel the agreement

V-49 ·

Procedures for amending the agreement during its intended term

 \square

Π

U

Π

 \square

 \Box

Disposition of property and equipment upon termination or cancellation

9. Other Provisions

The agency agreement is a custom document generated to reflect the unique needs of the specific circumstance and the environment in which the nutual agreement is being generated. Each agreement will include terms and conditions of an individual nature. Key among these are the following suggestions:

Method of resolving disputes; for example, arbitration, application of appropriate state or Federal law, etc.

- Identity of the court of jurisdiction for adjudicating disputes
- Definition of terms used in the agreement
- Statutory status and filing of the agreement
- Interrelationship of agreements.

* * * * *

The drawing of interlocal contracts and agreements to provide public services or hold public property for the mutual advantage of multiple governmental organizations may be subject to local, state, or Federal regulations and/or laws. These documents should be drawn up under the advice of legal counsel to insure their status as legal instruments. These system requirements for a digitally addressed trunked communication system (DATCS) for Salt Lake County, Utah evolved from the report "Analysis of Problesm and Requirements" prepared by Booz, Allen and Hamilton, June 18, 1980. This summation of DATCS system requirements is also included in the Project 16B Task 7 report "DATCS Design Concept-Salt Lake County/Utah County" prepared by Booz, Allen and Hamilton on September 19, 1980.

APPENDIX B

SYSTEM REQUIREMENTS

FOREWORD

	a angin te transformation A	the second second the Branch and the second second second second
		The system requ of the Task 6 report (June 18, 1980). Th of a meeting in Salt in the following sub
		1. WIDE AREA COVER
		Radio coverage testing at 800 MHz i precludes any reason station site will pr of radio propagation
		MHx must be conducte and other fixed site tions are based on p and from known cover
	(1) <u>Salt Lake</u>
		The Traver tend to partial City from Utah that north/sout in Salt Lake Cc
		Preliminar provide the req ridge.
	\square	2) <u>Utah County</u>
		L ake Mount site to cover U of Provo and Or
	51	3) <u>Canyons Eas</u>
		The east c suggest a talk-
		Wide area cover
		10, 1980, does not e County. However, it
		this system concept system. The followi
		engineering support
		Preceding page blank
\mathbf{L}		

APPENDIX B

SYTEM REQUIREMENTS

airements have been developed from a review "Analysis of Problems and Requirements" as summation of this review was the subject Lake City on July 22, 1980 and is reflected osections.

RAGE

experience at UHF and the results of limited in the Salt Lake County/Utah County area hable expectation that a single 800 MHz base rovide adequate coverage. Engineering studies h and antenna pattern characteristics at 800 ed before the location of the base stations es can be established. The following suggespreviously reported 800 MHz coverage tests rage patterns at VHF frequencies.

County

rse Mountains south of Salt Lake City Lly separate Salt Lake County and Salt Lake County and Provo/Orem areas. It is likely th shadow effects will require a base station bunty.

ry indications are that Mt. Vision will quired coverage north of the Traverse Mountain

cain is the primary choice of a base station Jtah County and the population concentration rem.

st of Salt Lake/Utah Counties

canyons of Salt Lake County and Utah County -back base station at Snowbird (Hidden Peak).

CONSIDERATIONS

rage is defined in the agreement of January extend beyond Salt Lake County and Utah is recognized that a statewide DATCS is Lake County/Utah County system described in is the initial implementation of a statewide ing addresses statewide configurations with nat it is advisory information only and no data was possible under the existing agreement.

Suggested Configuration 1)

The configuration suggested is a four cell DATCS system will the major cell in the Wasatch Front from Ogden south through Salt Lake City to Provo. A second cell in the soutwest includes Cedar City/St. George area. A third cell in the northeast covers Vernal and the fourth cell in the southeast includes the Moab/Monticello area. These four DATCS cells are independent of each other and their Coverage areas do not overlap.

2) System Engineering

The system of DATCS cells must be engineered to preserve the trunked radio insolation between cells. Antenna elevations, antenna patterns, and limitation of effective radiated power from each site must control this isolation. This engineering effort will be part of a later DATCS implementation program.

3) Base Station Sites

The following candidate base station sites in areas other than Wasatch Front are suggested based on reported radio coverage experience from existing VHF radio installations. The sites which are also terminal points for the existing 2GHz statewide microwave system are preferred:

- Ogden Area much of the Ogden area should be accessable to the DATCS base station at Mt. Vision but talk-back support (voted with Mt. Vision) might be needed from Little Mountain
- Brigham/Logan Area a DATCS base station at Mt. Logan would appear to offer adequate coverage and also protect the north Ogden area

St. George/Cedar City - A DATCS base station at Frisco Peak voted with a talk-back support station at Utah Hill would be primary candidates for southwest Utah DATCS; antenna pattern shaping will be required to restrict coverage north and northeast of Frisco Peak

Vernal - a DATCS base station at Asphalt Ridge is the primary candidate for northwest Utah DATCS

Moab - a DATCS base station at Bald Mesa is the primary candidate for southeast Utah DATCS; a talk-back support station at Abajo Peak voting with Bald Mesa may also be required.

4) Intermediate Sites

Bridging between DATCS cells can be accomplished by intermediate repeater sites strategically located between the DATCS cells. These sites are conceived as being one or more 800MHz channel repeaters which are accessed by the "fail soft" capability of a mobile which has gone beyond range of a DATCS control channel signal. Command and Control can be exercised by connecting these repeater sites to a control point through the existing 2GHz microwave system. Suggested candidates for these intermediate sites include:

Tabby Mountain Ford Ridge Cedar Mountain Dutch John Gap Red Spur Black Mountain (Delle) Moccasin Peak Monroe Peak Leran Peak

The decision to use any of these locations or any other locations as intermediate sites will require engineering study and field tests at 800 MHz to determine the antenna elevations, antenna patterns and effective radiated power needed for system integrity.

Although statewide coverage with DATCS appears feasible, the configuration of a statewide system must remain flexible in response to engineering evaluation and expanding DATCS technology. In all probability the existing 2GHz backbone microwave system will not provide the channel capacity needed to support a statewide DATCS. The microwave will require revision to a higher level of channel capacity or replacement of some microwave legs with new microwave equipment. Again a thorough engineering study will be required beyond the scope of this program.

MULTIPLE DISPATCH POINTS

3.

The existing public-safety, state government, and local government radio services in Salt Lake County/Utah County consist largely of independent radio channels; each gency having its own command and control requirements over its own fleet of mobiles.* In its operation DATCS will be a replica of this configuration. Therefore the DATCS concept will not only provide for a multiple of fleet and/or subfleet addresses but will also be programmed to direct these subfleets to their individual dispatch points by means of their address identity.

* Task Report, Salt Lake and Utah Counties - Mobile Radio Systems Description and Requirements, "Booz Allen, revised July 15, 1980.

The number of individual dispatch points for Salt Lake County and Utah County could include:

Salt Lake County Sheriff'

- Sheriff's Central Dispatch
- Pleasant Grove Police
- Utah Highway Patrol
- Salt Lake City Fire Department
- Salt Lake County Fire Department
- Salt Lake City and County Health Departments
- Salt Lake County Public Works
- Salt Lake County Attorney
- Salt Lake County Surveyor
- Salt Lake County Building Inspection
- Salt Lake County Security
- Utah Department of Transportation
- Salt Lake County Fleet Management Division

Brigham Young University Police

Utah University Police

This list assumes that Salt Lake City Police, the City of Provo municipal agencies, and City of Orem Municipal agencies will not participate in DATCS implementation according to the Task 4 participation.

In the final configuration each dispatch point must be independent of any other dispatch point without incidental interaction between them except on command and the master controller. The mobile units will not have the capability to modify address codes or change the dispatch point/

To respond to an incident which calls for a mobile command post and a mix of public safety/public service agencies responsible to the command post, the system controller can create a phantom group and the mobile units can be attached to that group control point by software instructions based on the mobile address codes. There would be no change needed in the mobile logic to create the phantom group or to return the units to their original group and control point assignment.

V-56

4. INTERAGENCY COORDINATION

particular colorida

And the second se

The survey of Salt Lake County/Utah County area agencies* resulted in a pattern of interagency coordination connectivity though by the agencies to be appropriate to their operating needs. This pattern of permissible connectivity between the agencies must be programmed into the controller software and provisions made for it to be modified by program instructions as needed.

It is expected that regional public safety and public service interagency coordination plans will include the regional disaster assistance plan developed in accordance with Federal Act PL93-288 (Disaster Relief Act of 1979) as amended.

5. CONVENTIONAL RADIO SYSTEM INTERFACE

In accordance with the system philosophy adopted for DATCS, public safety communications systems, as now configured in the Salt Lake and Utah Counties area, some public-safety agencies would not fully participate in the DATCS concept. Public safety radio system tend to be portable radio intensive. This represents a significant investment in relatively new VHF high band equipment which must be protected. Together with the questionable availability and cost of trunked radio portables, a judgement has been made to continue the operation of police and fire portable radios on conventional radio system channels where they now operate and provide any required coordination with DATCS by cross patching at the appropriate dispatch console. The Salt Lake City Fire Department Mobile Data Terminals will also remain outside DATCS until technical uncertainties related to data error rate degradation due to the DATCS/MDT multipath characteristics at 800 MHz can be resolved by engineering field tests and theoretic calculations that are not within the scope of this assistance program.

With the Salt Lake City Police radio system remaining outside of the DATCS concept, there is no requirement for CAD interface. However, the provisions for interagency coordination will probably include the Salt Lake City Police dispatch Center; the interconnection being made by a control station. The interface with the Salt Lake City Fire CAD will include only status reporting to the status display of the CAD.

6. EMERGENCY ACCESS PRIORITY

The capability for prompt access to the DATCS under conditions which justify the exercise of channel access priority when the DATCS system is fully occupied would be limited to public safety mobiles and certain selected public officials. The exercise of a priority access function when the DATCS is fully occupied requires a system frequency plan which designates at least one DATCS radio channel for control information.

* Task 4 Report "Salt Lake and Utah Counties - Mobile Radio Systems Description and Requirements," Booz, Allen, revised July 15, 1980.

The waiting queue priority method has been selected as the means of emergency access to the fully occupied DATCS. When denied access to DATCS, those mobiles with emergency access priority capability can activate this function manually and thereby move their request for channel access to the top of the appropriate queue. The public service/public safety agencies have indicated the desire for a 5 level priority structure. The activation of the emergency priority request would move the request to the top of waiting queue but not ahead of any other request for emergency priority from an agency mobile with an equal or higher priority classification.

The activation of this emergency function will also be recorded by the central controller and display at the control * station along with the identity of the mobile unit making the request. The central controller will cause the first available DATCS channel to be assigned to the priority request.

MOBILE-TO-MOBILE COMMUNICATION 7.

Mobile-to-mobile communication is inherent in the trunked radio system. Each mobile is assigned an individual address and a group address. The controller software determines what group code or codes will be accessed by each mobile on an initial call-up.

A mobile is programmed to communicate with any other mobile in the group through the repeater. By means of the individual address assigned to each mobile, the controller can also be instructed to activate individual units within the group and exclude all other units even those with the same group address.

If intergroup or interagency flexibility is needed, the dispatcher or terminal operator can, through software instruction to the controller, enable the mobile unit to call up any individual address, group address, or subgroup address within the capability of the trunked system.

8. SYSTEM FAILURE MODES

A DATCS must inherently protect itself against common failure modes. The following paragraphs address system features required for system failure protection. The discussion in the following paragraphs relates to the three base station configuration described for the Salt Lake County/Utah County area in which two of the base stations, Mt. Vision and Lake Mountain, have their own control channel and the base station at Snowbird (Hidden Peak) has no control channel transmitter. Radio frequency propagation tests may later prove that the roles of Mt. Vision and Snowbird should be reversed.

1) Central Control Failure

The total failure of the master controller at the central location, or any of its terminals, should not be apparent to field operations. The switching and control logic is resident in the site controllers. The central controller monitors the site controllers and initiates any

 \square

 Π

1 (

- · .

2) Site Controller Failure

The failure of a site controller should permit full DATCS operations in a degraded mode using the remaining two base station sites. There should be no need to switch the DATCS to conventional radio channel operation or "fail soft" operation. A failure of the site controller at Lake Mountain and Mt. Vision will be evident to the users as a temporary reduction in coverage due to shadow effect; the failure of the site controller at Snowbird (Hidden Peak) will probably create talk-back problems in the canyons. The failed base station site should also be muted to prevent interference on the voice channels.

Voting Selector Failures 3)

The failure of a voting selector should not be apparent to the field operations; one voice channel will be disabled and its traffic will be distributed among the remaining DATCS voice channels (control channels are not voted). A total failure of the voter selector system effecting all voice channels should enable DATCS to function in a degraded mode using one base station site; Lake Mountian or Mt. Vision whichever proves to have the greater area coverage. The remaining two sites should be muted.

4) Base Site Repeater Failure

The failure of a repeater at any base site should be transparent to field operations. The failed repeater will not be voted in the DATCS sequence of operation until it is repaired and returned to service. It should not be necessary to take any DATCS voice channel out of service due to the failure of any one of its three repeaters.

5) Antenna Subsystem Failure

The antenna subsystem at each base station site shall be configured so that a catastrophic failure of any one antenna, combiner, filter, or multicoupler will have a minimum effect on DATCS operations. To the degree possible these antenna subsystem elements will be standardized so that repairs can be made by replacement or interchanging elements.

6) Control Channel Failure

The DATCS concept includes two separate control channels, one on Lake Mountain and one on Mt. Vision. The failure of either channel will cause the site controller to assign a back-up voice channel at that site as a substitute control channel. The mobiles or control stations will not require

changes or corrections. The journal recorder should be switched to the output of the site controllers when the master controller fails, and the voter selector interface with the master controller should be neutralized. The only functions that can not be initiated are manual program changes.
programming modification to accept this substitution. If the loss of a voice channel is not desired due to traffic loads, a degraded mode of operation can be tolerated using a single control channel from either Lake Mountain or Mt. Vision.

Failure or degradation of individual control stations or mobile stations will not degrade DATCS system performance.

9. SYSTEM TESTING AND ALARMS

An alarm panel will be installed at the master control terminal location. This alarm panel shall provide failure alarms (performance below acceptable thresholds) for at least the following:

Excessive VSWR in any antenna subsystem transmission line

Loss or reduction in output power of any transmitter

Reduction in receiver channel signal-to-noise ratio below threshold

- Unauthorized carrier on any channel
- Loss of site controller integrity

Degradation in repeater receiver interface

Degradation in repeater transmitter interface

Failure or degradation of a voting selector.

The alarm system that alerts to any element of degraded performance should also provide preliminary diagnostic information. This level of diagnosis may be limited to the identity of the offending unit of equipment and the kind of offense being observed.

10. SYSTEM MANAGEMENT AND CONTROL

The high technology features of DATCS are centralized and require an integrated control. This implies central management and control for sustaining operation on behalf of the user organization. Key management requirements are discussed in the following subsections.

1) Journal Recorder/Printer

System operations, as reported by the site controllers and the voting selector, are recorded on a magnetic tape transport. This information will include:



Î

The maintenance of the DATCS must be uniform and standardized. The performance quality of each voice channel should be uniform with every other voice channel regardless of which of the three base station sites is used for each transaction. The skill and training required to achieve this maintenance requirement calls for a highly skilled, centrally managed maintenance staff.

System activity by each user agency control station System activity by each mobile stations

Total channel activity time

Channel use profile by hour, day, week, month

Use of special features such as emergency priority

System queue population and holding time in the queue by hour, day, week, month

All alarms by hour, day, week, month

Additional information entered manually by the master control terminal can include:

User complaint reports

Channel down time for routine maintenance

Channel down time due to failure

Maintenance reports - time/cost

Selected reports from the journal file can be programmed by the master controller and printed out periodically on the system printer. Other reports can be programmed and printed out off-line by batch process EDP operations.

Centralized System Control

Overall DATCS operation on behalf of the multiple users will be controlled from a master control terminals at a central location. This control point will be responsible for sustaining the quality of system operation, providing limited access to the address code structure from user agency terminals, making programming changes, and implementing temporary program modifications such as interagency coordination and setting up

Centralized Maintenance

4) Management

The recurring costs of operating, maintaining and expanding DATCS over a phased implementation program, and the high level of performance required of the system requires a dedicated and centralized management organization. This management should be at the level of a Division or Department of the State or local government in order to exercise the needed authority.

V-62

responsible for each system. \square 1 \square \square \square D D

 \square

. .

APPENDIX C

CURRENT SYSTEM INVENTROY

FOREWORD

This Appendix has been derived from the report "SALT LAKE AND UTAH COUNTIES - MOBILE RADIO SYSTEMS DESCRIPTION AND REQUIREMENTS" submitted by Booz Allen and Hamilton on March 31, 1980 and revised on July 15, 1980. These data represent the radio systems as configured at that time according to the reports of personnel responsible for each system.

ા હેર્~ેટ

B	Salt La
	The overall feasibility with a sing
	entire regio Federal regu radio, when
	Because addressed tr
	to revolutio local govern
	• A • Th pc
	re . Th
	The fir
	serving the present and findings of and develops
	implemented ments are de these two co
	Preceding page
د . مراجع د المحر	د و مراجع می برویستند در از از ا

INTRODUCTION

ake County, Utah was selected along with four other n the United States to participate in a model project. objective of this project is to demonstrate the of replacing several independent mobile radio systems le, integrated communications system that satisfied an on's local government mobile radio needs. Recent ulatory changes and advancing technology in mobile coupled with special planning, will permit an entirely h to local government radio communications.

e of its unique inherent capability, a digitally runked communication system (DATCS) offers the potential onize local government communications by providing nment users with:

greater number of operating channels on demand

he ability to directly communicate with dispatch bints, mobiles, and portables of all agencies, without egard to equipment compatibility

he opportunity to manage resources through communicaions in ways that were previously impractical.

rst step of an implementation planning process includes g users of the various communications systems currently counties of Salt Lake and Utah to determine their future requirements. This Appendix presents the the interviews conducted by Booz, Allen and Hamilton s a set of requirements to be met by any DATCS system in the Counties of Salt Lake and Utah. The requireesigned to meet the needs of all public agencies in ounties and will eventually apply statewide.

; blank



PRESENT SYSTEMS - SEVENTEEN INDEPENDENT MOBILE RADIO SYSTEMS IN SALT LAKE AND UTAH COUNTIES USING EXTENSIVE EQUIPMENT AND FREQUENCY SPECTRUM RESOURCES PROVIDE BASIC RADIO COMMUNICATIONS

Each Public Agency Operates Its Own Independent Mobile System With Individual Dispatch Centers and Base Stations - About 17 Dispatch Centers and 38 Base Stations Provide Coverage to Salt Lake and Utah Counties.

Sixty-six base stations and repeater stations strategically placed in 46 locations throughout the counties of Salt Lake and Utah provide mobile radio communications to the 17 agencies surveyed in these counties

> The four Law enforcement agencies, two fire departments and Utah DOT own and operate about 65% of all base stations and repeaters

Eight agencies use only one base station

The remaining use two or three base stations to achieve coverage in their operating areas.

Each agency, excepting Fleet management, owns and operates its own base stations and dispatch centers

Twenty-one Dispatch Centers manned by 104 full-time dispatchers coordinate the radio communications of the individual agencies - the public-safety agencies including Fire and the Utah DOT employ about 75% of all dispatchers.

The Agencies Surveyed Have a Total Inventory of About 1100 Mobile Radios and 325 Portable Radios - Their Channel Capabilities Range From 1 to 8 Channels (See Exhibit II)

> The 17 agencies surveyed use about 1100 mobile radio with channel capabilities ranging from one to eight channels to provide communications among vehicles and the dispatch center - the law enforcement agencies, the fire departments and the Utah DOT own about 75% of all mobile radios most of which are four channels

The law enforcement agencies, fire departments, and the Utah DOT also own over 65% of the approximately 325 portables used by the 17 agencies.

Thirty-three Channels Are Assigned to the 17 Agencies Surveyed - In Addition, the Agencies Share or Monitor About 40 Channels Belonging to Other Agencies (See Exhibit III)

> Thirty-three channels are assigned to the seventeen agencies surveyed

 \odot

Q12 1

-

[] \square \square

	Iegend	-											-										-	
	* Voting Comparator	DISP	ATCH											BZ	ASE	STZ	TIC	ON I	LOCZ	TIC	ONS			
•	** One base station	CEN	TERS																					ç1°4
	plus one satellite				·····					·····										-		15		i-
	receiver														Å							10	+	
				. · ·			ੱਮ				· · .				Гa									
·			្ល	۱.			Б.								ц			d)						
3.		a	Je -	ļ			ш	e L						г	ц Ц			ak						
		Ū.	្រុ				ö	ы Ц		1)		EJ	ď	ŏ	N I		ю. •	Ĥ	d	Ľ'n.				
		0a1	oat Sat	0.5	; ë		'n	လ်		jg	ທີ່	õ	Ľa]	dwc	z	63	ö	ц Ц	ភ្ល	ца	¥	×	•	
		[s]	ເຊິ່	Ľ,	- õ		n	ЧЧ	10	12	0	23	Þ	Ře	S	Ö	54	ល	ц ц	H	ับ เม	6	ge	
		Ë,	Ä	Ľ.		สร	je.	h	54!	ц.	Ĩ		μ		Ē	ñ		Ĩ	BO	ō.	Å	Å	ц.	
	ACENCTES	ч ч ч	4	" (ທິທີ	al,	Ţ	ល័		E.		S	Ĩ,	S	Ţ		М	S	ž	2	g	en	Ř	a)
	AGENCIES			z g	SQË	ΪV	10	Ő	В	ų.	Ē	8	Þ.	00	ц С	0	51	00	Se S	ģ	5	đđ	Ŋ	Ĥ
		o la	ġ	0	n o g	Ę	_	510	44	¶s	37	46	E.	34(-	45(6	23	ပြ	La]	Ë	Ξ÷	Ö	D
			·																					
				1																	· ·			
	.Law Enforcement																							
•	-Pleasant Grove P.D.	1	4	÷								2												
	-Sheriff's Central	,	· •																					
	Dispatch		20				2								: 1	2	1				3			
	-SLC Sheriri's Oir.	L T	22				2			7					т	2	<u> </u>	. 1	2	٦	1	1	2	٦
	-Utan Highway Patroi	-	20							1								т	2	Т.	Ξ.	· •	2	Ť
4	D!			.																				
ĩ	.Fire		- 0		-																			
5	-SL City Fire Dept.		TO	2	1			.	• •	•						, T								
	-SL County Fire Dep.	1 1	9					2	· 2·	•				,										
	.Public Services				,															۰.				
	-SL CITY & County	.,	1					7																
· .	Health		. Е Т	1	. 1			т																
• .	-SL PUBLIC WORKS	2	2	1	· 1	. 2		7																
	-SLC Attorney		2																					
	-SLC Surveyor	2	1	· ·				1																
3	-SLC BIG. Inspection		2					±.																
	GIG Aging Mranco-																н. 1 1.1.1							
• .	-SLC Aging Transp-	_	1	· ·	1																			
	ortation	L _	1		т																			
	The improved this of			1																				
	Universities	2					-																	
	-Bringham Young Only												2										•	
	-utan university	↓ ↓											. 2											
	011																							
	Uther			ľ						1									1	1	1	2	1	1
	-Utah DOT -Fleet Management	lδ	2							T									. .			<u>م</u>	-	-
		10											· · · · · ·			•								
	TOTAL	<u>د ۱</u>	Ť03												;									
																							-	



EXHIBIT II INVENTORY OF MOBILE AND PORTABLE RADIOS

		-	MOBI	LE RADIOS]	PORTABI	LE RADIOS	3	
AGENCIES		Number	of Cl	annels			-	Number	r of Cl	nannels	· · · · · · · · · · · · ·	
	1	2	3	4	8	Total	1	2	3	4	8	Total
. Law Enforcement					a			1				
- Pleasant Grove P.D.				17	2	19		3		10		12
- Sheriff's Central Disp.	1.1	-	÷ .	73	-	73				10		24
- SLC Sheriff's Office			¢٠	250		250	1 · · · ·			24		63
- Utah Highway Patrol				170	22	192				25		25
		1 A A			-	202				25	-	
. Fire	-		ļ							· · ·		
- SL City Fire Dept.		75			an a thu	75	22					22
- SL County Fire Dept.	75			=	-	75	50				-	50
	-				-							
. Public Services				-	1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -		-					1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -
- SL City & County Health	21					21	1					1
- SLC Public Works	-			110	30	140		3		P_{i}		- 3
- SLC Attorney				6		6	1.1	· · ·	· · ·	1.1		0
- SLC Surveyor	-	25				25		11				11
- SLC Building Inspection	20			-		20	1					1
- SLC Security	1					1.	40					40
- SLC Aging Transportation	14		Ĩ			14	1					.1
		* :				а. А.			1 1	-	1997 - 1997 - 1998 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -	1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 - 1997 -
. Universities									1			
- Brignam Young Univ.	-5		22	8	-	13	4			19	3	26
- Otan University			23			23		-	25			25
Other										-		
- Utab DOT		50		150		200		E.		10		15
- Fleet Management				130		200	-		ан (т. 1997) 1. т. 1997	10		0
ጥጋጥኔ ፐ.	136	150	23	711	32	1082	119	22	25	151	3	320

 $+ \left(\frac{\pi^*}{2} \left| \frac{1}{2} \right) \right)$

7

() S

EXHIBIT III CHANNEL ASSIGNMENTS

	ASSIGN	IED	SHAREI	D .	MONITO	RED
		Frequency		Frequency		Frequency
AGENCY	Activity	(MH <i>z</i> ,)	Assigned To	(MHz)	Assigned To	(MHz)
Law Enforcement						
- Pleasant Grove P.D.	Operations	46 520	Central Dignatch	154 860*	Central Dignatch	154 710vv
	Operations	465 475	Civil Defence	460 475	Orom P D	155 420
	operations	100.170	Highway Patrol	155 505*	OTEM T.D.	100.400
			Highway Patrol	155 745*		
			Interior	100.140		
- Sheriff's Central	Operations	154.860	Jeen Possee	47 500	Highway Patrol	155 625 **
Dispatch	Car-to-car	154 710	Mos Control	453 900	Provo P D	155 430
			County Boads	155 760	DVII	15/ 965 **
			County Fire	153 950	DIO	104.900
			Like	154 040		
			Sp. Fork	155 625		
			Sp. Fork	155 820		
		4	Highway Patrol	155 505	and the second	
			nighway Factor	100.000		
- SIC Sheriff's Office	Uniformed Officers	154 650	Search & Recove	45 500		
	Adm & Plain Clothes	154 785	Search & Rescue	45.660	and the second state of a	
	Traniry	155 370	Highway Patrol	155 505*		
	-mquity	133.370	Inginway ractor	100.000		
- Utah Highway Patrol	SLC	155 580		. 1	Local law	46 520 **
otan migimaj racioi	Tooele Co	155 910		1	onforcement	465 475 **
	Utah Co	155 625			agencies	15/ 710 **
	All counties	155 505	•		ageneres	154 650 **
	Motor Vehicles	155 785				154 785 **
	Car-to-Car	155 745				155 270 **
Fire		122.142				T02.210.""
- SL City Fire Dept	Dispatch	154 430		· · ·		
on erey tire bept.	Command	154 210	1			
	Paramedics	154 340				
	Radio Box	72.320				
	Mobile Printers	453 200				and a second second
	Mobile Printers	458 200				
	Car-to-Car	155.505		a.		
- SLC Fire Dept.	Dispatch	46.060			Sheriff Dispatch	454.600
	Command	46.100		ан. 10	Medical Network	154.340
Public Services						
- SL City & County Health	Operations	45.490				
- SLC Public Works	Sanitation	45.160			Search & Rescue	45.500
	Public Works	45.600				

1011010-0010-0010-00	Street recently and the second		 ~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~	<u> </u>
		7		

EXHIBIT III(2)

//

AGENCY		GNED	SHAR	ED	MONITORED			
	Activity	Frequency (MHz)	Assigned To	Frequency (MHz)	Assigned To	Frequency (MHz)		
	<u> </u>		· · ·					
Lic Services (cont.)								
LC Attorney	Operations	158.550			SLC Sheriff	154.650**		
					Highway Patrol	155.505**		
					Sandy City	155.310		
		÷						
LC Surveyor	Dispatch	158.955						
	Field Work	158,865						
	· ·							
C Building Inspection			SLC Security	158.820*				
C Security	Operations	158 820	10 (C)			1. A.		
C Securry	Operacions	100.020						
a Aging Transportation	Dispatch	45 280	•					
ac Adrug Transportacton	DISPALCII	73.200						
versities	Oneustiens	JEA OF	The Strenger States	154 015	Uishuan Datral	155 505 **		
ringnam young Univ.	Operations	104.965	Universities	T24.0T2	Districe D D	155.305		
			Statewide		Provo P.D.	155.190		
					Utan Co.Sheriii	154.860		
					Provo Fire Dept.	154.190		
		-						
tah University	Operations	154.725	Universities	154.815		and the second		
			Statewide			1.55 505		
ther	Operations	47.040			Utah Highway Patrol	155.625		
DOT	Operations	47.180				1		
Fleet Management	SLC Paging	156,120				4		
						· · · · · · · · · · · · · · · · · · ·		
Nency Band	ssigned	Shared*	Moni	tored				
Pand	10	3		2				
Band	20	10	1	9				
	3	2		2				
TOTAT	33	15	2	3				
					· · · ·			
t should be noted that	only 5 of the s long to other a	hared frequence gencies in the	cies are assigne e State.	d to the age	encies surveyed. The	e remaining		

Frequency Band Low Band		Assigned 10	Shared* 3	Monitored 2
VHF		20	10	19
UHF			2	$\frac{2}{23}$
	TOTAL	33	L T2	2.5

말을

T . Band Low Band VHF UHF Total (4) • • $\{2, 2\}$ total.

The four law enforcement agencies, the two fire departments, and the Utah DOT are the greatest users of the frequency spectrum

These agencies have about 70% of the frequencies assigned to the agencies surveyed

They share and use another 13 frequencies assigned to other agencies or about 90% of those frequencies found in our survey to be shared

They monitor 12 frequencies from other agencies or about 50% of all monitored frequencies in our survey

The VHF band is predominantly used by the agencies surveyed as shown below:

No. of Channels	<pre>% of Total</pre>
15	22
48	68
7	10
70	100

The Total Annual Operating Costs - Mainly Salaries for the 104 Dispatchers Employed by the Agencies Surveyed - and Maintenance Costs Exceed \$1.8 Million (See Exhibit IV)

> The total operating costs in 1979 was about \$1.7 million with about 60% paid by the public-safety agencies and 20% by the two fire departments and the Salt Lake City Department of Public Service

Maintenance costs for all agencies were about \$175,000 with the public service agencies responsible for about 80% of this amount. The Utah Highway Patrol alone spent about \$46,000 or about 35 percent of the

* * * * *

The seven largest agencies from the seventeen agencies surveyed were the four law enforcement agencies, the two fire departments and the Utah DOT. These agencies:

Own 65% of all base and repeater stations

Own about 75% of all mobile radios

Use 70% of all assigned frequencies.

v-71

EXHIBIT IV

			Maintenance 8	© Operating	Costs
0			OPERATING COSTS	5	
AGENCY	MAINTENANCE COSTS	No. of Dispatchers	Operating Hours	Cost	TOTAL COSTS
 Law Enforcement Pleasant Grove P.D. Sheriff's Central Dispatch SLC Sheriff's Office Utah Highway Patrol Fire SL City Fire Dept. SL County Fire Dept. Public Services SL City & County Health SLC Public Works SLC Attorney SLC Surveyor SLC Surveyor SLC Security SLC Aging Transportation Universities Brigham Young University Utah University Other Utah DOT Fleet Management 	\$ 3,000 2,088 12,250 45,657 6,000 10,000 2,100 12,000 700 3,900 1,300 1,100 1,250 5,000 8,000 19,000 750	4 8 22 20	24 24 24 24 24 24 24 24 7-5 24 8-5 24 8-5 24 8-5	\$308,235 118,945 410,000 269,360 120,000 140,000 140,000 25,000 10,000 22,000 40,000 10,000 55,000 63,000 33,333	\$311,235 121,033 422,250 358,140 126,000 150,000 7,100 82,000 25,700 13,900 23,300 41,100 11,250 60,000 71,000 52,333 750
TOTAL .	\$134,095	104		\$1,699,873	\$1,877,091

 $\mathbb{C}_{\mathbb{C}}^{2}$

 \odot

()

Legend * Dispatched by the Utah Highway patrol except in Utah County from 8AM to 5PM when DOT does its own dispatching

I 67

V-72

ن ي

3

 \bigcirc

 $\langle \rangle$

Der Co

11

1.1

Ś

Q.

00





SYSTEMS RECENT C	ARE INDEPENDENTLY MANAGED BY THE INDIVIDUAL AGENCIES: CENTRALIZATION OF EQUIPMENT MAINTENANCE AND REVIEW OF			the suppl
CAPITAL	PROCUREMENTS CONTRIBUTES TO REDUCING COSTS			(4) <u>Generally, Tele</u> ment Budgets an Replace Older
(1) <u>The</u> Indi	Mobile Radio Systems are Independently Managed by the Lvidual Agencies			. Most age
•	Each agency configures its system, installs its base stations and procures equipment to satisfy its com- munications needs			years beo their con severaly
•	Equipment types are procured on a need basis by each agency			. As a res aging eq radios.
•	Each agency selects its equipment supplier and often purchases equipment from multiple suppliers or through State of Utah state contracts		3	B. PROBLEMS AND LIMITA SYSTEMS PROVIDE BAS
•	Frequency coordination is done through the Utah Chapter of APCO's Frequency Coordination Committee consisting			SUCH AS PROVIDING I
	of AASHTO, APCO, and ISMA Coordinators plus a local government coordinator.			(1) The Individual munications Am
(2) <u>The</u> Mai Lak	Fleet Management Group in Salt County by Centralizing ntenance and Computer Records of Repairs Helps the Salt e County Agencies Reduce Cost Through Lower Repair Costs			. Transmis agencies
and Rep	Identification of Opportunities to lower Costs by lacing Equipment		area montana	. Channel in most
	The Fleet Management Group was established by Salt Lake County in order to provide equipment maintenance services on mobile radios for all the public agencies in the county			emergenc expresse - Br
•	A few county agencies in Salt Lake County continue to provide their own maintenance, for example Salt			- Th du
	In 1978 and 1979 the Fleet Management Group billed			– Th
	\$65,000 and \$80,000 respectively for maintenance services rendered. This is estimated to be about 20-25% lower than otherwise would have been spent	Contraction and Contraction		. Most age dependin frequenc operatio
•	The Fleet Management Group keeps records on each piece of equipment sent by the county agencies for mainten-			- Sh tw
	ance. Maintenance cost profiles are then developed for each equipment and used to determine replacement timing for the equipment			th - Sa
(3) Mar and	ny Agencies Elect to Employ Independent Radio Maintenance 1 Repair Facilities			ch of an
•	The two universities, the Utah Highway Patrol, Utah DOT, Pleasant Grove Police Department and Sheriff's			
	tenance on all radio equipment			

y agencies continue to service their consoles stations primarily through contracts with ier on local private firms

communications Take Low Priority in Governi as a Result the Agencies are Unable to quipment or to Expand Their Systems

cies interviewed indicate that in recent ause of budgetary restraint in the state munications capital procurement plans are curtailed

It the agencies have been unable to replace ipment, or to add needed mobile and portable

IONS OF PRESENT SYSTEMS - THE INDIVIDUAL AGENCY C COMMUNICATIONS BUT BECAUSE OF TECHNICAL UNABLE TO MEET CERTAIN OPERATIONAL NEEDS TER-AGENCY COMMUNICATIONS AND COMPLETE AREA

Agency Systems Provide Adequate Internal Comng Vehicles and Their Dispatch Center

ion quality was indicated by the surveyed to be acceptable within the coverage area

ccess delay was not indicated to be a problem nstances, except for some agencies during es or special events. The agencies which difficulties were:

gham Young and Utah Universities during cial campus evonts (e.g., sports)

Salt Lake City and County Fire Departments ing major fires

Utah Highway Patrol

cies operate on two or more frequencies on size and have assigned individual es to different operating units to facilitate s and improve channel access delay

riff's Central Dispatch has assigned its channels, one for general operations and second for car-to-car communications

t Lake City Sheriff's Office has its three nnels assigned one each for "uniformed icers", "administration and plain clothes" "inquiry"

	and a second	1	
			– The I
			its s
			each
	•		two f
	er e		are f
R			and f
		(2)	Inter-agency Comm
			Difficulties in t
			Limitations of th
			lack of con
n sen en la sen de la sen en la La sen en la			. The lack of
			agencies wa
			handicap.
			achieved th
			(See Exhibi
			- BYU n
			and H
			UILIC
			III UII L C
		•	- Some
			some
			، • فَظُنُ المَانِينَ المَانِينَ المَانِينَ المَانِينَ المَانِينَ المَانِينَ المَانِينَ المَانِينَ المَانِينَ الم
			• •
		and the second	
			• •
			System onei
			of each age
			useful by r
			improve ope
			•
		(3)	The Inherent Tech
			Preclude Operatio
n en			. A conventio
n han sena han an an ann an an ann an ann an ann an			
			are compati
			This approa
			radios for
			privacy by
			be unaccept
\sim , where \sim , \sim ,	البنية. 19-19 - 19-19	Literation of the second s	Taplinary developed a low of the construction
		n de la Antonio de la Composición de la Composición de la Composición de la Comp	
,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人 我们们的我们,我们就是我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的人,我们就是我们的			
	 And Anna Anna Anna Anna Anna Anna Anna A		
		·	and the second

Utah Highway Patrol has assigned four of six channels to geographic zones - one for Salt Lake City, Tooele County, and for Utah County, while the remaining two functionally assigned for "All Counties" for the Motor Vehicle Business Administration.

munications, System Adaptability and Coverage the Canyons Were Identified as Serious he Current Systems

Π

encies expressed problems of fading or total mmunications in most canyons

f a capability to communicate with other as indicated to be a serious operations Some inter-agency communication are hrough frequency sharing and monitoring it VI)

monitors the frequencies of the Provo Police Fire Departments, Utah County Sheriff's ce, the Utah Highway Patrol, and can comcate by mobile radio with these agencies

inter-communications are possible among agencies for example

Salt Lake City Public Works and Salt Lake City Sheriff

The Universities of Brigham Young and Utah

Pleasant Grove Police Department, Utah Sheriff Central Dispatch, Salt Lake City Sheriff and the Utah Highway Patrol

rational adaptability to organize field units ency by task or by geographic zone are deemed most agencies interviewed as a means to erations.

hnical Limitations of the Present Systems onal Needs From Being Met Economically

onal means to achieve improved inter-agency ions would be to permit other agencies to dios with increased channel capacity which ible with the other systems

ach would translate into more expensive these agencies and uncontrolled loss of the agencies. The loss of privacy would table to public safety agencies



-

1 To improve coverage in the canyons each agency would need to obtain and install additional base stations at considerable cost Improved system adaptability to permit greater flexibility in grouping mobile units on an assignment or qeographic basis could only be achieved through acquisition of additional channels which are scarce Additionally, for each frequency assigned, new base stations would need to be established and the existing mobile radios would need to be modified or new ones procured. TRENDS AND FUTURE PLANS - THE CURRENT AND PROJECTED REQUIREMENTS OF THE INDIVIDUAL AGENCIES IF MET BY SIMPLE EXPANSION OF THE CURRENT OPERATING SYSTEMS WOULD TAX THE AVAILABLE FREQUENCY SPECTRUM, CAUSE DUPLICATION OF INVESTMENT AND NOT YIELD THE NEEDED OPERATIONAL IMPROVEMENTS (1) The Projected 20% Growth in the Population of Salt Lake and Utah Counties Expected by 1985 Will Pressure Agencies to Expand Services (See Exhibit VII) The populations of the two counties has increased by about 30% in the last ten years Utah state and federal government statistics forecast continued population growth of about 20% by 1985 This population growth invariably will result in greater demand for services provided by the public agencies and corresponding increases in the use of telecommunications to provide these services. By 1985, the Radio Channel Requirements the Agencies in the (2) Two Counties are Expected to Double and the Number of Mobile Radios and Portables are Projected to Increase by About 40% (See Exhibit VIII) The agencies surveyed estimated an immediate need for 11 additional frequencies, an increase of over 30% to improve operations By 1985, an additional 42 frequencies, an increase of 130%, are projected which would increase the total channels in use to 75 The requirement for 75 frequencies by 1985 would severely tax the available frequencies allocated for public mobile radio use and provide limited opportunities for further expansion

V-78

4.

EXHIBIT VII Census Projection for Salt Lake and Utah Counties



1,100

1,000

900 -

800 -

700 -

600-

500-

400-

300-

200-

100-



		Channels		Mobile	Radios	Portabl	e Radi
Agencies	Present	Presently Needed	Required in 1985	Present	Required in 1985	Present	Requ: in 19
Law Enforcement							
. Pleasant Grove PD	2	2	3	19	19	10	1
. Sheriff's Central Dispatch	2	2	5	73	103	24	3
. SLC Sheriff's Office	3	5	8	250	375	63	9
. Utah Highway Patrol	8	8	10				
Fire							
SI City Fire Dent	2	3	5	75	75	22	2
. SL County Fire Dept.	2	3	5	75	75	50	5
Public Services						÷	
. SL City & County Health	1	2	2	21	21	1	
. SLC Public Works	2	2	3	141	180	3	1
. SLC Attorney	1			6	9		
. SLC Surveyor	2	2	2	25	30		
. SLC Building Inspection	1	2	3	20	20		-
. SLC Security		2	2		2	40	5
. SLC Aging Transportation	L	<u>д</u>	2	14	40	. <u> </u>	
Universities							
Brigham Young	2	3	3	13	15	26	3
. Utah University	2	2	3	23	30	25	3
Other							
Utah DOT	3	5	5	200	260	15	2
. Fleet Management		, J		2	200	±	
TOTAL	33	44	61	1082	1481	317	44

"

08**-**A

-



a second a s	 The second s			
	(* -			
			570 A	
		i i i i i i i i i i i i i i i i i i i	Gal .	
			(TT)	. The nu
				increa
			₩cn.#	 •
			17 • ≈ 1	. The nu
				500 ur
		I.		
			(3)	Improved Cov
			TT.	Improved Sec
	i			tant Require
				the Public
				^
				. covera
			L	preser
				· • • • • • • • •
				. increa
				as an
				operat
				Tmnwor
				. Improv
				Incere
			l m	mho ol
				. The at
			r~-1	agenc:
				LaCII
				Tmnwor
				. Improv
				emerge
				agenc
				Croate
				. Greate
				Dir Sys
				neces
			(1)	The Current
			(7)	ties to be H
				mente Becall
		and the second		mento becau.
				TO Dro
				• IO PIC
				within
				** {.,.1
				The nu
				at a
				provid
				statio
	ф. 	•		
	$H_{\mu} = \frac{1}{2} \left(\frac{1}{2} - \frac{1}{2} \right) \left(\frac{1}{2} -$			
		•	1 (1)	
		· · · · · · · · · · · · · · · · · · ·		
			ц.,	•
an Anglan an an air an ann an Anna an Anna. Anna an Anna an			STATISTICS AND COMPACTIVE STATISTICS FOR A COMPACT AND A C	ار این اور این
		· · · · · · · · · · · · · · · · · · ·		
			•	

The VHF band is almost fully utilized

The UHF band is about 50% used from the public safety standpoint

umber of mobiles needed by 1985 are projected to ase by about 400 units or 37%

umber of portables required in 1985 are about nits, an increase of 40%.

verage, Increased Inter-agency Communications, curity and Other Capabilities are seen as Imporements to Improve Operations and Services to (See Exhibit IX)

age of canyons is a particular difficulty at nt with all public agencies

ased Inter-agency Communications was identified important requirement and one which would enhance tion efficiency

ved communications security is of particular est to public safety agencies

bility to assign teams to a particular channel need basis is a strong need of all public safety ies, including fire departments, to improve and itate operations

ved channel availability, especially during ency situations is a strong need of all the ies surveyed

er system reliability and management information stem usage is attractive to some agencies and sary for others.

Systems Are Severely Limited in Their Capabili-Expanded to Meet the Present and Projected Requirese of Technical and Economic Constraints

ovide the 75 channels needed by 1985 to present ms would have to secure 37 additional frequencies in the two counties

umber of base stations would need to proliferate rate proportional to the channel requirements to de the current coverage. The additional base ons are estimated to cost about \$3000 each

		Oper	ationa	1	: 			Mana	gement	Requirements	
	Improved Reliability	Operational Reliability	Increased Channel Availability	Automatic Priority Capability	Improved Security	Intra-Agency Coordination	Other Functions (e.g., Radio-in service)	Management Infor- mation Recording	Common Equipment & Central Control		
aw Enforcement											
 Pleasant Grove PD Sheriff's Central Dispatch SLC Sheriff's Office Utah Highway Patrol 			•				0 0 0			Legend:	
ire . SL City Fire Dept.			•	•	0	•	0		0	• Necessary O Desirable	
. SL County Fire Dept.			•		0	•			U,		
ublic Services				0	0	e e e e e e e e e e e e e e e e e e e	ò	0	0		
Health SLC Public Works SLC Attorney SLC Surveyer SLC Building	0000	0000	• 0 0 •				• 0 0 0		0000		
Inspection . SLC Security . SLC Aging Trans- portation	00	0		0	00	0	0 0	0	0 0		
niversities											
. Brigham Young . Utah							0				
other . Utah DOT . Fleet Management	000	•			00				00		

.

-,

V-82

Ì

Γ



Expansion of the present systems would also not provide for many of the operational requirements of the individual agencies, for example:

Improve security

Greater interagency communications

Operation adaptability

Improved channel availability during emergencies.

DATCS REQUIREMENTS - THE SYSTEM AND MANAGEMENT REQUIREMENTS HAVE BEEN DERIVED TO MEET THE PRESENT AND FUTURE RADIO COMMUNICATIONS NEEDS OF THE AGENCIES SURVEYED AND TO PROVIDE FOR SYSTEM EXPANSION TO SERVE OTHER AGENCIES STATEWIDE (See Exhibit X)

The System Requirements Which Must be Met By A DATCS System Have Been Derived From the Trends In Radio Communications and Future Plans of the Agencies Surveyed

> The system requirements for DATCS proposed are intended to meet the growing radio communications needs as expressed by the individual needs and as implied by demographic trends

Other system requirements have been imposed because it is understood to be inherent capabilities of DATCS and can be of value to public agencies in improving operational efficiency

The system requirements have been expressed in terms consistent with the DATCS architecture.

The DATCS System Being Considered for Development has the Capability to Meet all the Requirements and Eliminate Duplication of Investment by the Individual Agencies

> Under DATCS System common base stations will be used by all the agencies transmitting and receiving at all frequencies. It is estimated that three of four sites could provide complete coverage in the two counties as opposed to those which would be required by 1985 with the present system configuration

The number of frequencies needed would be reduced with DATCS from the projected 75 under the present configuration to approximately 40 because of the improved channel utilization efficiency of DATCS

The DATCS system as presently conceived has the inherent capability to provide all the requirements needed or found desirable by the individual agencies to improve operations.

EXHIBIT X BASIC DATCS REQUIREMENTS

Ш

ARCHITECTURAL - The system initially must provide for at least 20 dispatch centers, geographically distributed, with several dispatch positions - The system must be expandable to include other geographically distributed dispatch centers from user agencies - The system must be expandable to include new user agencies. COVERAGE - The system must provide radio coverage throughout the counties of Salt Lake and Utah including all canyons - The system must be expandable geographically to provide regional coverage statewide coverage - Mobile radios beyond the range of the signaling system will be notified automatically. ADDRESSING -The system must be software adaptable to include or exclude specialized groups or discrete units or to recognize the group addresses -The system must permit at least 5 levels of priorities T V-84

EXHIBIT X (2)

- The system should permit priorities and addresses to be reassigned by keyboard entry on the controller.

SYSTEM PERFORMANCE

. .

.

- The system access time by any user must not exceed 0.5 seconds under normal busy hour traffic conditions

- The system must permit any group of mobile units to be transferred to any dispatch position within the system

- Privacy will be provided by discrete unit addresses and group addresses.

INTERCONNECTIVITY

- The system must be able to interface with the state, local or national data information systems such as NCIC.

- The system should be able to interconnect with the public telephone network

- The system should permit the use of conventional portable radios.

RELIABILITY

- Channel failure will be automatically detected and isolated from the system until restored

- Failure of the system controller will revert the system to a preplanned conventional mode of operation

- Failure of the signaling control channel will result in the automatic transfer of the signaling function to a usable voice channel.

EXHIBIT X (3)

M

Case 2

IJ

 \square MANAGEMENT - Overall system responsibility must be centralized - System management guidelines, controls and operational policies should be established - System maintenance including individual units, control units, base stations and consoles should be pre-determined and centralized - Each user agency's present and future responsibilities must be defined - Common training standards for dispatch, field personnel and users must be established - The system controller must monitor and record all message traffic including signaling by type user agency 81 . V-86

This Appendix is intended as a topical outline of a typical Statement of Work (SOW) for quick reference by the planner or system procurement manager. The SIP Guide (APCO Project 16B) provides supporting material for this topical outline.

The following outline adheres to the recommended order of topics for a high technology system SOW. The planner or system procurement manager is cautioned to employ only those elements of the outline which relate to the specific procurement and which can be accurately evaluated or measured.

APPENDIX D

STATEMENT OF WORK

FOREWORD

$\mathcal{O}_{\mathcal{O}}$, where $\mathcal{O}_{\mathcal{O}}$ is the second		I. INTRODUCTION
		The introdu a general overvi by the responder the following su
		1. Backgr
		A brie procurement
		2. Object
		A brie be achieved
	H	3. <u>Scope</u>
	([*])	services th example:
		- E
		• S
		.]
	1 fi ***	• 7 • N
		II. INSTRUCTION
		This section information need
		1. Procur
		The so bid request
	Ш_5 П ⁻ Т	• •
		•••• 1
		receding page blank
		enterenteringen over en
	ana ta sair	en e

APPENDIX D

STATEMENT OF WORK

uction to the Statement of Work (SOW) should provide iew of the project to enable a better understanding rs. Key elements of this section are outlined in ubsections of the Introduction.

round of the Project

ef historic description of the origin of the : and the constraints imposed on the design.

ive of the Procurement

ef discussion of the main system objectives to by the procurement.

of the Procurement

ic and brief outline of the hardware, software, and nat are being purchased. These may include for

Hardware

Software

System Design

Installation

Testing

Training

Maintenance.

NS TO BIDDERS

on of the SOW will provide the bidders with all the ded to respond correctly to the bid request.

rement Schedule

chedule time and dates for responding to the t will include:

Time and place of the pre-bid conference

Time and place bids and proposals are due

V--89

Bid and	proposal	opening	time,	date,	and	place
---------	----------	---------	-------	-------	-----	-------

- . Proposed contract award date
- . Expected date of contract completion.

2. Authorized Contracts

The persons will be identified who are authorized to respond to bidders requests prior to bid opening. These may be restricted to the:

Procuring officer

Technical consultant.

The statement should be made regarding how the issues raised by these contacts will be reported to the other bidders.

3. Bid and Proposal Requirements

Each of the items to be submitted as part of a responsive bid will be defined. Any limitations should be included in the description. These items would include for example:

- Itemized cost proposal
- . Technical proposal
- . Management proposal
- Provisions for alternate proposals
- Proposal limitation such as page count, standardized forms (cost proposal), proposal size, unnecessary artwork, and number of copies to be submitted.

4. Bidders Qualifications

The contractor for a high technology system must be qualified, technically and financially, to complete the required work. These qualifications should be expressed in the SOW by requiring reasonable responses to limits imposed on:

Bidders facilities and financial resources

- Performance history in terms of references and a list of similar and recent contract work
- Support facilites such as field maintenance shops, training facilities, and spare parts replacement depots

I. <u>INTRODUCTION</u> The introduction to the Statement of Work (SOW) should provide a general overview of the project to enable a better understanding by the responders. Key elements of this section are outlined in the following subsections of the Introduction.

A brief historic description of the origin of the procurement and the constraints imposed on the design.

2. Objectiv

1.

1.

8

6 1

A brief discussion of the main system objectives to be achieved by the procurement.

3. Scope of

services example:	tha
• • • • • • • •	На
• • • •	So
•	Sy
	Tn

V-90

APPENDIX D

STATEMENT OF WORK

Background of the Project

Objective of the Procurement

Scope of the Procurement

A basic and brief outline of the hardware, software, and ces that are being purchased. These may include for

rdware

ftware

stem Design

Installation

Testing

Training

Maintenance.

II. INSTRUCTIONS TO BIDDERS

This section of the SOW will provide the bidders with all the information needed to respond correctly to the bid request.

Procurement Schedule

The schedule time and dates for responding to the bid request will include:

Time and place of the pre-bid conference

Time and place bids and proposals are due

Bid and proposal opening time, date, and place

Proposed contract award date

Expected date of contract completion.

2. Authorized Contracts

The persons will be identified who are authorized to respond to bidders requests prior to bid opening. These may be restricted to the:

Procuring officer

Technical consultant.

The statement should be made regarding how the issues raised by these contacts will be reported to the other bidders.

3. Bid and Proposal Requirements

Each of the items to be submitted as part of a responsive bid will be defined. Any limitations should be included in the description. These items would include for example:

- Itemized cost proposal
- Technical proposal

Management proposal

Provisions for alternate proposals

Proposal limitation such as page count, standardized forms (cost proposal), proposal size, unnecessary artwork, and number of copies to be submitted.

4. Bidders Qualifications

The contractor for a high technology system must be qualified, technically and financially, to complete the required work. These qualifications should be expressed in the SOW by requiring reasonable responses to limits imposed on:

- . Bidders facilities and financial resources
- Performance history in terms of references and a list of similar and recent contract work
- Support facilites such as field maintenance shops, training facilities, and spare parts replacement depots

The quality and availability of personnel for management, supervision, and technical staff.

Proposal Evaluation Criteria

Criteria should be established for the comparative

Technical proposals

Cost proposals

5.

6.

7.

8.

related to:

]

evaluation of:

Management proposals.

These criteria should be clearly defined in the SOW so that each bidder will understand the emphasis placed on the requirements of the RFP.

Contract Award

The SOW will identify the type of contract to be awarded the successful bidder whether or not this contract will be awarded without pre-contract discussions.

Indemnification

The purchaser should require indemnification against damages imposed through intentional or unintentional acts of the contractor. High technology system procurements will need indemnification against the improper use of patented items or processes. All procurements which include services such as installation and test will be well advised to require indemnification from damages resulting from the actions of contractors' employees or agents.

Titles and Rights

Rights of the purchaser and ownership of titles will be clearly defined by the SOW. These items include statements

The transfer of title to equipment and software

Rights to data and software developed or provided as part of the contract

Rights and title to the submitted proposals

The right to reject proposals

The right to negotiate with any bidder

The right to cancel negotiations or the contract.

Legal counsel should be consulted for the scope, content, and wording of this subsection of the SOW.

9. Bonds, Insurance, and Warranty

Each governmental entity will have its own limits for, and description of, the bonds, insurance and warranty provisions for contractor provided material and services. Frequently there will be state regulations which also must be adhered to. These requirements will provide for:

- Performance bonds
- Workman's compensation
- Comprehensive liability insurance
- Warranty on workmanship and materials.

The purchasing department of the major governmental agency included in the program will provide counsel as to how these requirements are to be included in the SOW.

10. Costs of Proposal Preparation

The SOW should state clearly that any costs related to the preparation of bids or proposals can not be included in the proposed cost of the project or billed back to the purchaser in any manner.

11. Contract Disputes

and the second second

At the option of legal counsel, the SOW may indicate the means by which the governmental agency resolves contract disputes. This will be a function of the governmental body and will not be an option of project management. These provisions may include:

> The procedure and method of resolving contract disputes

Termination of contracts for cause

- The assessment of liquidating damages
- The identity of the court of jurisdiction.

12. Applicable Federal/State Laws and Regulation

When state or Federal laws/regulations affect the system design, its operation, or allowable use, these laws or regulations should be identified in the SOW.



III. SCOPE OF WORK This section of the SOW defines the items of work expected of the contractor; the specific tasks required by the contractor are described in detail. 1. Services to be Provided High technology systems will require professional services to be performed by the system contractor in addition to the delivery of hardware and software items. Typical services to be provided are: System design which can include system engineering. radio propagation engineering software development, and hardware development Installation and debugging of hardware and software Assistance in procuring required licenses and permits Developing and conducting performance tests and acceptance tests The preparation of sites and facilities which can include construction, HVAC, emergency power, lighting, installation of utility services, etc. 2. Deliverable Items Each deliverable item should be identified in the SOW. Deliverable items include hardware, software, and services. The list will be unique to the complexity of the system to be procured. However, some common items on all lists will include the following: Each item of electronic or radio hardware Each item of computer software Maintenance support Spare parts and spare module provisions Towers, buildings and other structures Manuals and documentation such as test plans, test procedures, maintenance manuals, operators manuals, user manuals, training aids, installation drawings, and test reports Progress reports and deficiency reports. V-93

Delivery Schedules 3.

Each deliverable item should be supported by a delivery schedule. The SOW will require a formal and contractually obligated schedule for:

Each item of electronic or radio hardware

Each item of computer software

Each tower or building

The completion of all facility preparations

Delivery of all manuals and documents

The time and place of technical progress reviews

The delivery of reports including progress reports, test reports, discrepancy reports, and the final report or acceptance test report.

IV. SYSTEM SPECIFICATION

The system specification is a complete technical system description which includes the system performance, system interfaces, and system configuration. It places technical limits on these parameters and adds reliability and/or maintainability requirements where appropriate.

Hierarchy of Specification and Standards 1.

Industry or regulatory specifications or standards frequently apply to the specification of a system. These should be identified. The hierarchy of these documents, as they apply to the system specification, should be defined so that conflicts between them can be resolved.

2. System Description

The system description includes the configuration of the system elements and their interconnection with each other. It is recommended that this be done graphically with supporting written clarification. The system description is derived from the system concept design.

System Interfaces and Supporting Services 3.

The means by which the system interconnects with the world outside of the system should be fully described. These interfaces are relatively unique to each individual system but some key interfaces will include:

V-94

		-1	. – –
	•	En	ner
		se	erv
		se	erv
4.	Syst	em	So
	The	sys	ste
tion	of t	he	op
requi	ired	to	pr
5.	Syst	em	Op
	The	spe	eci
is ta	ailor	eđ	to
in th	ne sy	ste	em
inclu	ide a	CC	pmp
	•	Tl	ıe
		sj	yst
		ar	ıđ
		pe	erf
	•	тł	ie
		op	per
		ma	ina

6.

Frequently limitations on funding or other resources will prevent the initial implementation of a sophisticated system from realizing the full potential of system operation, scope, and performance. When these expansion requirements are part of the initial system procurement, they must be fully described in the system specification.

7.

High technology systems may exceed the apparent limits of Federal or state licensing regulations and extraordinary procedures may be needed to acquire the necessary permits or licenses. These requirements should be clearly defined in the system specification. They will be unique to each specific system configuration and intended operation.

External communication systems such as mobile radios, private telephone systems, data processing, digital data systems, alarms, microwave transmission systems, and similar facilities

Utility systems such as public access telephone systems commercial power systems, and water supplies

gency operations such as emergency power ices, fire protection, emergency medical ices, and civil disaster prevention operations.

ftware

m specification will include a complete descriperations software and the applications software ovide the needed system performance.

eration and Performance

fication of system operation and performance meet individual system requirements as developed concept. Typically these specifications will lete description of:

functional modes of system operation for routine em performance, emergency system performance allowable fail-soft or degraded levels of ormance

calculated levels of personnel staffing for the ating staff, maintenance staff, and system gement.

Expansion Requirements

System Licenses and Approvals

V. EQUIPMENT SPECIFICATIONS

The equipment specifications define the quality and performance characteristics of each item of hardware and software needed to meet the system specifications. The identity of each hardware and software item is derived from the system configuration diagram which was developed as part of the system concept.

1. General

This subsection of the equipment specifications accumulates all items of the individual equipment specifications requirements that are common among all items of equipment or software. It also defines the quality and quantity of each item of hardware and software that is required to meet the system configuration. Typical items that would appear under the general heading are:

Applicable industry standards or publications and their hierarchy for this program

 \square

 \square

- Standard environmental conditions such as temperature, humidity, primary power, vibration, shock, dust, and dirt
- Deliverable equipment lists showing the quantity required of each item including spares
- . The required level of quality and workmanship for each hardware item.

2. Individual Item Specifications

Each equipment item will have an individual technical specification which reflects its construction and performance as required to meet the system specification. These are derived individually from the system performance requirements and should be no more severe than that needed to meet the system performance specifications. Software specifications are typically combined with the equipment specification for the computer subsystem or the switching subsystem in which they are resident.

APPENDIX A

EXAMPLE PROBLEM ANALYSIS REPORT

DRAFT REPORT

T

1

Submitted to:

Mr. John E. Simmons APCO Project 16B Project Director Phoenix Fire Department 313 North Ninth Street Phoenix, Arizona 85006

1.5

Task 6

ANALYSIS OF PROBLEMS AND REQUIREMENTS

June 15, 1980

BOOZ · ALLEN & HAMILTON Inc.

Communications and Information Technology Division

776 SHREWSBURY AVENUE TINTON FALLS, NEW JERSEY 07724 747-9303 AREA CODE 201

e e e e e e e e e e e e e e e e e e e		
	c	
т	סייזא ד	
• • •	T 14 T 14	.00
1.1 1.2	Purp Scop	e
II.	SYSI	EM
$\begin{array}{c} 2.1\\ 2.1.1\\ 2.1.2\\ 2.1.3\\ 2.1.4\\ 2.1.5\\ 2.1.5.1\\ 2.1.5.1\\ 2.1.5.2\\ 2.1.5.3\\ 2.1.6\\ 2.1.6.1\\ 2.1.6.2\\ 2.1.6.3\\ 2.1.6\\ 2.1.6.3\\ 2.1.7\\ 2.1.7.1\\ 2.1.7.1\\ 2.1.7.2\\ 2.1.7.3\\ 2.1.8\\ 2.1.8.1\\ 2.1.8.1\\ 2.1.8.2\\ 2.2\\ 2.2.1\\ 2.2.2\\ 2.2.3\end{array}$	Majc Mult Oper Sate Mult Fail Syst Dist Syst Char Syst Test Emer Ruth Disp Wait Venc Mult Syst Mob Man	or ipt ilprice of the second s
III.	OPE	RA!
3.1 3.2 3.3 3.4 3.4.1 3.4.2	Uni Inte Dis Fie Vol Sta	t a pai ld uni tus
· • • •		
		, 'A
	•	

T

Τ

1.00

L'Z

-

2

TABLE OF CONTENTS

	Page
ODUCTION	1
ose of the Requirements Analysis e of the Requirements Analysis	1 1
EM REQUIREMENTS	1
rr System Problems iple Base Stations ation with Portable Radios ilite Voting Receivers iple Dispatch Points ure Mode Operation eem Control Failure e-out Timers ributed Logic Failure e-out Timers ributed Logic Failure em Testing and Alarming nel Quality Evaluation eem Test Alarms rgency Priority hess Preemption batcher Alert ring Queue Priority ersystem Operations dor Compatible Equipment riple Vendor Incompatibility tems Considerations MDT Interfaces ile-to-Mobile Communication	2 2 3 4 5 5 5 6 6 6 7 7 7 7 8 8 9 9 9 10 10 10 11
RATING REQUIREMENTS	12
t and Group Address eragency Coordination patcher Work Load ld Unit Status unteer Status Report by the Mobile tus Interrogation	12 13 13 14 14 14

ii.

TABLE OF CONTENTS (Continued)

		Page
3.5	Restricted Access	14
3.6	Dispatch Point Displays	15
3.7	Mobile Unit Displays	15
3.8	Mobile-to-Mobile Talk Around	15
3.9	Public Telephone System Interconnect	16
3.10	Management Considerations	16
3.10.1	Operation Management	16
3.10.2	Maintenance	17
cv.	FUTURE FUNCTIONAL STANDARDIZATION	17

was approved on January 10, 1980.

I.

INTRODUCTION

The purpose of this requirement analysis is to define the functions and processes that should be satisfied by a trunked radio system for Phoenix and adjacent communities. To this end the discussion of the system requirements accepts the data output from Task 4 "Documentation of Existing Radio Systems and Reported System Requirements" and from this data develops the dimensions of those system requirements and planned solutions to the reported problem areas which impinge on the system concept.

1.2 SCOPE OF THE REQUIREMENTS ANALYSIS

This requirements analysis is limited in scope to those items that bear directly upon the development of a system concept. The system concept will be a baseline document intended to resolve the question of the feasibility of trunked radio techniques in solving public safety and public service land mobile radio problems in the Phoenix area. Areas of radio equipment design and the deveopment of a system procurement document are beyond the scope of this requirements analysis. These are to be addressed in later documentation.

categories:

System requirements

Operating requirements

Future considerations.

The discussion of system requirements is further divided into those classified as major problem areas critical to the basic functional integrity of the communications service and those system considerations which can be satisfied by trunked radio system basic capabilities.

II. SYSTEM REQUIREMENTS

and the

The system requirements (as reported through the system survey data) present some major system problems which must be resolved if a trunked system is to be considered feasible. There are other system considerations which, although of serious concern, can be accommodated more routinely. These are addressed individually in the following subsections.

This document responds to Task 6 of the Support Agreement and Work Plan for Phoenix, Arizona under APCO Project 16B which

1.1 PURPOSE OF THE REQUIREMENTS ANALYSIS

The following paragraphs divide the requirements into three

2.1 MAJOR SYSTEM PROBLEMS

The following problem areas and their proposed solutions are presented as being critical to the successful implementation of a trunked radio system in the Phoenix area. The potential solutions offered reflect state-of-the-art system technology and hopefully do not seriously violate the state-of-the-art hardware and software offerings that can be made available in the near future when competitive procurement documentation has been developed.

2.1.1 Multiple Base Stations

Radio coverage experience at UHF in the Phoenix area precludes any reasonable expectation that a single 800 MHz base station site will provide adequate coverage. At a minimum, base stations will be needed at North Mountain and South Mountain. Coverage of Paradise Valley, Tempe and Mesa will probably require a third base station site east or southeast of the Salt River. At this time Bell Butte or Tempe Butte are likely candidates but a location further east, such as the Mesa Water Tower may be preferred.

The technique of simulcasting from multiple locations could offer a solution. However, simulcasting requires the precise equalization of frequency, phase and FM deviation among all transmitters at each frequencey (there could be a total of 60 transmitters at 3 sites). Although simulcasting under these conditions is technically feasible, in practice it is extremely expensive to accomplish and requires an unacceptably high cost of maintenance to retain specified system performance.

The alternate solution for a dedicated control channel type of trunked system is to vote the received signal from the multiple locations and, by means of the voter comparator, also select the transmitter at the base station site which has the better received signal. Conventional hardware is available for accomplishing this and cost can be minimized if the backbone microwave system can be used for most of the interconnect transmission requirements.

The control channel, which is transmitting data continuously to all mobile units, can not be adapted to voter comparator steering of selected transmitter sites. A possible solution is to simulcast only the control channel from the multiple transmitter sites. An alternate solution is to transmit control information only from one site which can be reasonably expected to put a realiable data signal into all parts of the greater Phoenix area. The North Mountain site appears to be a possible candidate to meet this requirement.

If simulcasting the control channel is not an acceptable solution, and North Mountain will not provide adequate data coverage for the Greater Phoenix area, two or more separate control channels will be required to cover the service area. The mobiles will have to home on a different control channel depending upon what part of Phoenix they happen to be in. Control channel selection can be automated within the mobile logic but could create a problem for mobile-tomobile communications across Phoenix.

2

A preliminary observation of the 3 candidate base station site configuration suggests the possibility of a two cell trunked configuration. The primary cell consists of North Mountain and South Mountain with North Mountain transmitting the dedicated control channel signal. The secondary cell consists of the southeastern base station site that covers Mesa, Tempe and Paradise Valley; this site would transmit its own dedicated control channel which would differ in frequency from that of the primary cell. The cost in "overhead" control channels would be 10 percent of a 20 channel system.

If a distributed logic type of trunked system is selected in which control information is transmitted on the "idle" channel, the frequency assignments for each base station site may be different (all three candidate sites are within propagation range of each other and could mutually interfere with each other). This would divide the available channels of the trunked systems (probably 20 channels) into three separate groups with the possibility that no one group would have enough channels to handle its peak traffic load. An alternate frequency control system which would provide better frequency utilization permits frequencies to be duplicated among the three base station sites and the central controller can be programmed to inhibit the use of a frequency at any site other than the site where it is first assigned as result of a mobile request.

The technical proposal of the potential system contractors will have to explore the feasibility and cost effectiveness of these system alternatives within the confines of each supplier's hardware and software offerings.

2.1.2 Operation with Portable Radios

The availability of trunked portable radios will depend upon the successful development of a technically acceptable frequency synthesizer and associated control system switching logic each of small size and with small power requirements. These circuits together with the usual 800 MHz portable radio subassemblies, and a battery pack with sufficient capacity to permit at least 8 hours of high performance service, must be made available in a small handheld portable radio case. Estimates by industry sources place the earliest availability of such a unit in the 1983-85 time frame. Until such portable radios are available, trunked radio system equipment configurations will include only base stations, control stations and mobile stations.

Public safety radio systems in Phoenix are portable radio intensive. This is also true of Glendale, Mesa, and Tempe public safety agencies. Scottsdale Police Department is also heavily dependent upon handheld portable radios. The significant investment in this mix of VHF high band/UHF portable radio equipment must be protected. These existing equipments must be permitted to operate with the trunked radio system with no modification or, at the most, a very minimum of modification.

Conventional portables can interface with the trunked radio system through a cross-band repeater. This repeater would have a two frequency half-duplex VHF or UHF channel on one side and a frequency synthesized trunked radio control station on the other side. In the trunked system logic the cross-band repeater will have a single unit address for a cluster of portable radios all having the same VHF or UHF channel. Each such cluster will have its own cross-band repeater.

Trunked systems can not pass the system control information through to a cross-banded portable. Therefore, the portable will not be able to detect an "all channels busy" signal and will not know why a channel request has been denied. To correct this the trunked control station side of the cross-band repeater can cause an audible tone to be sent on the receive channel of the cluster of portables when trunked channel access has been denied. The denied request for channel access from a portable radio should not be placed in the queue for the control channel type of trunked system even if a "channel available" signal is also transmitted to the cluster of portables. Because the cross band repeater can not individually address each portable in the cluster, the portable radio queue will have to be a contention queue. With all portables in a contention queue for trunked channel access, the portable radio will have to manually repeat the request for a channel whether or not a "channel available" signal is provided.

The users of portable radios cross-banding into a trunked system will also have to learn to "quick call" the system for a channel request and wait for the request to be acknowledged before they start to speak. This technique has proven difficult in the past. Also, the "hold time" of the repeater/base station combination will require special adjustment to prevent dropping the trunked channel during normal press-to-talk switching through the repeater. These techniques have proved troublesome for field personnel to master.

The preferred solution to UHF/VHF portable operation is to continue using the portable radios on conventional VHF/UHF radio channels and manually cross patch these channels into the trunked system at the dispatcher's console when crossband coordination is required. Until trunked portable radios become available and have been field tested, the recommended solution for the Phoenix area is to keep the conventional VHF/UHF frequencies for portable radio operation and use the trunked system for the mobile radios.

2.1.3 Satellite Voting Receivers

Radio propagation experience with 800 MHz trunked systems indicates that base-to-mobile and mobile-to-base coverage has been similar. The effective radiated power of the two types of stations is very much the same when the installation of both has been well engineered. Therefore, a base station/control station/mobile trunked system without trunked portables would need no satellite receivers or possibly need only one or two remote satellite receiver sites to correct unique trouble spot and to vote with the base station receivers in a transmiter steering configuration. It will be the obligation of the system contractor to conduct radio propagation tests to minimize the need for any satellite voting receivers.

If trunked portable radios are eventually added to the system, the need for satellite receiver installations throughout the Greater Phoenix area will have to be re-examined at that time.

2.1.4 Multiple Dispatch Points

1.4.14

Caro

5ª

The Phoenix land mobile radio system, including participation by all appropriate governmental agencies of Phoenix, Glendale, Mesa, Scottsdale, and Tempe, will require a number of dispatch points to maintain the existing level of command over the field units. The survey data shows a total of 46 dispatch positions in 14 dispatch centers throughout the Greater Phoenix area. In addition there are a total of 154 remote control positions at 43 locations throughout the total area. While this number of dynamic dispatch points can be reduced by system planning, the remaining number is going to be large and will remain dynamic. The assignment of an individual mobile unit to a specific dispatch point will be information that resides only in the system control equipment and can not be changed or initiated from the mobile unit.

In concept, the mobile unit, in transmitting a request to access the trunked system, would only transmit its own identity and possibly its assigned group address. From this address code, the controller can determine from its data file which agency the mobile belongs to, the dispatch point the mobile reports to at that particular time, and the group (or subgroup) the mobile will be permitted to include in its call-up. It is also conceived that some mobiles will also have the limited capability of modifying the group request; this feature should be permitted for the singular purpose of reducing the dispatcher work load.

To respond to an incident which calls for a mobile command post and a mix of public safety/public service agencies responsible to the command post, the system controller can create a phantom group and the mobile units can be attached to that group by software instructions based on the mobile address codes. There would be no change needed in the mobile logic to create the phantom group or to return the units to their original group assignment.

2.1.5 Failure Mode Operation

The trunked system must inherently protect itself against common failure modes. The following paragraphs address system features recommended for system failure protection.

2.1.5.1 System Control Failure

An individual base station repeater is installed for each channel licensed to the trunked system. Radio coverage problems and the system configuration may require the installation of repeaters for all channels at all base station sites. These individual repeaters are interconnected and controlled by a controller. If the controller fails completely, the repeaters can operate independently as conventional repeaters. They can be accessed by mobile stations and control stations that are operating on the assigned frequency for

each channel of the trunked system. The mobile stations can be preprogrammed to home on an assigned channel if there is a loss of control signal or idle tone (depending upon the selected trunked system); the available 800 MHz conventional channels will be distributed among the mobile fleet units according to need and coordination requirements. Access to the repeater from the dispatch point can be achieved by control station, wireline, or microwave.

Conversion to conventional mode of operation from a trunked operation can also result in a severe shortage of frequency resources needed to conduct routine operations. To avoid this inconvenience, it is recommended that redundant controllers be installed in a hot standby configuration to improve system reliability.

Failure of the control system in any mode other than a total catastrophic failure should result in a graceful degradation of the system capability. The failure should be limited to the loss of control over individual channels. This limited failure mode should be transparent to the system users. The loss of one or more channels should not effect the operation of the remaining channels.

2.1.5.2 Time-out Timers

A repeater configured land mobile radio system such as a trunked system can be paralyzed by accidental or intentional prolonged mobile or control stations transmissions. To protect the trunked system from this failure mode, all transmiters of the base stations, control stations, crossband repeaters, and mobile stations will be equipped with time-out times similar to the timers used in conventional repeater type systems. Two categories of timing circuits should be specified; the first permitting an absolute maximum time to hold the channel with or without modulation, the second should provide a limited channel hold time without modulation.

2.1.5.3 Distributed Logic Failure

The inboard logic functions assigned to control stations and mobile stations should be minimized; critical system logic and data files must be concentrated at the central control point location. Failure of the logic functions or RF functions at the mobile or control station should effect only that individual unit; such failures need not be alarmed back to the central control point.

2.1.6 System Testing and Alarming

The complexity of a trunked radio system that will provide adequate communication services for Phoenix will require a built-in capability to evaluate the individual channel quality and alarm back to the central control point any degradation in channel quality below established threshold limits or other significant system degradations. This requirement is not unlike the remote testing and alarm capability used routinely in private microwave systems.

methods: to the central control point. 2.1.6.2 System Test trunked radio system.

15/11

11

6

Printer a

2.1.6.3 Test Alarms

.

The alarm systems that alerts to any element of degraded performance should also provide preliminary diagnostic information. This level of diagnosis may be limited to the identity of the offending unit of equipment and the kind of offense being observed.

2.1.7 Emergency Priority

The trunked radio system operating in a public safety environment should have the capability of providing prompt channel access to mobile units exposed to life threatening situations or other incidents which require the immediate response of emergency services. The general approaches to emergency priority capability for a trunked radio system are discussed in the following paragraphs. The conditions under which initiation of any of these priority functions would be authorized include all of the following:

> A routine request for channel has been denied by the trunked system because all channels are in use and

There has been a waiting queue established for the next available channel and

The activation of the priority function has been conscionable decision; i.e., there has been a manual switch closure

2.1.6.1 Channel Quality Evaluation

To minimize catastrophe channel failures, individual channel quality should be continuously monitored by one of the following

Idle channel noise threshold (no signal)

Busy channel $\frac{S+N}{N}$ threshold (signal or idle tone).

Departure from a predetermined threshold limit will be alarmed

The ability of the trunked system to respond to mobiles/control stations and to process their message content should be tested periodically. It is recommended that a programmable frequency synthesized test transmitter with known radiation characteristics and test message content (multi-frequency tone and/or digital message) be centrally located in Phoenix at a location that provides nominal 800 MHz converage of the Phoenix area. This test transmitter can be programmed to transmit its test signal on each trunked radio channel sequentially and the received test signal at each fixed receiver location in the Phoenix area can be compared by quality and message content with a standard. The standard can be established at the time the test transmitter system is installed. Any degradation below established limits will be alarmed to the central control point. It is preferable that this test system be peripheral to the

These conditions presume the implementation of a dedicated control channel type of trunked radio system. If all channels are in use in a distributed logic system, the mobile will be in a frequency search mode and will have no capability of requesting emergency priority considerations and there will be no capability of developing a waiting queue with or without priority. Therefore the following three basic alternatives of Ruthless Preemption, Dispatcher Alert, and Waiting Queue Priority refer only to a dedicated control channel trunked system.

2.1.7.1 Ruthless Preemption

Ruthless Preemption will result in the seizure of any radio channel (except the Control Channel) from its current user. For a reasonable and automated decision to be made, the assumptions must be that:

> The controller has a positive indication that the seized channel was not carrying equally urgent emergency traffic at the time it was seized

The controller is able to select a channel carrying noncritical traffic.

There is no assurance that these assumptions can be realized. The normal mobile/control station identifier or preamble will contain only the unit address. There is no information in the identifier that would signify the urgency of a given communication even among those mobile units who, by their identifier, are known by the controller to have preemption capability.

To permit the ruthless preemption of an in-use channel appears to be a very difficult equipment problem, programming problem, and operational problem. This approach to emergency priority is not recommended.

2.1.7.2 Dispatcher Alert

The activation of a unique and manually initiated emergency priority request through the control channel can be recognized by the Controller. An alert alarm message will be directed to the dispatch point assigned to the mobile unit; this message will also contain the identity of the mobile unit making the request. If the emergency priority capability is restricted to public safety mobiles, it is conceivable that this dispatch point will have one or more of the system channels in use at that time the priority alert is received and can release on of these channels to the mobile that requested the priority. This assumes that:

> The dispatcher has information relative to the identity of the trunked radio channel being used by his agency and can identify this channel to the Controller so that the mobile can be instructed to move to that specific radio channel and

The dispatcher can "freeze" the channel long enough so that the mobile can be switched into it before the system drops the channel (otherwise the next mobile in the waiting queue will capture the idle channel).

8

It is possible to meet those requirements through hardware and software development. The complexity and estimated cost of the development constrains the recommendation of this method of enabling an emergency priority.

2.1.7.3 <u>Waiting Queue Priority</u>

A more easily implemented approach to emergency priority is to move the priority request for channel access to the top position in the waiting queue without attempting to seize a channel. The success of this method depends upon the following assumptions:

> The trunked system is properly engineered and, in complying with FCC liecensing criteria, the probablity of blockage at peak busy hours is minimal and

Channel discipline is such that channel hold time for the average communication transaction is no greater than a few seconds (hopefully less than 5 seconds).

This type of emergency priority is inexpensive to implement in a dedicated control channel trunked system when compared to the other alternatives. Channel access to a well managed trunked system will normally be made within a few seconds even when the first call has been blocked. The Waiting Queue Priority system will insure that additional lost time is not incurred due to the size of the waiting queue. This method of emergency priority is recommended for the Phoenix trunked system.

2.1.8 Intersystem Operations

A mobile unit equipped with a trunked radio system configured for operation in the Greater Phoenix area can have difficulty communicating in other Arizona jurisdictions outside the Phoenix area each of which would have its own unique trunked system. Two of the more probable situations will be discussed briefly in the following paragraphs, one in which the adjacent trunked systems are provided by the same supplier, and the other in which the adjacent systems are provided by different suppliers. It is assumed that all mobile trunked radios have 800 MHz frequency synthesizers.

2.1.8.1 Vendor Compatible Equipment

1025

Top Determine

When adjacent jurisdictions have procured trunked radio systems from the same vendor, the signaling methodology and protocols are assumed to be compatible. It is possible with coordinated planning between the cooperative jurisdictions for a mobile from one jurisdiction to be able to communicate within the trunked system of the other jurisdiction. However, the following must also be addressed:

> It must be assured that the visiting mobile be able to identify and access the host control channel frequency

The host trunked system controller will have to recognize and accommodate any limitation in the visiting moble radio channel configuration. For example, a mobile from a 10 channel trunked system can not fully accept channel instructions from the controller of a 20 channel system

The problem of a visiting mobile communicating directly with its own dispatcher in its owned trunked system through the facilities of a host trunked system also involves long haul transmission media. To automate such an interconnection will involve additional compatible switching logic at the controllers of both trunked systems and may also require the use of private backbone microwave systems or, alternatively, the use of the public access telephone system. This may require special waivers of the FCC Rules and Regulations regarding interconnect services.

The difficulties presented by the casual unplanned intrusion of a visiting trunked mobile unit into a host trunked system are formidable and go unresolved by vendor offerings at this time. In the Greater Phoenix area it is recommended that all trunked radio facilities of the several candidate jurisidictions be planned as part of the same radio system supplied by the same contractor.

2.1.8.2. Multiple Vendor Incompatibility

At this time the hardware and software offerings of the key vendors of trunked radio systems are incompatible with each other. Although the radio frequencies, as assigned by the FCC for trunked systems, can be accessed by the mobile frequency synthesizers of competitive suppliers, the channel control methodology and protocols are incomprehensible between the competitor's equipment. Visiting mobiles intruding upon a host trunked radio system supplied by another vendor will remain completely out of communication with the host system.

2.2 SYSTEMS CONSIDERATIONS

The following system considerations will be addressed during the system concept development of the Phoenix trunked radio system.

2.2.1 CAD/MDT Interfaces

The public safety agencies in Phoenix plan a continuing expansion of land-mobile data transmission by means of mobile data terminals. These terminals will be interfaced with the public safety data files that serve the police and fire departments CAD systems. Mesa plans to expand Public Works MDT installations. Therefore, the engineering design of the trunked radio system, as proposed by the system contractor, must be based upon the bit error rate limitations of the digital system rather than analog voice intelligibility. This manifests itself in these areas:

10

Data rate error limitations will reduce the expected radio coverage range by approximately 6dB from that calculated for analog voice traffic according to past experience

The error detection, error correction, and message retransmission coding of the digital MDT system must be compatible with the multipath deep fade characteristics of the digital trunked radio communication system at 800 MHz (MDT data rates could be advisely affected)

The mobile terminal call-up and address protocols will have to be compatible with the switching delays introduced by the call-up and address protocols of the trunked system

If automatic status updating is included in the trunked system, the CAD system must be programmed to recognize the unit address and relate that address to a specific incident report in the journal file.

The system contractor for the trunked radio system must be required to conduct sufficient tests to insure compatiblity between the existing CAD/MDT systems and the trunked radio system being offered.

2.2.2 Mobile-to-Mobile Communication

Mobile-to-mobile communication is inherent in the trunked radio system. Each mobile is assigned a group address and this group address is accessed by the controller using the address code of the mobile unit; the controller software determines what group code will be accessed by each mobile on an initial call-up. A single mobile unit can also be considered as a special subgroup within the assigned group. A mobile can be programmed to communicate with any other mobile in the group through the repeater.

If intergroup or interagency flexibility is needed the mobile unit can also be programmed to access the trunked system controller through the dispatcher or controller terminal operator. The dispatcher or terminal operator can, through software instruction to the controller, enable the mobile unit to call up any individual address, group address, or subgroup address within the capability of the trunked system.

This arrangement keeps the mobile logic as simple as possible and provides a higher level decision filter for permissible mobileto-mobile communications. It is recommended for the Phoenix trunked system.

2.2.3 Management Information

i i

Ш

All of the switching process for the trunked radio system is accomplished at the central control location; the mobile unit or control unit input consists only of its identity code and group address. Statistical information relative to system operation and performance can be made available from this central control point. This system management information should consist of at least the following:

> System activity by each user agency System activity by each mobile or control station

Channel use profiles by hour, day, week, month

Use of any special features such as emergency priorityor cross agency coordination

Total activity time for each channel

System queue status

- Profile by hour, day, weeks, and month

- Average channel excess time from the queue.

Additional management data can be entered by the terminal keyboard. This data can include:

User complaint reports

- Channel down time
 - Due to catastrophic failure

Due to maintenance activity

- Automatic test system performance deficiency alarms
- Maintenance reports
- Time and material expended
- Installation and test man power costs.

III. OPERATING REQUIREMENTS

The basic operating requirements for the trunked radio system have been gleened from the individual survey reports of the potential user agencies and from documentation reported from APCO Project 16A. The basic operating requirements are discussed in the following sections.

3.1 UNIT AND GROUP ADDRESS

For considerations of cost, system complexity, and system flexibility it is evident that control and switching logic should be held to the absolute minimum in mobile equipment and control station equipment. This will also enhance the ongoing and future development of hand held trunked portable equipment in which size, weight, and power drain become critical design limitations.

It is conceived that the only information needed from a mobile or control station for the controller to identify the unit, identify the dispatch point for the unit, and identify the other members of the group to be included in the communication, is the two part address code of the calling mobile or control station. This address will consist of the unit identity and the group identity. All of the routing and control information can be programmed into the Controller data memory and activated upon recognition of the unit address. Routing codes and group access codes can be changed at any time by keyboard entry.

12

To provide a level of flexibility and thereby reduce the dispatcher work load, selected mobile units may also be provided with the capability for manual modification of the programmed group address. These are conceived as only permitting subgroup level changes in the performance of routine assignments. An address modifier can also be used by the mobile to request an interconnection with another agency or another group within the mobile's own agency. The dispatcher or terminal operator can permit this variance from routine operation by keyboard or function key instruction to the Controller.

In response to situations that require a multiagency emergency coordination with possibly a mobile command post as a dispatch point, the central control point can create a phanton group address by keyboard entry. Any number or type of mobile units within the trunked radio system can be assigned to this phantom group by keyboard entry without any change being required of the mobile address or mobile identity code.

To accommodate this level of operational flexibility, it is recommended that the Phoenix trunked system require all routing and dispatch point logic be located at the central control point and that the minimum of switching and address logic be located in the trunked mobile/control stations.

3.2 INTERAGENCY COORDINATION

The survey of Phoenix area agencies resulted in the development of particular pattern of interagency coordination connectivity thought by the agencies to be appropriate to their operating needs. This pattern of permissible connectivity between the agencies must be programmed into the controller software and provisions made for it to be modified by program instruction as needed. This capability of the trunked system must be required of the system contractor by procurement specification.

3.3 DISPATCHER WORK LOAD

The recommended group address configuration for mobiles and their dispatcher is that all mobiles and/or control stations that are assigned to a dispatch point be also assigned the same group address. All mobiles in the group will routinely monitor all the traffic between the dispatcher and the fleet and between mobiles within the fleet.

This group configuration is similar to the conventional repeater radio channel operation now used except that non-group co-channel users will not be heard by the group on the trunked system. If all members of the group are not simultaneously aware of the traffic between group mobiles and the dispatcher it is probable that the dispatcher will receive simultaneous multiple requests for service, to which a single dispatcher can not respond. Communication blockage will occur because of dispatcher overload even though each requestor is able to access a separate radio channel to the dispatcher. The less cost effective alternative is to add multiple dispatchers to each active dispatch point.
3.4 FIELD UNIT STATUS

It is desirable that the status of the mobile field units be reported in coded form rather than by analog voice. The present methods of status indications take two general forms; one a status interrogation by the group dispatch point and the other a volunteer status report by the mobile unit.

3.4.1 Volunteer Status Report by the Mobile

A device to encode preestablished mobile status codes and transmit these to the dispatch point need not, and probably should not, be integral to the mobile trunked radio system. An external encoder when activated can key the mobile trunked radio and, when channel access to the dispatcher has been achieved, will cause the appropriate status code to be transmitted. The trunked mobile radio will automatically provide the vehicle identity.

A companion decoder at the dispatch point will provide status information associated with the vehicle identity. This decoder output can be interfaced with a display board or with existing CAD systems to provide automatic logging of the information and to possibly update the CAD incident report.

3.4.2 Status Interrogation

An automatic device that could interrogate the group mobiles as to their last reported status need not, and probably should not, be an integral part of the trunked radio system. The device need only to key up the control channel or idle channel with the group address and, having accessed the radio channel, transmit the encoded polling sequence to the mobile units within the group. Decoding the mobile response can also be outboard to the trunked system.

The mobile units in the group will require additional logic capability in order to respond to automatic status polling. Each mobile must recognize the initiation of the unique polling sequence and respond individually and uniquely with previously recorded encoded status when it recognizes its own individual unit address code. This will increase the cost of the logic in each mobile which is expected to respond to an automatic status interrogation.

A mobile responding to an automatic status interrogation can only respond with a retransmission of the status previously transmitted manually. This is not fresh information and adds little to what the system knows to be the status of the mobile unit. Therefore, automatic status interrogation is not recommended for Phoenix on the basis of it being not cost efficitive.

3.5 RESTRICTED ACCESS

Under the stipulation that all group assignments, routing, and switching is done by controller software in response to the mobile unit address, it follows that any mobile unit can be programmed out of the system by keyboard instruction. A mobile unit address that has been locked out of the trunked system will receive no communications other than the mute control channel or idle tone and will be unable to capture a communications channel.

3.6 DISPATCH POINT DISPLAYS

Ĩ

Each dispatch point in the trunked radio system will receive only those mobile unit calls that are assigned specifically to that dispatch point by the controller software. The Phoenix survey shows that the identity of dispatch points and the location of active dispatch points change significantly with the change of work shifts and, in some cases, with the time of day. When dispatch positions are shared with other dispatch positions or dispatch points, the traffic can be merged with these other dispatch operations by software instruction. The dispatch position will display the following information for the mobile units assigned: Code identity of the mobile unit in communication with the • dispatcher Code identity of the mobile unit exercising an emergency . priority option. 3.7 MOBILE UNIT DISPLAYS For operational integrity the trunked mobile unit should have the following displays: A handshake indicating a request for channel has been received by the system An indication that the request for a channel has been . accepted and a channel has been made available An indication that the request for a channel has been . denied and the request has been placed in a waiting queue (not applicable to distributed logic systems) An indication that the mobile unit is beyond the range of the system base stations. These displays can be visual, audible, or a combination of both. 3.8 MOBILE-TO-MOBILE TALK AROUND When outside the range of the system base stations the mobile units should have the ability for the half-duplex talk-around between mobile units. The factors that should be specified in procuring the system include the following: One specific channel frequency must be selected for this operation The initiation of talk-around should require a manual switch selection in the mobile to prevent the mobile from automatic selection of the talk-around mode when experiencing deep fade multipath conditions within the range of the base stations The talk-around mode will be inhibited when the mobile is within the useful range of the base station signal. 15

3.9 PUBLIC TELEPHONE SYSTEM INTERCONNECT

The need for interconnection with the public telephone system was not expressed by the potential user agencies during the survey but has been included in the published requirements of APCO Project 16A. It is advisable that the trunked radio system not be configured for mobile access to the public telephone system without dispatcher or control operator intervention. The system specifications must conform to the FCC Rules and Regulations in effect at the time apllication is made for the system license.

3.10 MANAGEMENT CONSIDERATIONS

The management of a trunked radio system will require the adherence to standards of quality, performance, maintenance, and administration not previously required of conventional radio systems. The delivery of acceptable communications services to a group of user organizations having diverse operational needs implies centalized management and operation of the trunked radio system.

3.10.1 Operation Management

The trunked radio system is a centralized radio system with unified control that requires centralized management for its efficient operation. Some of the centralized functions include:

- Financial Considerations
 - Acquisition of implementation and expansion funds
- Procurement of joint-use hardware and software
- Preparation of recurring operating budgets
- Distribution of system costs to user agencies
- Technical Considerations
 - Development of interagency compatible software
 - Hardware installation and test
 - Unified vendor/supplier relationships
 - Regulatory agency procedures
 - . Uniform system performance standards
 - Uniform maintenance standards

Pers	sonnel Co
-	Acquisi
-	Acquisi
_	Trainir
-	Trainir operato

It is recommended that a centralized management be governed by a body which represents the user agencies in proportion to their participation in the trunked system.

3.10.2 Maintenance

A

R

The quality of the maintenance of all base station, control stations, mobile stations and/or cross band repeater equipment used in the trunked radio system must be standardized and should be accomplished by one highly trained and well equipped centralized maintenance organization preferably administered by the City of Phoenix.

IV. FUTURE FUNCTIONAL STANDARDIZATION

The expansion of the trunked radio system concept beyond independent local systems into interjurisdictional and interregional coordinated systems will depend in part upon basic standardization of methodology, procedures, control functions and control protocols that is not now evident in the vendors offerings. The lack of such standards will also prevent competitive modular procurement, post implementation modular expansion, and interjurisdictional expansion of even the presently conceived independent local systems.

There is concern that implementation and expansion of wide area trunked radio systems without the opportunity for competition modular expansion offered by national standardization will not be in the best interest of the public safety community.

onsiderations

ition and training of control operators ition and training of software staff

ng of system maintenance personnel

ing of agency dispatchers and mobile



FOREWORD

This Appendix to the Draft System Implementation Plans for Participating Communities contains the complete final report to Bucks County, Pennsylvania pertaining to their planning for a DATCS implementation. Following their review of the system problems and requirements, it was their decision to pursue the upgrading of their conventional radio system and to defer the digitally addressed trunked communications system planning until a later time.

and a second s		:			
		di seconda de la constante de			•
		÷ 6	÷	m	BOOZ · ALLEN &
			ſ		
				m	
9			0		
				1 m	
				and a second	
	T I				
	•				
		•		a province and a second se	Mr. Charles McGill
					Director of Emergency S
(1, 2, 3) is the set of the se					Bucks County Administra
					Room 107
					Doylestown, Pennsylvani
					Door Mr. Macille
					Dear Mr. MCG111:
					Since the APCO Pro
					ends officially on Nove
				π	for Booz, Allen to prov
	•				the completed activitie
					Although DATCS planning
					the project in Bucks Co
					In this letter report w
					under our support agree
				67	to the county as a resu
			•		
					1. ACCOMPLISHMENTS
					The Cuppert Agreem
					Project 16B outlined te
	No.				Booz, Allen support eff
					Allen and Bucks County
					were completed. The mo
					. Conduct neces
					Document curr
	•				
$\gamma_{\rm s} = 1000$ s $\gamma_{\rm s} = 1$. Analyze probl
				1 4.1	
					The first activity
					basis for documenting c
					the interviews, a data
					Approvimately 20 organi
					services were interview
				4	and future requirements
	مراجع المح ن ة المحافظ ال				systems.
			•	لاسكا []	
			•	1	
$r_{\rm eff}$,					
				هي ا	
				1	an and a set of the set
$\mathbf{L}_{\mathbf{k}}$ is a set of the first set of the set of					

& HAMILTON Inc.

776 SHREWSBURY AVENUE TINTON FALLS, NEW JERSEY 07724 747-9303 Area code 201 November 14, 1980

06992-005-001

Services ation Bldg.

ia 18901

pject 16B support contract with Booz, Allen ember 15, 1980, it is appropriate at this time vide to Bucks County a final letter report on es and accomplishments of this project. If was conducted only through the problem I that positive results were achieved and that bunty was as successful as in other communities. We would like to summarize the accomplishments ement and identify some of the major benefits alt of participation in the APCO 16B Project.

ment and Work Plan for Bucks County under APCO en activities to be accomplished during the fort. With the combined efforts of Booz, project personnel, six of these activities ore significant activities were as follows:

ssary interviews and review available documents cent radio communications systems and requirements Lems related to DATCS and conflicting requirements.

y was a data gathering effort to form the current systems and requirements. Prior to collection guide (another of the six activities ed for use in the data gathering process. izations including police, fire, and rescue wed throughout the county to collect current s and to obtain information on current radio Mr. Charles McGill November 14, 1980 Page Two

Following the data collection effort, a report was prepared that documented current mobile radio systems and requirements. This report provided a baseline description of the police, fire, and rescue mobile radio systems now in use, presented a list of current and future mobile radio requirements, and presented functional requirements for a new DATCS. This report provided the foundation for the DATCS problem analysis and for developing various alternative system design concepts.

The analysis of problems and conflicting requirements was performed and documented in a problem analysis report. This report identified a number of potential problems, developed three alternative mobile radio system concepts, and provided an analysis of the technical and operational impacts of each alternative, including budgetary cost estimates for system procurement. The three alternatives included a new UHF conventional mobile radio system, a partial DATCS (Lower Bucks with DATCS, Upper Bucks with existing conventional), and a county-wide DATCS. The outcome of our analysis indicated that some serious technical and operational deficiencies would exist with any form of DATCS implemented in the near future.

Our report advised the county that if UHF frequencies could be obtained to operate a new conventional UHF system (the county had a draft plan to license 21 UHF frequency pairs in the commercial UHF TV channel spectrum for a new public safety repeatered mobile radio system), then this solution would be certainly more desirable than any near-term DATCS solution for the county.

Upon consideration of this recommendation, the county decided to pursue the UHF conventional radio system solution. Upon receiving the necessary licenses and waivers to begin implementation of a conventional UHF system, DATCS planning under APCO 16B was terminated.

2. BENEFITS FROM COUNTY PARTICIPATION IN APCO 16B

Although the original objective to produce DATCS implementation planning documentation was not attained, Bucks County did, however, derive a number of benefits from participation in the APCO 16B project. Some of the more significant benefits realized are:

- The county had the opportunity to explore the trunked radio system concept as it compared to a new UHF conventional system.
- The results presented in the problem analysis report were instrumental in obtaining the necessary licenses and waivers from the FCC for the new UHF conventional system.

Mr. Charles McGill November 14, 1980 Page Three

The documents prepared for Bucks County (requirements report and problem analysis report) and for the entire APCO 16B project (the SIP Guideline manual) provided Bucks County communications planners with insights into the proper planning and implementation of new public safety radio systems, both trunked and conventional.

Basic problems and requirements have been identified and established for a trunked radio system, should the county decide to implement a DATCS in the far term when trunked systems may be sufficiently advanced to satisfy the problems and requirements identified within the County.

In light of the accomplishments and benefits resulting from participation in the APCO 16B Project, we feel that the Bucks County portion of the project should be viewed as a successful endeavor. Although DATCS will not immediately be implemented in Bucks County, the concept was successfully investigated and the investigation led to a viable solution to provide better public safety mobile radio communications.

It has been a pleasure working with you during the APCO project, and we wish you every success in the planning and implementation of your new UHF radio system. Please feel free to call upon us should you require further assistance in any aspects of telecommunications planning and implementation.

Gerald W. Bernas Principal

Approved by:

llin

William F. Stasior Senior Vice President

cc: Bruce Fisher

Martin Ficke Frank Grausso

Very truly yours, BOOZ /ALLEN & HAMILTON Inc.

